

Durham E-Theses

Molecular hydrogen in galaxies

Wilkinson, David Adam

How to cite:

Wilkinson, David Adam (1987) Molecular hydrogen in galaxies, Durham theses, Durham University. Available at Durham E-Theses Online: http://etheses.dur.ac.uk/6657/

Use policy

 $The full-text\ may\ be\ used\ and/or\ reproduced,\ and\ given\ to\ third\ parties\ in\ any\ format\ or\ medium,\ without\ prior\ permission\ or\ charge,\ for\ personal\ research\ or\ study,\ educational,\ or\ not-for-profit\ purposes\ provided\ that:$

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in Durham E-Theses
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full Durham E-Theses policy for further details.

Academic Support Office, The Palatine Centre, Durham University, Stockton Road, Durham, DH1 3LE e-mail: e-theses.admin@durham.ac.uk Tel: +44 0191 334 6107 http://etheses.dur.ac.uk

MOLECULAR HYDROGEN IN GALAXIES

by

David Adam Wilkinson, B.Sc.,

The copyright of this thesis rests with the author. No quotation from it should be published without his prior written consent and information derived from it should be acknowledged.

A Thesis submitted to the University of Durham

for the

Degree of Doctor of Philosophy



这、同心的

To the glory

of my friend and Lord

Jesus Christ

'He is the image of the invisible God, the first born of all creation; for in him all things were created, in heaven and on earth, visible and invisible, whether thrones or dominions or principalities or authorities, all things were created through him and for him. He is before all things and in him all things hold together. He is the head of the body, the church; he is the beginning, the first born from the dead, that in everything he might be pre-eminent. For in him all the fulness of God was pleased to dwell, and through him to reconcile to himself all things, whether on earth or in heaven, making peace by the blood of his cross.'

(Colossians 1:15-20)

ABSTRACT

This study aims to understand the key role played by molecular hydrogen in the evolution of galaxies, with a view to constraining its radial distribution in the Galaxy and the $CO \longrightarrow H_2$ conversion factor \mathfrak{A}_{20} .

The star formation rate is shown to be correlated with the surface density of H_{2} . A correlation between the molecular hydrogen fraction and the metallicity of a region allows the time evolution of H_{2} to be described. This leads to a modified 'Schmidt Law' of the SFR which explains quite naturally the production of galactic metallicity gradients and the constancy of the SFR in the absence of infall.

A consistent closed model of the chemical evolution of the Galaxy is proposed to solve the G-dwarf problem, the stellar age-metallicity relation and the metallicity gradient, leading to the prediction of some initial amount of pre-disc processing of gas into visible and dark matter. It is found that a constant yield of metals is more appropriate than a yield proportional to metallicity. Possible time variations of the returned fraction, the dark matter fraction and the SFR are also studied. For consistency, we suggest that dark matter in the solar neighbourhood could be totally baryonic provided the Miller-Scalo IMF is modified at the lower end, that is, the dark matter resides in low mass stars or brown dwarfs. The production of metallicity gradients in spiral galaxies is shown to be a direct consequence of the radial variation of the total surface density of matter and the age of the disc.

The role of molecular gas in the evolution of the Oort Cloud of comets is examined. It is shown that comet showers with a mean interval of \cong 30My cannot be produced using perturbations of the Oort Cloud by known stars or molecular clouds. If there is indeed an apparent 30My periodicity in the terrestrial mass extinction and geological records, we argue that astronomically induced processes are unlikely to be the primary cause.

Evidence is presented that the lifetime of the molecular gas phase is $\measuredangle 2.10^{\circ}$ y, and arguments, particularly from CO observations of the Virgo galaxy cluster, favouring longer lifetimes are shown to be not well founded. We suggest that the ICM in Virgo reduces the value of \oiint_{20} as compared to isolated galaxies.

From the above considerations, the radial distribution of H₂ in the Galaxy is derived and shown to agree in the inner Galaxy with that derived from &-ray analysis. In the solar neighbourhood we find $\alpha_{20} = 2.5\pm0.5$, and present evidence that α_{20} varies as a function of Galactocentric radius and from galaxy to galaxy.

PREFACE

The work presented in this thesis was carried out between 1984 and 1987 while the author was a research student under the supervision of Professor A.W. Wolfendale, FRS, in the Department of Physics, University of Durham.

Parts of Chapters 2, 3 and 4 were carried out in collaboration with Dr. N.C. Rana (Tata Institute of Fundamental Research, Bombay), but the calculations reported represent the work of the author. The results and formalism of Chapter 5 were developed as part of a collaboration with Dr. M.E. Bailey (Department of Astronomy, University of Manchester) and Professor A.W. Wolfendale. Parts of Chapter 6 were carried out in collaboration with Professor A.W. Wolfendale but the calculations reported represent the work of the author.

Some of the results contained in this thesis have been discussed more extensively in the following publications: Rana, N.C. and Wilkinson, D.A., 1986. Mon. Not. R. astr. Soc., 218, 497-532.

Rana, N.C. and Wilkinson, D.A., 1986. Mon. Not. R. astr. Soc., 218, 721-728.

Rana, N.C. and Wilkinson, D.A., 1986. Il Nuovo Cimento, <u>9C</u>, 714-724.

Rana, N.C. and Wilkinson, D.A., 1987. Mon. Not. R. astr. Soc., <u>226</u>, 395-422.

Rana, N.C. and Wilkinson, D.A., 1987. In: Astrochemistry, IAU Symp. No. 120, eds. Vardya, M.S. and Tarafdar, S.P., p.323-324, Reidel, Dordrecht, Holland.

Rana, N.C. and Wilkinson, D.A., 1987. Mon. Not. R. astr. Soc., (submitted).

Bailey, M.E., Wilkinson, D.A. and Wolfendale, A.W., 1987. Mon. Not. R. astr. Soc., 227, 863-885.

-ii-

CONTENTS

CHAPTER 1 INTRODUCTION 1		
1.1	Galactic structure and components	1
1.2	The mass and radial distribution of $ extsf{H}_{ extsf{L}}$ in the	
	Galaxy	4
	1.2.1 Determination from CO	5
	1.2.2 The M-ray method	9
	1.2.3 Other methods	13
1.3	The role of molecular hydrogen in the Galaxy	15
CHAP	TER 2 THE STAR FORMATION RATE	21
2.1	Introduction	21
2.2	Previous formulations	22
2.3	Star formation rate and gas density in the Galaxy	27
	2.3.1 SFR in relation to the surface density of gas	27
	2.3.2 SFR in relation to the volume density of gas	32
2.4	The time evolution of the SFR in the Galaxy	36
2.5	Star formation in external spiral galaxies	48
	2.5.1 Star formation rate and molecular hydrogen	48
	2.5.2 Molecular hydrogen and metallicity	52
	2.5.3 The dust-to-gas ratio	57
CHAP	TER 3 GALACTIC CHEMICAL EVOLUTION I - THE MODEL	64
3.1	Introduction	64
3.2	Formulation	65
	3.2.1 The simple model	65
	3.2.2 Additions to the simple model	70
3.3	The G-dwarf problem and finite initial metallicity	76
3.4	The age-metallicity relation	84

i

-iii-

The variation of the SFR	90		
The metallicity gradient 9			
TER 4 GALACTIC CHEMICAL EVOLUTION II - CONSISTENCY	Z		
AND CONSEQUENCES	103		
Introduction	103		
The present values of the dark remnant fraction and			
the yield	104		
4.2.1 The remnant mass function	105		
4.2.2 The initial mass function	109		
4.2.3 Brown dwarfs	112		
4.2.4 The production of heavy elements	114		
Is the yield of metals a function of the			
metallicity?	116		
4.3.1 Variations in $\wp(m)$	118		
4.3.2 Variations in p _{zm}	119		
4.3.3 Variations in m_{fl} and m_{yl}	121		
4.3.4 Other possibilities	123		
The nature of the dark matter in the solar			
neighbourhood	126		
Metallicity gradients in spiral galaxies	133		
The radial distribution of $H_{\hat{\mathbf{Z}}}$ in the Galaxy	140		
4.6.1 A relative distribution of H_2 from SFRs in			
the Galaxy	141		
4.6.2 A tentative distribution of H_2 in the Galaxy	146		
Radial gradients in gas, stars and metals	150		
PTER 5 PERTURBATIONS OF THE OORT CLOUD AND			
TERRESTRIAL MASS EXTINCTIONS	158		
Introduction	158		
-iv-			
	The variation of the SFR The metallicity gradient PTER 4 GALACTIC CHEMICAL EVOLUTION II - CONSISTENCES AND CONSEQUENCES Introduction The present values of the dark remnant fraction and the yield 4.2.1 The remnant mass function 4.2.2 The initial mass function 4.2.3 Brown dwarfs 4.2.4 The production of heavy elements Is the yield of metals a function of the metallicity? 4.3.1 Variations in p_{XM} 4.3.2 Variations in p_{XM} 4.3.3 Variations in m_{fl} and m_{fl} 4.3.4 Other possibilities The radial distribution of H_2 in the solar neighbourhood Metallicity gradients in spiral galaxies The Galaxy 4.6.1 A relative distribution of H_2 in the Galaxy 4.6.2 A tentative distribution of H_2 in the Galaxy A.6.2 A tentative distribution of H_2 in the Galaxy PTER 5 PERTURBATIONS OF THE OORT CLOUD AND TERRESTRIAL MASS EXTINCTIONS Introduction		

	5.1.1 Periodicity in the terrestrial record	160
	5.1.2 The Galactic model of terrestrial mass	
	extinctions	164
5.2	The definition of an extinction shower	168
	5.2.1 The proportion of shower to background comets	169
	5.2.2 The critical semi-major axis	171
5.3	The signal problem	176
5.4	The signal⇔to-noise and phase problems	183
5.5	The survival of the Oort Cloud	191
CHAPT	TER 6 THE LIFETIME OF GIANT MOLECULAR CLOUDS AND	
	H_{2} IN THE VIRGO CLUSTER SPIRALS	194
6.1	Introduction	194
6.2	The lifetime of giant molecular clouds	195
	6.2.1 The mass of H_2 in the inner Galaxy	195
	6.2.2 The formation and growth of GMCs	204
	6.2.3 Spiral arms	206
	6.2.4 Other considerations	210
6.3	The lifetime of the molecular phase from gas	
	stripping in the Virgo galaxies	212
	6.3.1 The HI deficiency	213
	6.3.2 The likelihood of ram pressure stripping	216
	6.3.3 Comparison of the CO and HI results	218
6.4	The $CO \longrightarrow H_2$ conversion factor in the Virgo galaxies	221
	6.4.1 The effects of the inter-cluster wind on GMC	221
	6.4.2 The star formation rate	222
	6.4.3 IR results	228
	6.4.4 The evolution of galaxies in the Virgo	
	cluster	234

- v -

CHAPTE	<u>R 7</u> CONCLUSIONS	236	
7.1 5	Summary of the present work	236	
7.2 1	Implications for the CO \dashrightarrow H_{2} conversion factor and		
t	the mass of $ extsf{H}_{2}$ in the Galaxy	243	
7.3 E	'uture work	248	
APPENI	DIX	252	
A.1.1	Recent values of the CO to H_2 conversion factor,		
	$lpha_{2\odot}$, published in the literature	252	
A.1.2	Total Galactic H ₂ mass estimates for		
	2 < R _G < 10kpc	254	
A.2.1	Basic data for gas distributions and star		
	formation rate in the Galaxy	255	
A.2.2	Determinations of the metallicity gradient		
	(dex kpc ⁻¹) for the Galaxy	256	
A.2.3	Sources of data for Figure 2.8 (Section 2.5.2)	257	
A.2.4	2.4 The calibration curve of oxygen abundance against		
	the observed ratio ([OII] + [OIII])/H		
	regions	258	
A.2.5	Is the dust-to-gas ratio a function of the		
	metallicity?	258	
A.4.1	Best fitting model parameters for different sets		
	of δ_{i} and μ_{o}	263	
A.5.1	Short term periodicities claimed in the		
	terrestrial record	264	
REFERE	INCES	265	

ACKNOWLEDGEMENTS

282

CHAPTER ONE

INTRODUCTION

1.1 GALACTIC STRUCTURE AND COMPONENTS

The question of the origin and evolution of galaxies is still one of the important and unsolved problems in contemporary astrophysics. Tinsley (1980) noted that, 'essentially everything of astronomical interest is either part of a galaxy, or from a galaxy, or otherwise relevant to the origin and evolution of galaxies'. This is perhaps the main reason for our still simple understanding, the fact that galaxies are the most complicated configurations of astronomical bodies, and their study requires knowledge of almost every field of physics and the synthesis of universal cosmology with the astrophysics of individual bodies.

Galaxies may be considered to consist of three basic components, a visible round spheroid, a much flatter disc and a dark halo. The proportion of disc to spheroid determines a galaxy's position along the Hubble sequence, ellipticals being pure spheroid while spirals have both significant spheroids and discs. Spheroids are essentially purely stellar systems, while spiral discs contain typically about 10% of gas by mass. In spiral galaxies, the dark halo contains most of the mass in the galaxy in some inert (as yet unknown), collisionless form such as elementary particles, very low mass stars or black holes.

In the conventional cosmological picture, the primordial gas emerging from the Big Bang some 15-20 Gy ago

consisted almost entirely of hydrogen and helium (Wagoner 1973). Despite the expansion of the Universe, density perturbations led to some regions becoming locally bound and collapsing. The dark halo is thought to have undergone a non-dissipational collapse, with the dark matter already in its present form. The spheroid and disc formed from the collapse of matter initially in gaseous form. The matter that formed stars during the collapse results in the spheroid as there is no dissipation of energy. The matter that remained gaseous results in a flat disc, as collisions between gas clouds dissipate energy parallel to the net angular momentum vector. Subsequent star formation would then occur in the gaseous disc. The collapse of the disc occurs in a time which is short compared to the age of the galaxy, and therefore the evolution of the disc is relatively independent of the other components of the galaxy.

This thesis is concerned with the properties and evolution of spiral discs after the initial collapse and in particular the role of the gas. Observations of our own Galactic disc provide important information on this (roughly) 10% of its total mass.

The interstellar medium (ISM) has long been recognised as an important component of the Galactic disc. The total amount of gas is composed of \simeq 70% hydrogen, \simeq 28% helium and \simeq 2% elements heavier than helium. The gas provides the sites of star formation and contains the ashes of thermonuclear burning from previous generations of stars.

Following the work of Cowie et al. (1981) and Sanders et al. (1985), the ISM can be divided into four phases which

-2-

can be labelled as in Table 1 on the basis of different mean temperaturcs (T). The ionised hydrogen (HII) is produced as the result of star formation or supernova explosions

Table 1 : Components of the ISM

		Т(К)	Density (cm 3)	Filling factor
Hot	HII	3.10 ⁵	1.6.10 ³	0.74
Warm	HII	8000	0.25	0.23
Cool	HI	80	40	0.02
Cold	H2	10	300	0.008

in the ISM, but most of the mass of gas resides in atomic hydrogen (HI) and molecular hydrogen (H_2) . The amount of gas and its distribution, both in space and between these components constitute some of the most important parameters in understanding star formation, the chemical evolution of the Galaxy and the evolution of the Galaxy as a whole.

The distribution and amount of HI throughout the Galaxy has been known relatively precisely for some years. It is readily detectable through its 21cm emission line and the observed line intensities can be converted to column densities provided assumptions are made concerning the variations of spin temperature and optical depth. This has led to agreement about its distribution in the inner Galaxy (in the present work the distance to the Galactic Centre is taken to be 10kpc) although in the outer Galaxy there are some problems concerning the appropriate rotation curve to adopt (e.g. Burton and Gordon 1978; Henderson et al. 1982; Li et al. 1983; Blitz et al. 1983; Kulkarni 1987). However,

-3-

it has only recently been possible to estimate the H_2 distribution. It is generally agreed that a substantial fraction of interstellar gas is in the form of H_2 but the actual distribution has become a subject of much debate. The next section describes that debate and the principal methods used previously to determine the distribution of H_2 .

1.2 THE MASS AND RADIAL DISTRIBUTION OF Ha IN THE GALAXY

The reason for the controversy surrounding the distribution of H_2 , is that the molecule itself is not generally directly detectable. The lowest rotational transitions are far greater than the typical kinetic temperatures found in molecular clouds, in addition to which the rotational transition probabilities are very small since the molecule has no permanent dipole moment. It can only be observed directly in some special limited situations, either by absorption in the UV Lyman bands if the extinction is not too high (Bohlin et al. 1978) or by emission from shock heated or UV pumped gas (Beckwith et al. 1978; Shull and Beckwith 1982). The first method is restricted to the local ISM, denser and more distant regions containing H₂ cannot be observed in this way since the extinction becomes overwhelming. Shock heated emission gives a useful probe of denser regions where violent events are occurring but an observation of an unbiased sample of clouds is not possible. As the bulk of the gas is believed to be in dense cold clouds, which are not accessible to these techniques, any global survey of H₂ in the Galaxy must use indirect techniques.

-4-

1.2.1 Determination from CO

The most widespread indirect method is to use the 2.6mm $J = 1 \longrightarrow 0$ transition of the CO molecule as a tracer of H_2 . CO is the most abundant interstellar molecule with a permanent dipole moment and is readily excited by collisions with H_2 , for H_2 densities $\ge 100 \text{ cm}^3$, even in clouds with low kinetic temperatures (e.g. Scoville and Solomon 1974; Goldreich and Kwan 1974; van Dishoeck and Black 1987). The first mm observations of CO (Wilson et al. 1970) were followed by early studies establishing a qualitative relationship between CO emission and the presence of H_2 in a variety of structures (Liszt 1973; Dickman 1975; Knapp and Jura 1976). The presence of H_{2} is required for efficient formation of CO in most chemical schemes (Williams 1985) and CO has become a trace molecule for determining the H_2 distribution.

The first large-scale surveys of CO in the inner Galaxy (Scoville and Solomon 1975; Burton et al. 1975; Gordon and Burton 1976) showed the existence of a 'molecular ring' at approximately 6 kpc from the Galactic Centre, and that most of the emission was confined to discrete clouds, which contrasted to the HI distribution which was rather flat as a function of radius with little tendency to be clumped. More recent surveys of ¹²CO have broadly confirmed these results (e.g. Sanders et al. 1984; Sanders et al. 1986a; Bronfman et al. 1987), as well as observations of the J = 1- \approx 0 line of ¹³CO (Liszt et al. 1981; Despois and Baudrey 1983; Liszt et al. 1984).

The problem however, is an obvious one: how to transform the observed quantity, the velocity-integrated

-5-

¹²CO line temperature along a given line of sight $(\int T(^{12}CO)dv)$, to $N(H_2)$, the column density of H_2 in molecules cm⁻²? Theoretical arguments are not conclusive (e.g. van Dishoeck and Black 1987) and so it has become fashionable to determine the conversion factor empirically, where

$$\ll_{20} (10^{20} \text{ atoms } \text{cm}^2 \text{ K}^1 \text{ km}^1 \text{ s}) = \frac{2N(H_2)}{\sqrt{T(\sqrt{12}CO)} dv}$$
(1.1)

(It is to be noted that some other authors define their conversion factor in terms of molecules rather than atoms and therefore will state $pprox_{20}/2$ - often denoted 'X'). It is hoped that if a large enough area is sampled, $pprox_{20}$ will be more or less insensitive to cloud properties and structures and consequently a global value can be used for this and other galaxies.

The early surveys (Scoville and Solomon 1975; Solomon, Sanders and Scoville 1979) adopted $\alpha_{20} > 10$, giving rise to the conclusion that H_2 was the dominant component in the inner Galaxy. Much of the controversy concerning the mass of H₂ has arisen because of different estimates of α_{20} , but before these estimates are described, it is important to point out that the major CO surveys themselves have resulted in considerable disagreement about the quantity and distribution of the ¹²CO. In particular, Figure 1.1 shows the distribution of CO emissivity in the Galactic plane as a function of Galactocentric radius (R_{c}) from the work of the Massachusetts-Stony Brook group (Sanders et al. 1984, hereafter referred to as SSS) and the Columbia group (Dame 1984). This quantity is obtained from observations of the antenna temperature $T^{\mathfrak{P}}_{{\boldsymbol{\beta}}}(1,b,v)$ by an unfolding procedure based on an adopted rotation curve for the Galaxy and a

-6-

model for the vertical distribution of the ¹²CO. The difference between the two curves is primarily due to a



Figure 1.1: The distribution of ¹² CO emissivity for the Northern Hemisphere as a function of Galactocentric radius from the work of Dame (1984) and Sanders et al. (1984). The emissivity plotted is the peak value, that is allowing for the warp in the CO disk (from Mayer 1986).

different weighting used to obtain the radial emissivity and to a 20% lower CO intensity calibration (Bhat et al. 1986; Solomon and Rivolo 1987; Bronfman et al. 1987). Even if the same e_{20} was used, each curve would lead to different radial distributions of H₂. Therefore in any comparison of e_{20} determinations from different groups, reference needs to be made to the original CO data.

Nevertheless, estimates of $pprox_{20}$ have been made in a number of different ways both locally and throughout the Galactic disc.

Locally, the integrated CO emission has been measured for a number of regions with high visual extinction, and $N(H_{\odot})$ estimated from measures of visual extinction and the

-7-

extrapolation of a 'standard' gas-to-extinction ratio. The visual extinction A_{M} (defined as referring to 0.55µm) is derived from star counts (Dickman 1978) or infra-red extinction (Frerking et al. 1982). This involves major uncertainties, in particular star count estimates are usually very inaccurate and the extrapolation of the gas-to-extinction ratio determined from UV measurements of lightly reddened stars, N(H)/ A_{\forall} = 1.59.10²¹ cm² mag¹ (Savage et al. 1977) has been questioned (Rouan and Leger 1984). This method can also be compared with a 13 CO column density in regions where the 12 CO is optically thick (and therefore the observable line intensity is not linearly related to the number of CO molecules) while the ¹³CO is optically thin. This requires however, a conventional LTE analysis which is uncertain due to theoretical arguments (van Dishoeck and Black 1987; Maloney 1987) and uncertain isotopic ratios (Langer 1977; Bally and Langer 1982). Alternatively, SSS argue that LTE assumptions are unnecessary and propose that the 13 CO integrated intensity bears a simple relation to A_V and therefore $N(H_2)$. Problems with applying this method to a Galactic survey have been discussed extensively by Williams (1985) and Bhat et al. (1986). It is sufficient here simply to state that the determinations of \mathfrak{A}_{20} by different groups using this method range from 2.2 (Lebrun and Huang 1984) to 15 (Blitz 1978), and are listed in Table A.1.1 (Appendix).

An alternative method, used primarily on clouds in the 6 kpc molecular ring, is to estimate the mass of molecular clouds by application of the virial theorem. The simplicity of this method is that the observed line width can be

-8-

related with the cloud size to the cloud mass, which can be compared to the CO integrated emission and α_{200} derived. There is substantial observational evidence for assuming the clouds satisfy the virial theorem (Larson 1981; Sanders et al. 1985; Myers 1985), and Solomon et al. (1987a) using this method show an impressive empirical relation between the virial mass and the integrated CO emission for ≈ 200 clouds in the 6kpc molecular ring, giving $\alpha_{200} = 6.0$. However there is much controversy concerning the physical interpretation of the line widths (e.g. Kutner and Leung 1985; van Dishoeck and Black 1987) and Bhat et al. (1986) have argued that the virial theorem using the more appropriate optically thinner lines gives $\alpha_{200} \simeq 3.0$ locally, with some evidence of even smaller values in the inner Galaxy.

It is in view of these problems that 5-rays have been used recently to constrain \mathfrak{A}_{20} , and as this had led to important results it will be discussed in more detail.

1.2.2 The &-ray method

Diffuse \forall -radiation, with energy ~100MeV - 5GeV, in the Galactic plane is produced by cosmic ray interactions with the ambient hydrogen nuclei. The \forall -ray flux can thus be used to measure the total gas concentration (HI + H₂) provided the cosmic ray flux and its variations are known (e.g. Black and Fazio 1973; Li et al. 1982, 1983; Mayer-Hasselwander 1983; Lebrun et al. 1983; Riley et al. 1984; Bhat et al. 1984, 1985; Blitz et al. 1985). Outside the solar circle, CO is weak, and therefore HI is presumed to be rather more abundant than H₂. The 21cm line intensity can therefore be used to fix the zero point in the \forall -ray

-9-

are formed by the cosmic ray interactions, thus the pre-requisite of determining the mass of gas is to know how the cosmic ray intensity varies on a small scale (i.e. whether they penetrate molecular clouds) and on a galactic scale (i.e. is there a Galactocentric gradient in cosmic ray intensity?).

There seems to be no evidence either theoretical (Cesarsky and Völk 1978) or observational (Houston and Wolfendale 1985) that cosmic rays of energy sufficient to generate S-rays above 35MeV do not penetrate clouds and therefore 'see' all the gas. Studies of Orion and other local clouds by the Durham group (Issa and Wolfendale 1981; Li et al. 1983; Houston and Wolfendale 1985; Bhat et al. 1985) have yielded values for c_{20} of 2.3-3.7, lower than the value of the COS B collaboration (Bloemen et al. 1984) of $c_{20}^{4} = 5.2$ for the same region. This is due to the use of a lower local V-ray emissivity value in the latter work. However, all these values are lower than the $c_{20}^{4} = 6$ quoted recently by Solomon et al. (1987a,b) and Scoville and Sanders (1987) and much lower than the early estimates of c_{20}^{4} .

The evidence for large scale gradients in the proton component of the cosmic rays has been somewhat ambiguous (Bloemen et al. 1984; Bloemen 1985; Bhat et al. 1985) but recent work seems to confirm its existence (Strong et al. 1987; Bloemen 1987; Mayer et al. 1987). The \forall -ray, emissivity is the product of gas density and cosmic ray intensity; thus neglecting an increase of cosmic ray intensity with decreasing R_{G} will lead to an overestimate of \ll_{20} .

-10-

Arguing in favour of a cosmic ray gradient, Bhat et al. (1985) derived $\ll_{20} \approx 1.5$ for the SSS CO-distribution at $R_{fr} = 6 kpc$, that is a decrease by a factor of two from the local value. This decrease was attributed, following Blitz and Shu (1980) to the increase in metallicity observed in the inner Galaxy (Pagel and Edmunds 1981). Specifically $\mathfrak{C}_{\mathcal{R}_{\mathcal{A}}} \mathfrak{C} \left[\mathsf{O}/\mathsf{H} \right]^{\sim}$ gave good agreement with the 'Z-ray' $\mathsf{H}_{\mathcal{Z}}$ distribution. However, there are arguments, both theoretical (Kutner and Leung 1985) and observational (Young et al. 1985b) that metallicity cannot be a major influence on α_{20} . The true situation will be very complex since the gas phase abundances are likely to be controlled by a balance between accretion and shock disruption of dust grain surfaces (Williams and Hartquist 1984). More recent work on external galaxies (Bhat et al. 1986 and references therein) points strongly toward large-scale variations in \$200 but factors other than metallicity could be causing these changes. In particular, the detailed theoretical models of Kutner and Leung (1985) suggest that temperature is the dominant factor in determining \$20, such that

$$\alpha_{20} \propto T_{\rm bc}^{-1.3}$$
 (1.2)

(see also Maloney and Black 1987). If the clouds in the inner Galaxy are hotter than the clouds in the outer Galaxy, as claimed by Kutner and Mead (1981), then α_{20} should fall with $R_{\rm fr}$.

It should be noted, that the COS B collaboration (e.g. Lebrun et al. 1983; Bloemen et al. 1986) find $\alpha'_{20} \simeq 5.6$, constant with Galactocentric radius. However, this uses the Columbia CO survey (curve D in Figure 1.1), so that when the mass of H₂ is calculated for the inner Galaxy (shown

-11-

explicitly in Table A.1.2 (Appendix)) the results from the Durham group and the COS B/Columbia group are very similar.



Figure 1.2:

The time history of estimates of the mass of gas in the Galaxy, shown by the ratio of the estimated mass of H_2 to that of HI in the Galactocentric distance range $R_{G} = 4-8 \text{kpc}$ (from Bronfman et al. 1987; Osborne et al. 1987).

Mass-Stony	Brook	S:Solomon, Sanders & Scoville
		(1979); Solomon, Scoville
		& Sanders (1979); Solomon
		& Sanders (1980).
		Sa: Sanders (1981)
	5	SSS: SSS (1984)
		SR: Solomon & Rivolo (1987)
Columbia		C:Cohen et al. (1980)
		D:Dame (1983)
		TD:Thaddeus & Dame (1984)
		B:Bronfman et al. (1987)
		ST: Strong et al. (1987)
Durham	в1,82,	B3:Bhat [°] et al. (1984,1985,1986)
		0:0sborne et al. (1987)

Figure 1.2, illustrates the situation, showing the mass ratio of H_2 to HI between $R_6 = 4-8$ kpc as a function of time! It can be seen that the Columbia group, recently using an \ll_{20} from COS B, are similar to the Durham group, consistently below the estimates of Solomon and his collaborators. The massive amounts of H₂ claimed by the early surveys have been considerably reduced, although there is still significant disagreement.

1.2.3 Other methods

Apart from the main methods, described above, to determine \bowtie_{20} and M_{M_2} for the Galaxy, there are a number of other possible avenues.

Arguments from theoretical chemical models of the ISM (Duley and Williams 1984) together with observations of different molecules in molecular clouds are unable at present to be used for surveying the ISM on a large scale. However, current models of interstellar chemistry provide no support for the use of a canonical $CO:H_2$ ratio in dark clouds and the ratio may vary from point-to-point in a cloud, and between clouds reflecting a number of different factors (Williams 1985).

Optical extinction methods to determine \mathfrak{el}_{20} locally are not available in the inner Galaxy. Bhat et al. (1986) have summarised other techniques to calibrate \mathfrak{el}_{20} in the inner Galaxy. Most of the Galactic X-ray sources exhibit a low energy cut-off in their spectra because of absorption of soft X-rays through photoelectric interactions with chiefly atoms of oxygen, carbon, nitrogen and hydrogen (Adams 1980). Therefore by comparing the observed X-ray spectrum with the production spectrum an estimate of the total gas can be made. There are major uncertainties in the production spectrum, but Bhat et al. (1986) derive $\mathfrak{el}_{20} = 3.8$ locally, decreasing towards the inner Galaxy.

-13-

Unlike the optical region, the extinction by interstellar dust at near infra-red wavelengths is sufficiently small to enable the inner Galaxy to be seen. By examining the degree of reddening suffered at these wavelengths, an estimate can be made of the gas column density along various directions under the assumption of a large scale uniformity in the Galactic dust-to-gas ratio. Again there exists considerable uncertainty in this method,



Figure 1.3: Summary of estimates of ∞₂₀ as a function of Galactocentric radius. The points shown are taken from the various methods of Bhat et al. (1986), suggesting a gradient in ∞₂₀. Also shown for comparison are the constant values of ∞₂₀ claimed by SSS (1984), the COS B/Columbia collaboration (Bloemen et al. 1986) and the recent paper of Strong et al. (1987).

but Bhat et al. (1986) have argued from this method for a low value of α_{20} at $R_{6} \cong 6 \text{kpc}$.

The situation can be summarised by Figure 1.3, taken from Bhat et al. (1986), showing estimates of α_{20} as a function of R_{G} . In the inner Galaxy, the scatter is large, but taken with results for the Galactic centre (Bhat et al. 1984, 1985; Blitz et al. 1985), the data are compatible with

-14-

the argument that \bigotimes_{20} varies from place to place in the Galaxy.

These results for a 'low' \ll_{20} and low H₂ have been further strengthened by a study of IR emission in the inner Galaxy and Galactic centre (Parkinson et al. 1987; Osborne et al. 1987). This method uses fluxes at 150 μ m, 250 μ m, 300 μ m and IRAS wavelengths to determine the mass of dust and then uses an assumed dust-to-gas ratio to determine the total gas.

It must be stressed that all of the above methods contain assumptions and uncertainties, and although Bhat et al. (1986) claim consistent results on the value of \mathfrak{A}_{20} , these results are still the subject of much debate (e.g. Solomon and Rivolo 1987; Scoville and Sanders 1987).

1.3 THE ROLE OF MOLECULAR HYDROGEN IN THE GALAXY

From the discussion above, it is clear that the controversy concerning the mass and radial distribution of H_2 , and the $CO \rightarrow H_2$ conversion factor \ll_{20} is not yet concluded. In an extensive review, van Dishoeck and Black (1987) conclude that 'theoretical justifications of the empirical relations are still scarce and the range of validity of the conversion factors is still poorly understood'. Further work is in progress on the empirical determination of \ll_{20} (e.g. Solomon et al. 1987a; Strong et al. 1987; Richardson and Wolfendale 1987) as well as observations and study of molecular cloud chemistry (e.g. van Dishoeck and Black 1986; Maloney 1987). However, there is an alternative way of constraining the mass of H_2 in

-15-

galaxies. This involves attempting to understand the global physics of H_2 and its role in the many processes in galaxies. This study should not only be helpful in understanding the global evolution of galaxies, but could shed light on H_2 and therefore α_{20} .

Figure 1.4 shows the surface density of hydrogen as a function of Galactocentric radius. Following the discussion



Figure 1.4: Surface density of H_2 and HI as a function of radius from the centre of the Galaxy. HI is the average of the determinations of Burton & Gordon (1978) and Li et al. (1983), while two contrasting H_2 distributions are shown.

of Section 1.2 the H_2 distributions of SSS (1984) and Bhat et al. (1985) are shown in comparison with HI. Bhat et al. (1984, 1985, 1986) pointed out that such a dramatic decrease in the estimate of H_2 would have implications for star formation, Galactic chemical evolution and even perturbations of the Sun's Oort Cloud of comets. It is the intention of this work to study in detail those initial remarks, to understand the processes involved and to constrain the H_{\Re} distribution.

In the Galaxy, virtually all known regions of star formation activity are associated with molecular clouds (e.g. Shu 1985; Scoville and Sanders 1987). In Chapter 2, the relationship of the star formation rate to the mass of H_2 is investigated and an empirical law of star formation is suggested. Evidence from external spiral galaxies from IR and CO fluxes is used to further examine this relationship. The dust-to-gas ratio in spiral galaxies is then used to constrain \mathscr{A}_{20} .

The chemical evolution of galaxies studies the processing of hydrogen through stars into heavier elements and their distribution in galaxies. It is a subject of great contemporary interest particularly with respect to metallicity gradients observed in spiral galaxies (Pagel and Edmunds 1981). Bhat et al. (1984, 1985) suggested that lower masses of H₂ would provide better fits to metallicity gradients in the Galaxy and other spirals. Tosi and Diaz (1985) did show that in their models, 'metallicity corrected' H₂ distributions did give better agreement with observations of spiral galaxies, while Güsten and Mezger (1982) in fact, chose a distribution similar to that proposed by Bhat et al. (1985) in their model of the chemical evolution of the Galaxy. However, these authors did not take into account the very important role of H₂ in star formation.

Chapters 3 and 4 present a self-consistent model of the

-17-

chemical evolution of the Galaxy, using the star formation rate as derived in Chapter 2. Chapter 3 presents the model and compares its predictions with the observational problems of Galactic chemical evolution. The production of a metallicity gradient is shown to depend quite sensitively on the distribution of H_2 . Chapter 4 examines the consistency of the model with respect to the yield of metals from stars, and the nature of the dark matter in the solar neighbourhood. In addition, model predictions for metallicity gradients in external spirals are presented. Finally the distribution of H_2 is derived from the observed metallicity gradient and its time evolution is explained.

One of the results of the early studies of CO, was the realisation that the majority of H_2 was contained in Giant Molecular Clouds (GMCs) with mass $\simeq 10^5$ - 3.10⁶ M_@ (Sanders et al. 1985), which are therefore the most massive objects in the Galaxy. Dynamical arguments also suggested massive GMCs (Stark and Blitz 1978), and the observed mass spectrum means that most of the mass in the ISM is contained in the largest clouds (e.g. Sanders 1981; Tereby et al. 1986). GMCs therefore become important in dynamical processes in the disc, for example the disruption of open clusters (Spitzer 1958; van den Bergh and McClure 1980; Lynga 1982; Wielen 1985; Terlevich 1987) and globular clusters (Grindlay and Hertz 1984), the disruption of wide binaries in the solar neighbourhood (Weinberg et al. 1987) and the evolution of the velocity distribution of stars in the Galactic disc (Lacey 1984; Villumsen 1985; Carlberg 1987). Of current interest is the claim (Clube and Napier 1984a,b; Rampino and Stothers 1984a) that perturbations of the Oort Cloud of

-18-

comets by GMCs are the cause of the claimed 30My periodicity in the record of terrestrial mass extinctions. Chapter 5 reviews the claims of periodicity and shows that perturbations by GMC (or any other known astronomical bodies) cannot be the cause of a 30My periodicity. Bhat et al. (1986) suggested further, that reduced GMC masses would not totally disrupt the Oort Cloud in the way that had been understood previously (e.g. Clube and Napier 1982a; Napier and Staniucha 1982). This suggestion is re-examined with particular reference to the masses of GMC.

The final part of this work examines the question of GMC lifetimes and H_2 in other spiral galaxies. Scoville and Hersh (1979) claimed GMC lifetimes > 10^9 y and linked this to large H_2 masses in the inner Galaxy. Chapter 6 examines these claims and the link to H_2 masses.

At present there exist single dish CO measurements in over 100 spiral galaxies and well sampled radial distributions exist for 23 Sc and 19 Sb/Sbc galaxies (Young 1986; Scoville and Sanders 1987). The conventional approach has been to assume that the value of α_{20} determined locally, applies universally (Young and Scoville 1982a; Dickman et al. 1986). Thus all radial variations of CO within a galaxy and overall differences between galaxies have been attributed solely to variations in the amount of H₂. Apart from the controversy concerning the value of \ll_{20} locally, Maloney and Black (1987) have recently used sophisticated modelling techniques to show that this assumption could lead to an order of magnitude error in the estimated H_{2} mass due to differences in temperature, mean cloud density and metal abundance.

-19-

Alternatively, Thronson et al. (1987b) have suggested the use of IR data to calculate the mass of dust in a galaxy. Then, assuming a constant dust-to-gas ratio (taken as the solar neighbourhood value), the mass of H_2 can be determined if the HI mass is known. The weakness of this method is the assumption of a constant dust-to-gas ratio and whether all the dust in a galaxy is sampled. Nevertheless, if this IR method could be used to calibrate \approx_{20} in external galaxies there would be great advantages.

Using the assumption of a constant $\[mathbb{C}_{20}\]$, there have been claims about H₂ content and lifetime in galaxies in the Virgo cluster (Kenney and Young 1986; Stark et al. 1986; Knapp et al. 1987). Chapter 6 reviews these claims, and then examines the effect on $\[mathbb{C}_{20}\]$ of the cluster environment using data from other wavelengths. In addition, the constancy of the dust-to-gas ratio between galaxies is investigated. (A similar investigation is also carried out in the context of the star formation rate in Chapter 2).

Chapter 7 summarises the main conclusions and gives an outline for future work.

-20-

CHAPTER TWO

THE STAR FORMATION RATE

2.1 INTRODUCTION

Our poor understanding of star formation is one of the main obstacles preventing us quantifying how galaxies evolve. There are two functions related to the process of star formation which are of fundamental importance and application to many fields of current research, the frequency distribution of stellar masses at birth and the spatially averaged star formation rate (SFR) in a galaxy. Despite extensive observational and theoretical investigations, the question of the physical mechanisms involved in star formation and their relationship to these two functions remain conspicuously obscure. It has been recognised for a long time that the diffuse component of the interstellar gas had to be initially compressed in some way before gravitational forces could take over, star formation occurring by the fragmentation of these self gravitating clouds (Goldreich and Lynden-Bell 1965). Although there have been many attempts to relate cloud-scale processes to the galaxy scale (e.g. Seiden and Gerola 1982; Comins 1984; Freedman and Madore 1984; Struck-Marcell and Scalo 1987), the problem still remains of a general theory or parameterisation of the SFR in disc galaxies. This is vital in understanding or predicting the gross physical, chemical and photometric properties of galaxies (Tinsley 1980).

In this Chapter, previous global formulations of the

-21-

SFR are reviewed, followed by an attempt to relate empirically the SFR to gas density and metallicity in the Galaxy. The properties of this new formulation are qualitatively discussed and evidence for its application is sought from other spiral galaxies. Finally the effect of the gas-to-dust ratio is discussed in relation to the SFR and the amount of H_2 in galaxies.

2.2 PREVIOUS FORMULATIONS

The star formation rate has generally been thought to be related to the surface or volume density of the total interstellar gas. However, there is no prima facie reason why it should be related to the density of gas rather than the magnitude of the initial density contrast and the scale size of the spatial inhomogeneity. It is also not clear whether the SFR should be physically more related to the volume density of gas (ρ_{g}) than the surface density of gas (Σ_{g}) for a thin but non-uniform disc such as the Galaxy. The conventional wisdom has, however, been to look for an empirical power-law relationship between the SFR ($\frac{\psi_{g}}{v_{g}}$ or $\frac{\psi_{y}}{v_{g}}$) and the corresponding density of gas, that is

$$\Psi_{s} \quad e \in \Sigma_{H_{2}}^{k} \qquad (2.1)$$

$$\Psi_{v} \quad e \in \rho_{H_{2}}^{d} \qquad (2.2)$$

where Ψ_{S} is the SFR in $M_{\odot}pc^{-2}$ Gy⁻¹, Ξ_{S} is the surface density of the total gas (atomic and molecular) in $M_{\odot}pc^{-2}$, Ψ_{V} is the SFR in $M_{\odot}pc^{-3}$ Gy⁻¹, ρ_{S} is the volume density of the total gas in $M_{\odot}pc^{-3}$, and k and <u>a</u> are the corresponding exponents.

Schmidt (1959, 1963) made an analysis of the variation of star formation with height above the Galactic midplane

and proposed the relationship (2.2) with $\underline{a} \cong 2$ (see also Mathis 1959; Salpeter 1959). This 'law' was supported on the basis of a theoretical treatment of fragmentation by Field and Saslaw (1965). Sanduleak (1969) proposed that in the case of a thin disc, $\underline{\mathbb{S}}$ could be used to represent ρ , and there then followed many observational tests of the law in local galaxies (van Hoerner 1960; van Genderen 1969; Hartwick 1971; Einsato 1972; Emerson 1974; Madore et al. 1974; Hamajima and Tosa 1975; Azzopardi and Vigneau 1977; Smith et al. 1978; Bruck 1980). The correlations obtained were not good, and the value of k had a wide range, 0.5 -3.5 for different determinations and galaxies.

There are a number of objections to the above 'Schmidt law' of the SFR. Firstly, Madore (1977) pointed out that in many observational tests the observed quantity is the number of stars formed, not the SFR, and showed that this led to an overestimate of k. Secondly, the fact that k appears to vary within and among galaxies indicates that the total gas density is not the only relevant quantity (Larson 1977; Tinsley 1980). More recently, Donas and Deharveng (1984) and Donas et al. (1987), although finding some evidence for a correlation between total SFR and total HI mass, find only a very weak correlation in terms of surface densities. In particular the Small Magellanic Cloud (SMC) is exceedingly gas rich and yet produces only about 20% of the mass of stars per unit mass of gas that the Large Magellanic Cloud does (Lequeux 1984). It must be pointed out that all of the above determinations have simply used HI as the total gas, neglecting H_2 , a point that will be discussed shortly.

The final reason for rejecting the Schmidt law concerns

-23-

the apparent constancy of the SFR with time in this and other spiral galaxies. Miller and Scalo (1979), from a detailed study of the mass spectra of stars in the solar neighbourhood, in order to fulfill the constraint that the initial mass function (IMF) be continuous, required $rac{4}{8}$ to be roughly constant over the age of the Galaxy. Although subsequent studies of the IMF have weakened this continuity constraint (Scalo 1986; Larson 1986; Rana 1987a and Chapter 4), the constancy of $\psi_{\mathbb{S}}$ with time in the solar neighbourhood has been supported by the luminosity function of white dwarfs (Liebert et al. 1979; Rana 1987b), the distribution of initial masses of planetary nebulae (Tinsley 1978), the metallicity distribution and velocity distribution of nearby F and G stars (Twarog 1980; Vader and de Jong 1981; Meusinger 1985; Fuchs and Wielen 1987) and data on stellar lithium abundances and calcium emission line strengths (Scalo 1986). It also seems that star formation has proceeded at a relatively constant rate over the lifetime of most late-type disk galaxies (Searle et al. 1973; Mayor and Martinet 1977; Rocca-Volmerange et al. 1981; Kennicutt 1983b; Gallagher et al. 1984; Sandage 1986; Hunter and Gallagher 1986), while Tinsley and Danly (1980) argued for a constant v_{s} from the point of view of star formation in the Universe. Now comparing the total mass of the Galactic disc to the present mass of the gas, it is seen that the mass of gas (assuming that at the time of the formation of the disc, the majority of the mass was in the form of gas) has been depleted by a factor of ≈ 10 . Relationship (2.1) with k ≥ 1 therefore predicts a very sharp decrease of the SFR with time which contradicts the above results.

-24-

In the light of these objections, a number of different alternatives have been explored. Some workers have ignored the first two objections and have continued using relations (2.1) and (2.2) (e.g. Smith et al. 1978; Güsten and Mezger 1982; Caimmi and Dallaporta 1982; Chiosi and Matteucci 1982, 1984; Güsten 1986; Matteucci and Greggio 1986; Arimoto and Yoshii 1986, 1987). With this form for ψ_S , infall of gas from the halo to the disc has been invoked to keep \mathbb{Z}_S and therefore ψ_0 roughly constant with time (e.g. Lacey and Fall 1983, 1985; Clayton 1984, 1985a,b, 1986, 1987).

Other workers have suggested that the star formation rate depends not only on gas density, but the gas density multiplied by some other quantity such as the frequency with which a parcel of gas encounters a density wave (Talbot and Arnett 1975), the volume of gas (Caimmi 1978), the dust-to-gas ratio (Viallefond et al. 1982), the velocity dispersion of the gas (Brosche and Leutes 1985), the temperature of the gas (Caimmi and Secco 1986) and the total mass (i.e. gas and stars) of a region (Dopita 1985). The model of Dopita (1985) has received particular attention (e.g. Maddox 1985), but once again neglects H_2 , both in theory and in correlation plots.

There have also been many who have rejected the empirical forms (2.1) and (2.2), and inserted a SFR independent of gas density into models of chemical or photometric evolution (e.g. Tinsley 1980; Tosi 1982). Either a constant $\frac{1}{28}$ has been taken (e.g. Twarog 1980; Twarog and Wheeler 1982, 1987) or the time evolution left as a free parameter, varied to fit observational results (e.g. Larson and Tinsley 1978; Tosi and Diaz 1985). Recently,

-25-

Larson (1986) has used bimodal star formation in order to attempt to solve the dark matter problem in the solar neighbourhood, the basic idea being that massive and low mass stars are formed in distinct locations in space and/or time. Using the constraint on the high mass mode that the model should predict enough mass in remnants to account for the unseen matter near the Sun, he finds a SFR which decays with time as exp(-t/3.4 Gy). Wyse and Silk (1987) have extended this argument, having stars > 2 M_® forming at an exponentially declining rate, while lower mass stars form at a constant rate. Güsten and Mezger (1982) had also proposed a bimodal IMF, massive stars forming primarily in spiral arms, while all masses of stars forming in the inter-arm regions. Indeed there is growing observational evidence for bimodal star formation, such as the discontinuous slope of the IMF (Larson 1986; Rana 1987a), the observation of nearby dark clouds with a high abundance of low mass pre-main sequence stars but virtually no high mass stars (Scoville and Sanders 1987) and that the largest HII regions and most massive GMCs are confined to spiral arms (Dame et al. 1986).

However, none of the above formulations explicitly take account of molecular hydrogen. It is well known that the giant molecular complexes are the primary seats of star formation, and a simple law linking the star formation rate to the density of H_2 would be intuitively satisfying. Talbot (1980) investigated the relationship of the SFR to the density of gas assuming the emission per unit area from ionised hydrogen to be proportional to the number of massive stars and therefore to the rate of formation of massive stars $\frac{1}{10}$. He then plotted $\frac{1}{10}$ against the surface densities

-26-

of total gas, atomic hydrogen ($\Sigma_{M\Sigma}$) and molecular hydrogen (Σ_{M_2}) and concluded that the rate of formation of massive stars is more significantly correlated with Σ_{M_2} than with $\Sigma_{M\Sigma}$ or even $\overline{\Sigma}_3$, for both the Galaxy and M83. Subsequently DeGoia-Eastwood et al. (1984) confirmed this conclusion from a study of NGC 6946. In the light of the contemporary debate concerning the distribution of H_2 , the following sections re-examine this conclusion and show how a SFR proportional to the surface density of H_2 solves quite naturally the initial objections to the Schmidt law.

2.3 STAR FORMATION RATE AND GAS DENSITY IN THE GALAXY

2.3.1 SFR in relation to the surface density of gas

The analysis of Talbot (1980) for the Galaxy has been repeated using different available distributions of \mathbb{Z}_{ME} , \mathbb{Z}_{M_2} and \mathbb{Z}_{3} , and an independent SFR taken from Smith, Biermann and Mezger (1978, hereafter referred to as SBM). The total gas \mathbb{Z}_{3} is defined as

 $\Sigma_{Q} = x^{-1} [\Sigma_{M_{Z}} + \Sigma_{M\Sigma}],$ (2.3) X being the chemical mass fraction of hydrogen in the ISM ($x^{-1} = 1.36$). SBM derive a SFR from the rate of emission of Lyman continuum photons from giant HII regions. Other estimates of the SFR have been made by Mezger (1978) and Güsten and Mezger (1982) using different assumptions concerning the lifetime of HII regions and spiral structure but a recent compilation of SFRs based on supernova remnants and pulsars (Lacey and Fall 1985) seems to favour SBM rather than the other two. Talbot (1980) has systematically underestimated the SFR and the radial dependence is also

-27-
somewhat different, compared with that of SBM.

Table 2.1 shows the results of plotting the SFR ($\sqrt[n]{3}$) against different distributions of \mathbb{Z}_{MC} , \mathbb{Z}_{M_0} and \mathbb{Z}_{3} and using

> Table 2.1: Correlation of SFR and surface densities of gas. (SFR from SBM 1978)

I _g in	relation	Source of	data	k	^R c
(2.1)	replaced by				

^E HI	Li et al. (1982)	-7.3 <u>+</u> 2.4	-0.74
	Burton & Gordon (1978)	3.5 <u>+</u> 1.6	0.61
^۲ н ₂	Bhat et al. (1985)	1.6 <u>+</u> 0.3	0.90
£	SSS (1984)	1.1 <u>+</u> 0.2	0.91

 Image: Kg g g
 H2
 HI

 Bhat et al. (1985)
 Li et al. (1982)
 -0.9±2.5
 -0.13

 SSS (1984)
 Burton & Gordon (1978)
 2.7±0.7
 0.81

 Bhat et al. (1985)
 Burton & Gordon (1978)
 3.8±0.8
 0.85

a least squares fit to calculate k, the standard errors and the corresponding correlation coefficients R_c . The two surface density distributions for H_2 , SSS (1984) and Bhat et al. (1985) are used to illustrate the sensitivity of the results. The surface density distributions, scale heights and SFR are given in Table A.2.1 (Appendix). The corresponding plots for $\frac{4}{5}$ against these surface density distributions, after eliminating their explicit dependence on Galactocentric radius (R_c) between 4.5 and 14.5kpc are shown in Figure 2.1. Of course the statistics of the data are not very good as at most only 10 points were available for each plot, which means that the derived value of R_c should not be taken too seriously. Nevertheless a number of factors may be noted in view of the above analysis.

-28-



Star formation rate as given by SBM is plotted against the various surface density distributions of gas (given in Table A.2.1) after eliminating $R_{\mathfrak{G}}$. $\mathbb{Z}_{\mathfrak{M}}$ and $\mathbb{Z}_{\mathfrak{S}}$ show virtually no correlation, but both the H₃ distributions show some degree of correlation with k = 1.1\pm0.3 for SSS (1984), and k = 1.6\pm0.3 for (1984), and $k = 1.6\pm0.3$ for the Bhat et al. (1985) H_2 distribution. Figure 2.1:

-29-

(i) Even with the uncertainties involved, $\sqrt[4]{S}$ seems to be better correlated with $\mathbb{Z}_{M_{\mathfrak{A}}}$ than \mathbb{Z}_{g} or $\mathbb{Z}_{M\mathbb{X}}$, as found earlier by Talbot using independent data. The lack of any correlation with HI is also found for external galaxies (Kennicutt and Kent 1983; Hunter et al. 1982) and is consistent with the observation that the HI distribution is often extended relative to the places where stars are currently forming (Bosma 1981b). The correlation with H_{2} is further strengthened by the observation that stars form in dense H_{χ} clouds (e.g. Pudritz 1986; Myers et al. 1986; Waller et al. 1987) and the z-dependence of $H_{\rm R}$ is similar to that of young stars (Fich and Blitz 1982; SSS 1984). In fact, within 3 kpc of the Sun only one GMC has been found without traces of star formation (Blitz 1987), indicating that once a GMC forms, star formation is almost inevitable and takes place very quickly.

(ii) Although the estimated distributions in total gas are widely different, no relationship of the form $\psi_s \ll \Sigma_g^k$ seems to exist, which is consistent with the poor correlations and widely different estimates of k made earlier for external galaxies (Section 2.2). Fall (1987) has cautioned, however, that present uncertainties in the estimation of ψ_s and Σ_g do not exclude $\psi_s \ll \Sigma_g^k$.

(iii) A relationship of the form $\frac{1}{2} \ll \frac{1}{2}^k$, if existing at all, will perhaps be relevant only to Σ_{M_2} . But, uncertainties mean that at the present stage nothing conclusive can be said about the more acceptable of the surface density distributions of H_2 as they finally lead only to different values of k, with roughly the same

-30-

correlation coefficient. The difference in the value of k will however, become apparent in the chemical evolution model of Chapter 3.

(iv) Bhat et al. (1984) had observed that their $\mathbb{Z}_{M_{2}}$ as a function of R_{g} is very similar to that of HII regions (Lockman 1976), pulsars (Taylor 1979) and supernova remnants (Huang and Thaddeus 1986), all having a peak around $R_{g} = 6$ kpc. In addition to which OH/IR stars (Baud et al. 1979) and cool giant stars (Ito et al. 1977) also have a peak at $R_{g} \cong 6$ kpc. All of these Population I tracers reflect enhanced star formation in the recent past (Lester et al. 1985) and the further link between the SFR and distributions reinforces that all these processes are physically related.

It must be stressed that the above surface densities of gas and their ensuing conclusions are preliminary as the gas distributions used for $H_{\mathfrak{P}}$ are averaged over the incomplete rings around the centre of the Galaxy. In particular the measurements of $\Sigma_{M_{\mathcal{P}}}$ outside the solar circle given by SSS (and therefore derived by Bhat et al.) are incomplete, being only for the northern declinations. Recent results (Robinson et al. 1984) may suggest more H_g in the outer Galaxy than that given by SSS (1984). Therefore an average of the SFRs given by Lacey and Fall (1985) and SBM (1978) was taken and a correlation looked for in the inner Galaxy where the data are complete. Figure 2.2 shows the result, the points in the left half of the Figure corresponding to the outer Galaxy, the dotted lines being extrapolations from the inner Galaxy. It can be easily seen that the points in the outer Galaxy do not fit the extrapolation, possibly due

-31-



Figure 2.2: SFRs averaged from SBM (1978) and Lacey and Fall (1985) plotted against the surface density of molecular hydrogen. A good correlation (R_c = 0.91) is found for the inner Galaxy (shown by the solid line) with k = 0.7 ± 0.2 for the SSS distribution and 1.2 ± 0.2 for the Bhat et al. distribution. If the line is extrapolated to the outer Galaxy (dashed line) where the H₂ data are incomplete, the values of Σ_{H_2} seem to be underestimates.

to an underestimate of \mathbb{Z}_{\aleph_2} . This is an important point which will be investigated further in Chapter 4.

This more detailed study shows roughly the same correlation, but a reduction in the value of k to 0.7 \pm 0.2 for the SSS distribution and 1.2 \pm 0.2 for the Bhat et al. distribution of Σ_{M_2} .

2.3.2 SFR in relation to the volume density of gas

Guibert et al. (1978) studied such a relationship as given by equation (2.2) and concluded that the exponent <u>a</u> is 1.2 - 2.5. At present such an analysis can be extended only for the incomplete gas distributions, using scale heights for H₂ from SSS (1984) and for HI from Kulkarni et al. (1982). From $\frac{4}{5}$ as given by SBM, one can possibly derive $\frac{4}{5}$, assuming that the scale heights for $\frac{4}{5}$ are the same as

-32-

those for the H_2 distributions, because it is known that star formation takes place primarily, if not necessarily, inside the molecular complexes. The results of fitting a relationship (2.2) to the different distributions of ρ_{M_2} , ρ_{MX} and ρ_{R} are shown in Table 2.2. The values of the

Table 2.2: Correlation of SFR and volume densities of gas.

	(SFR from	SBM 1978)		
ρ_g in relation	Source of data		<u>a</u>	Rc
(2.2) replaced by				
٩HI	Li et al. (1982)	4.6 <u>+</u> 1.8	0.66	
	Burton & Gordon (1978)		3.0 <u>+</u> 0.6	0.86
^р н _о	Bhat et al. (1985)	1.5 <u>+</u> 0.3	0.89	
6	SSS (1984)	1.0 <u>+</u> 0.2	0.90	
Рд	^H 2	ні		
	Bhat et al. (1985)	Li et al. (1982)	3.1 <u>+</u> 0.6	0.86
	SSS (1984)	Burton & Gordon (1978)	1.7 <u>+</u> 0.4	0.83

exponent \underline{a} and the correlation coefficient R_{c} , derived from least-squares fitting are calculated as before.

Bhat et al. (1985) Burton & Gordon (1978) 2.2+0.3 0.93

It can be seen that, as for the surface densities, $\frac{4}{10}$ is once again correlated with ρ_{M_2} rather than ρ_{M_2} . The values of <u>a</u> and <u>k</u> are similar as the scale heights of H₂ stay roughly constant over the range of R₆ considered. Therefore, most of the conclusions that are noted previously from Table 2.1 are still valid in the present context. Fits are found to be equally good for both the SSS and Bhat et al. volume density distributions of H₂, but leading to two different values of <u>a</u>, namely, 1.0 <u>+</u> 0.2 and 1.5 <u>+</u> 0.3 respectively. However, the volume density of the total gas

-33-

is also well correlated with $\frac{1}{10}$. This is largely due to the small contribution of HI (as its scale height is appreciably larger) compared to that of H₂, which means that the distribution of H₂ 'controls' the ρ_3 distribution and hence $\frac{1}{10}$ is primarily controlled by H₂ densities. The large value of <u>a</u> for the ρ_3 correlation is highly inconsistent with the fact that the SFR has stayed fairly constant as has been argued previously.

Scoville et al. (1986) claim that locally the number density of giant HII regions is proportional to the square of the density of H_2 , and suggest that this implies that OB stars form as a result of cloud-cloud collisions (Brosche 1970; Larson 1977). They also suggest that lower mass stars show a linear correlation implying that they are formed from individual clouds. The results of Table 2.1 and 2.2 possibly suggest a gross average over both high mass and low mass star formation with respect to H_2 .

The conclusion of the above analysis is that if a phenomenological relationship of star formation rate as a power of gas density exists, then it is only relevant to H_2 . Between the surface densities and volume densities the existing data do not make a sharp distinction. Mathematically, equations (2.1) and (2.2) will be compatible only if $k = \underline{a} = 1$ or if the scale-height is constant with R_6 . Some may feel that physically the SFR should be related to the volume density rather than surface density. But the surface density gives the mass of gas in a particular region and therefore the total number of clouds. As Clayton (1986, 1987) has pointed out, if star formation proceeds at a fixed rate in molecular clouds then it is the number of clouds

-34-

that will determine the SFR. The surface density is to be preferred to the volume density also for practical reasons. Any measurement of ρ_0 depends strongly on the spatial resolution of the observations (Tinsley 1980). In external galaxies, the observations measure \mathbb{Z} directly, while a knowledge of the scale-height is needed for ρ . As the scale-height is not well known, \mathbb{Z} is a much better quantity to work with in the study of external spiral galaxies. The equations of chemical evolution (Chapter 3) can also be solved more naturally for surface densities. The rest of this work will therefore adopt a SFR expressed in terms of the surface density of H₂, that is

 $\psi_{s} \ll \sum_{M_{2}}^{K}$ (2.4) where a larger value of k is favoured by the Bhat et al. (1985) distribution than the SSS (1984) distribution of H₂.

Finally, the delineation of spiral arms by active regions of star formation has led many workers to believe that spiral arms are an important trigger of star formation in galaxies (e.g. Baade and Mayall 1951; Roberts 1969; Toomre 1981; Gerola and Seiden 1978). However, Elmegreen (1987c) has questioned this view, arguing that star formation simply follows the gas with no preferential trigger related to the density wave. Comparisons between galaxies with and without density waves reveal no significant differences in the SFR or metal abundances (Elmegreen and Elmegreen 1986; McCall 1986). From a study of supernovae in external galaxies (McCall and Schmidt 1986) it appears that density waves do not enhance the efficiency of massive star formation, while some irregular galaxies have high $\frac{1}{2}$ but no spiral arms (Hunter and Gallagher 1986).

-35-

Elmegreen (1987c) suggests that spiral arms simply re-organise the matter within the discs without actually enhancing the SFR. As McCall (1986) points out, "if the underlying details of density waves are indeed inconsequential to star formation the task of identifying the physical properties controlling the chemical evolution of spiral galaxies is greatly simplified". No doubt the real situation is quite complex, but over large scales and long time-scales a relationship such as (2.4) may be applicable.

2.4 THE TIME EVOLUTION OF THE SFR IN THE GALAXY

It has been shown above that there is an empirical correlation in the disc of the Galaxy between the SFR and the surface density of molecular hydrogen, to be understood at a simple physical level that it is from ${\rm H}_{\rm R}$ that stars form. The next question is: can equation (2.4), derived empirically from the radial distributions, be used to predict the time evolution of the SFR? Any such law of star formation is in any case possibly a smoothed average over stochastically occurring bursts, but can it predict the gross variation of the SFR over the age of the Galaxy? Although it has been known for some years that stars form out of H₂, and Talbot (1980) proposed the relationship (2.4), the problem faced by evolutionary models is how to deal with the evolution of $H_{\mathcal{R}}$ quantitatively. In chemical evolution models for example, the absolute values of the SFR are important for studying the age-metallicity relation for stars in the solar neighbourhood (Twarog 1980) while its

-36-

radial distribution with time will manifest itself in shaping the abundance gradient of the processed elements in the Galaxy.

Molecular clouds in the Galaxy differ from the diffuse atomic clouds in that they are optically thick to UV radiation and visible continuum and that they are self gravitating. Their formation has been ascribed to shock compression in the ambient gas, agglomeration of smaller clouds and spontaneous instabilities in the gaseous disc (e.g. Elmegreen 1987a,b). These formation mechanisms seem to explain some of the observed cloud complexes in the Galaxy, although there is still no general agreement on which is the dominant mechanism. In addition, the current formation mechanisms do not at present distinguish between H_g and HI. The lifetime of GMCs is another controversial issue, which will be examined in detail in Chapter 6. At this time, it seems very difficult to link these processes to the global time and radial variation of H_g in the Galaxy.

Wyse (1986) has attempted to understand the difference in H_2 and HI profiles in galaxies in terms of enhanced GMC formation in local gravitational potential perturbations such as spiral arms, proposing

 $\Sigma_{M_2} \ll \Sigma_{M_1}^2 \cdot (\Omega(R_G) - constant)$ (2.5) where $\Omega(R_G)$ is the local angular frequency. If CO is a good tracer of H₂, using a constant ω_{20} , this form provides good fits for the Galaxy, NGC 2841, M51 and M31 but not too good for M101. It also explains the steep fall off of the molecular gas in the outer regions reflecting that of the angular frequency. However, Stark et al. (1987) have shown no obvious correlation between the surface brightness of CO

-37-

and arm class for over 100 spiral galaxies, which suggests that spiral arms are not required to form H_2 in a galaxy. From their data it appears that molecular clouds form continuously, whether or not a spiral density wave is present in a galaxy. In addition, apart from the assumption of a constant \mathscr{C}_{20} in this law, the main problem against it is its time evolution. In order to solve analytically the equations of chemical evolution, Σ_{M_2} needs to be expressed as a function of the total mass of gas rather than Σ_{M_3} . In order to do this the following argument will attempt to connect the amount of H_2 to Σ_9 through the abundance of metals in the ISM.

If the SFR has not changed appreciably since the formation of the disc (Section 2.2), one must argue that, in spite of the constant depletion of H_2 because of continued star formation, the amount of H_2 has somehow been continually replenished. It is known that HI continually evolves into ${\rm H}_{\chi}$ in the presence of grains (Savage and Mathis 1979; Duley and Williams 1984) and this may be the reason. The formation of H₂ is critically dependent on the availability of interstellar grains, unless the density of gas is as high as 10^{9} H atoms cm⁻³ (Williams 1985). Theoretical studies by Hollenbach and Salpeter (1971) showed that approximately one-third of H atoms striking grains formed ${\rm H}_{\rm S}$ and observational support has been put forward by O'Donnel and Watson (1974) and Jura (1975). It is also known that for catalysis of H2 formation, the grains must have a temperature of approximately 20K (Schmidt 1967). Recent infra-red studies at 60 and 100 µm (Burton et al. 1986) reveal that interstellar grains do have a temperature

-38-

20-25K in the plane, but also the surprising result that the scale-height of the dust is nearer that of HI than H_{2} (Burton and Deul 1987). This latter result does not fit the general picture of H_{2} formation and is not easy to



Figure 2.3: A qualitative scheme of re-cycling HI through H_2 in the ISM. Molecular clouds evolve into stars which eject metals and HI into the ISM. Metals segregate into grains which catalyse formation of H_2 . This H_2 finally agglomerates into molecular clouds, thus completing the cycle. Every generation of star formation leaves some stars behind, dead or alive.

understand unless there is some temperature effect perpendicular to the plane. Measurements at larger wavelengths, where more of the dust is sampled, are needed before firm conclusions can be drawn. It is these uncertainties in temperature as well as the actual number and size distributions of the grains at various distances from the Galactic Centre that make it a formidable task to rigorously handle the process of $HI \longrightarrow H_2$ conversion in the ISM. A further uncertainty involves the estimation of the so-called returned fraction which essentially determines how much matter in the form of HI is returned in the end to the ISM, assuming that all H_2 went initially into star

-39-

formation. Therefore only a qualitative scheme of recycling HI through H_2 is presented in Figure 2.3.

Knowing that grains do play a vital role in forming H_{g} from HI, let us investigate whether the fractional abundance (p_{g}) of H_{g} in the total amount of available hydrogen, in a given region, is related to the average metallicity (Z) of that region. Actually a link should be sought with the abundance of grains rather than metallicity in the gas phase, but the abundance of grains is assumed to roughly follow the metallicity (Viallefond et al. 1982; Franco and Cox 1986). In fact metals in the gas phase may themselves be important in forming H_{g} clouds as they act as coolants (van den Bergh 1981) and increase opacity (Franco and Cox 1986). Following the discussion of Section (2.3.2), P_{g} is defined in terms of the surface densities

$$p_{2} = \frac{\Sigma_{M_{2}}}{(\Sigma_{M_{2}} + \Sigma_{M\Sigma})}$$
(2.6)

The available distributions of Σ_{MZ} (as given in Table A.2.1) have been averaged to derive $\overline{\Sigma}_{MZ}$ and a probable range of error in its estimates. Then substituting this value in (2.6), $\overline{p_2}$ has been computed for both the SSS and Bhat et al. distribution of H_2 . The variation of p_2 with Galactocentric radius is shown in Figure 2.4, the errors reflecting the uncertainties in HI. Both of the distributions behave similarly, although the absolute values are quite different. The fraction of molecular gas flattens out below $R_{c_1} = 6-7$ kpc and falls off towards the outer edge. The sharp change in the trend just above $R_{c_2} = 10$ kpc could be due to the underestimate of H_2 in the outer Galaxy (Section 2.3.1).

Elmegreen and Elmegreen (1987b) have noticed a similar

-40-

radial dependence of this molecular fraction in individual



Figure 2.4: The variation of the molecular fraction of gas (\overline{p}_2) with Galactocentric radius R. Two estimates are made of \overline{p}_2 for the different distributions of H₂, as given by SSS (1984) and Bhat et al. (1985). The forms are found to be similar although the absolute values are quite different.

'super cloud' complexes, consisting of HI and H_2 , with masses $\geq 10^6 M_{\odot}$. The total mass of the complexes appears to be constant as a function of R_{c} , which suggests that the molecular cores to these clouds become less massive with increasing R_{c} , while their atomic envelopes become more massive. This is ascribed to an increase in the average pressure of the clouds in the inner Galaxy, leading to larger densities and molecular line shielding of the background UV radiation becoming more important. However, one may expect increased metallicity (and/or grains) to have the same effect (Section 6.2).

Figure 2.5 shows the variation of metallicity or more precisely log(O/H)+12, with R_G , essentially compiling all

-41-

the available data from a number of sources. There are few



Figure 2.5: A compilation of the observational data on the variation of oxygen abundance measurements with Galactocentric radius. The objects comprise HII regions, planetary nebulae and some main sequence stars in the solar neighbourhood. The gradient from a least squares fit for $R_{G} > 6$ kpc is shown and is $\simeq -0.09 + 0.01$ dex kpc⁻¹.

observations for R_{G} < 6kpc. Although the individual estimates have a considerable spread in the ordinate there seems to exist a general trend of metallicity with R_{G} , with a gradient

$$d \frac{\log(O/H)}{dR_{fo}} = -0.09 + 0.01 \text{ dex kpc}^{-1}$$
(2.7)

extending between $R_{G} = 6$ and 14 kpc. Abundance gradient determinations by different authors using different objects are compiled in Table A.2.2 (Appendix). There are large uncertainties in abundance determinations from HII regions and this is probably the main cause for the spread in metallicity at any particular R_{G} . However, there may be

-42-

also some genuine evolutionary effects. Stars themselves (as shown by the solar neighbourhood) show an increase in metallicity with time of formation, while the metallicity spread in HII regions could be due to the locking of metals into grains (e.g. Mathis 1986). For planetary nebulae, Peimbert (1978) argues that for most the O enrichment due to their evolution has not been considerable and that differences are related to gradients present in the ISM at the time of formation of the progenitor stars (Peimbert and Serrano 1980). This will be discussed in more detail in Chapter 3.

Assuming that $Z \ll O/H$, O being the major contributor to the metallicity (Peimbert and Torres-Peimbert 1974; Peimbert et al. 1986), p₂ is investigated as a function of Z, aiming to fit a relationship of the form

$$p_2 \ll z^b$$
 (2.8)

In Figure 2.6, $\overline{p_2}$ is plotted against metallicity after eliminating R_{c_1} , for 6 < R_{c_2} < 14 kpc only. Errors shown are only formal errors and could be larger due to systematic effects.

Considering the errors involved, fairly good correlations are obtained for least-squares fits between the data and equation (2.8), yielding the exponent b as

 $b = 1.3 \pm 0.3$ for the Bhat et al. distribution of H₂,

= 1.4 \pm 0.3 for the SSS distribution of H₂. They now agree well with each other within the large error bounds. If there is more H₂ in the outer Galaxy than presently estimated, as suggested earlier, this will have the effect of reducing b although due to errors in HI this will be a small effect. Preliminary though this correlation

-43-



Figure 2.6: The molecular gas fraction \overline{p}_2 is plotted against the metallicity (assuming Z C O/H), for $6 < R_G < 14$ kpc in the Galactic disc. Both H₂ distributions are shown and a correlation is found for $\overline{p}_2 \ll Z^b$, with b = 1.3-1.4.

between the molecular hydrogen fraction and metallicity is, it would nevertheless mean that metallicity may not monotonically increase right up to the centre of the Galaxy and it may eventually flatten off for $R_{\oplus} < 6$ kpc as the Galactocentric variation of $\overrightarrow{p_2}$ indicates. The metallicity in the very central regions is highly uncertain, in the range 12 + log(O/H) = 9.0 - 9.6 (Mezger et al. 1979; Shaver et al. 1979; Wink et al. 1983) but a value of $Z \cong 3Z_{\odot}$, i.e. a flattening out of the gradient, is not unreasonable (Güsten and Ungerechts 1985).

Now, using (2.8), a more explicit functional form of the SFR may be obtained as follows. Let μ be defined as the gas fraction of the total mass available in the form of gas and stars per unit area of the disc

$$\mu = \frac{\Sigma_{\rm q}}{\Sigma_{\rm r}} \tag{2.9}$$

where $\Sigma_{\gamma} = \Sigma_{g} + \Sigma_{s}^{\dagger}$, (2.10)

 Σ_{s}' being the total surface density of matter locked in stars, dead or alive. The measured rotation curve of the Galaxy yields Σ_{T} (the estimated values are given in Table A.2.1 (Appendix)). Combining equations (2.4), (2.6), (2.8) and (2.9) one obtains

$$\Psi_{S} \circ \mathcal{L} (\Sigma_{T} \times_{\mu} Z^{b})^{k} .$$
 (2.11)

The SFR given by (2.11) has the following significance. (i) This form of the SFR has primarily been derived from phenomenological considerations of the spatial variations of various quantities with Galactocentric radius. But the form (2.11) is now such that it can be used to study the time variation of $\frac{1}{\sqrt{2}}$ for any given annulus, provided that the evolution of μ , $\overline{\omega}_{\gamma}$, X and Z with time is known. When the disc first stabilised, its rotation curve and therefore the

-45-

distribution of \mathbb{Z}_{γ} presumably also stabilised (except for local variations due to the propagation of density waves). The chemical fraction of hydrogen, X, is a very slowly decreasing function of time. So for a given region, such as that near the Sun, assuming no infall, $\Sigma_{T} X$ may be regarded as a constant. Consequently, the evolution of $\frac{1}{2}$ in that region may be studied by the evolution of the quantity $(\mu z^{\flat})^{\natural}$. At the time of formation of the disc, the gas fraction $\mu_0 \leqslant 1$ and non-zero. In order therefore for a non-zero $\[mathscale{\%}\]$, the initial metallicity of the disc, Z $_{\!0}$, cannot be indefinitely small, a suggestion that was first made in the context of the chemical evolution of the solar neighbourhood (Truran and Cameron 1971). In fact, if 🖔 remains constant with time and the gas is depleted at most by a factor of 10, log (Z_{1}/Z_{0}) cannot exceed \cong 0.7 $(Z_{1}$ being the present metallicity in the solar neighbourhood.) (ii) With the ageing of the disc, μ decreases whereas Z increases due to star birth and death (van den Bergh 1958), thereby effectively balancing one another and leading to a constant SFR. Equation (2.11) is then a modified Schmidt In fact Talbot and Arnett (1973) in attempting to law. model the chemical evolution of the solar neighbourhood had suggested that star formation was enhanced by the presence of metals and had multiplied the mass of gas by a term dependent on the metallicity in the standard Schmidt law. The metallicity term in equation (2.11) is due to the consideration of molecular hydrogen in the role of star formation.

(iii) Since the value for the exponent b in (2.8) has been derived for 6 < R_{c_r} < 14 kpc one must be careful in extending

-46-

the application of (2.11) outside the above domain. However, presumably when the disc was formed, μ_{\odot} , X_{\odot} and Z_{\odot} were constant as a function of R_{\oplus} . It was therefore the variation of Σ_{Υ} with respect to R_{\odot} that determined the initial value of the SFR, $(\frac{N}{2})_{\odot}$. Since Z_{Υ} increases sharply towards the centre of the Galaxy, $(\frac{N}{2})_{\odot}$ should have been very high in the inner part of the Galaxy - thus leading to faster evolution there and thereby initiating the future metallicity gradient as well as a sharp gradient in μ . Clayton (1987) has re-emphasised the point that a large abundance gradient cannot be modelled if $\frac{N}{2}/Z_{\odot}$ is a constant and so uses infall of gas to the disc to produce the gradient. The SFR given by (2.11) predicts a non-constant $\frac{1}{N}/Z_{\odot}$ with radius and time and therefore will lead to an abundance gradient.

(iv) The main problem of the original Schmidt law, that of a constant SFR, can be overcome quite naturally by the time dependence of $(\mu Z^{b})^{k}$. Previous workers (Larson 1972; Clayton 1984, 1985, 1986; Lacey and Fall 1985) with a SFR proportional to gas density, needed infall to show why the SFR has practically stayed constant over the age of the Galaxy. However, a SFR proportional to a power of the surface density of H₂ resolves this problem without infall.

The above qualitative points will be examined quantitatively in the context of chemical evolution in Chapters 3 and 4. But before calculating it in detail for the Galaxy, evidence from other spiral galaxies needs to be examined in order to determine how universal the application of equation (2.11) can be.

-47-

2.5 STAR FORMATION IN EXTERNAL SPIRAL GALAXIES

2.5.1 Star formation rate and molecular hydrogen

The detection of CO in external spiral galaxies (Rickard et al. 1975; Solomon and de Zafra 1975) has allowed investigation of the star forming component of the ISM in these galaxies. However, just as in the case of the Galaxy, the problem is the conversion factor to ${\rm H}_{\rm 2}$, and just how much H₂ is really there. In a sense, knowledge of α_{20} for external galaxies is even more problematic than for the Galaxy, as the U-ray method (Section 1.2.2) cannot (yet) be used to calibrate \$20. Many past attempts (e.g. Young and Scoville 1982a) have assumed that the local value of α_{20} applies universally, thereby determining H_2 . Although Young and Sanders (1986) have argued that this method is correct to a factor of 2, many have questioned this (Rickard and Blitz 1985; Bhat et al. 1986; Lo et al. 1987a; Stark et al. 1987) even to the extent of suggesting that radial distributions of H_2 in galaxies cannot be determined from the CO data with any confidence at all (Blitz 1985).

In the light of this uncertainty concerning α_{20} and the effects of temperature and metallicity, any rigorous testing of $\oint_{S} \propto \sum_{M_2}^{k}$ and in particular the value of the exponent k is fraught with difficulty. However, we do see a qualitative correspondence, certainly between CO (and probably H₂) and the tracers of recent star formation.

The Population I tracers which are extremely similar radially to CO in the Galaxy (Mihalas and Binney 1981) have the same trend in M51 (Scoville 1983) and M33 (Freedman

-48-

1986). Young and Scoville (1982a,b) have shown that the CO luminosity (L_{CO}) follows that of the blue luminosity (L_{CO}) with radius in NGC 6946 and IC 342, and the existence of a linear correlation between L_{CO} and L_{CO} for the central 5 kpc of Sc galaxies, both isolated and in the Virgo cluster (Young et al. 1985b; Young 1985a). If the CO traces the H₂ and L_{CO} is taken to be mostly from Population I stars, thereby indicating the amount of star formation over the last 2.10^O y (Scoville and Young 1983) this implies k \cong 1 in the $\frac{V}{V}$ of $\Sigma_{M_2}^{k}$ relation. This correlation has been discussed extensively by Bhat et al. (1986) who argue that the correlation is improved by introducing a metallicity correction to α_{20} . From their Figures 13 and 14, the effect of a metallicity correction implies k > 1, in agreement with the Galactic analysis (Section 2.3.1).

A possibly more direct measure of the young stellar content of a galaxy is the infra-red luminosity (e.g. Telesco et al. 1986). It has been argued that the far infra-red luminosity gives a direct measure of the energy output from recently formed stars (Cox et al. 1986; Bushouse 1986) based on the fact that Galactic regions of star formation are always associated with dusty and relatively dense molecular clouds, with nearly all the luminosity emerging in the FIR (e.g. Knapp et al. 1977). Therefore, as expected, CO is found to correlate with the FIR luminosity, within galaxies (Scoville and Young 1983) and between galaxies (e.g. Young et al. 1985a). If the FIR luminosity is a measure of $\frac{44}{5}$, then Rickard et al. (1985) found k = 1.48 in the centres of nearby spirals. The availability of IR fluxes from the IRAS satellite for external galaxies has

-49-

seen a study of CO and IR emission in dwarf irregular galaxies (Tacconi and Young 1987), isolated and interacting galaxies (Rickard and Harvey 1984; Rengarajan and Verma 1986; Young et al. 1986b), IRAS bright galaxies (Young et al. 1986a; Sanders et al. 1986b), radio bright spirals (Sanders and Mirabel 1985) and ultra-luminous IR galaxies (Sanders et al. 1987). These studies have confirmed the correlation between IR and CO over six orders of magnitude, but there is a wide dispersion, which appears to be due in part to temperature effects (Young et al. 1986a; Solomon et al. 1987b) or variations in the dust-to-gas ratio (Rydbeck 1985). Interacting galaxies seem to have a higher efficiency of star formation (higher IR luminosity for the same CO) than isolated galaxies, but both seem to follow the same trend with k ☆ 0.75 (Young et al. 1986b).

Figure 2.7 summarises the situation for the IRAS bright galaxy sample of Young et al. (1986a) and the isolated and interacting galaxy sample of Young et al. (1986b). The mass of H₂ has been calculated from maps of the CO emission using $\alpha_{20} = 8$. The SFR, $\frac{4}{5}$, is calculated following Scoville and Young (1983) using

$$(M_{\odot}y^{-1}) = 7.7.10^{-11} (L_{\Sigma R} + L_{R})$$
 (2.12)

where the blue luminosity $(L_{\textcircled{0}})$ has been derived from RC2 (de Vaucouleurs, de Vaucouleurs and Corwin 1976) and the total FIR luminosity $(L_{\fbox{0}})$ is given by

 $L_{IR} (L_{\odot}) = 4.10^{5} CD^{2} (2.58S_{\odot} + S_{\odot}). \qquad (2.13)$ In the above equation, S_{\odot} and S_{\odot} are the 60µm and 100µm fluxes (in Jy), D is the distance to the galaxy (in Mpc),

and the correction factor C is taken from the IRAS

-50-

Extragalactic Catalogue (Lonsdale et al. 1985). As Young et al. (1986a) point out, these estimated SFRs do not account for either the formation of low mass stars (as it assumed the observed luminosities are produced primarily by O,B and A stars) or for the re-cycling of gas in the ISM, although to some extent these two effects may balance each other. For galaxies with Hs flux measurements (Kennicutt and Kent 1983), the above $\frac{10}{5}$ has been compared to the $\frac{10}{5}$ derived from Hs (Kennicutt 1983b) and found to agree within 50%.



Figure 2.7: SFR determined from the blue and infra-red luminosities plotted against the total mass of H₂ (using $\alpha_{20} = 8$, H₀ = 50kms⁻¹ Mpc⁻¹) for IRAS bright galaxies (Young et al. 1986a) and isolated and interacting galaxies (Young et al. 1986b). Excluding the interacting galaxies a good correlation is found (R_c = 0.92) with slope = 0.60 + 0.05. Arrows show the effect of a metallicity correction to α_{20} having the effect of increasing the slope.

Excluding the interacting galaxy sample, a good correlation

is found (correlation coefficient = 0.92) with k = 0.6 in $\psi_0 \ll M_{M_2}^{(k)}$. The interacting galaxies seem to show the same trend but have systematically higher SFRs indicating a different efficiency (cf. Kennicutt et al. 1987). The arrows on certain points illustrate the effect of introducing a metallicity correction to \ll_{20} (Bhat et al. 1984; 1985) where metallicities have been taken from a variety of sources (Matteucci and Chiosi 1983; Hunter et al. 1982; McCall et al. 1985; Hunter and Gallagher 1986). With only a few metallicities available for these galaxies, it can only be said that a metallicity correction seems to increase k. The Galactic analysis gave k = 0.7 - 1.4, which is consistent with the above value.

Finally, as mentioned in Section 2.3.1, the H& emission which is an indicator of the star formation rate correlates with CO for the Galaxy and for M51 (Young 1985a). Talbot (1980) using a metallicity corrected H₂ distribution found $k \simeq 1.2$ for M83, while DeGoia-Eastwood et al. (1984) found $k \simeq 0.8$ for NGC 6946 in the $\sqrt[6]{s} \ll \sum_{M_2}^{K}$ relation.

Therefore, apart from large uncertainties concerning the estimation of $\frac{1}{5}$ and $\frac{1}{2}_{M_2}$ in external galaxies, $\frac{1}{5} \ll \frac{1}{2}_{M_2}^k$ is certainly consistent with present data with k in the range 0.6 - 1.5 (this range being possibly entirely due to uncertainties in the estimation of $\frac{1}{5}_{S}$ and $\frac{1}{2}_{M_2}$).

2.5.2 Molecular hydrogen and metallicity

In Section 2.4, a relationship of the form $p_2 \ll Z^{b}$ was found for the Galaxy, with $b = 1.3 \pm 0.3$. This Section compiles the available data on the radial distributions of HI, H₂ and (O/H) for nine other spiral galaxies in order to see whether this relationship is applicable in external

-52-

galaxies.

The references from which the data have been taken are shown in Table A.2.3 (Appendix). The HI distributions are taken from the sources listed in the second column of Table A.2.3, as used by Diaz and Tosi (1984) and Tosi and Diaz (1985) with their adopted distances. The H_{2} distributions are inferred from the observed CO distributions along the respective galactic discs, except M33 where the CO distribution was assumed to follow the blue luminosity profile (Diaz and Tosi 1984) and normalised according to the CO nuclear detection of Young and Scoville (1982b). Two possible radial H_2 distributions are derived because of the controversy surrounding \mathfrak{A}_{20} . The first \mathbb{Z}_{\aleph_2} (S) follows SSS (1984) assuming a constant $\alpha_{20} = 7.2$. The second Σ_{N_2} (B) follows Bhat et al. (1984, 1985, 1986), reducing \$200 in the solar neighbourhood, and introducing a metallicity correction, M_{χ} , so that

$$\sum_{M_2} (B) = \sum_{M_2} (S) \frac{2.7}{7.2} \cdot \frac{1}{M_z}$$
(2.14)

where log $M_Z = 12 + \log (O/H) - 8.9$ (2.15) The data on the galactocentric gradients of (O/H) used in deriving M_Z are primarily those calibrated and used by Diaz and Tosi (1984) and Tosi and Diaz (1985). Dopita and Evans (1986) have recently examined the semi-empirical abundance diagnostic ratios used to determine metal abundances (Pagel et al. 1978; Edmunds and Pagel 1984a; McCall et al. 1985) and suggested lower abundances particularly for the most metal rich HII regions (see also Pagel 1986c and Evans 1986). In Figure A.2.4 (Appendix), the calibration curves for the oxygen abundance against the ratio ([OII] + [OIII])/H β are shown. But until there is agreement, this

-53-

work will follow Tosi and Diaz (1985) using the curve of Edmunds and Pagel (1984a).

Having obtained two alternative distributions of \mathbb{Z}_{W_2} , the corresponding p_2 values were calculated using the same distribution of \mathbb{Z}_{W_2} . Each metallicity measurement at a particular galactocentric radius of any particular spiral galaxy, with its quoted error where possible has been plotted against p_2 for that radius and galaxy, so eliminating the explicit radial dependence of the same quantities for individual galaxies. This is shown in Figure 2.8 where the calculation of the two sets of p_2 has been based on $\mathbb{Z}_{W_2}(S)$ and $\mathbb{Z}_{W_2}(B)$ respectively. From this Figure the following points may be noted:

The points shown seem to have a large spread, (i) nevertheless a general trend is apparent. For a higher observed metallicity, the higher is the molecular fraction of the gas in the respective region of the galaxy. This means that metal poor spirals will have in general little H_{ij} in comparison with HI. This may be the reason why gas rich irregulars (Elmegreen et al. 1980; Tacconi and Young 1985), irregular dwarf galaxies (Israel and Burton 1986) and the LMC and SMC (Israel 1984) show extremely weak CO signals. These galaxies have low gas metallicities (Rocca-Volmerange 1984; Hunter and Gallagher 1986) and this could be the reason for the apparent low H_2/HI ratio. In fact, Tacconi and Young (1987) find that dwarf irregular galaxies have low $\rm H_{g}\,/\rm HI$ ratios and qualitatively ascribe this as possibly due to low metallicity. However, the situation is complicated by the fact that α_{20} may be dependent on metallicity and other factors so that low CO emission does not necessarily

-54-



Figure 2.8: The fraction of molecular hydrogen, $p_2 = \sum_{M_2} / (\sum_{M_2} + \sum_{M_3})$ plotted against the metallicity for different regions of various spiral galaxies. The upper plot uses a constant α_{20} in determining \sum_{M_2} , while the lower plot uses a metallicity dependent α_{20} . Shown on both plots are the corresponding $p_2 \ll Z^{\flat}$ relation deduced for the Galaxy (Figure 2.6).

imply low H₂ content (Young et al. 1984; Bhat et al. 1984; Hunter and Gallagher 1986; Israel et al. 1986; Maloney and Black 1987; Thronson et al. 1987a).

(ii) Shown on both plots is the corresponding $p_2 \ll 2^{\circ}$ relation deduced for the Galaxy (Section 2.4). Data from the other spirals seem to confirm this trend. In fact if the data are taken together the value of b = 1.35 \pm 0.10 for p_2 (S) and b = 1.30 \pm 0.10 for p_2 (B) are obtained, in agreement with our previous determination of b in the Galaxy. There is a suggestion that the trend may flatten off at 12 + log(O/H) > 9.1, but the errors are large and if in fact the new calibration of Dopita and Evans (1986) is used (Figure A.2.4) the high metallicity measurements are reduced.

(iii) It should be noted that the errors in the above compilation of data are rather large. They reflect the controversial aspects of the $CO \rightarrow H_2$ conversion ratio and the large uncertainties in the derivation of the O/H ratios for the HII regions. It could even be argued that the above correlation is due to a possible correlation between the intensity of CO emission and the abundance of oxygen. However, that the correlation between p_2 and Z still holds, and in fact improves for the metallicity corrected H_2 , suggests against this possibility.

Therefore, this preliminary study of external spiral galaxies seems to support the empirical law of star formation of equation (2.11). At present the empirical correlations do not distinguish between the competing distributions of H_2 , although the Bhat et al. (1985) type of distribution has a marginally better correlation. However,

-56-

the application of this empirical law in models of the chemical evolution of the Galaxy, presented in Chapters 3 and 4, will show the importance of the powers k and b, from which the distributions can be distinguished. To conclude this Chapter, the dust-to-gas ratio in spiral galaxies is briefly examined, in order to confirm the above law of star formation rate, but also to possibly constrain \mathscr{O}_{20} .

2.5.3 The dust-to-gas ratio

Young et al. (1986a) have used the 60 μ m and 100 μ m fluxes from IRAS observations of external galaxies to estimate the mass of dust (M₀) in those galaxies. They found M₀ to be correlated with the mass of H₂ in a relation M_{M₂} \propto M₀^{1.2}, but also concluded that variations in the relative amounts of HI and H₂ in a galaxy are not obviously correlated with any other galaxy properties.

This work has previously emphasised the importance of the role of dust in the evolution of $HI\longrightarrow H_2$. In the Galaxy it was more practical to use the metallicity (assuming it to be proportional to the dust) in correlating against the fraction of molecular gas in the total gas. However, for external galaxies the situation is almost reversed. Very few, apart from nearby spirals have published metallicities, whereas many have had 60μ m and 100μ m fluxes measured by the IRAS satellite. The calculation of the mass of the dust however, has large uncertainties.

Following Young et al. (1986a) one first assumes a λ^{-1} emissivity law and calculates the dust temperature, T_{\bigcirc} , using the 60 /100 µm flux ratio. The physical interpretation of this temperature when referring to a whole galaxy is not clear (Thronson et al. 1987b), but in the Young et al.

-57-

(1986a) sample T_{\odot} is in the range 30-46K. The following includes the colour correction to the temperature (King 1986) and uses the fluxes as given by Young et al. (1986a). The 100 μ m flux can now be used to calculate the mass of warm dust. Longer wavelengths are required to sample the colder dust. The mass of dust from a flux density S_{\Im} at frequency \Im is given by

$$M_{\mathbb{D}} = Q_{\mathbb{V}}^{\dagger} \left[\frac{S_{\mathbb{V}} D^2}{B(\mathbb{V}, T_0)} \right]$$
(2.16)

where Ω_{∇}^{l} is taken from Hildebrand (1983) and B(ϑ , T_D) is the blackbody intensity.

The suggestion of Section 2.3 is that the fraction of molecular hydrogen should be correlated in some way with the amount of dust. Defining p_2 this time in terms of masses rather than surface densities, and taking HI and H₂ masses (with $@(_{20} = 8)$ from Young et al. (1986a), p_2 has been calculated for the IRAS bright galaxy sample. Total galactic masses (M_T) and HI masses for the isolated galaxy sample of Young et al. (1986b) have been taken from the literature (Rogstad and Shostak 1972; Bosma et al. 1977; Shostak 1978; Crutcher et al. 1978; Reakes 1980; Fisher and Tully 1981; Bottinelli et al. 1982; Huchtmeier 1982; Hunter et al. 1982; Matteucci and Chiosi 1983; Rubin et al. 1982; Hunter and Gallagher 1986).

No correlation is found for p_2 against M_{D}/M_{T} , but a rough correlation is found for p_2 against M_{D} . However a much better correlation is found for p_2 against the dust-to-gas ratio (M_{D}/M_{Q}) , where of course

 $M_{g} = 1.36 \ (M_{W_{2}} + M_{MX}),$ (2.17) the factor 1.36 accounting for He. This is shown in Figure 2.9. Although once again the scatter is large, the general

-58-



Figure 2.9: The fraction of molecular hydrogen, $p_2 = M_{M_2}/(M_{M_2} + M_{M_2})$, as a function of the dust-to-total gas ratio for the IRAS bright galaxies of Young et al. (1986a) and the isolated galaxies of Young et al. (1986b). The mass of dust is calculated from the 100µm flux and therefore only represents the warm dust component. Arrows represent the effect of metallicity correction to M_{M_2} .

trend is that the greater the dust-to-gas ratio of a galaxy, the more gas is in the form of molecular hydrogen. Of course this is the $M_{W_2} \ll M_D^{1.2}$ trend found by Young et al. (1986a), but if confirmed would suggest that the relative amounts of HI and H₂ in a galaxy are determined by the dust-to-gas ratio. Even more interesting, ignoring the possibly important distinction between surface densities and total integrated mass, this is the general trend of relationship predicted by $p_2 \ll z^b$ if Z is proportional to M_p/M_q . A gradient of 1.3 has been fitted by eye to the points in Figure 2.9 to illustrate this. In fact the effect of metallicity corrections to M_{W_2} (shown by arrows in the Figure) in general improves the correlation.

However, the biggest drawback to this analysis is that the 100 µm flux does not sample all of the dust in a galaxy

-59-

(Young et al. 1986a). This can be readily seen from Figure (2.9) as all galaxies are at least an order of magnitude below the canonical dust-to-gas ratio of 10^{-2} determined in the solar neighbourhood (Spitzer 1978).

Chini et al. (1986) have recently published 1300 µm measurements of 26 spiral galaxies. The 1300 µm fluxes are more sensitive to colder dust and therefore should sample more of the dust than 100 µm. Chini et al. (1986) argue that their observations cannot be explained by dust emitting at a uniform temperature, and adopt a warm and cold component model with corresponding temperatures T_W and T_c . Following Chini et al., the dust mass for these galaxies is calculated using ${\tt T}_{\rm C}$ and the 1300 $\mu{\tt m}$ flux. Measurements of CO emission, either total or for the central area, were available for nineteen of these galaxies (Young et al. 1983; Rickard et al. 1985; Sanders and Mirabel 1985; Verter 1985; Kenney and Young 1986; Young et al. 1986a,b; Sanders et al. 1986b, 1987; Knapp et al. 1987) and HI masses were taken from the previously described literature. There are now two problems, one is to convert the CO emission for certain galaxies which do not have maps but only one central measurement to total CO emission, and secondly to then reduce the gas distributions to the area sampled by the 90" beam of the 1300µm measurements. In the absence of detailed HI and H_2 maps, a crude attempt has been made to do this. It is well known that in general HI distributions are all very similar and have relatively flat profiles over the disc (Rogstad and Shostak 1972), apart from a few galaxies with pronounced low R_{G} 'holes', and therefore HI masses are scaled by the ratio of the area of the beam width to the

-60-

optical size of the galaxy. However, in many spiral galaxies CO is peaked and follows the exponential luminosity profile (Young 1985a). The CO emission is therefore assumed to fall off exponentially with a scale-length of 5 kpc (Young and Scoville 1982a). This has been used by Young et al. (1985a) and discussed extensively by Verter (1983) who concludes that this type of extrapolation is correct to within a factor of 2-3. Thus, taking account of the inclination of the galaxy, integrating over the beam width of the CO observation and over the entire galaxy, the fraction of CO emission in the 90" beam is calculated. The mass of H_2 for that area is then determined using $C_{20} = 8$ (following Young et al. 1986a).

The dust-to-gas ratio as determined from the 1300µm flux in the central 90" of a galaxy is shown as a function



Figure 2.10: The dust-to-gas ratio as determined from 1300µm flux measurements for the galaxies of Chini et al. (1986), as a function of the fraction of molecular gas to the total gas. The dust-to-gas ratio for the solar neighbourhood is also shown.

of p_2 in Figure (2.10). The following points may be noted from this Figure:

(i) There is no evidence of the correlation with p_2 seen

for the warm dust. This could possibly be due to only the warm dust taking an active role in H_2 formation.

(ii) With $\mathfrak{A}_{29} = 8$, the central 90° of these galaxies is dominated by H_2 , assuming an exponential profile. (iii) The distribution of dust-to-gas ratios has a mean value close to the solar neighbourhood value suggesting that the 1300µm samples most of the dust, but even so, spans an order of magnitude. Therefore the use of a constant dust-to-gas ratio must be questioned.

However, in the central regions of spiral galaxies we (iv) may expect a dust-to-gas ratio, 2-3 times higher than the solar value. If the dust-to-gas ratio follows the metallicity (see Appendix A.2.5), the dust-to-gas ratio in the centre of the Galaxy should be ≈ 3 times the solar value. In this case only 3 or 4 galaxies in Figure 2.10 seem to have 'normal' dust-to-gas ratios. If 🖾 20 < 8 and not constant then the other galaxies could be increased in their dust-to-gas ratio. For instance in order for NGC 660 to have $M_0/M_q > 2.10^{-2}$ in its central regions $\alpha_{20} < 1$. It seems therefore for M_{\odot}/M_{q} to be constant in different galaxies, \$20 must vary from galaxy to galaxy. Alternatively a constant α_{20} implies a large variation in M_{0}/M_{q} .

(v) The uncertainties in the above analysis are very large. A factor of at least 2 or 3 is involved in the assumption of an exponential profile of the H_2 distribution. Detailed radial maps for both HI and H_2 for the Chini et al. (1986) sample of galaxies are required in order to improve this aspect. The use of T_{W} rather than T_c in the calculation of the mass of dust, decreases the dust mass by at least 50%.

-62-

Once again more 1300 mm measurements are required both of these galaxies and more to enlarge the sample.

In conclusion, a study of the dust-to-gas ratio in external spiral galaxies gives at present ambiguous evidence for the claim that the amount of dust controls the relative amounts of HI and H_2 in a galaxy. However, there seems to be more evidence, either for a variable $@_{20}$ or a variable dust-to-gas ratio from galaxy to galaxy (or both). More observations, especially at 1300 μ m are needed to confirm this. This subject will be further examined in the context of the Virgo cluster of galaxies in Chapter 6.
CHAPTER THREE

GALACTIC CHEMICAL EVOLUTION I - THE MODEL

3.1 INTRODUCTION

The chemical evolution of galaxies studies the chemical processing of hydrogen and helium through stars to produce metals (defined as all elements heavier than helium). At an early epoch galaxies condensed out of a primordial gas which consisted by mass of \sim 78% H, \sim 22% He and traces of D, ³He and Li. This chemical composition is quantitatively explained by nuclear reactions taking place in the early universe (Yang et al. 1979, 1984). At the present time galaxies consist of ~70% H, ~28% He and ~2% heavier elements including C, N and O. It is this process of enrichment of metals and their distribution with which chemical evolution studies are concerned. Burbidge et al. (1957) showed that this chemical evolution could be explained as a byproduct of nuclear reactions in stars which form out of the ISM and in their final evolutionary stages return heavier elements to the ISM. Therefore it depends critically on the properties of the constituent stars, their mass spectrum, formation rates and mass and composition of the processed material returned to the ISM. As these questions cover such wide areas, a wealth of literature and models have grown up since the pioneering work of Schmidt (1959, 1963). The subject has been comprehensively reviewed many times (e.g. Lynden-Bell 1975; Tinsley 1980; Pagel 1981,

-64-

1986a, 1987; Chiosi and Jones 1983; Güsten and Mezger 1983; Mould 1984; Peimbert 1985; Güsten 1986) and such a review is outside the scope of the present work. This Chapter discusses a self consistent model of the chemical evolution of the Galactic disc component based on the adopted SFR of $\psi_S \ll \Xi_{M_2}^k$ and shows how such a model can explain the G-dwarf problem, the age-metallicity relation of disc stars, the constancy of the SFR and the metallicity gradient. The consistency of the model and its consequences particularly for the nature of the dark matter and distribution of H₂ will be discussed in Chapter 4.

3.2 FORMULATION

3.2.1 The Simple Model

Any model of Galactic chemical evolution is to some extent reliant on a number of important assumptions, these assumptions being partially justified by comparison with observation (Pagel and Patchett 1975). The so called simple model (Searle and Sargent 1972) assumes:

- (i) The gas is chemically homogeneous at all times.
- (ii) A region can be modelled as a closed system.
- (iii) The system has started with 100 per cent metal-free gas.
- (iv) The initial mass function of stars is invariant.

Every one of these assumptions has since been questioned but the simple model still provides a good framework for study.

Two further assumptions will be used below. Firstly, it is assumed to first order that spiral discs are

-65-

axisymmetric, allowing the disc to be divided up into a set of concentric annuli with uniform properties around each annulus. Stars are born on roughly circular orbits and therefore their migration has a negligible effect on chemical evolution and the processed gas follows the trajectory of the progenitor. Consideration of a disc annulus is an accepted assumption (Clayton 1986) based on the idea that mixing within the annulus by differential rotation and turbulence may homogenise the ISM within it whereas angular momentum barriers inhibit radial mixing.

The second assumption is commonly referred to as the instantaneous recycling approximation which assumes that stars dominating chemical evolution evolve so rapidly in comparison with Galactic timescales that ejecta are immediately returned to the ISM (Talbot and Arnett 1971; Searle and Sargent 1972). This greatly simplifies the calculation and removes the need of explicitly specifying the initial mass function. Clayton and Pantelaki (1986) have shown that this simplification is surprisingly satisfactory unless one is interested in the very early evolution when the differential evolution times of the first stars matter or in late evolution when the mass of gas is so small that it is dominated by return from small old dwarfs (Pagel and Patchett 1975). The instantaneous recycling approximation is no longer useful in the study of time dependent abundance ratios or when the SFR is a strongly decreasing function of time (Tinsley 1980) or for predicting integrated colours of galaxies (Arimoto and Yoshii 1986). However, for the discussion below instantaneous recycling is adequate and allows analytic solutions, following the spirit of Clayton

-66-

(1986) that 'nothing illustrates the ideas of chemical evolution of galaxies as well as a system of differential equations plus their solutions'.

Following Tinsley (1980, 1981) the basic analytical equations for simple models of galactic chemical evolution, using the instantaneous recycling approximation, take the following form,

 $\Sigma_{T} = \Sigma_{g} + \Sigma_{s} + \Sigma_{p}, \qquad (3.1)$

$$\mu = \Sigma_{0} / \Sigma_{T})$$

$$\sigma = \Sigma_{s} / \Sigma_{T})$$

$$\delta = \Sigma_{0} / \Sigma_{T} ,)$$
(3.2)

$$\frac{\mathrm{d}\Sigma_{g}}{\mathrm{d}t} = - (1-R)\psi_{g}, \qquad (3.3)$$

$$\frac{d\mathbb{Z}_{S}}{dt} = (1-R-D)\psi_{S}, \qquad (3.4)$$

$$\frac{d\Sigma_0}{dt} = D\psi_S, \qquad (3.5)$$

and
$$\frac{d(Z\Sigma_{g})}{dt} = -Z(1-R)\psi_{g} + (1-R-D)y_{\chi}\psi_{g}$$
, (3.6)

where Σ_{Γ} is the total mass density of the system, Σ_{S} is the total mass density locked up in visible stars, Σ_{S} is the total mass density of gas and ψ_{S} is the star formation rate (in $M_{\odot} pc^{-2} Gy^{-1}$).

The so-called returned fraction R is the fraction of matter that is returned in the form of gas to the interstellar medium by a generation of stars compared with the total initial mass in that generation that went into stars. Following Tinsley (1981) and Rana and Wilkinson (1986a, hereafter referred to as Paper I) the above equations also include an explicit expression for dark matter. Previous formulations (e.g. Tinsley 1980) do not make any distinction between the visible stars and invisible or so-called dark remnants. Observations for the G-dwarf

metallicity distribution or the age-metallicity relation of the disc stars are relevant only to visible stars. If the proportion between the dark remnants and the visible stars varies from place to place or from time to time, it is better to express the dark matter explicitly. Also there may be an amount of pre-disc dark matter incorporated into the disc. Therefore, D represents the dark matter fraction.

Since R has already been defined to be the returned fraction, the fraction (1-R) represents the mass locked into stars visible or invisible. The latter fraction is now divided into two parts (1-R-D) as the visible stellar fraction and D the dark matter fraction (for example, cold dwarfs, neutron stars and black holes). The mass density of dark matter is now $\Sigma_{\mathbb{D}}$ and μ , σ and \mathfrak{S} the fractions of gas, visible stars and dark matter compared to the total mass of the region.

The metallicity by mass fraction of the interstellar gas (Z) is linked to the mass of gas and SFR through the yield (y_{Z}) defined as the ratio of total mass of new metals ejected by a generation of stars and the total mass of visible stars. It should be noted that the above yield, defined for analytic convenience, differs from the standard definition of the yield by the relationship

$$y_{Z} = \frac{(1-R)}{(1-R-D)} y'$$
 (3.7)

where y' is the value used by all previous workers (e.g. Tinsley 1980).

For completeness it is necessary to define some of the above parameters which depend on the initial mass function (IMF), p(m). This frequency distribution of stellar masses at birth is usually normalised according to

-68-

$$\int_{m_0}^{m_{\rm W}} (t) m_0(m) dm = 1$$
 (3.8)

where $m_{\hat{l}}$ and $m_{\underline{M}}$ represent the lower and upper mass limits of the range of possible stellar masses, m being the mass of a star in solar units. The returned fraction is therefore defined by

$$R(t) = \int_{m_0}^{m_M} (t) (m - w_m) \mathcal{D}(m) dm \qquad (3.9)$$

where m_1 is the turn-off stellar mass corresponding to the main sequence lifetime of the star bounded by the age of the disc t, and $w_{\overline{m}}$ is the consequent remnant mass for a white dwarf, neutron star or black hole depending on the value of intial mass of the star.

The dark matter fraction is given by

$$D(t) = \int_{m_2(t)}^{m_M(t)} w_m \phi(m) dm + \int_{m_l(t)}^{m_0} m \phi(m) dm \qquad (3.10)$$

 m_2 being the mass above which the star collapses only to a neutron star or black hole, but not a white dwarf and m_0 being the mass limit below which the star ends up as a brown or cold dwarf. The yield is therefore given by

$$y_{\chi} = \frac{1}{(1-R-D)} \int_{m_{1}}^{m_{M}(t)} m_{\chi_{m}} \phi(m) dm$$
 (3.11)

where p_{Zm} is the mass fraction of a star of mass m that is converted to metals and rejected.

Usually, R, D, y_Z and ψ_S are externally specified, either ad hoc or by assuming specific forms for $\mathcal{O}(m)$, remnant masses, lifetimes, production of metals and SFRs. Basically one attempts to solve equations (3.3) and (3.6) for $\mathbb{Z}_{9}(t)$ and Z(t) respectively using the rest as auxiliaries.

3.2.2 Additions to the simple model

In view of difficulties the simple model had in reproducing the observations of the G-dwarf problem and abundance gradients, each of the inherent assumptions of the model have been questioned:

In their model of metal enhanced star formation Talbot (i) and Arnett (1973) used inhomogeneities in the gas on the scale of individual star forming regions in an attempt to solve the G-dwarf problem. But in order to work this requires large inhomogeneities in the gas when the metallicity is very low which is difficult to obtain. It has been argued that disc annuli are not closed (ii) systems and that gas flows are important in the chemical evolution (Larson 1972). Dynamical collapse models of spiral galaxy formation suggested the gradual and delayed build up of the disc by settling halo gas, this gas acting as infall of metal poor gas onto the disc (Tinsley and Larson 1978). It has also been claimed that gas could be accreted from the general inter-galactic or inter-group medium (Hunt 1971; Gunn and Gott 1972). Lynden-Bell (1975) and Tinsley (1975) showed that infall could solve the G-dwarf problem in the solar neighbourhood and the effect has been incorporated into a large number of investigations of chemical evolution (e.g. Fowler 1972; Searle 1972; Sciama 1972; Quirk and Tinsley 1973; Biermann and Tinsley 1974; Audouze and Tinsley 1976; Chiosi 1980; Chiosi and Matteucci 1982; Peimbert and Serrano 1982; Tosi 1982; Yokoi et al. 1982; Yoshii 1984).

However, one of the problems with infall is that it is not fixed observationally. Oort's (1970) interpretation of

-70-

high velocity clouds (HVC) as evidence for infall implying an infall rate of $1-2M_{\odot}y^{-1}$ was strongly contested (Davies 1974; Verschur 1975). Bregman (1980) instead interpreted HVC within a Galactic fountain model where hot gas rises up several kpc from the disc and then condenses into clouds which fall back to the disc. Of course, if clouds return to their original position the standard equations need no modification. If the clouds rementer at larger R_{ff} then this would amount to a radial gas flow. Güsten (1986) argues that their chemical composition, which within uncertainty is consistent with local disc abundances, and their velocity field imply that HVC are probably not related to the early halo gas at all. Mirabel and Morras (1984) have argued that at least some of the clouds do represent infall of a total rate $\approx 0.2 M_{\odot} y^{\neg}$ to the disc but the situation is far from clear (cf. van Woerden et al. 1985; Kaelbe et al. 1985).

Diffuse gas would almost certainly be highly ionised and emit X-rays in a cooling flow as it approached the disc. From a set of detailed hydrodynamical calculations and observations of the soft X-ray background, Cox and Smith (1976) derived the present infall rate in the solar neighbourhood to be $\leq 1M_{\odot}pc^{-2}Gy^{-1}$. It therefore seems that even if infall was important in the past it has in fact faded away, which calls into question temporally constant infall models (e.g. Tosi 1982; Diaz and Tosi 1984; Tosi and Diaz 1985; Twarog and Wheeler 1982, 1987). In fact, in order to avoid this problem, decaying infall rates have been invoked (Vader and de Jong 1981; Güsten and Mezger 1982; Lacey and Fall 1983, 1985; Clayton 1984, 1985a,b, 1987) with even the suggestion of terminated infall or periodic infall

-71-

(Clayton 1986). Further evidence against infall being a global event in the evolution of disk galaxies stems from the failure to detect directly the source or reservoir of infalling gas in external spirals (Bothun 1985).

The evidence for infall has primarily been indirect, that is it was claimed it was needed to solve chemical evolution problems. Chapter 2 has already pointed out that with $\bigvee_{S} \Subset \, \widehat{\mathbb{Z}}_{\mathsf{M}_{9}}^{\, \mathsf{K}}$, infall is not required to explain the constancy of the SFR over the age of the Galaxy. There were suggestions (Larson 1972; Twarog 1980; Larson et al. 1980; Tinsley 1980) that the present constant SFR in spiral galaxies implied that the ISM would be used up on too short a timescale if there was not gas replenishment by infall. This problem however can be easily dealt with also by $\psi_{s} \in \Sigma_{M_{s}}^{K}$. Both of these points will be shown explicitly in Section 3.6. (It should be noted that the bimodal model of star formation of Larson (1986) also avoids these problems removing the need for infall). Although problems in the solar neighbourhood can be solved by infall alone (Twarog 1980), abundance gradients across the disc are much more difficult. Tosi and Diaz (1985) have claimed good fits to the abundance gradients using infall, but other workers using more realistic SFRs and infall rates have needed supplementary assumptions such as variable yield (Peimbert and Serrano 1982; Güsten and Mezger 1982) or radial gas flows (Lacey and Fall 1985) for the same fits.

In fact the situation may be quite complex with radial gas flows (Mayor and Vigroux 1981; Lacey and Fall 1985) and gas outflows from the disc (Hartwick 1976; Vigroux et al. 1981; Wyse and Silk 1985) being important. Arimoto and

-72-

Yoshii (1986, 1987) have extensively studied the sudden ejection of gas heated by supernovae explosions from elliptical galaxies and shown good agreement with photometric and chemical evolution.

In view of the above discussion, the interesting question is, can the Galactic disc be successfully modelled as a closed system, neglecting the effect of infall, outflow and radial gas flows? Of course the neglect of gas flows simply means the assumption that the rate of flow is small compared to the SFR (Tinsley 1977). After all, the disc has to initially form by accreting matter from the surroundings. A case without infall may be interpreted as to imply a small time constant for the formation of a stable disc. In fact from dynamical considerations a galactic disc can form as quickly as $\sim 2.10^{3}$ y (Vader and de Jong 1981; Burkert and Hensler 1987). Therefore, in the following, the Galactic disc will be assumed to form quickly and then evolve as a closed system with gas flow rates small compared to the SFR. The third assumption of the simple model may not be (iii) correct if the gas which forms the disc was pre-enriched in metals by a previous generation of stars outside the disc (Truran and Cameron 1971). Chapter 2 pointed out that for the new formulation of the SFR to be non zero at t = 0, an initial metallicity Z_{ω} was required. This point will be discussed more extensively in the context of the G-dwarf problem (Section 3.3). However, assuming a Z_{0} and gas fraction μ_{0} (\leq 1) when the disc formed, the derived SFR of Chapter 2 can be written as

$$\Psi_{S}(R_{G},t) = C \left[\Sigma_{\gamma}(R_{G}) X(R_{G}) \underline{\mu}(R_{G},t) \left(\frac{Z(R_{G},t)}{Z_{O}} \right)^{b} \right]^{b}$$
(3.12)

-73-

showing the explicit dependence on time and Galactocentric radius, C being a constant. It will be the properties of this SFR that will be studied in the following chemical evolution model.

(iv) It has been argued that in order to successfully model the chemical evolution of the disc, the initial mass function of stars varies (Schmidt 1963). Many dependences have been proposed such as relatively more massive stars produced at smaller R₆ (Quirk and Tinsley 1973; Güsten and Mezger 1982) or at low Z (Terlevich and Melnick 1984; Melnick and Terlevich 1986) or at high Z (Peimbert and Serrano 1982). (Of course a high Z_{Θ} may be interpreted as a variable IMF between disc and halo stars). Such a variation in the IMF leads inevitably to a variable yield, an addition to the simple model which has been used recently by a number of workers. Peimbert and Serrano (1982) proposed that the yield is linearly related to the metallicity, Güsten and Mezger (1982) used a yield dependent on Galactocentric radius and Edmunds and Pagel (1984a) suggested that the yield was a function of the total mass surface density. Paper I proposed an ad hoc assumption of the yield proportional to the metallicity. In this work, this assumption will be generalised to

$$y_{z} = KZ^{n}$$
(3.13)

where K is a constant and n is a power, with n = 0corresponding to the simple model (with finite initial metallicity) and n = 1 corresponding to the assumption of Paper I. (In fact, Caimmi (1978) had previously suggested $y' \ll z^{0.5-1.0}$ for the halo population). The equations of chemical evolution will be derived for all values of n but

will be solved only for the cases n = 0 and n = 1 for comparison. The strict validity of this assumption and in particular the value of n from the point of view of the IMF and stellar nucleosynthesis will be discussed in detail in Chapter 4. It needs to be emphasised that although a variable IMF implies a variable yield, a variable yield does not necessarily imply a variable IMF. The variation in the yield could be due to details of stellar nucleosynthesis.

It is now easy to see that equations (3.1) - (3.13) will allow the following solutions

$$\underline{\mathcal{M}} = 1 - \frac{(1-R)}{(1-R-D)} \left(\frac{\underline{\mathcal{O}} - \underline{\mathcal{O}}_{\underline{\mathcal{O}}}}{\underline{\mathcal{M}}_{\underline{\mathcal{O}}}} \right)$$
(3.14)

$$\delta - \delta_0 = \frac{D}{(1 - R - D)} (\sigma - \sigma_0) = \frac{D}{(1 - R)} (\mu_0 - \mu)$$
 (3.15)

For convenience G is defined as

$$G = \frac{(1-R)}{K(1-R-D)}$$
(3.16)

and then the fraction of gas is related to the metallicity by

$$\mu = \mu_0 \exp \left[\frac{G}{(1-n)} \left(Z_0^{1-n} - Z^{1-n} \right) \right] \qquad n \neq 1 \qquad (3.17a)$$

$$\mu = \mu_0 \left(\frac{Z}{Z_0}\right)^{-6} \qquad n=1 \qquad (3.17b)$$

All the quantities with subscript '0' represent the initial values. Finally, the age-metallicity equation is given by

$$\frac{dZ}{dt} = \frac{C(1-R)}{G\mu_0} \sum_{T}^{k-1} \frac{z^{kb + n}}{Z_0} \exp\left[\frac{(k-1)}{(1-n)} G(Z_0^{1-n} - Z^{1-n})\right] \qquad n \neq 1 \quad (3.18a)$$

$$\frac{dZ}{dt} = C\left(\frac{1-R}{G\mu_0}\sum_{v}^{k-v}\left(\frac{Z}{Z_0}\right)^{-G(k-1)+k_0} \qquad n=1 \quad (3.18b)$$

In the following sections these equations will be applied to problems in the solar neighbourhood and across the Galactic disc.

-75-

3.3 THE G-DWARF PROBLEM AND FINITE INITIAL METALLICITY

van den Bergh (1962) and Schmidt (1963) originally discovered a problem concerning the metallicity distribution of long-lived disc dwarfs in the solar neighbourhood. The problem is that there are far fewer metal poor stars than are predicted by the simple model of Galactic chemical evolution. This is termed 'the G-dwarf problem' and has proved to be one of the most powerful constraints on models for the solar neighbourhood. The simple model (for example, following Pagel and Patchett 1975) predicts the fraction of stars (visible and invisible, however) with metallicities not exceeding Z, to be given by

$$S(Z) = \frac{M_{S_1}}{M_{S_2}} = \left(\frac{1 - \mu_A}{(1 - \mu_A)}\right)$$
(3.19)

where the subscript '1' (and henceforth everywhere) refers to the value of the parameter at present, $M_{\$}$ being the visible stellar mass. This expression is obtained on the assumption that $Z_0 = 0$ and $y_z = \text{constant}$. The observational data and the above prediction are shown in Figure 3.1. The metallicity curve deviates quite significantly at the lower end of the range. It is relatively insensitive to the value taken for the gas fraction μ_1 (Tinsley 1980) so that the controversy over H_2 estimates in the solar neighbourhood (Chapter 1) does not provide a solution. A similar metallicity distribution has since been confirmed for M dwarfs (Mould 1978).

There have been two main solutions to this problem. Lynden-Bell (1975) proposed a solution by using infall of

-76-

gas from the halo to the disc. Alternatively Truran and Cameron (1971) postulated a pre-galactic burst of massive



The cumulative metallicity distribution for 133 Figure 3.1: G dwarfs in the solar neighbourhood. The data with errors are represented by the solid curve, reproduced from Tinsley (1980). The abscissa is a measure of the metallicity (Fe/H) inferred from the measurements of UV excess (Pagel and The prediction of the simple Patchett 1975). model, which is far in excess of the observational numbers at lower metallicities, is shown by the dashed line. The model with finite initial metallicity, variable yield and no infall (and also duly corrected for the dark matter fraction) is shown by a dot dashed line. The low-metallicity tail extending beyond the initial metallicity (Z_{Θ}) may correspond to the pre-disc population (which is about 5% of the total as indicated by the ratio σ_0/σ_1).

stars leading to a finite initial metal abundance Z_{\odot} . As the SFR of Chapter 2 already predicted a finite initial metallicity, the latter solution was followed in our previous work (Paper I; Rana and Wilkinson 1987a, hereafter referred to as Paper IV), where a more extensive discussion is given.

From equation (3.17) the mass fraction of visible stars with metallicity no more than Z is given by

$$S(Z) = \underbrace{\mathscr{G}(Z)}_{\mathscr{O}_{1}} = \underbrace{\operatorname{G'-exp}[\operatorname{G}(Z^{l-n} - Z_{0}^{l-n})/(1-n)]}_{(\operatorname{G'-}\mathcal{M}_{1})} n \not\models 1 \quad (3.20a)$$

$$= \frac{G^{\circ} - \mu_{0} (Z/Z_{0})}{(G^{\circ} - \mu_{0})} \qquad n=1 \quad (3.20b)$$

where G' is defined as

 $G' = \mu_0 + \phi_0 GK$ (3.21)

and
$$(GK)^{-1} = 1 - \frac{(S_1 - S_2)}{(\mu_2 - \mu_1)}$$
 (3.22)

There are now two possible cases.

<u>Case A</u>. This is a special case with $\mu_0 = 1$ which (i) implies that during the formation of the disc, all material was in the form of gas only. Therefore $\mathcal{O}_0 = \mathcal{O}_0 = 0$ and G' = 1. For a particular n the problem is reduced to supplying the values of μ_i and δ_i and fitting the G-dwarf curve. This will determine G, Z_1/Z_{\odot} , K and the ratio D/(1-R). From dynamical arguments Bahcall (1986c) estimated \mathcal{S}_{1} = 0.55 <u>+</u> 0.10 and this value for the dark matter fraction has been used. There is now some debate however that this value is an overestimate (e.g. Gilmore and Wyse 1987) and this will be discussed more extensively in Chapter 4. Irrespective of the gas distribution used, best fitting models give $\log_{10} (Z_1/Z_0) \approx 0.57$, implying that the pre-disc metallicity $Z_{\odot} \cong Z_{\downarrow}/4$. A representative plot of the model for n = 1 is shown in Figure 3.1. The n = 0 model is similar to that of Truran and Cameron (1971) with a slightly worse fit than n = 1 at the high metallicity end of the curve.

(ii) <u>Case B</u> The general case with $\mathbb{G} \neq 0$ and $\mathbb{G} \neq 0$, that is allowing that when the disc formed some matter was already in visible stars or dark matter. There are now essentially five independent parameters for a particular n, namely μ_0 , \mathfrak{G}_0 , μ_1 , \mathfrak{G}_1 and the ratio $\mathbb{Z}_1/\mathbb{Z}_0$ with which to fit the observed G-dwarf curve, allowing a maximum deviation of 5% to 21 points. As the actual scatter in the ordinate and the abscissa is likely to be no less than \pm 15% (Tinsley 1980) then fits for $\mathbb{Z}^3 < 50$ are acceptable. The fraction of visible stars at present in the solar neighbourhood was taken as $\mathfrak{G}_1 = 0.3 \pm 0.1$ (Bahcall 1986c). The value of μ_0 was varied between 0.2 and 0.8, while $\mathfrak{G}_0 = 0 - 0.1$ (Vader and de Jong 1981; Bahcall et al. 1983). The range of μ_4 was chosen to include various estimates of the mass of gas (Chapter 1).

For n = 1, that is variable yield, the best fitting parameters gave $\frac{1}{2}$ < 21 with $G_0 = 0.05$, log $(Z_1/Z_0) =$ 0.57 ± 0.05 and G = 1.31 ± 0.04 roughly independent of the chosen gas fraction, while G and μ_0 followed the empirical relationships (Paper I)

G'≊ 6.6µ, (3.23)

$$\mu_0 \cong 5.7 \mu_1.$$
 (3.24)

Therefore, the constant K in $y_Z = KZ$ is found to be 1.0 or 1.4 while D/(1-R) \simeq 0.25 or 0.56 for the Bhat et al. or SSS distribution of H₂ respectively.

One may recall from the analysis of the SFR (Chapter 2) that the near constancy of the SFR had suggested two things - little or no infall and a lower bound on the initial metallicity - to be given by $\log_{10} (Z_1/Z_0) < 0.7$. Both are found to be satisfied here, the first one by choice and the second one from the solution of the G-dwarf problem.

-79-

Interesting information is also obtained on Co, Mo and 8. That 5% 2 0.05, means that about 5% of the total matter in the solar neighbourhood has metallicity not exceeding Z_{α} , they are the visible component of the pre-disc population. The observed fraction of metal-poor or halo objects in the solar neighbourhood is in fact about 0.04 - 0.06 (Vader and de Jong 1981; Bahcall et al. 1983). The small tail to the distribution in Figure 3.1 therefore could be due to a pre-disc population. The value of μ_0 depends on μ_1 (equation 3.24) and for $\mu_0 = 0.08 - 0.14$, $\mu_0 = 0.46 - 0.80$ and therefore $S_0 = 0.49 - 0.15$. (Note that $\mu_1 = 0.09 \pm 0.01$ for the Bhat et al. distribution of H₂ and $\mu_1 = 0.13 \pm 0.01$ for the SSS distribution of H_2). If $Z_0 \approx Z_1/3.7$ is required, it is not surprising that the pre-disc population in the solar neighbourhood had to produce Spe 0.45, $\mathcal{C}_{0} \simeq 0.05$ with $\mu_{0} \simeq 0.5$ for $\mu_{1} = 0.09$.

For n = 0, that is a constant yield, the fit to the G-dwarf curve is not as good, best fitting parameters having $\mathcal{K}^2 > 25$ for $\mathfrak{S}_1 = 0.55$. However for $\mathcal{H}^2 < 50$, once again log $(\mathbb{Z}_1/\mathbb{Z}_0) = 0.57$, $\mathfrak{G}_0 = 0.05$ are still satisfied for $\mu_0 > 0.45$. For a particular μ_1 (e.g. = 0.09) the fit improves as μ_0 is chosen to be higher. As μ_0 increases from 0.5 - 0.8, K (and therefore $y_{\mathbb{Z}}$) increases from 0.020 to 0.028. The ratio D/(1-R) also increases sharply as μ_0 increases, from 0.24 to 0.56 for the same range in μ_0 . These values for the yield and dark matter fraction will be discussed later in Sections 4.1 and 4.2.

Therefore, for a finite initial metallicity solution to the G-dwarf problem, a yield proportional to metallicity improves the fit but a constant yield case is still

-80-

allowable. The finite initial metallicity for the best fit (with due regard for $\mu_0 \neq 1$) is $Z_0 \approx Z_1/3.7$. It is now useful to examine briefly whether there is any physical ground for this condition.

Olive (1986) has remarked that provided there exists a mechanism, finite initial metallicity or prompt initial enrichment (PIE) models are the most straightforward perturbations of the simple model. It is recognised that there was some production of metals by pre-galactic star formation but this is restricted to having produced only a tiny fraction of metals in the disc by observations of very low metallicity stars in the spheroid (Hartwick 1983; Bessel and Norris 1984; Beers et al. 1986).

There have been however, many attempts to explain PIE by the pre-disc generation of stars. Truran and Cameron (1971) used an extreme form of variable IMF in postulating only large mass stars form until Z ≈ 0.25 Z₀. It has also been argued that the metal yield was high in low metallicity stars (Matteucci and Tornambe 1985; Tornambe and Matteucci 1985; Jura 1985), and the time-scale fast enough to set up a Z_{Ω} . Wyse and Silk (1987) and Olive (1986), on the basis of a bimodal SFR suggested a high rate of massive star formation in the early stages of disc evolution acting as Initial enrichment of the disc before the commencement PIE. of star formation due to prior activity in the halo (e.g. Hartwick 1976) can be taken as a legitimate initial condition for the chemical evolution of the disc on its own (Pagel 1987). Cayrel (1986) from a detailed study of the first stars concluded the disc would be enriched to $Z \approx 0.25 Z_{\odot}$. Ostriker and Thuan (1975) considered a two

-81-

component disk-halo model in which matter shed by dying halo stars accumulates in a smaller more rapidly rotating disc. However, Pagel and Patchett (1975) pointed out that the essential feature of these models is the PIE (of $Z_{\odot} \approx Z_{\downarrow}/5$) due to enrichment by matter shed in a short initial period from the most rapidly evolving massive stars in the halo while the subsequent inflow of enriched material from small stars of the halo has only a negligible effect. Dynamical collapse models (Schmidt 1975; Larson 1976; Tinsley and Larson 1978) may also lead to a situation which resembles PIE and which may be treated as such for some purposes, but this could be dangerous in problems where time is of importance, for example cosmochronology (Pagel 1987).

Furthermore, recent consideration of the so-called Galactic thick disc metallicity structure provides a physical explanation of the pre-enrichment of the classical thin disc (Gilmore 1984; Gilmore and Wyse 1986; Wyse and Silk 1987). Eggen, Lynden-Bell and Sandage (1962, hereafter ELS) argued on the basis of the correlation of UV excess, orbital eccentricity and z-motions that halo stars formed first, born during free-fall collapse in a short time with little or no dissipation and that the residual gas dissipated energy in a comparable time to form the disc in which the more metal rich stars have been continually forming ever since. The situation now is more complex. Norris et al. (1985) pointed out a kinematic bias in the ELS data and discovered stars with low Z and low eccentricity. Zinn (1985) finds two groups of globular clusters readily distinguished by their distributions in [Fe/H], the group with [Fe/H] > -1 being a flattened population of disc

-82-

globular clusters with intermediate Z. These observations have been identified with the in situ observations at high Galactic latitudes of Gilmore and Reid (1983) (cf. Bahcall 1986d; Yoshii et al. 1987; Sandage 1987; Norris 1987), who found a stellar component of the Galaxy with kinematics and metallicity distribution intermediate between those of the extreme spheroid and thin disk (Gilmore and Wyse 1985; Wyse and Gilmore 1986). This component has a scale height larger than the already identified thin disc and so is referred to as the thick disc. The ELS picture is now modified, with the halo formed in free fall followed by extended dissipative phases leading ultimately to the present thin disk in which most of the residual or recycled gas is confined, while the bulge, thick disc and halo are fossilized remnants conveying records of a past history. Gilmore and Wyse (1986) have argued that initial enrichment due to rapid infall from the extreme halo alone cannot provide a resolution of the thin disc G-dwarf problem, but that pre-enrichment and then star formation in the thick disc leads to a mean [Fe/H] for the thick disc of -0.6 + 0.1 which agrees well with the metallicity determined for the oldest, most metal poor G-dwarfs (Gilmore and Wyse 1985), and also the Z_1/Z_0 determined above.

While the above scenario resulting from a fit to the cumulative metal abundance curve for the local G-dwarfs appears to be quite interesting, one must nevertheless be aware of the following limitations. First, the cumulative abundance curve is at present based on a sample of only 133 local G-dwarfs (Pagel and Patchett 1975). There is a possibility (Pagel 1987) that the low metallicity stars have

-83-

such large scale-heights that they have been missed altogether in the local samples even after weighting by velocity perpendicular to the plane. Secondly, metallicity has been inferred from the UV excesses which are assumed to reflect the relative abundance of iron (Fe/H). The above derivation of Z_{1}/Z_{0} assumed $Z \ll Fe/H$, the validity of which is discussed briefly later and in Paper I. Thirdly, G dwarfs constitute only a small fraction of the total mass locked up in visible stars ($\mathbb{Z}_{\mathbb{S}}$). The validity of the assumption that $\mathcal{O}(Z) \subseteq \mathbb{N}_{\mathbb{Q}}(Z)$, where $\mathbb{N}_{\mathbb{Q}}$ = number of G dwarfs sampled, for all Z is highly questionable. Finally, the G-dwarf problem may be a peculiarity of the Sun's position in the Galactic disc and not represent a universal state of affairs. Pagel (1987) has summarised the position with respect to the halo and disc globular clusters (Zinn 1985) and Galactic bulge stars (Rich 1986) and shown that they fit well the predictions of the simple model, so that any proposed solution to the G-dwarf problem must take this into account and not appeal to a universal law. He suggests therefore that inflow is potentially acceptable, but models appealing to a law that true yields are always high at low Z are not.

3.4 THE AGE-METALLICITY RELATION

Network and the second

One can obtain some knowledge of the past evolution of the Galaxy and principally the solar neighbourhood by studying the variation of surface composition of unevolved stars with stellar age, assuming that the surface composition of an unevolved star represents the ISM out of

-84-

which it is formed. Twarog (1980) showed that disc stars in the solar neighbourhood exhibited a systematic increase of metallicity with the estimated time of their formation, the metallicity having only increased by about a factor of 3 to 4 over the lifetime of the disc, as shown in Figure 3.2. However, determinations of stellar ages and compositions have important uncertainties and Carlberg et al. (1985) have re-analysed part of Twarog's basic data leading to results



Figure 3.2: The stellar AMR in the solar neighbourhood for disc stars. Observations are from Twarog (1980), the same data recalibrated by Carlberg et al. (1985) and the independent data set of Nissen et al. (1985). The predictions of this work with a constant or variable yield are contrasted with the infall model J of Lacey and Fall (1985).

that are significantly different. Also shown in Figure 3.2 are results from the independent survey of Nissen et al. (1985) which apart from significant scatter seem to follow the trend of Twarog's points. Preliminary results from an extensive uvby photometric survey of F stars also seem to reinforce the Twarog result (Strömgren 1987), and therefore

-85-

in this work χ^2 will be calculated for the Twarog (1980) points.

Unlike the situation in the G-dwarf problem, the study of the age-metallicity relation (AMR) requires an explicit knowledge of the SFR. Twarog (1980) pointed out that a good fit to the AMR was obtained by a roughly constant SFR. Therefore, previous workers using a SFR proportional to the total gas density have required infall, to keep $\frac{1}{5}$ roughly constant in order to derive the AMR (e.g. Güsten and Mezger 1982; Lacey and Fall 1983). One of the best models of Lacey and Fall (1985) is shown in Figure 3.2 in order to illustrate the additional difficulty of infall models for the region t < 5Gy, especially if the results of Carlberg et al. (1985) are used.

With the SFR of Chapter 2 (equation 3.12) and with a finite initial metallicity Z_0 , the AMR relation has already been given in equation (3.18). Its solution is given by,

$$z = z_0 \left(1 - \frac{t}{\xi}\right) \qquad n = 1 \quad (3.25b)$$

where the function F(k,G,Z,n) and the constants A, T^{-1} and J have been defined for convenience as

$$F(k,G,Z,n) = \exp \left[\frac{(1-k)GZ^{l=n}}{(1-n)} \right]$$
 (3.26)

$$A = G - k(G - b)$$
 (3.27)

$$\overline{U}^{\sim} = \underline{A} \overline{\Sigma}_{\overline{Y}}^{k-1} \underline{X}^{k} \underline{C} (1-\underline{R}), \qquad (3.28)$$

and

$$J = \frac{\nabla^{\circ} F(k_i - G_i Z_{\odot i} n)}{A Z_{\odot}^{k_0}}$$
(3.29)

and the second second second

The constants \overline{C}^{-1} and J can of course be alternatively defined from the normalisation when at $t = t_0$, $Z = Z_0$ giving

$$\mathbf{\tilde{C}} = \mathbf{t}_{\mathbf{i}} \begin{bmatrix} \mathbf{A} \\ \mathbf{I} - \begin{pmatrix} \mathbf{Z}_{\mathbf{0}} \\ \mathbf{Z}_{\mathbf{i}} \end{bmatrix}$$
(3.30)

$$J = \frac{1}{t_{i}} \int_{Z_{0}}^{Z_{i}} \frac{dZ}{n+kb}$$
(3.31)
$$Z = F(k,G,Z,n)$$

where t, is the age of the disc. In the following t, is taken to be 15Gy, which is consistent with recent estimates of the ages of globular clusters (Janes and Demarque 1983; Vandenberg 1983; Ratcliff 1987). In particular, Zinn (1985) found that the most metal rich globular clusters form a disc system whose age as judged by the two prototype disc globular clusters, M71 and 47Tuc is the same as the age of the halo clusters, that is at least 15Gy. Therefore using these equations, with k and b determined from Chapter 2, the set of best fitting parameters (in particular $log(Z_1/Z_0) = 0.57$) are all found to give fits to Twarog's 12 data points with $x^2 < 12$. Following Paper I x^2 has been computed for the standard deviations of Twarog's points. Of course, strictly using standard errors 2^2 will be larger but in the light of possible systematic errors the models give a good fit. This is illustrated in Figure 3.2 where the best fitting n = 0 and n = 1 models are shown. Several comments may be made.

(i) Models with constant yield, that is n = 0 give better fits to the AMR of Twarog (1980) than a yield proportional

- 8.7-

to metallicity (typically $\mathcal{Y}^2_{MRR} \leq 3$ for n = 0, while $\mathcal{Y}^2_{MRR} \leq 6.5$ for n = 1). This of course is in contrast to the G-dwarf problem where the opposite is the case, but once again observational uncertainty does not rule out either model at present.

(ii) Marginally better fits are obtained with k and b determined using the Bhat et al. distribution of H_2 , but the difference cannot be used to rule out either distribution. (iii) Models with $y_2 = KZ$ have a more concave AMR than the constant yield model. This is not reflected in the points of Twarog or Nissen et al. (1985) but does appear in the Carlberg et al. (1985) results. However the Carlberg et al. (1985) points have systematically higher metallicities especially when t is small.

(iv) If the data points are taken together, then they suggest that the disc should have an initial metallicity in the range log $(Z_0/Z_{\odot}) \cong 0.4 - 0.6$ consistent with the requirement needed to solve the G-dwarf problem. However the considerable spread in the points especially at t ≤ 6 Gy does not at present distinguish between the case for infall (Lacey and Fall 1985) and no infall.

(v) The above discussion has assumed $Z \in C$ Fe/H, in that the model predicts Z_1/Z_{\odot} , while the metallicity measurements for the AMR refer to Fe/H. This assumption may be questionable, and more consideration of it is given in Paper I and the next section. However on the basis of this assumption the same set of values which solves the G-dwarf problem solves the AMR within the errors of observation.

Finally from equations (3.26) - (3.31), the returned fraction in the solar neighbourhood may be evaluated

-88-

independent of its source equation (equation 3.9). One obtains

$$(1-R) = \underbrace{\sum_{T} G_{\mu_{0}} Z_{1}^{k} F(k,G,Z_{0},n)}_{\Psi_{1} \mu_{0}^{k-1} t_{1}} \int_{Z_{0}}^{Z_{1}} \frac{dZ}{n+kb} n \neq 1 \quad (3.32a)$$

$$(1-R) = \frac{\sum_{\Gamma} G \mu_{1}^{K}}{A \sqrt[k]{t} \mu_{0}^{K-1}} \left(\frac{Z_{0}}{Z_{0}} \right)^{K} \left[1 - \left(\frac{Z_{0}}{Z_{0}} \right)^{K} \right] \qquad n=1 \quad (3.32b)$$

where all values refer to the solar neighbourhood and ψ_1 is the present star formation rate. For numerical computation, $(\Sigma_{\gamma})_{\odot} = 75M_{\odot}pc^{-2}$ (SSS 1984) and $(\psi_1)_{\odot} = 4 \pm 1M_{\odot}pc^{-2}Gy^{-1}$ (SBM 1978; Turner 1984; Lacey and Fall 1985). For the range of parameters obtained by fits to the G-dwarf problem and the AMR, the range of resulting R is 0.3 - 0.65, with the Bhat et al. gas distribution giving higher values of R than that of SSS. This is in good agreement with R = 0.35 (Twarog 1980; Chiosi and Matteucci 1983), 0.48 (Tinsley 1980) and 0.5 - 0.8 (Gusten and Mezger 1982). It strongly depends, of course, on $(\psi_1)_{\odot}$. A larger value of $(\psi_1)_{\odot}$ implies a larger value of R, otherwise one would expect to see a greater mass in visible stars than implied by $\sigma_1 \Sigma_{\tau}$.

Once R is determined, D, the dark matter fraction becomes fixed. Assuming of course $\mathfrak{S}_0 = 0.55$, for n = 0, $D \cong 0.10 - 0.24$, while for n = 1, $D \boxtimes 0.1 - 0.39$, this time higher values of D being favoured by the SSS gas distribution. The consistency of these figures and the consequences for dark matter in the solar neighbourhood will be discussed in Chapter 4.

-89-

3.5 THE VARIATION OF THE SFR

The evidence supporting the claim that the SFR has stayed roughly constant over the lifetime of the Galaxy in the solar neighbourhood, as well as in some external spiral galaxies was reviewed in Section 2.2. Olive et al. (1987) have further tried to estimate limits on the SFR by nucleosynthesis data, but conclude that present uncertainties in the yield for massive stars do not allow any rigorous constraints. Studies in nucleocosmochronology which might in principle be used to give information on the history of the SFR are also restricted by uncertainties associated with the input nuclear physics, the meteoritic abundance data, and the nucleosynthesis calculations themselves (Thielemann and Truran 1985; Meyer and Schramm 1986).

However, from the above chemical evolution model it is now possible to study the time evolution of the SFR in the solar neighbourhood. Defining \bigvee_{0} as the SFR at t = 0, and treating $(\Sigma_{T}X)$ as a constant (i.e. no infall) equation (3.12) becomes

$$\left(\frac{\Psi_{I}}{\Psi_{O}}\right)_{O} = \left(\frac{\mu_{I}}{\mu_{O}}\right)_{O} \left(\frac{Z_{I}}{Z_{O}}\right)_{O}$$
(3.33)

which through equation (3.17) can then be expressed purely as a function of μ or Z and then evaluated as a function of time.

For both n = 0 and n = 1, the range of parameters to fit the G-dwarf problem and AMR give,

$$\left(\frac{\Psi_i}{\Psi_i}\right) \approx 1.0 \pm 0.5 \tag{3.34}$$

for the whole range of $(\mu_i)_{ij}$ defined by the different gas

-90-

distributions. Therefore, according to the above model the SFR in the solar neighbourhood has not practically changed throughout the age of the disc, which is consistent with the evidence for a constant SFR presented in Section 2.2. Further, infall is now not required to explain the constancy of the SFR in the context of a 'Schmidt' type law of star formation. Larson (1986) in constructing his IMF to fit the observed dark matter fraction with dead stellar remnants requires the total SFR to decrease by a factor of 18 between t = 0 and t = 15Gy. For none of the above parameters is this possible for a SFR given by equation (3.33).

It may be argued that if $(\frac{1}{2}, \frac{1}{2})_{\odot}$ has remained practically constant for the past 15Gy, it cannot go on for long as the Galaxy will shortly run out of gas (Larson 1972; Tinsley 1980; Kennicutt 1983b; Pagel 1986b). In fact Larson et al. (1980) in commenting on the short gas depletion timescales inferred from constant SFRs in external spirals, had suggested infall as a possible way out of the situation. This is a possible problem with a $y_{\chi} \propto Z$ model which would predict the Galaxy running out of gas in another 3Gy due to an increasing rate of change of metallicity with time. However, Figure (3.3) shows the time variation of Ψ/Ψ_{0} and μ/μ_{0} for a model with constant yield (with $\mu_0 = 0.5$, Paper IV). In the region between t = 0 and t = 15Gy, $\frac{9}{9}$ peaks at around 9Gy but the variation is much less than a factor of 2 and $\sqrt[4]{6} \simeq 1$. If the AMR is extrapolated to times greater than 15Gy (which may be dangerous due to the breakdown of the instantaneous re-cycling approximation), the model predicts a decreasing SFR with time which flattens out, therefore keeping $\mu/\mu_{\odot} > 0$

-91-



Figure 3.3: The fractions $\psi(t)/\psi$ and $\mu(t)/\mu_0$ as a function of time from the formation of the disc of the Galaxy at t = 0, extrapolated to t > 15Gy, the assumed present age of the disc. This is the prediction of a model with constant yield and $\mu_0 = 0.5$ (Paper IV).

for times up to 25Gy. Although more detailed modelling is needed, it appears that a star formation rate proportional to the mass of H_2 , in a chemical evolution model with constant yield is able to explain a slowly varying SFR in the past and also avoids a 'catastrophic' end to the Galaxy without having to invoke infall.

3.6 THE METALLICITY GRADIENT

Abundance gradients were first detected in nearby spiral galaxies from optical spectroscopy, Searle (1971) interpreting the radial gradients in HII region line strengths ratios as being caused by the decline in the abundance of metals with galactocentric radius. This

-92-

interpretation has been confirmed and abundance gradients studied in this and other galaxies (e.g. Pagel and Edmunds 1981; McCall et al. 1985; Garnett and Shields 1987). Table A.2.2 (Appendix) lists values derived for the Galaxy of

$$\frac{d\log(Z)}{dR_{G}} = -(0.05 - 0.13) \text{ dex kpc}^{\circ}$$
(3.35)

over $R_{f_0} \cong 5 - 15$ kpc for least squares fits.

Early attempts to explain these radial gradients involved the SFR following the total mass surface density of the disc (Talbot and Arnett 1975) leading to greater enrichment in the inner parts of the Galaxy, or the effect of spiral arms on the SFR leading to a similar effect (Jensen et al. 1976). This latter explanation has major difficulties, both theoretically (Casse et al. 1979), and from the observation that some galaxies have metallicity gradients without spiral arms (McCall 1986; Elmegreen 1987c; Section 2.2).

The prediction of the simple model (Schmidt 1963; Searle and Sargent 1972) with constant yield, no infall and $Z_0 = 0$ gives

$$Z_{i} = Y' \ln\left(\frac{1}{\mu_{i}}\right)$$
(3.36)

As Pagel (1987) has re-emphasised, if Σ_{H_2} follows Σ_{T} (as in the distribution of SSS 1984), μ_i is roughly constant with R_{f_i} and no metallicity gradient results. It was this apparent failure of the simple model which led to two main modifications in order to explain metallicity gradients.

Firstly, it is claimed a constant yield with vertical infall of gas will produce a metallicity gradient. Tosi and Diaz (Tosi 1982; Diaz and Tosi 1984; Tosi and Diaz 1985) assume temporally and spatially constant infall and vary the

-93-

SFR as a free parameter. However, the generation of the metallicity gradient in this model is largely a geometrical effect, the geometry of the disc being such that a uniform infall rate of halo gas per unit area of the disc, having little or no metallicity will cause more dilution of metals at its periphery than near the centre. The main weakness is the assumption of a spatially uniform infall rate as one would expect the rate to increase towards the centre (Hunt 1975; Pagel 1987).

In fact, other workers using more realistic SFRs and infall rates have been unsuccessful in explaining the metallicity gradient (Lacey and Fall 1983; Güsten 1986). The additional factors of radial gas flows (Lacey and Fall 1985) or variable yield (Peimbert and Serrano 1982; Güsten and Mezger 1982) have had to be invoked in addition to infall to solve the problem. Clayton (1987) by studying six different analytic models of galactic chemical evolution has shown that infall models can be characterised by

$$Z(\mu) = \frac{y'}{2} \left\{ \ln(1/\mu) + \ln[\ln(1/\mu) + 1] \right\}$$
(3.37)

Both this model and the simple model (equation 3.36) fit reasonably well the observational 12 + log(O/H) versus $ln(\mu)$ plot for M81 (Garnett and Shields 1987). However, as Garnett and Shields point out, their μ is calculated only using the HI gas, neglecting H_2 . If the H_2 follows the distribution of H_2 in similar spirals the observed Z-ln(μ) relation will not fit the above models.

The second modification of the simple model used in an attempt to fit the abundance gradient is that of variable yield based on a bimodal or otherwise variable IMF (e.g. Quirk and Tinsley 1973; Güsten and Mezger 1982). Peimbert

-94-

and Serrano (1982) postulated a yield dependent on the metallicity, while Edmunds and Pagel (1984a) used a yield linear with total mass $\mathbb{Z}_{\mathbb{Y}}$. The situation concerning a variable IMF is far from clear, for example the variable IMF model of Terlevich and Melnick (1984) predicts even flatter gradients than the simple model.

It is immediately obvious however, that with the Bhat et al. (1985) distribution of H_{λ} for the Galaxy, a gradient exists in μ which through equation (3.36) leads to a gradient in Z (Paper I) therefore allowing the possibility that the simple model could still be applicable to predicting abundance gradients. However, this explanation of the metallicity gradient is essentially phenomenological, that is the variation in Z is derived from a given variation already existing for μ . Even if that is done successfully, it does not explain the existence of a radial gradient in μ_{i} starting from an initial assumption that M_0 and Z_0 are constant with Galactocentric radius at time of formation of the disc. Furthermore, any calculation of a metallicity gradient by this method depends crucially on μ_i which in turn depends on $\mathbb{Z}_{N_{n}}$ which is a poorly known observational quantity, especially in external galaxies (Chapters 1 and 2). McCall (1982) and Edmunds and Pagel (1984a) pointed out an observational relationship between 12 + log(O/H) and Σ_{T} (determined from the rotation curve) of regions in spiral galaxies. As Σ_r can be better estimated than μ_i , it seems reasonable to express the metallicity as a function of $\Sigma_{\mathbf{r}}$ in order to test any model against observation. It is for these reasons that the following formalism is developed.

Equation (3.18) studies the evolution of metallicity in

-95-

any region of the Galaxy for which \mathbb{Z}_{Γ} is available and the exponents k, b and G are thought to be applicable. Solving this equation with $Z = Z_1$ at $t = t_1 = 15$ Gy, the variation of the present metallicity Z_1 for different regions of the Galaxy can be obtained as

$$\int_{Z_0}^{Z_1} \frac{dZ}{n+kb} = J' \mathbb{Z}_{\gamma}^{k-1} \qquad n \neq 1 \quad (3.38a)$$

$$Z = F(k,G,Z,n)$$

$$\log\left(\frac{Z_{i}}{Z_{0}}\right) = -\frac{1}{A}\log\left(1 - B \sum_{i}^{k-i}\right) \qquad n = 1 \quad (3.38b)$$

where once again F, A, B and J' have been defined for convenience,

$$B = \frac{CX^{k}At_{1}(1-R)}{G\mu_{0}}$$
(3.39)

and

$$J' = \frac{B F(k, -G, Z_0, n)}{kb}$$

$$A Z_0$$
(3.40)

Now the question is, should Z_{Θ} (and hence μ_{Θ}) be regarded as constant for all R_{G} or did the disc have an initial metallicity gradient? This is not improbable as one finds an appreciable metallicity gradient in Fe/H in the radial distribution of globular clusters in our Galaxy and M31 (Sharov and Lyutyi 1984; Pilachowski 1984) although there is some controversy. There is a possibility that consideration of the thick disc metallicity structure could lead to an initial metallicity gradient but further work needs to be done. An initial metallicity gradient however, is not a particularly interesting case because it would partly or wholly explain the present metallicity gradient ab initio, that is, through the choice of initial assumption only. In the first instance therefore Z_{Θ} will be taken as a

-96-

constant throughout the disc, and a metallicity gradient will be generated by enrichment in the disc alone.

The constants B and J' can now be evaluated for the solar neighbourhood,

$$B = \left[1 - \left(\frac{Z_{i}}{Z_{o}}\right)_{\odot}^{A}\right] \left(\Sigma_{\gamma}\right)_{\odot}^{i-\kappa}$$
(3.41)

$$J' = (\Sigma_{\gamma})_{\odot}^{i \sim k} \int_{Z_{\Theta}}^{Z_{1_{\Theta}}} \frac{dZ}{n+kb}$$
(3.42)
$$\int_{Z_{\Theta}}^{z} F(k,G,Z,n)$$

and applied to the rest of the disc, assuming that X, the chemical fraction of hydrogen is only a very slowly varying quantity with time and R_{G} , and hence regarded as a constant.

It can now be seen that the variation of \mathbb{Z}_{i} with \mathbb{R}_{G} is determined simply by the variation of $\sum_{\widehat{V}}$ (equation 3.38) which can be determined independently of μ_{i} . (For the determination of $\sum_{\widehat{V}}$ see Caldwell and Ostriker 1981 and Bahcall et al. 1983). Furthermore, the value of the parameter k, the index expressing the degree of dependence of the SFR on $\sum_{N_{2}}$ becomes important in determining the gradient. If k = 1, the gradient vanishes irrespective of whether the yield is variable or not. For k > 1, the metallicity increases towards the Galactic Centre, but for k < 1 the gradient is in the wrong direction.

In order to illustrate this on the compilation plot (Figure 3.4) of metallicity against R_{G_1} , the quantity $(Z_1/Z_{\odot})_{\bigodot}$ needs to be represented in terms of 12 + log(O/H). This quantity has been determined for the G-dwarf problem and AMR using Fe/H measurements. Paper I made the assumption that [O/H] = [Fe/H] \ll Z and discussed in detail its validity. We also argued that the spread in

-97-

12 + log(O/H) at a particular R_{g} , was due not only to observational error but also due to the dependence of the metallicity with time of formation of stars and the locking or depletion of metals into grains in HII regions.



Figure 3.4: The metallicity gradient in the Galaxy shown by (O/H) abundance observations. The sources of data are given in Figure 2.4. Assuming a constant initial metallicity Z_0 as shown by the solid baseline, a predicted fit to the upper envelope of points for a constant yield model with dlog(Z_1)/dR₆ = -0.06 dex kpc⁻¹ is shown.

Therefore, the present metallicity Z_1 is given approximately by the upper envelope to the points in Figure (3.4). This forced us to draw a baseline for Z_{\odot} , if possible, without leaving any point below the line, while satisfying the condition of $\log(Z_1/Z_{\odot}) = 0.57$ at $R_{\odot} = 10$ kpc which also should not exclude any point above Z_1 at the same R_{\odot} . Making some compromise on both sides, we preferred to draw the baseline at 12 + $\log(O/H) = 8.30$ as shown in Figure 3.4.

-98-

This fixes the present metallicity in the solar neighbourhood as $12 + \log(O/H) = 8.87$. Since the solar $12 + \log(O/H) = 8.80 + 0.12$ (Rana and Wilkinson 1986c, and references therein) and Edmunds (1984) has adopted 8.92, this upper bound is perhaps justifiable. It is interesting that the spread in (O/H) observations at $R_{fr} = 10$ kpc is very similar to (Z $_{\rm I}$ /Z $_{\odot}$) determined from the G-dwarf problem and At R_{ff} > 12 kpc there are about half-a-dozen AMR. measurements of (O/H) that lie below the baseline. It may be that these objects show an abnormal depletion in the gas phase, possibly because of substantial segregation into grains, or may just be due to observational error. These points could be easily included by the use of a baseline reflecting an initial gradient in Z_0 , but in the first instance is unnecessary. The production of a gradient of -0.06 dex kpc^{-1} is shown in Figure 3.4 for the case of a constant yield (Paper IV and a similar plot for variable yield can be found in Paper I). The question of whether Z, is represented by the upper envelope of the distribution or the distribution itself is a complex one concerning the value of Z, in the solar neighbourhood and the validity of the assumption [O/H] = [Fe/H] (cf. Clegg et al. 1981; Chiosi and Matteucci 1983; Gratton 1985; Laird 1985; Pagel 1987). The important point however is the value of the gradient predicted, that is $dlog(Z_1)/dR_{G_1}$ and that the gradient of the upper envelope is virtually the same as that of the $12 + \log(O/H)$ distribution itself.

Therefore, in the light of the above discussion, the various gradient determinations in Table A.2.2 (Appendix) and following Lacey and Fall (1985), the observational

-99-
gradient is assumed to be

$$\frac{d\log(Z_{i})}{dR_{i}} = -0.06 \pm 0.01 \text{ dex kpc}^{-1}$$
(3.43)

Using equations (3.38) - (3.42) and a spatial variation of Σ_{γ} as given in Table A.2.1 (Appendix) the metallicity gradient was evaluated for the parameters which give good fits to the G-dwarf problem. The value of k was then derived by imposing equation (3.43). For n = 1 the required range of k for providing the observed metallicity gradient is $\approx 1.10 - 1.15$; but for n = 0, in order to produce the same gradient k = 1.30 - 1.60. The value of k depends also on μ_{Θ} , the lower the value of μ_{Θ} , the lower is the required value of k for producing the same gradient. Hence, k increases from $1.36^{\oplus 0.09}_{-0.05}$ for $\mu_{\Theta} = 0.5$ to about $1.50^{\oplus 0.25}_{-0.15}$ for $\mu_{\Theta} = 0.8$. The value of k is found to be virtually independent of S_{μ} .

A number of points may be noted from this analysis. (i) The empirical correlation $\sqrt[k]_{S} \ll \sum_{N_{2}}^{k}$ of Chapter 2 gave $k = 1.2 \pm 0.2$ for the Bhat et al. (1985) distribution of H₂ in the Galaxy while $k = 0.7 \pm 0.2$ for the SSS (1984) distribution. Consequently the SSS distribution leads to no metallicity gradient or a gradient in the wrong sense, while the Bhat et al. distribution is able to produce the observed gradient. Although this result was expected on the basis of the simple model, it also applies to models where the yield is proportional to the metallicity. Therefore, it is concluded that the Bhat et al. distribution is superior to that of SSS. In fact the argument may be inverted, and fixing k by the observed metallicity gradient \sum_{N_2} may be derived from the SFR ($\frac{N_{S}}{S}$) as a function of Galactocentric radius, independent of the controversial CO surveys. This

-100-

will be done in Chapter 4.

The metallicity gradient has been expressed in terms (ii) of \mathbb{Z}_{T} rather than the uncertain μ_{1} , and the formalism developed is sufficiently general for its possible application to other spiral galaxies. The existence of abundance gradients in other spirals could be linked to their variation in \mathbb{Z}_{r} (cf. Talbot and Arnett 1975). A relationship between metallicity and \mathbb{Z}_{τ} is actually observed (McCall 1982; Edmunds and Pagel 1984a; Bothun et al. 1984) and once again this will be examined in detail in Chapter 4. (iii) Given an initial density profile of the total mass distribution in the disc, it has been shown that with time the Galaxy may naturally develop a radial gradient in the gas fraction (M_1) and the metallicity (Z_1) even though it begins with a constant initial gas fraction (μ_0) and a constant initial metallicity (Z₀) at all radii, provided the exponent k > 1. This metallicity gradient is not generated by infall or radial gas flows or necessarily variable yield (although this helps). Even for the case of constant yield and no infall a metallicity gradient is produced, the reason being (as noted by Lacey and Fall 1985 and Clayton 1987) the 'efficiency' of star formation (ψ_{s}/Σ_{q}) varies as a function of R_{G} (equation 3.12). This SFR has, of course, been derived empirically rather than adopted to provide the correct result.

This Chapter has shown that using the SFR derived in Chapter 2, using an explicit form for dark matter, a finite initial metallicity of the disc and no infall, problems such as the G-dwarf problem, the AMR, the constancy of the SFR and the Galactic metallicity gradient can be solved quite



easily, provided that the H_2 distribution in the Galaxy follows that of Bhat et al. (1985) rather than SSS (1984). In the next Chapter, the consistency and consequences of this model will be examined.

CHAPTER FOUR

GALACTIC CHEMICAL EVOLUTION II - CONSISTENCY AND CONSEQUENCES

4.1 INTRODUCTION

The previous Chapter outlined a model of chemical evolution of the Galactic disc using a form of the SFR, $\psi_s \ll \Sigma_{M_2}^k$, in addition to an explicit dependence for dark matter, finite initial metallicity and no appreciable infall over the majority of the age of the disc. It was shown that with a yield of metals taken to be a constant or proportional to the metallicity, a self-consistent model could be constructed to predict the G-dwarf problem, the age-metallicity relation, the constancy of the SFR in the solar neighbourhood and the Galactic metallicity gradient in good agreement with observations. The aim of this Chapter is to further test the consistency of this model and outline some further predictions.

Firstly, the model will be examined in the light of the available forms for the initial mass function of stars and details of stellar nucleosynthesis, in order to check values of the yield, the dark matter fraction and the returned fraction (Section 4.2). These considerations will then be used to see whether the assumption of a yield proportional to metallicity is justified (Section 4.3). The consequences of these results for the dark matter fraction will be discussed, leading to a consideration of the possible form

-10,3-

of dark matter in the disc (Section 4.4). The predictions of the model for abundance gradients in external spiral galaxies will be compared with observation (Section 4.5) and the amount and distribution of H_2 will be examined in detail (Section 4.6). Finally, the evolution of radial gradients in gas, stars and metals will be illustrated (Section 4.7), leading to a possible explanation of the shape of the H_2 distribution in the Galaxy and other spirals.

4.2 THE PRESENT VALUES OF THE DARK REMNANT FRACTION AND THE YIELD

The dark remnant fraction (D) was defined in Chapter 3, in terms of the IMF(ϕ) and its upper and lower mass limits, m_u and m_g, as

$$D(t) = \int_{m_{2}}^{m_{M}} (t) w_{m} \phi(m) dm + \int_{m_{1}}^{m_{0}} m_{0} \phi(m) dm \quad (4.1)$$

where a star of mass m leaves behind a remnant mass w_{m} , m_{2} being the mass above which the star collapses only to a neutron star or black hole but not a white dwarf and m_{0} is the mass limit below which the star ends up as a brown dwarf. Similarly the returned fraction R(t),

$$R(t) = \int_{m_{1}}^{m_{1}} (t) (m - w_{m}) \phi(m) dm \qquad (4.2)$$

where m, is the turn-off stellar mass corresponding to the main sequence lifetime of a star bounded by the running age the disc, and the yield, y_R ,

where p_{ZM} is the mass fraction of a star of mass m that is converted to metals and ejected.

1.5.1

In the solar neighbourhood, D,R and $y_{\mathbb{Z}}$ were estimated, independently of the IMF, by solutions to the G-dwarf problem and age-metallicity relation. It was found that for n = 1 (i.e. yield proportional to the metallicity) and for n = 0 (i.e. constant yield),

$$\frac{D}{(1-R)} = \frac{\delta_0 - \delta_0}{\mu_0 - \mu_0} = 0.1 - 0.45, \qquad (4.4)$$

$$R = 0.3 - 0.6, \qquad (4.5)$$

$$y_{\gamma} = KZ^{\prime\prime} = 0.019 - 0.025$$
 (4.6)

assuming $\S_1 = 0.5$ (Bahcall 1986c), $Z_{\textcircled{O}} = 0.02$ and $\bigvee_{1}^{0} = 4M_{\textcircled{O}}pc^{-2}Gy^{-1}$, the range due to the different assumptions concerning the yield and the uncertainty concerning Σ_{M_2} in the solar neighbourhood. Using different models, R has been calculated to be 0.35 - 0.8 (e.g. Twarog 1980; Tinsley 1981; Güsten and Mezger 1983), in agreement with the above values. The above values of the yield in the solar neighbourhood are somewhat higher than previous estimates $y_2 = 0.014 \pm 0.005$ (Pagel 1981) and ≈ 0.012 (Peimbert and Serrano 1982).

It is now important to compare these values, obtained from model-fitting in the solar neighbourhood, with the values obtained from their definitions (equations 4.1 -4.3). In order to do this, explicit forms for w_{m} , $\oint(m)$, m_{2} , m_{0} and $p_{\pi m}$ need to be critically examined.

4.2.1 The remnant mass function

Previous workers (e.g. Tinsley 1980; Tosi 1982) have represented the function w_{fm} in a very simple way: any star

-105-

with initial mass lower than $4M_{\odot}$ will become a white dwarf with $w_{\widetilde{m}} = 0.7M_{\odot}$ and any star with initial mass greater than $4M_{\odot}$ becomes a neutron star with $w_{\widetilde{m}} = 1.4M_{\odot}$. More recently, other workers (e.g. Larson 1986; Rana 1987a) have used

 $w_{col} = 0.38 + 0.15m$ (4.7)

which is predicted by stellar evolution calculations assuming a standard mass loss rate (Iben and Renzini 1983).

A crucial part of our understanding of the final fate of stars of differing initial mass concerns recent work on so-called intermediate mass stars (IMS) with the range of mass of 2.3 - $8M_{\odot}$. Figure 4.1 shows schematically the final fate of such stars. The upper limit for IMS, M_{up} , defined as the limiting mass between degenerate and semi-degenerate C-ignition, marks a sharp discontinuity in the evolutionary behaviour of stars (Renzini 1984). For initial masses $m < M_{up}$, stars experience the asymptotic Giant Branch (AGB) phase and eventually die, either as C-O white dwarfs having failed to ignite C, or ignite C under degenerate conditions thus experiencing a C-deflagration which causes the total disruption of the star (Nomoto 1984a). The latter is termed a supernovae event of type I⁴/₂ (Iben 1980).

For stars with $m < M_{WO}$ the final product is a C-O white dwarf remnant. From a theoretical point of view, the value of M_{WO} is uncertain owing to a poor knowledge concerning the mass loss process in stars, but values in the range 4-6 M_O have been widely used (Renzini 1977; D'Antona and Mazzitelli 1985). Observations of white dwarfs in young Galactic clusters seem to indicate that all single stars up to $(7 \pm 1) M_{\odot}$ become white dwarfs (Weidemann 1984; Koester and Reimers 1985). However, it has been noted (Iben 1983) that

-106-



Figure 4.1: The final fate of intermediate mass stars $(\approx 2-12M_{\odot})$. M_{WD} , M_{Up} and M_{Up} are defined in the text.

 M_{WO} may be a function of the metallicity Z, if the mass loss rate is a function of Z.

For stars with $m > M_{up}$, stars ignite carbon either semi-degenerately if $m < M_{u_0}$, or non-degenerately thus failing to experience the AGB phase and eventually undergo a core collapse giving rise to a supernova event of Type II, leaving a neutron star remnant (Nomoto 1984b,c; Woosley et al. 1984; Yahil 1984). Although Hillebrandt et al. (1985) suggested that less massive progenitors could lead to this type of collapse, this work follows Tornambe and Matteucci (1985) in identifying supernovae of Type II events with stars of initial mass m > M_{up} , thereby fixing the m_2 of equation (4.1) as M_{up} . As with M_{wp} , M_{up} has been proposed to be a function of Z (Becker and Iben 1980; Tornambe 1984; Chieffi and Tornambe 1984; Fujimoto et al. 1984). Recent suggestions (Matteucci and Tornambe 1985; Tornambe and Matteucci 1985; Castellani et al. 1985; Renzini et al. 1985; Bertelli et al. 1986;) indicate that the inclusion of

-107-

overshooting from convective cores and semi-convection in evolutionary models of IMS limits the mass $M_{_{UP}}$ to a value somewhat lower than $8M_{_{O}}$ and $M_{_{UP}}$ possibly decreases with decreasing Z for log Z > -4. In particular, for Z = Z_O, $M_{_{UP}} \cong 6M_{_{O}}$ whereas for lower metallicities (Z $\cong 10^{^{-1}}$), $M_{_{UD}} \cong 4-5M_{_{O}}$ (Tornambe and Chieffi 1986).

There is one further complication for the remnant mass function, w_{Θ} . Calculations of supernova collapse and explosion have shown that supernova explosions are expected for stars with initial mass at least up to $50M_{\odot}$ (Wilson et al. 1986). However, Tornambe and Matteucci (1985), computing a rough estimate of the current Type II supernovae rate by assuming that all stars more massive than $M_{U_{\odot}}$ become Type II require, in order to alleviate discrepancy between the theoretical and observational rates, that stars with initial mass > $50M_{\odot}$ collapse to black holes. In fact, Schild and Maeder (1985) from a consideration of the pulsar distribution, and helium abundance support this view. It is not easy to incorporate this effect into the remnant mass function.

In the light of the above discussion, three possible cases, as described in Table 4.1 are taken. White dwarf

Table 4.1: Parameters for the remnant function, Wm .

 $M_{up}(M_{\odot})$ M_{en} (M_G) M_{BH} (M₀) Case 6 100 5 A б 7 50 В С 5 6 100

masses are taken roughly in the region discussed by Weidemann and Koester (1983), linear from $1M_{\odot}$ to M_{WD} . Case

-108-

A follows Tinsley (1980) in assuming that for $M > 5M_{\odot}$, $w_{m} = 1.4M_{\odot}$. Case B increases w_{m} linearly between M_{up} and $M_{\odot M}$ where the minimum remnant mass for a black hole is assumed to be $3M_{\odot}$ (McDowell 1985). Case C is the relationship used by Larson (1986) and Rana (1987a), equation (4.7) with M_{tMD} and M_{up} as shown in Table 4.1.

4.2.2. The initial mass function

The frequency distribution of stellar masses at birth (IMF) can be estimated from the observed luminosity function if an assumption is made about the time dependence of the SFR (see Scalo 1986 for a comprehensive review). The constraint that the IMF should not show an unphysical discontinuity was used by Miller and Scalo (1979) to argue for a nearly constant SFR and an IMF which is a monotonically declining function of stellar mass. This conventional picture of the IMF has recently been challenged by evidence for a dip at $0.7M_{\odot}$ in the luminosity function (Upgren and Armandroff 1981; Scalo 1986). Larson (1986) has used this and new data compiled by Scalo (1986) to propose a bimodal model of the IMF. This model, where the IMF is double peaked consists of two components of similar form representing separate low-mass and high mass modes of star formation. Using the constraint on the high mass mode that the model should predict enough mass in remnants to account for the Bahcall (1984) amount of unseen matter near the Sun, he finds a reasonable match with Scalo's data if the SFR decays with time as exp(-t/3.4Gy). Rana (1987a) has used a different set of data on the luminosity function, scale heights and main-sequence lifetimes which differ substantially from Scalo (1986). Although his data are

-109-



Figure 4.2: The total number §(log m) of stars per unit log mass ever formed during the past 15Gy in a column of cross-section lpc⁻² through the Galactic plane, for different sets of data, and assumptions about the time-dependence of the SFR. The solid curve is the case of constant SFR for the data of Miller and Scalo (1979). The curve from Rana (1987a) uses a different compilation of data and a SFR from Figure 3.3 of this work. The data of Scalo (1986) are represented by two cases, a constant SFR case and the model of Larson (1986) which has a decreasing SFR with time.

closer to that of Miller and Scalo (1979) he also finds a bimodal, or even multi-modal IMF, which he interprets as possibly evidence of bursts of star formation giving birth to stars of a specific mass range. Due to disagreement concerning the data and the time-dependence of the SFR, four different IMFs, illustrated in Figure 4.2 will be contrasted in the following. With an assumed age of the disc to be 15Gy, the case for a constant SFR is shown for the data of Miller and Scalo (1979). The data of Scalo (1986) are

-110-

represented by the case for constant SFR and the model of Larson (1986) which has a decreasing SFR with time. The fourth IMF is derived from the data of Rana (1987a) using the SFR for the constant yield case of Chapter 3.

Neglecting the contribution to the dark remnant fraction due to brown dwarfs by setting to zero the second term on

		$m_{\ell} = 0.1 M_{\odot}$			$m_{\chi} = 0.007 M_{\odot}$		
Ø (m)	₩m	D	R	D/ (1-R)	D	R	D/(1-R)
Miller & Scalo(1979)	A	0.030	0.43	0.053	0.060	0.41	0.102
	В	0.014	0.44	0.025	0.045	0.42	0.078
	с	0.038	0.44	0.068	0.068	0.43	0.120
Rana (1987a)	A	0.025	0.43	0.044	0.060	0.40	0.100
	B	0.015	0.43	0.026	0.050	0.41	0.085
	с	0.036	0.44	0.064	0.070	0.42	0.121
Larson (1986)	A	0.050	0.63	0.135	0.051	0.63	0.138
	в	0.025	0.65	0.071	0.025	0.65	0.071
	ç	0.068	0.64	0.189	0.068	0.64	0.189
Scalo (1986)	A	0.012	0.30	0.017	0.014	0.30	0.020
	B	0.007	0.31	0.010	0.009	0.31	0.013
	С	0.018	0.32	0.026	0.020	0.32	0.029

Table 4.2: Dark remnant and returned fractions.

the right-hand side of equation (4.1) and integrating using $m_{fl} = 0.1 M_{\bigodot}$ and $m_{fk} = 100 M_{\bigodot}$ (Terlevich 1982) the resulting R and D for the three cases of w_{ml} and each IMF is shown for $m_{fl} = 0.1 M_{\bigodot}$ in Table 4.2.

In evaluating R, the age of the disc is assumed to be 15Gy, which using the stellar mass-lifetime relation of

-111-

Scalo (1986), gives $m_0 = 0.91 M_{\odot}$. The value of R turns out to be virtually independent of the three different functions of w_{\odot} but varies with different IMFs from $\approx 0.3 - 0.65$. These values are consistent with that determined from fitting the observational problems of chemical evolution.

However, apart from Larson's IMF with w_{fin} functions A and C (which has been constructed of course to give this result) the values shown for D are inconsistent with the predictions from fitting the G-dwarf problem and other parameters in the chemical evolution of the Galaxy. It should be noted that with function B for w_{fin} , which is probably the most physical representation, even Larson's IMF produces too little dark matter. Therefore, the question is can the dark matter fraction be increased by the contribution from brown dwarfs?

4.2.3 Brown dwarfs

Brown dwarfs are stars and planets with masses lower than that required for the onset of nuclear burning (Salpeter 1977). The maximum mass for a 'Jupiter-like' object, m_{\odot} , determined by the onset of nuclear burning is thought to be $0.08M_{\odot}$ (D'Antona and Mazzitelli 1985; Bahcall 1986b). From theoretical arguments, calculations for the lower limit of the Jeans mass have led to different estimates of m_{β} , in the range $0.004 - 0.007M_{\odot}$ (Low and Lynden-Bell 1976; Rees 1976; Silk 1982; Palla et al. 1983). Both $m_{\beta} = 0.007M_{\odot}$ and $m_{\beta} = 0.004M_{\odot}$ have been examined, but finding little difference the former will be used. In order to perform the calculation for D, including the second term on the right hand side of equation (4.1), the IMF needs to be extrapolated form 0.1 to $0.007M_{\odot}$. This is not

-112-

unjustifiable as m_{\odot} is simply the stage at which nuclear burning stops and has nothing to do with cloud fragmentation (Bahcall 1986b).

The column corresponding to $m_{\beta} = 0.007 M_{\odot}$ in Table 4.2, shows that even with the contribution from brown dwarf objects, D/(1-R) is still very small compared to the value obtained from fitting the G-dwarf problem. The extrapolation of the IMFs of Larson and Scalo contribute almost no extra dark matter as both decrease very steeply at m < 2M_{\odot}. Rana's IMF has some discontinuities at low mass which makes the extrapolation similar to that of Miller and Scalo (1979). Even the form of the Miller-Scalo IMF, which falls off at m < 0.1M_{\odot}, means that the classical suggestion of brown dwarfs as dark matter can have a meaning only if the IMF is very different at low masses (Tinsley 1981).

There is however, severe difficulty in estimating the IMF at low masses. The determination of the luminosity function is based on few stars, and it is not clear whether at low luminosity it flattens off, decreases or increases (e.g. Gilmore and Reid 1983; Gilmore et al. 1985; Scalo 1986; Liebert and Dahn 1986; Robin and Crézé 1986; Reid 1987). Recently Hawkins (1986) from a very deep R-band UK Schmidt survey for M stars to low luminosity and an analysis of the R-I colours and proper motion in 15 survey plates, has claimed the first direct detection of very low-luminosity stars. There seems an indication of a steadily increasing population into the regime of substellar masses which would of course increase the dark remnant fraction through brown dwarfs. Further uncertainty in determining the IMF comes through the scale-height to

-113-

magnitude relation and the mass to magnitude relation (Rana 1987a). In fact, D'Antona and Mazzitelli (1986), adopting the mass-luminosity relation from their models of low mass stars find that there is no decline in the IMF down to $0.1M_{\odot}$, a trend which if continued to lower masses would once again considerably increase D. However, an IMF similar to Miller and Scalo (1979) and not remarkably different at m < $0.1M_{\odot}$ would leave only the bimodal model of Larson (1986) able to produce the amount of dark matter required by the model of chemical evolution. This question will be discussed in detail in Section 4.4.

4.2.4 Production of heavy elements

The production of heavy elements by massive stars of constant mass has been studied by Arnett (1978) who followed the evolution of He bare cores. More recent studies have included the important property of mass loss (e.g. Maeder 1981, 1984; Prantzos et al. 1986). Transforming y' to $y_{\rm g}$ (Section 3.2.1), Maeder (1984) evaluating equation (4.3) obtains $y_{\rm g}$ = 0.012 - 0.02 using a Miller-Scalo IMF below 25M₀ and that of Garmany et al. (1982) above, and R = 0.17. Chiosi and Matteucci (1984) in fact quote a range of $y_{\rm g}$ = 0.0035 - 0.069 for different stellar IMFs, mass loss rates, overshooting and nucleosynthesis data.

In Figure 4.3, Maeder's (1984) variation of p_{ZM} as a function of initial mass, m, is reproduced for the intermediate case of mass loss. The minimum in the metal production at the limit of IMS and massive star models, found by various authors, is thought to be real (Audouze and Tinsley 1976; Mallik 1980, 1981).

With functions A, B and C for w_{6n} and the above values

-114-

of R and D, the Miller-Scalo IMF gives $y_Z = 0.025$ in good agreement with the value derived for the solar



Figure 4.3: The mass of metals, mp_{ZM}, returned to the ISM as a function of the initial mass of the star, reproduced from Maeder (1984). The equivalent curve for helium production is also shown for comparison.

neighbourhood. The IMF of Rana gives a slightly higher yield $y_z \simeq 0.031$ while that of Scalo gives a low $y_z \simeq 0.015$. The IMF of Larson (1986) gives a yield four times higher than that of Miller and Scalo. In his model he argues that this is not a constraint as the upper limit of metal production (m_{ZM}) is uncertain. He requires $m_{ZM} \lesssim 16 M_{\odot}$ to give $y_{\chi} \simeq 0.025$. If this is the case then stars more massive than $16M_{\odot}$ rather than $50M_{\odot}$ (cf. Section 4.2.1) must evolve directly to black holes with no ejection of metals. Twarog and Wheeler (1982) argued for an upper cut-off of $50M_{\odot}$ or oxygen would be overproduced in the solar neighbourhood, but Matteucci (1986) has suggested there is no clear evidence requiring m_{ZM} to be much lower than 80-100 Although the observed high &Y/&Z ratio (Pagel et al. M_o. 1986) may suggest a low mgu (Peimbert 1986; Pagel 1987),

-115-

such a low value of $m_{ZM} \simeq 16 M_{\odot}$ is potentially problematic. Olive (1986) describes the following problems. Arnett (1978) claimed that the typical heavy element producing stars are O stars with masses greater than $26M_{\odot}$. The typical source of metals in Larson's bimodal model is a star of $\approx 14M_{\odot}$. In addition, it has been noted (Truran 1984) that halo stars show abundance patterns which are typical of Another potential difficulty, in the context of 40M_@. abundance ratios, is that the bimodal model overproduces Ne and Mg by factors 2 and 4, respectively. Although there are considerable uncertainties in the yields, the new calculations of Woosley and Weaver (1986a,b) make the situation worse. Olive (1986) in order to relax this constraint has shifted the high-mass mode of Larson's IMF to lower mass, which fits the dark matter limit and yield with $m_{ZM} \leq 42M_{\odot}$. Olive et al. (1987) argue that uncertainties at present are too great to place severe constraints on the model.

Therefore, for certain IMFs and assumptions about the upper mass cut-off for metal production the present yield is consistent with that derived from the model of the chemical evolution of the solar neighbourhood. However, is the yield proportional to the metallicity or is it constant?

4.3 IS THE YIELD OF METALS A FUNCTION OF THE METALLICITY?

In many models of the chemical evolution of the Galaxy the yield has been taken to be a constant quantity. However, in order to solve problems associated with the simple model some workers have used a yield varying with

-116-

Galactocentric radius (e.g. Gusten and Mezger 1982). Edmunds and Pagel (1984a) argue for some evidence from spiral and irregular galaxies that the yield may be constant in time in any one region but depends on local conditions such as the total surface density. However, they do state (Edmunds and Pagel 1984b) that the yield could be metallicity dependent if the evolution of stars is significantly influenced by metal abundance, for instance through varying mass loss rate.

Peimbert (1984) has contended that a yield independent of metallicity, even with infall, does not agree with the observations of the yield of primary elements derived from HII regions located in spiral, irregular and dwarf galaxies. A yield increasing with the abundance of heavy elements, as proposed by Peimbert and Serrano (1982), better explains these observations than a constant yield. Serrano and Peimbert (1983) also require a yield increasing with metallicity to explain the N/O versus O/H diagram. However, other workers (White and Audouze 1983; Matteucci and Tosi 1985; Diaz and Tosi 1986) have argued that when the actual details of stellar nucleosynthesis are taken into account, a yield independent of Z is consistent with HII region observations for both spirals and irregulars.

Chapter 3 made an ad hoc assumption of $y_z \ll z^n$ and showed that both n = 0 and n = 1 could give adequate models for the chemical evolution of the Galactic disc. The consequences for the dark matter fraction and the mass of H_2 , however, are dependent on the value of n (Paper I and IV). Furthermore, the question still remains, is it reasonable to explain a yield proportional to the

-117-

metallicity from our present knowledge of the IMF and stellar nucleosynthesis in the Galaxy? The variation of the yield over the lifetime of the Galaxy will be given by the time dependence of the parameters in equation (4.3) namely $\int_{\infty}^{\infty}(m)$, $p_{_{2CM}}$, $m_{_{R}}$ and $m_{_{M}}$.

4.3.1 Variations in ()(m)

There is much debate concerning whether the IMF is constant in space and time. Garmany et al. (1982) suggested, based on counts of early type stars, that there were spatial variations in the upper IMF. Humphreys and McElroy (1984) subsequently showed that this result was due to a selection effect and the data favour a constant IMF. Observations of the IMF in other galaxies and even in globular clusters (e.g. Berkhuijsen 1982; McClure et al. 1986) do not show a large variation from the Galactic IMF. Terlevich and Melnick (1984, see also Terlevich 1985; Melnick and Terlevich 1986) by studying the properties of a sample of extragalactic HII regions, suggested a dependence of the slope of the IMF (represented by a power law of the mass) on the metal content of such objects given by

 $x = \log 2 + 5.05$

(4.8)

for stars in the mass range 0.1-100M_O. As the metallicity of a galaxy increases with time after the formation of the disc (Twarog 1980) this implies a time-dependence of the IMF. Although the work of Boisse et al. (1981) may be interpreted as implying variations with ambient conditions, Turner (1984) and Scalo (1986) have contended with very different arguments that the IMF is grossly the same everywhere. Furthermore, Matteucci and Tosi (1985) have shown that the Terlevich-Melnick IMF does not reproduce the

-118-

observed data in dwarf irregular galaxies, suggesting that the IMF dependence on the metallicity, if any, should be much weaker than that given by equation (4.8).

Therefore, significant variations in the IMF may be rare enough for the assumption of a universal function to be a reasonable first approximation (Tinsley 1980; Lacey and Fall 1985). In any case, even if the IMF is variable it is not clear how this variability should be parameterised or how it affects chemical evolution (Pagel 1987). In the following $\phi(m)$ will be assumed to be constant in time, and once again the four IMFs of Section 4.2.2 will be contrasted.

4.3.2 Variations in pRM

The minimum in p_{ZM} is at the limit of the IMS and massive star models and therefore corresponds to M_{Up} . Any time dependence of M_{Up} may affect the shape of the curve. From Figure 2 of Tornambe and Matteucci (1985), M_{Up} can be approximated for 2 in the range 10^{-3} to 10^{-1} over the age of the Galaxy using the expression

 $M_{up} = 8.0 + 0.8 \log Z$ (4.9) so that for Z = 0.02, $M_{up} = 6.6M_{\odot}$ (cf. Tornambe and Chieffi 1986). Therefore, the minimum in Maeder's curve (Figure 4.3) should be at $6.6M_{\odot}$, not $9.5M_{\odot}$ and should move with metallicity. As it would be a very complicated calculation to produce curves of p_{ZM} for different Z rigorously, a crude approach is adopted, adequate for our purposes. Fixing points at m = $1M_{\odot}$ and m = $50M_{\odot}$, the abscissa is then re-scaled to match with the required position of the minimum. If m" corresponds to the mass for a certain p_{ZM} given by Maeder (1984), then let m be the corresponding mass

-119-

for the metallicity-dependent revised curve. Now $M_{u_0}^{"} = 9.5 M_{\odot}$ for Maeder, M_{u_0} is given by equation (4.9), and a simple transformation is adopted relating m and m" to be given by

$$m = (M_{up} - 1) (m'' - 1) + 1 \qquad m'' < M_{up}'' \qquad (4.10a)$$

and
$$m = (50 - M_{up})(m^2 - 50) + 50$$
 $m^2 > M_{up}^2$ (4.10b)
(50 - M_{up}^2)

The change of metallicity with time in the Galaxy, changes M_{up} , thereby altering the shape of the $p_{y_{max}}$ curve. It should be stressed that this simple scaling of the yield for different values of Mup might be dangerous, since the actual effects of overshooting on the size of the stellar envelope and on the corresponding nucleosynthesis has not yet been computed (Renzini 1984; Greggio and Tosi 1986). However, using this approach in the following, equation (4.9) does not predict a very large effect, that is for Z to increase from ≈ 0.005 to 0.02, Mup increases from ≈ 6.2 to 6.6 Mo. The AMR of Twarog (1980) is taken with log $(Z_{i}/Z_{O}) = 0.57$ (from Section 3.4) for the Z versus t relation over the lifetime of the disc. This change in Mus will have an effect on the remnant mass function, but because the change in M_{up} is small, w_m will remain roughly constant with time. Case A is used for simplicity.

The turn-off stellar mass corresponding to the main sequence lifetime of a star bounded by the age of the disc t, m,, essentially controls the number of stars which contribute to chemical evolution after the formation of the The mass-lifetime relation from Scalo (1986), has disc. been used to evaluate m_{μ} as a function of age of the Galactic disc.

-120-

4.3.3 Variations in mg and mg

The upper mass limit of the IMF, m_{A} , is assumed to be 100M_® at present (Terlevich 1982; Humphreys and McElroy 1984). If the contribution of brown dwarfs is neglected, m_{fl} may be taken as $0.1M_{\odot}$ (Larson 1981). Figure 4.4 shows the variation of y_{Z} , D and R as a function of time with m_{fl} and m_{A} constant with time, for each of the IMFs described previously. This results in the yield, the returned fraction and the dark remnant fraction being fairly constant with time. Also plotted is the yield as a function of the metallicity in order to look for a possible relation $y_{Z} \ll Z^{n}$ using the AMR of Twarog (1980). If such a relation does exist then the value of n would be quite small, $n \cong 0.0 - 0.1$.

While modelling stellar atmospheres, Kahn (1974) proposed that the upper mass limit m_{μ} could vary with metallicity Z as

$$m_{\rm h} \propto 2^{-0.5}$$
 (4.11)

In Figure 4.5, the solid curve shows the results for the Miller-Scalo IMF of keeping $m_{\hat{I}}$ constant at 0.1M₀ while m_{M} is varied as in equation (4.11), once again using the AMR of Twarog (1980) and $\log(Z_1/Z_0) = 0.57$. Therefore, m_{M} decreases from 197 M₀ when the disc formed to 100 M₀ at present. The resulting variations are very similar to Figure 4.4, except that the metal yield is slightly increased at early times (less than 5Gy) suggesting an overall $n \cong -0.1$. Because the IMF is falling rapidly at high m, this variation in m_M has little effect. The other IMFs show similar shapes and therefore only the Miller-Scalo IMF will be used to illustrate the following discussion.

-121-



Figure 4.4:

The variation of R, D and $y_{\mathbb{Z}}$ with time, and $y_{\mathbb{Z}}$ with metallicity. The case of time independent $m_{\hat{\mathbb{N}}}$ and $m_{\underline{M}}$ neglecting the contribution of brown dwarfs (i.e. $m_{\hat{\mathbb{N}}} = 0.1 M_{\Theta}$) for each of the IMFs described in Section 4.2.2. Solid curve: Miller and Scalo (1979); dotted curve: Rana (1987a); short dashed curve: Larson (1986); long dashed curve: Scalo (1986). Silk (1977) proposed that the lower mass limit could vary as

$$m_{g} \approx Z^{2}$$
 (4.12)

Taking the variation of metallicity as before, this means that $m_{\hat{l}} = 1.38 \ M_{\odot}$ when the disc formed and 0.1 M_{\odot} at present. This variation is significant because in this region of initial mass the IMF is maximum. The dotted curve in Figure (4.5) shows that R, D and $y_{\mathbb{Z}}$ decrease with time giving a yield roughly inversely proportional to the metallicity (n \cong -1.0), which is just the opposite of the usual assumption made for a variable yield (Paper I; Peimbert and Serrano 1982).

However, if the contribution of brown dwarfs is included, with a variation in m_{R} given by (4.12), m_{R} decreases from 0.097 M_☉ to the assumed present value of 0.007 M_☉. In this range of initial mass the IMF is once again falling off and therefore the m_{R} variation has little effect on R, D and y_{Z} . The two other plots in Figure (4.5) correspond to the variation in m_{u} and m_{R} described above but with $m_{R} = 0.007 M_{\odot}$. It is observed that this variation has little effect, except for the absolute value of D.

4.3.4 Other possibilities

It is clear from the above that a 'conventional' variation of the upper and lower mass limits will not justify an assumption of $y_z \ll 2$. In fact, if a relationship of the form

 $y_{g} \ll Z^{n}$ (4.13) exists at all then n g 0 is favoured with n g -1 for a Silk variation of the lower mass limit of the IMF. Therefore in



Figure 4.5: The variation of R, D and y_Z with time and y_Z with metallicity. The effect of varying m_{\parallel} and m_{μ} as a function of metallicity for the Miller-Scalo IMF. Solid curve: $m_M \ll Z^{-0.5}$, $m_{\parallel} \approx 0.1M_{\odot}$; dotted curve: $m_M \ll Z^{-0.5}$, $m_{\parallel} \approx 0.007M_{\odot}$; short dashed curve: $m_M \ll Z^{-0.5}$, $m_{\parallel} \approx 0.007M_{\odot}$; long dashed curve: $m_M \approx 100M_{\odot}$, $m_{\parallel} \ll Z^{-2}$ (including brown dwarfs).

order to produce a relationship of the form (4.13) with $n \ll 1$, there must be more radical changes with metallicity either of stellar nucleosynthesis or the IMF.

A yield inversely related to the mass loss rate in massive stars has been shown by Chiosi and Caimmi (1979), but observations seem to indicate that the metallicity increases the mass loss rate (Maeder 1980) that is, again the wrong way, n = -1. Mallik and Mallik (1985) by varying the upper cut-off mass in p_{ZM} due to the total collapse of stars into black holes have shown that a variable yield with $n \le 1$ can be achieved. However, this assumption is difficult to justify particularly for low cut-off masses in p_{ZM} . Section 4.3.2 showed that recent work on IMS may suggest p_{ZM} changes with Z, but a stronger dependence than that given in equations (4.9) - (4.10) is required to give n = 0 These remain possibilities but more work needs to be done before firm conclusions are drawn.

Peimbert and Serrano (1982) give two possibilities to change the IMF, either the relative amount of more massive stars increases with Z, or the relative amount of low mass stars decrease with Z. They argue from the work of Lequeux et al. (1980) that the former is not likely and further cannot be important because they claim it would increase the $\Delta Y/\Delta Z$ ratio with Z contrary to observations. Terlevich and Melnick (1984)'s variation of the IMF through equation (4.8) would flatten the IMF at high mass early in the evolution of the Galaxy, that is decreasing the relative amount of low-mass stars with Z. A change in the relative amount of low mass stars is more likely but there are conflicting

-125-

views concerning how this changes with metallicity. Silk (1977) and Reddish (1978) predict that the relative number of low-mass stars decreases with Z while Larson (1981) predicts exactly the opposite.

The situation concerning the dependence of the yield on metallicity is therefore confused with a possible relation such as (4.13) being valid, but the exponent n having possible values from -1 to 1. However, n = 0, that is a constant yield, seems to be strongly favoured and is probably the best assumption to make at present.

4.4 THE NATURE OF THE DARK MATTER IN THE SOLAR NEIGHBOURHOOD

The suggestion that the Universe contains substantial dark matter, that is mass without corresponding light, is now generally accepted although the two key questions of what fraction of the mass is dark and what is its nature remain controversial (e.g. Rees 1985; Tayler 1986). This so called dark matter problem occurs on a variety of scales; and each scale may provide a distinct problem.

On a cosmological scale, if the cosmological density parameter \mathbf{A} is believed to be unity (e.g. Guth 1981) then a large fraction of the total mass of the Universe must be non-baryonic since primordial nucleosynthesis requires the fraction of \mathbf{A} due to baryons < 0.2 (Yang et al. 1984). From virial mass estimates a dark matter problem appears in galaxy clusters (Cowie et al. 1987). The existence of flat rotation curves-in the Galaxy and other spirals points towards the halo consisting primarily of dark matter (Faber

-126-

far se

and Gallagher 1979). Heqyi and Olive (1986) have discussed the observational and physical difficulties with the dark halo mass being in the form of baryons and conclude that the halo dark matter must be primarily non-baryonic (cf. McDowell 1985; Hills 1986). A popular candidate is therefore cold dark matter (Bond and Szalay 1983), a class of weakly interacting Majorana-mass fermions. However, it should be noted that the argument of Hegyi and Olive (1986) against low mass stars or brown dwarfs depends crucially on the assumption that the halo IMF is not substantially different from the present day disc IMF, an assumption which may or may not be true. Stars with very low stellar masses have been suggested for some time for the dark matter in the halo (e.g. Ostriker et al. 1974; Gott 1981). Recently, Carr and Lacey (1987) have examined the possibility of the existence of $10^{6} M_{\odot}$ dark clusters composed of low mass objects in the halo, in order to explain the age-velocity dispersion relation of disc stars. This relation could possibly also be explained by the halo consisting of $\sim 10^{\circ} M_{\odot}$ black holes (Ipser and Semenzato 1985; Lacey and Ostriker 1985).

Recently the existence of dark matter in the Galactic disc has caused considerable interest. Bahcall (1984) determined the total mass density in the solar neighbourhood by solving for the gravitational potential in the disc and found $(\Sigma_{\overline{v}})_{\odot}$ may be anything between 65 and 135 M_☉pc⁻², but the most preferable value is quoted to be 75 M_☉pc⁻². The stellar contribution to this mass density is uncertain (Rana 1987a). Bahcall (1986c,d) suggests that the ratio of the dark to luminous matter, that is $S_{0}/(\mu_{1} + \sigma_{1})$ lies in the

-127-

range 0.5 = 1.5, and most of the unseen component must be distributed with a scale height no larger than that for old disc stars.

This dark matter is most probably baryonic, as its concentration towards the Galactic plane suggests that energy dissipation has occurred which is difficult to understand if weakly interacting particles are involved. It is not clear however what form the baryons take. The halo dark matter makes a negligible contribution to the disc (Hegyi et al. 1986; Bloemen and Silk 1987) and observations rule out the dark matter residing as ordinary baryonic gas (Spitzer 1978). Small, solid particles such as dust grains, interstellar meteors or interstellar comets can be ruled out from a variety of arguments (e.g. Tayler 1986; Hills 1986). Hegyi et al. (1986) have argued from X-ray emission and chemical enrichment that the dark matter is not in the form of black holes, which seemed to be consistent with the study of the survival of wide binaries in the solar neighbourhood (Bahcall et al. 1985) implying that the unseen disc objects must have a mass < 2 M_{\odot} . This interpretation of wide binaries has been subsequently challenged by Wasserman and Weinberg (1987) who conclude that no useful upper mass limit can be set on the unseen objects as yet. Hills (1977) however, has pointed out that if the dark matter was in the form of massive black holes, exchange collisions would cause conspicuous accumulation in binary stars.

The model of chemical evolution outlined in Chapter 3 and examined in more detail earlier in this Chapter, includes an explicit form for dark matter. In this context, an examination of the possible nature of the dark matter in

-128-

the solar neighbourhood is now extended. A constant yield will be used, following the results of the previous section, and for ease of presentation the Bhat et al. (1985) value of Σ_{M_2} in the solar neighbourhood will be used, that is $\mu_1 = 0.09$.

The value of the returned fraction R, which is computed in the model independently of the IMF and the remnant fraction $w_{_{ini}}$ (see equation 3.32) is directly related to the residual amount of gas $(\hat{\mathbb{Z}}_{g})_{\odot}$, and the amount of stars formed during the lifetime of the disc if the present rate of star formation continued in the past, that is $(\forall_i)_{i=1}$, apart from a weak dependence on μ_0 . The value of $(\Psi_0)_{\otimes}$ is not a well-determined quantity ranging between 0.5 Mpc² Gy⁻¹ (Talbot 1980) to 5.0 $M_{\odot}pc^{-2}Gy^{-1}$ (SBM 1978). Since R cannot be negative, from this present work a lower limit can be placed on $(\psi_1)_{\odot}$ to be given by $(\psi_1)_{\odot} > 1.7 M_{\odot} \text{ pc}^{-2} \text{ Gy}^{-1}$ for $t_1 = 15$ Gy. The upper limit to $(\psi_1)_{o}$ depends on how close to unity the value of R can be pushed. Using a Miller-Scalo IMF and the usual forms for w_m , it was shown in Section 4.2.1 that R should be about 0.4, which from equation (3.32) gives

$$(\psi_1)_0 = 3 M_0 p c^2 G y^{-1}$$
 (4.14)

and this value has been chosen for developing the rest of the present section.

Noting the uncertainty quoted by Bahcall (1986c), a range in the present dark fraction, S_1 , has been taken as 0.2 - 0.7. For each value of S_1 several values of μ_{\odot} have been chosen and k constrained in order to produce the required (observed) metallicity gradient, dlog Z/dR = $-0.06 \text{ dex kpc}^{-1}$. Table A.4.1 (Appendix) shows all

-129-

relevant quantities in the chemical evolution model for certain selected cases, where \mathscr{K}_{4D}^2 and \mathscr{K}_{AMR}^2 are computed for the G-dwarf problem and AMR as outlined in Chapter 3. Figure 4.6 is a plot of the initial gas fraction μ_0 (= 0.95 - S_0 , as \mathscr{C}_0 = 0.05) versus the present dark matter fraction S_1 . The region corresponding to $S_1 < S_0$ is



Figure 4.6: The allowed values for μ_0 , δ_0 and δ_1 in order to produce a metallicity gradient of -0.06 dex kpc⁻¹. The region $\delta_1 < \delta_0$ is excluded as the dark matter fraction is an increasing function of time in the case of no infall.—The solid lines represent the loci of allowed parameters requiring constant yield, y_z , and the dotted ones represent the required ratio D/(1-R).

excluded, as the dark matter fraction is an ever increasing function of time for a model that does not invoke infall. The solid lines represent the loci of combinations of S, and μ_0 requiring constant yield y_Z , and the dotted areas represent the required ratio D/(1-R). This Figure suggests the following points:

(i) Using Maeder's curve for p_{Rim} and the Miller-Scalo IMF

-130-

smoothly extended to include brown dwarfs, it has already been found $y_{\mathbb{Z}} = 0.024$ and D/(1-R) can be no larger than 0.1. Now in Figure 4.6, the appropriate curves do not overlap implying from the point of view of chemical evolution, $y_{\mathbb{Z}} \cong 0.024$ is inconsistent with the required value of D/(1-R), which is too high possibly greater than 0.3. They would be consistent if $y_{\mathbb{Z}} = 0.015$ and D/(1-R) = 0.1, the former, however cannot be ruled out considering uncertainty involved in evaluating $p_{\mathbb{Z}}$ as a function of m.

The IMF of Rana is even worse. For $D/(1-R) \approx 0.1$, $y_{\chi} \approx 0.03$ which requires D/(1-R) from Figure 4.6 to be ≈ 0.5. For Larson's IMF consistency might be achieved with an upper mass cut-off to metal production less than 16 M_{\odot} to give $y_{g} \cong$ 0.02, and D/(1-R) \cong 0.2. This would predict \mathbb{S}_{1} = 0.57 and μ_{0} = 0.47, but there is every indication that m_{ZM} should be > 16M_® (Olive 1986). The IMF for the constant SFR case of Scalo (1986) gives too little dark matter compared to Bahcall's value in the solar neighbourhood. To satisfy the condition $y_{\chi} = 0.024$ strictly for the (ii) Miller-Scalo IMF it would imply the following. Depending on the value of μ_0 , δ_1 should lie in the range 0.5-0.6. If μ_0 is taken to be somewhat less than unity, say $\mu_0 = 0.65 \pm 0.15$ leaving enough room for producing the initial metallicity by a pre-disc generation, $\boldsymbol{\delta}_{i}$ is found to be 0.55 \pm 0.05 which suggests $(\Sigma_{b})_{0} = 41 \pm 4 M_{0} pc^{2}$ and $(\Sigma_s)_0 = 30 \mp 4 M_0 \text{pc}^2$ for $(\Sigma_T) = 75 M_0 \text{pc}^2$. Since, like the metallicity, the change in § from $\mathfrak{S}_{0}(\cong 0.25)$ to \mathfrak{S}_{1} during the evolution of the disc is comparable to S_{\odot} , the dark matter in the solar neighbourhood is possibly totally baryonic. The required value of D/(1-R) is then

-131-

= 0.4 + 0.1. This condition can be met with the hypothesis of a large population of brown dwarfs, provided the IMF at the lower mass end is remarkably different from the one suggested by Miller and Scalo. The hypothesis of black holes can hardly do the job without changing R and $\boldsymbol{y}_{\!\mathcal{R}}$ appreciably. For brown dwarfs, changes can always be made in the lower end of the mass spectrum (Section 4.2.3) in order to upgrade the ratio D/(1-R) and the revised value of R, if any, can be readjusted with the value of $(n_k)_{\odot}$. (iii) If y_{χ} turns out to be \cong 0.015 (calculated from $p_{\chi_{\text{PA}}}$ and the IMF) it would basically support the Miller-Scalo form of the IMF which advocates very few brown dwarfs. This solution will call for similar or somewhat higher values of μ_{0} , but δ_{1} has to be less than 0.45. The required value of S_{0} is however \mathfrak{A} 0.35 which must exist prior to the formation of the disc. This implies a remarkably different IMF for the pre-disc population, if S_{∞} is baryonic. From theoretical derivations of the IMF for zero-metal stars, Yoshii and Saio (1986) suggest that the IMF could be substantially different, but preliminary estimates of dark matter produced by this IMF do not provide the required $\mathbb{S}_{\mathbb{O}}$ (Rana 1987a). It is difficult to imagine how a substantial fraction of So could be non-baryonic because of the difficulty of confining the matter into the shallow potential well of the disc.

(iv) For the more recent IMFs the situation varies. Rana's IMF would follow the above discussion but would require a modest mass cut-off to metal production in order to give $y_{\mathbb{Z}}$ < 0.03. The IMF of Larson (1986) can provide $D/(1-R) \cong 0.2$ in the form of dead stellar remnants rather

-132-

than brown dwarfs. However, a small upper cut-off mass to metal production, < 16 M_☉, is needed for consistency in the solar neighbourhood, which as described previously may be unrealistic. A further uncertainty concerns the cooling time of white dwarf remnants (e.g. Olive 1986; D'Antona and Mazzitelli 1986). A modification of Larson's bimodal IMF may overcome these problems (Olive 1986). The IMF of Scalo (1986) with constant SFR cannot provide dark matter in baryonic form to solve the dark matter values given by Bahcall (1986c).

(v) Finally, it should be noted that the actual estimation of $\$_1$ at present is very uncertain. Some estimates of $(\$_1)_0$ and $(\$_8)_0$ at present in the solar neighbourhood may imply an overestimate of $\$_1$ by Bahcall, suggesting little dark matter at all (e.g. Gilmore and Wyse 1987; Bienayme et al. 1987). Furthermore, Rana (1987a) in a detailed study of the IMF and an extension of the above discussion, argues that it is possible to satisfy the observational constraints on the local dynamical mass with very little dark matter. If this is the case then bimodal IMFs and other modifications are not necessarily required.

4.5 METALLICITY GRADIENTS IN SPIRAL GALAXIES

The existence of abundance gradients in spiral galaxies provides a good observational test for many theories of the chemical evolution of galaxies. However, the need for high quality data for the observed mass distributions and metal abundances restricts the number of external spirals for which reliable results can be obtained. It was shown in the previous Chapter, that given an SFR of the form $\sqrt[n]{8} \ll (X \gtrsim_{\gamma} \mu Z^{(k)})^{(k)}$, the metallicity gradient is an outcome of k # 1 and the radial variation of \gtrsim_{T} for a closed model of the evolution of the Galactic disc. Section 4.3 showed that a constant yield was at present preferable to a metallicity dependent yield, and so $y_{(2)} = \text{constant}$ (or n = 0) will be used in the following. The variation of the present metallicity, Z_1 at different $R_{(2)}$ was then derived, and is now re-written as

$$\int_{Z_{\odot}}^{Z} \frac{dZ}{kb} = E't \Sigma_{r}(R_{G}) \qquad (4.15)$$

$$Z \exp[(1-k)GZ]$$

to show explicitly the time dependence. The constant, E' can be evaluated in the solar neighbourhood at present (t, = 15 Gy),

$$E' = \frac{(1-R)CX^{k}}{G\mu_{\Theta}Z_{\Theta}^{k\Theta}} \exp[(k-1)GZ_{\Theta}]$$
(4.16a)

$$= \frac{1}{t_{1}(\mathbb{Z}_{7})^{k-1}_{\odot}} \int_{Z_{0}}^{Z_{1}_{\odot}} \frac{dZ}{kb}$$
(4.16b)
$$Z_{0}_{\odot} = \exp[(1-k)GZ]$$

This allows the calculation of the metallicity gradient at any time t during the evolution of the Galactic disc. It was shown that for $t_1 = 15$ Gy, the present metallicity gradient of - 0.06 dex kpc⁻¹ was produced for k \approx 1.4, b = 1.3 (from Chapter 2) and G derived from the fit to the G-dwarf problem (Paper IV and Table A.4.1).

Previous workers have attempted to derive metallicity gradients in spiral galaxies in terms of the gas fraction (e.g. Clayton 1987; equations 3.36 and 3.37). However, this is not particularly satisfactory due to uncertainty in estimating the gas density $\mathbb{Z}_{\mathfrak{g}}$. For a number of galaxies only

-134-

 $\mathbb{Z}_{ extsf{MS}}$ is available, neglecting the importance of $\mathbb{Z}_{ extsf{M}_2}$. Even in galaxies that have been mapped in CO, the $CO = >H_{\lambda}$ conversion factor is uncertain by a large factor. Therefore, the use of μ to predict abundance gradients must be used tentatively. Tosi and Diaz (1985) in their models which show reasonably good agreement with metallicity gradients in nearby spirals, recognise that these derived gradients are strictly related to the gas fraction and thus depend on the methods employed in the derivation of the gas distributions. Furthermore, the derivation of the metallicity gradient from A does not illustrate its production from a flat initial metallicity distribution, it simply derives a gradient from an already existing μ gradient. It was for this reason that the form (4.15) was adopted linking the present metallicity to the variation in the better determined quantity \mathbb{Z}_{r} . Of course the metallicity is related to the gas fraction (equation 3.17), and later $\mathfrak{D}_{\mathfrak{M}_2}$ will be derived from the observed metallicity gradient rather than vice-versa.

Lequeux et al. (1979) and Kinman and Davidson (1981) found that the abundance of metals in irregular galaxies increased systematically with total mass, in qualitative agreement with equation (4.15) if k > 1. Although it has been argued that this holds for only a few low mass irregular galaxies (Hunter et al. 1982), a relationship between metallicity and total surface density across the discs of spiral galaxies has been observed (McCall 1982; Edmunds and Pagel 1984a; Garnett and Shields 1987). In Figure 4.7a, a plot of abundances against surface mass density G_{cl} of stars in the disc is reproduced from Edmunds and Pagel (1984a). The surface mass densities have been

-135-


Figure 4.7

(a) Plot of oxygen abundance against surface mass density of stars, reproduced from Edmunds and Pagel (1984a). The eye-fit curve to the Scd data is repeated in the plot for early spirals for comparison.

(b) The metallicity at time t after formation of a galactic disc with initial metallicity Z_{\odot} , as a function of the total surface mass density. This is the constant yield model, which gives good fits to the Galactic disc, for k = 1.4.

derived by McCall (1982) following a method due to Monnet and Simien (1977) which assumes that the mass distributions in the bulge and disc follow the corresponding light distributions; with this assumption the rotation curve is fitted with just two parameters, namely the mass-to-light ratios of the bulge and the disc respectively. As Edmunds and Pagel (1984a) point out, this means that the effect of unseen halo material is not explicitly taken into account except through its indirect influence on the derived mass to light ratios. This probably does not impair the physical significance of *G* because in typical observed spirals there is a nearly unique relation between the surface brightness and total mass surface density (Petrou 1981).

Figure 4.7a shows that there is a well defined average line for Scd galaxies (shown by the eye-fit curve of Edmunds and Pagel 1984a) and that this same line essentially forms a lower envelope to the abundances found in galaxies of earlier morphological type. At a given of early type galaxies appear to have systematically higher abundances than late type galaxies (McCall 1982). Now, the fitting of the G-dwarf problem in the solar neighbourhood at $t_1 = 15$ Gy implied $\log(Z_1/Z_{\odot}) = 0.57$ and assuming Z_{\odot} was a constant with radius at time of formation of the galactic disc, equations (4.15) and (4.16) give Figure 4.7b where $\log(Z(t)/Z_0)$ is plotted against Σ_{γ} for different values of This uses parameters determined to give good fits to t. Galactic chemical evolution. Comparison of Figures 4.7a and 4.7b suggests the following points:

(i) Not surprisingly, since the model is fitted to the Galactic disc, the present trend between metallicity and

-137-

total surface density is reproduced. This is due primarily to the adopted SFR (Chapter 2) with k = 1.4. The fact that other galaxies follow this trend further suggests that this law of star formation is applicable in external spirals (cf. Section 2.5.2).

(ii) The shape of the Z(t) versus $\mathbb{Z}_{\mathbb{T}}$ relationship is determined by the age of the disc, t. For any particular, Σ_r , Z increases with t, that is the age-metallicity relation of Section 3.4, but the rate of change of Z with time is a function of \mathbb{Z}_{r} . For t > 15 Gy, Z begins to saturate for high \mathbb{Z}_{γ} , while for t = 10 Gy in the same regime Z is still increasing rapidly. This may provide a tentative explanation of the difference in the relationship for Scd and early spirals. The Scd galaxies follow very well the curve for t ≈ 10 Gy, which may indicate that these galactic discs formed more recently than the Galaxy. There is much more scatter in the data for early spirals, but there does appear to be a flattening at higher Σ_r indicative of t \geq 15 Gy (for example the galaxies NGC 4321 and M83). The spread in metallicities in Figure 4.7a could be (iii) therefore due simply to different lengths of time of chemical enrichment. Of course, the large error associated with observation and then calculation of abundances, in addition to poorly known depletion factors will contribute a large amount to the scatter. For example, in some galaxies at a particular 9_{d} there is a spread in 12+log(O/H) of \approx 0.5 dex. It was argued in Section 3.6, that the upper envelope of metallicity observations in the Galaxy gave the correct distribution of the present metallicity Z_{I} . This may be also the case in Figure 4.7a, but the situation is

-138-

far from clear. The problem once again is how to translate the computed value of Z/Z_{\odot} onto the 12+log(O/H) plot. It may be an oversimplification to assume that all of these galactic discs formed with roughly the same initial metallicity. Even so, the adoption of an initial metallicity of $12 + \log(0/H) = 8.3$ taken for the Galaxy, would leave only a few points in the Scd galaxies below this baseline (and within 2g). It is interesting that both sets of data seem to flatten off as $\varpi_{1} \rightarrow 0$, to 12+log(O/H) \approx 8.3. In these outer parts of spiral discs little star formation (and therefore little chemical enrichment) is observed and therefore observations could be quite close to the initial metallicity. If this were true then the higher 12+log(O/H) at a particular Σ_{γ} , the longer that region has been undergoing star formation.

However, there is a second important uncertainty and that is are $\Sigma_{\hat{\Gamma}}$ and $\mathfrak{S}_{\hat{C}}$ directly comparable? Edmunds and Pagel (1984a) discuss the uncertainty in determining $\mathfrak{S}_{\hat{C}}$ especially in the outer parts of spiral discs where it is low, and suggest that it is possibly underestimated in this region. The value of $\Sigma_{\hat{\Gamma}}$ is the total mass, that is visible and invisible mass. It is possible to calculate $\Sigma_{\hat{S}}$ (the surface density of visible stars) as a function of galactic radius and t from the chemical evolution model (Section 4.7) and the resulting $\log(\mathbb{Z}/\mathbb{Z}_{\hat{O}})$ versus $\log \mathbb{Z}_{\hat{S}}$ plot shows the same shape as Figure 4.7b with a shift in the x-axis. These difficulties in fixing $\mathbb{Z}_{\hat{O}}$ and $\mathfrak{S}_{\hat{O}}$ in Figure 4.7a, mean that Figure 4.7b has not been overlayed. These uncertainties however, do not change the shape of the relationship, which is determined by t.

-139-

(iv) Tosi and Diaz (1985) constructed detailed models of nine of these galaxies, and were able to obtain reasonable fits by varying the e-folding time of the SFR and the infall rate (assumed constant and uniform) for each Galaxy. This present work suggests that metallicity gradients in spirals can be explained in the context of a closed model, with a SFR of $\Psi_{S} \ll \Sigma_{M_{2}}^{1.4}$ simply from the variation of $\mathbb{Z}_{\mathbb{T}}$. The actual gradient in any one galaxy will depend on a number of factors, but it is tentatively suggested that the age of the disc may be important in determining the shape of the metallicity gradient. Detailed studies and better abundance data of individual galaxies are required before firm conclusions are drawn, but if this is the case, then the relationship between metallicity gradient and total surface density may indicate the age of a galactic disc.

4.6 THE RADIAL DISTRIBUTION OF H2 IN THE GALAXY

The SFR on the above model is related to the surface density of H₂ by the relationship $\psi_{S} \ll \sum_{W_{2}}^{k}$, with the index k being independently determined from the required fit to the observed metallicity gradient. So this value of k may then be used to infer either $\Sigma_{W_{2}}$ or ψ_{S} provided the other is known. Using a constant yield, it has already been shown that in order to produce a Galactic metallicity gradient of -0.06 dex kpc⁻¹, k is required to be in the range 1.35-1.48 with the permitted choices for δ_{1} and μ_{0} . This Section assumes k = 1.40 and uses the available radial distributions of the star formation rates ψ_{S} to derive the corresponding radial distribution of $\Sigma_{W_{2}}$ in the Galaxy. This distribution

-140-

ين. محمد ريم اليالي معني أجالي is then compared with the existing controversial estimates of Σ_{M_2} derived from the observed CO emissivity distributions, and implications derived for the universality of the CO-+>H₂ conversion factor, \mathcal{A}_{20} .

4.6.1 A relative distribution of H₂ from SFRs in the Galaxy

In Figure 4.8, available data on the radial distribution of the present SFRs normalised to their respective values in the solar neighbourhood are plotted as a function of Galactocentric radius. Values of $rac{1}{26}$ derived from studies of Lyman continuum emission from HII regions are shown for SBM (1978) and Mezger (1978). The two works sometimes disagree at the 20 level because of large discrepancies in the individual estimates of the lifetime of HII regions and of dust absorption corrections. (Both sets of data have been included to show the actual uncertainties involved in the estimation of $\frac{1}{2}$ from HII regions). Also shown are the SFRs derived from the pulsar distribution (Lyne et al. 1985) and the supernova remnants (Guibert et al. 1978). These surveys are obviously limited mainly to the solar side of the Galaxy, but reasonable models for the pulsar and SNR distributions over the whole Galaxy have been used to infer the SFRs averaged over Galactocentric rings of different radii. However, since the mass ranges of the progenitor stars that lead to the formation of pulsars and SNRs are poorly known, the absolute values of the SFRs cannot be ascertained with certainty, and so radial distributions relative to the value in the solar neighbourhood are used (as suggested by Lacey and Fall 1985).

Now the same plot can be used to represent the radial

-141-

variation of $\log (\mathbb{Z}_{M_2}/\mathbb{Z}_{M_2})$ provided the relation $\mathbb{V} \subseteq \mathbb{Z}_{M_2}^k$ is valid, except that the ordinate needs to be scaled up by a factor of k. Assuming k = 1.40, the same distribution of



Figure 4.8: The present radial variation of the SFR, $\frac{1}{15}$, normalised to its value in the solar neighbourhood. The same plot can be used to represent the present radial variation of $\hat{\Sigma}_{M_2}$, simply by re-scaling the ordinate by a factor of k, which has been taken to be 1.40. The radial variations of $\hat{\Sigma}_{M_2}$ derived from various CO surveys are shown by the family of curves marked B, C, R and S. Dotted lines show where surveys are incomplete.

data points now refers to the predicted radial distribution

-142-

of \mathbb{Z}_{M_2} in the Galaxy relative to its value in the solar neighbourhood.

Superimposed on this plot are curves representing some of the different estimates of the distribution of $\hat{\mathbb{Z}}_{M_n}$ derived from the CO emissivity measurements in the Galaxy. Each distribution has been normalised to its guoted value in the solar neighbourhood. This naturally eliminates the existing differences in the absolute values of $(\mathbb{Z}_{N_0})_{of}$ arising partly from differences in the absolute values of the CO emissivity and partly from differences in the adopted values of the $CO \longrightarrow H_{\chi}$ conversion ratio (Chapter 1). In this plot, only the shapes of various radial distributions of $\mathfrak{T}_{M_{n}}$ are relevant. The four continuous curves labelled B, C, S and R refer to ${\rm H}_{\lambda}$ distributions in the inner Galaxy averaged for the north and south (that is, averaged along the Milky Way on either side of the centre of the Galaxy), respectively given by Bhat et al. (1985), Cohen et al. (1985), SSS (1984) and Robinson et al. (1984). These distributions were chosen to illustrate and encompass the range of the many other published distributions. The last three distributions are derived independently of one another, each group using an independent CO survey of its own. The Bhat et al. distribution uses the CO survey of SSS (1984), adopts a lower \mathfrak{A}_{2m} in the solar neighbourhood and applies a differential metallicity correction to α_{20} . The observations for the outer Galaxy (that is, for. $R_{f_{r}} > 10$ kpc), are fewer and are shown by dotted lines where the sampling is incomplete. Only SSS give an observational distribution of $\mathbb{Z}_{\mathbb{H}_{2}}$ for the outer Galaxy as far as 14.5 kpc but for the northern declinations only. Using a metallicity

-143-

correction to $\&_{20}$, Bhat et al. suggest values that are represented by the dotted curve. Robinson et al. give values for the average of north and south which extend only up to $R_{\mathfrak{S}} = 11$ kpc. Now with reference to Figure 4.8 the following points may be noted:

Within the solar circle, almost all of the data points (i) lie within the bounds of all CO-based estimates thereby justifying the result of $\sqrt[4]{8} \ll \Sigma^{1}_{M_{0}}$ with k $\simeq 1.40$. In fact, since the vertical spread of the SFR data is considerably narrower than that found among the different CO surveys, these results may further be used to constrain them. (ii) Outside the solar circle, the agreement between the SFR data and those from the CO-surveys is rather poor. This is probably due to an underestimate of $\mathbb{Z}_{M_{2}}$ in the outer Galaxy from the CO-surveys, for a number of reasons. The CO survey data in the outer Galaxy is incomplete whereas the observed SFRs derived from the pulsars, SNRs and HII regions include the regions the SSS CO survey does not ($g = 170-330^{\circ}$). When the CO survey data is complete, there is a possibility that the average of the north and south will rise to the level required by the observed activity of star formation in the ring as a whole. In fact, the trend of Robinson et al.'s measurement of the north and south beyond the solar circle suggests such a possibility.

Kutner and Mead (1981) had argued for a similar underestimate of H_2 in the outer Galaxy due to the large molecular complexes being shifted upwards from the plane, but their analysis was questioned by Solomon et al. (1983b). However, there is still reason to believe an underestimate may be due to uncertainties in \mathfrak{A}_{20} . Bhat et al. (1985)'s

-144-

use of a metallicity correction still underestimates $\Sigma_{\rm H_2}$ in Figure 4.8, but they point out that this could still be an underestimate because in the quiescent clouds in the outer Galaxy a large fraction of CO would be locked into grains. van Dishoeck and Black (1987) further point out that strong CO line emission is excited efficiently only if the density exceeds a certain critical value. If the mean cloud densities are low in the outer Galaxy, there may be significantly more H₂ than is suggested by the weak CO line emission. The underestimate of H₂ may also be linked to a Galactocentric temperature gradient affecting α_{20} (Maloney and Black 1987).

ana 's

i konger det en galgere

n transformer and gene

1.25-4.16-12

Therefore, whether the H_2 is underestimated because of incomplete CO surveys, or because of an even larger \mathscr{A}_{20} in the outer Galaxy than adopted by Bhat et al. (1985), at the moment this will be left as a prediction.

(iii) Among the four CO-based H₂ distributions for the inner Galaxy, the SFR data seem to favour the Cohen et al. and Bhat et al. distributions in the inner Galaxy. The data however fails to be consistent with the shape of Robinson et al.'s distribution, but marginally fits the SSS distribution.

(iv) If, in view of the above point, the CO distribution of Cohen et al. (1985) is correct, the shape of their H_2 distribution is such that their values do not require any appreciable radial gradient corrections for e_{20} such as metallicity or temperature gradients. On the other hand, if the CO survey of SSS is correct, they would require a gradient correction, either in metallicity and/or temperature in order to make their distribution fully

-145-

م. مربقا مار الأرسية تولي ر

and the state of the second

compatible with the SFR data. For example, a metallicity correction may be introduced by assuming

$$\alpha_{20} \propto 2^{-n'}$$
 (4.17)

where Bhat et al. (1985) have used n' = 1. If n' were to be determined from the requirement of compatibility with the SFR data in the inner Galaxy, the SSS CO distribution would require n' \boxtimes 0.8 - 0.9. Upon the same consideration, Robinson et al.'s curve would require a metallicity correction with n' > 1.5. Similarly, a suitable radial gradient for the gas temperature T_K following

$$\mathfrak{A}_{20} \ll \mathfrak{T}_{\mathcal{N}}^{1\cdot 3}$$
 (4.18)

(Kutner and Leung 1985) may also be studied for each distribution leading to different values of the exponent. It must be stressed that at present it is difficult to distinguish between the metallicity correction and that of the temperature variation. Either or both factors may be responsible. At present all that can be said is that some radial gradient in \mathfrak{C}_{20} is needed for the SSS and Robinson et al. CO-surveys in deriving H₂, assuming that the value k = 1.40 is reasonably accurate.

(v) Previous work (Rana and Wilkinson 1986b) carried out a similar analysis to the above, but in the context of a variable yield model of Galactic chemical evolution which required k = 1.12. The conclusions were the same as those given above except that a smaller metallicity correction to the SSS data was required, i.e. n' \cong 0.7. However, it has been previously argued that the case of constant yield is preferable and therefore a higher value of n'.

4.6.2 A tentative distribution of H₂ in the Galaxy

Apart from the shape, it is also possible to predict an

-146-

absolute distribution of $\mathbb{Z}_{W_{2}}$ from the SFR data in Figure 4.8, provided the value in the solar neighbourhood, $\mathbb{Z}_{W_{2,\Theta}}$ is known. Chapter 1 summarised the uncertainties in the CO emissivities and the determination of the local value of \ll_{20} . In view of this controversy, we therefore prefer to estimate $\mathbb{Z}_{W_{2,\Theta}}$ independently of the CO surveys.

It has long been realised that the Galaxy appears to contain too much gas than could be observed from the present SFR in the Galaxy (Turner 1984). If large gas clouds are primarily responsible for star formation, they would gravitationally collapse on a time-scale of 10^{6} y, and in that case (in order to fit the observed SFR of $\approx 5 \text{ M}_{\odot}\text{y}^{-1}$) there should not be more than about 10^7 M_{\odot} of gas in the Galaxy in the form of large clouds. Now the central question is, how much H₂ is there in the Galaxy? The original CO surveys (Scoville and Solomon 1975) had claimed a huge amount of H₂ in the Galaxy, $M_{M_2} \approx 5.10^{6} \text{ M}_{\odot}$ and SSS (1984) still claim it to be > $3.6.10^{6} \text{ M}_{\odot}$ in spite of the fact that from the $\frac{6}{3}$ -ray analysis and other arguments, Bhat et al. (1985, 1986) could hardly allow for $M_{H_2} > 10^{6} M_{\odot}$.

Theoreticians trying to understand the process of star formation in the Galaxy had to face the problem of stabilising GMCs against self gravitation. If the average lifetime of a GMC is typically a few times 10^{7} y, this would imply an SFR of $\approx 100 \text{ M}_{\odot} \text{ y}^{-1}$ for the whole Galaxy if the SSS (1984) M_{Mg} is assumed to be correct. Turner (1984) however showed that under some circumstances a GMC may break up into smaller molecular clouds which may once again be recycled through the Galactic spiral density waves on time-scales $\approx 2.10^{8}$ y to form more GMCs. Having delayed the star

-147-

. روم مع د ال و

formation this way, he obtains a theoretical SFR of $\simeq 20 \text{ M}_{\odot} \text{y}^{-1}$ for $\text{M}_{\odot_2} = 4.10^9 \text{ M}_{\odot}$, which he took from SSS. If Turner's model of Galactic star formation is a viable one, the present observed SFR of 5 $\text{M}_{\odot} \text{y}^{-1}$ would suggest that $\text{M}_{\text{M}_2} \simeq 1.10^9 \text{ M}_{\odot}$. Interestingly enough, a recent study (Rengarajan 1984, 1986) of the star formation efficiency in molecular clouds from the FIR observations also independently supports the above ratio of $\frac{\text{M}_2}{\text{M}_2}$ to M_{M_2} .

Now suitably binning the data points in Figure 4.8 and estimating the integral for the total amount of H_2 in the Galaxy up to 16 kpc, it is found that $(\Sigma_{M_2})_{\textcircled{O}} = 1.30 \pm$ 0.20 $M_{\textcircled{O}}pc^{-2}$ for a tentative value of $M_{\textcircled{M}_2} = 10^{9} M_{\textcircled{O}}$. Recently, Dame et al. (1987) have obtained the same value by summing the masses of the largest clouds within 1 kpc of the Sun but assuming $(\mathfrak{A}_{20}) = 5.4$. The value derived from the SFR analysis is independent of the value of \mathfrak{A}_{20} but can be used in comparison with the CO to determine $(\mathfrak{A}_{20})_{\textcircled{O}}$. When applied to the CO data of SSS (1984), $(\mathfrak{A}_{20})_{\textcircled{O}} = 2.5 \pm 0.5$, which is close to the value advocated by Bhat et al. (1985) from the $\overset{\checkmark}{}$ -ray analysis.

Using $(\Sigma_{M_2})_{\odot} = 1.30 \ M_{\odot} pc^{-2}$, Figure 4.9 presents a radial distribution of Σ_{M_2} derived exclusively from the SFRs in the Galaxy. Error bars are derived from the spread of the values of Ψ_S derived from different indicators. Of course the true distribution is subject to change by a constant factor for all R_G , for any revision of the normalisation in the solar neighbourhood. It should be noted that Güsten and Mezger (1982) have claimed a Galactic SFR of 7.5 - 11 $M_{\odot}y^{-1}$ which in the above argument would raise $(\Sigma_{M_2})_{\odot}$ by a factor of two. However, their radial

-148-

distribution appears not to follow the other indicators of star formation such as the pulsars and SNRs. Furthermore, Chapter 6 will show that a lifetime of H_2 of a few times 10° y is probably an upper



Figure 4.9: A tentative radial distribution of \mathbb{Z}_{N_2} as suggested by the present work. The value of \mathbb{Z}_{N_2} in the solar neighbourhood has been so adjusted that the total mass of H_2 in the Galaxy between $R_6 = 1$ and 16kpc corresponds to $10^9 M_{\odot}$ for the present SFR of $5M_{\odot}y^{-1}$. The true distribution of \mathbb{Z}_{N_2} is of course subject to change by a constant factor for all R_{\odot} for any revision of the above normalisation.

estimate. The use of a constant yield leading to k = 1.40, does not affect the results of this analysis greatly. A variable yield with k = 1.12 leads to $(\Sigma_{W_2})_{\odot} = 1.15 M_{\odot} \text{pc}^{-2}$, $(\ll_{20})_{\odot} = 2.2$ and a radial distribution very similar to Figure 4.9 (Paper II).

With the above reservations, this method gives a

-149-

possible way of estimating Σ_{M_2} in the Galaxy independent of the controversial CO surveys. Of course, it depends on a model of the SFR as $\Psi_3 \ll \Sigma_{M_2}^{\&}$ and the fixing of k by the observed metallicity gradient, but Chapters 2-4 have argued for the validity of these points.

4.7 RADIAL GRADIENTS IN GAS, STARS AND METALS

In the previous section, the radial distribution of H_2 has been derived from the observed SFR through $\bigvee_{6} \ll \sum_{M_2}^{k}$, fixing k by the required metallicity gradient. As in the case of the metallicity gradient, this does not illustrate the evolution of the SFR radial distribution (and therefore the H_2 distribution) with time. It may be further argued that due to uncertainties in the SFR itself, especially in external spirals, it would be helpful to see whether the gas distributions can be derived directly from the total surface density, $\Sigma_{\rm F}$, (in a similar manner to the metallicity gradient) so that specifying this quantity for any galaxy will predict its distribution of metals, stars and gas.

The chemical evolution model, described above, is used with the simplest of assumptions, that is, constant yield and no infall. The case n = 0, $\delta_1 = 0.5$, $\mu_{\Theta} = 0.6$, b = 1.3and k = 1.40, described in Table A.4.1 (Appendix) is used in the following.

The radial distribution of \mathbb{Z}_{γ} is shown in Figure 4.10a and taken from Table A.2.1 (Appendix) following SSS (1984) between $R_{\mathbb{G}} = 3$ and 15 kpc. The assumption is then made that the initial metallicity ($Z_{\mathbb{Q}}$) and gas fraction ($\mathcal{A}_{\mathbb{Q}}$) are constant with $R_{\mathbb{G}}$. In reality this may not be strictly true,

-150-

but for simplicity it is a good approximation. The metallicity gradient is then derived using equations (4.15)-(4.16) and shown in Figure 4.10b, each line representing a time step of 5 Gy after formation of the disc at t = 0. Of course k has been chosen to give the observed metallicity gradient of -0.06 dex kpc⁻¹ at t = 15 Gy, and the present metallicity in the solar neighbourhood has been taken as $12+\log(O/H) = 8.90$ to normalise the plot.

The value of k is now fixed by the observed metallicity gradient, b is determined by the empirical correlation between p_2 and Z (Section 2.4) and all other parameters are fixed by the model of chemical evolution in the solar neighbourhood. Therefore, the radial distribution of SFR, gas, density of stars and dark matter can now be predicted.

The SFR shown in Figure 4.10c is given by equations (3.12) and (3.17) as

$$\Psi_{S} = CX^{k} \left[\Sigma_{T} \left(\frac{Z}{Z_{0}} \right)^{b} \exp \left[G(Z_{0} - Z) \right] \right]^{k}$$
(4.19)

where X has been taken as a constant. The value of CX^{k} can then be evaluated in the solar neighbourhood at $t_{1} = 15$ Gy,

$$CX^{k} = \frac{(\psi_{1})_{0}}{(\Sigma_{T})_{0}^{k}} \left(\frac{\psi_{0}}{\psi_{1}}\right)_{0}^{k}$$
(4.20)

where $(\Psi_{i})_{\odot} = 3 M_{\odot} \text{pc}^{-2} \text{Gy}^{-1}$ and $(\Psi_{i} / \Psi_{o})_{\odot}$ is taken from Table A.4.1 (Appendix).

The total gas density, \mathbb{Z}_{g} , is shown in Figure 4.10d and derived using equation (3.17), that is

$$\mathbf{\Sigma}_{\mathbf{Q}} = \mu_{\mathbf{Q}} \sum_{\mathbf{T}} \exp\left[\mathbf{G}(\mathbf{Z}_{\mathbf{Q}} - \mathbf{Z})\right] \tag{4.21}$$

The surface densities of visible stars and dark matter are then given by

$$\Sigma_{s} = \sigma_{0}\Sigma_{T} + \frac{(1-R-D)}{(1-R)}(\mu_{0}\Sigma_{T} - \Sigma_{s})$$
(4.22)

-151-

an an the second states and the second states and the



(a.)

Figure 4.10:Radial and time evolution of the Galaxy in a
model with no infall, constant yield and
 $V_{S} \ll \Sigma_{M_{2}}$. Shown are curves corresponding to
times t after disc formation. \longrightarrow t = 0
 \longrightarrow t = 0
 \implies t = 10Gy
 \implies = = = t = 15Gy

20Gy

t =

-152-

and
$$\Sigma_{d} = \Sigma_{T} - \Sigma_{S} - \Sigma_{S}$$
, (4.23)

and shown in Figures 4.10e and 4.10g. Finally, \mathbb{Z}_{M_2} can be derived from $\overset{ heta}{V_2}$ by

$$\Sigma_{M_2} = \left(\underbrace{\mathbb{N}_3}_{\mathbb{N}_8} \right) \quad (\Sigma_{M_2})_{\otimes} \qquad (4.24)$$

where for the solar neighbourhood $(\Sigma_{M_2})_{\otimes} = 1.30 \text{ M}_{\odot} \text{pc}^{-2}$ has been taken at present. The value of Σ_{MI} then follows

$$\mathbb{Z}_{MZ} = \mathbb{Z}_{g} X - \mathbb{Z}_{M_{Z}}$$
(4.25)

These radial distributions shown in Figure (4.10) suggest the following points:

(i) The metallicity gradient is a direct indicator of age for any exponential disc with no initial metallicity gradient (cf. Section 4.4).

(ii) The surface density of star formation \oint_S is exponential for the first 7-8 Gy, slightly increasing, but then declining for t > 10 Gy, especially in the central parts. The total SFR for the Galaxy has been estimated between R_{G} = 3 kpc and 15 kpc,

and is shown as a function of time in Figure 4.11. It can be seen that between t = 0 and t = 15 Gy, the SFR has stayed practically constant, declining from its original value by less than a factor of three at t = 15 Gy. This is consistent with the work of Kennicutt (1983b) and Gallagher et al. (1984) which suggests a roughly constant SFR in late-type spiral galaxies.

-153-



Figure 4.11: The total SFR in the disc of the Galaxy as a function of time after formation of the disc. (iii) The model reproduces quite nicely the presently observed radial distributions of ψ_{S} , Σ_{Q} , Σ_{MX} and $\Sigma_{M_{N}}$ (that is a Bhat et al. type of H_2 distribution), though not perfectly. The predicted peak of the 🖏 and 🖾 distributions occurs at ~ 8 kpc whereas observations suggest closer to 6 kpc. However, the fact that such an extra-ordinarily simple model predicts a molecular ring for spirals is quite important. The apparent 'hole' in the H2 and total gas distributions in this and other galaxies has been difficult to reproduce in other models of chemical evolution. For example, Lacey and Fall (1985) find it extremely difficult using infall and radial gas flows to produce this hole. Only with a $\sqrt[n]{s} \in \mathbb{Z}_q^2$ and with the inflow velocity rising sharply close to the centre of the Galaxy is it possible. Young and Scoville (1982a) had suggested that the hole was due to either orbital disruption at the inner Lindblad resonance (cf. Stark and Blitz 1978) or to gas having been exhausted in the nuclear bulge. Young (1985a)

-154-

concludes that the nuclear bulge hypothesis is favoured by data on CO distributions in external galaxies. Sanders (1977) had earlier suggested that the hole was due to gas inflow into the central 600 pc of the bulge, and Jog and Ostriker (1987) have more recently suggested an inward drift of clouds due to viscous interaction thus depleting gas within 3 kpc. The chemical evolution model, described in this work, however, produces a central hole in the gas distributions quite naturally without the effect of the bulge being taken into account. It is simply due to a high SFR in the central parts.

Furthermore, there have been a number of attempts to explain the relative radial distributions of HI and H₂ using, for example, density waves (Roberts and Burton 1976), spiral arms (Wyse 1986) or stochastic self propagating star formation (Seiden 1983a,b). The present work re-asserts that these distributions can be understood in the context of a simple model of Galactic chemical evolution with $\frac{1}{15} \ll \frac{1}{16}$ and the fraction of gas in molecular form simply controlled by the metallicity.

(iv) Pagel (1987) pointed out a possible problem with $\Psi_{S} \ll \Sigma_{M_{Z}}^{k}$, and the Bhat et al. (1986) distribution of $\Sigma_{M_{Z}}$, arguing that if Σ_{S} was roughly constant with R_{G} at present, while Ψ_{S} was highest in the 4-6 kpc ring the gas in that ring will not last much longer. If the gas is not replenished from an outside source, he predicted a 'wave' of gas exhaustion spreading out from the central regions. This is seen clearly in plots (d), (f) and (h) of Figure 4.10. However, it is not clear why this should be a problem. It can be seen that there is still a sizeable amount of total

-155-

gas remaining up until 15 Gy. The presently observed hole could be an indication of this gas exhaustion moving out through the Galaxy. If \mathbb{Z}_{3} presently followed the shape of \mathbb{Z}_{7} , as contended by the H₂ distribution of SSS (1984), then gas would remain in the 4-6 kpc ring for a longer time, but it then would be unclear as to what the reason for the < 4 kpc hole was.

الهاما المغامين فالمحمد المحمد ال

e en an ar an an

(v) The surface densities of visible stars and dark matter remain roughly exponential in shape at all times.

Finally, one may ask can this model be used to describe the evolution of other spirals. A detailed discussion of this is outside the scope of the present work, but some initial remarks may be made. As a first attempt, one could simply use the different total surface mass distributions and keep all the other parameters (determined for the Galaxy) the same. Of course, the normalisation of quantities may be different, but the shapes of the radial distributions will be determined by Σ_{T} and t. Total surface mass densities have been taken from Diaz and Tosi (1984) and Tosi and Diaz (1985) for three galaxies, M51, M31 and NGC6946 as examples. Determining the metallicity gradient between $R_{C} = 6$ and 14 kpc after $t_{T} = 15$ Gy, the predicted and observed metallicity gradients are compared in Table 4.3.

Table 4.3: Predicted and observed metallicity gradients.

dlog(2)/dR_G Predicted Observed Reference M51 -0.067 -0.081±0.027 Tosi & Diaz (1985) M31 -0.050 -0.030±0.010 Diaz & Tosi (1984) NGC6946 -0.068 -0.077±0.047 Tosi & Diaz (1985)

-156-

We have been the cost of a second second

As can be seen the agreement is reasonable considering the uncertainty in observations.

The surprising result however, is that for each of these galaxies, the predicted $\mathbb{Z}_{M_{2}}$ distribution exhibits a central hole, similar to the Galaxy. The observed CO distributions (Scoville and Sanders 1987) show only a central hole for M31, while M51 and NGC6946 have CO distributions increasing towards the centre. Although a metallicity correction as proposed by Bhat et al. (1985) flattens the observed ${\rm H}_2$ distribution, it does not produce central holes. Therefore, either the above model is too simple to apply to other galaxies, or the CO does not successfully trace the H_{χ} distribution. No doubt such a simple model will need some modification, but the fact that the metallicity gradients are reproduced could reinforce the growing feeling that the use of a constant conversion factor an external spirals is not valid (e.g. Bhat et al. 1986; Lo et al. 1987a; Maloney and Black 1987). Chapter 6 will further examine the issue of α_{20} in external galaxies.

-157-

CHAPTER FIVE

PERTURBATIONS OF THE OORT CLOUD AND

TERRESTRIAL MASS EXTINCTIONS

5.1 INTRODUCTION

The subject of mass extinctions of organisms has aroused interest for many years, and recently has been linked to the existence and properties of giant molecular clouds (e.g. Clube and Napier 1984a,b; Rampino and Stothers 1984a). It is the intention of this Chapter to re-examine this link in an attempt to constrain the masses of GMC.

Possible links of mass extinctions and other terrestrial geological events with astronomical phenomenon were postulated centuries ago (see the reviews of Clube and Napier 1982c and Bailey et al. 1986) but it is only fairly recently that a specific hypothesis and a particular astronomical phenomenon - cometary or meteoritic impacts has come to the fore (Öpik 1958; Urey 1973; Clube 1978; Hoyle and Wickramasinghe 1978; Napier and Clube 1979; Hoyle 1981; Hills 1981; Rampino and Stothers 1984a).

In particular, Alvarez et al. (1980) showed that deep sea limestones showed a marked increase in iridium concentration above the background level at precisely the time of the mass extinctions associated with the K-T boundary, which included the extinction of dinosaurs. This they ascribed to the impact of a 10km diameter extra-terrestrial body, injecting about sixty times its own mass into the atmosphere and so affecting photosynthesis and

-158-

food chains. This iridium layer has now been detected worldwide (Hsu 1980; Saito et al. 1986; Alvarez 1987) with other signatures of explosions such as microspherules and shocked minerals (Keller et al. 1983; Smit and Kyte 1984; Bohor et al. 1984). There has even been a claim of a global layer of graphite at the K-T boundary, due to the impact igniting worldwide fires (Wolbach et al. 1985). Similar iridium layers have also been claimed, associated with 3 other periods of mass extinctions (Ganapathy 1982; Alvarez et al. 1982; Playford et al. 1984; Xu et al. 1985) suggesting other impacts as causes of terrestrial mass extinctions.

Although the evidence for coincidence of apparent impact signatures with extinction events is extremely impressive (Jablonski 1986) a number of doubts with the above picture have been expressed. Some impact signatures are found mid-extinction and there is difficulty in the dating of palaeobiological data in order to derive coincidences (Crawford 1985; Padian and Clemens 1985). Furthermore, detailed studies of environmental change at the time of major extinctions may suggest gradual changes rather than the expected abruptness of an impact event (Stanley 1984; Clemens 1986; Sloan et al. 1986; Norman 1986; Donovan 1987). Officer et al. (1987) have recently reviewed the alternative case for volcanic activity being the cause of mass extinctions.

Luck and Turekian (1983) in examining osmium isotope ratios at the K-T boundary suggested comet showers rather than a single impact being the cause. It is these comet showers that have further been invoked to explain a possible

-159-

periodicity seen in the terrestrial record. 5.1.1 Periodicity in the terrestrial record

요즘 것 같아?

Traditionally it was thought that each mass extinction event was essentially unique and could be explained independently by changes in climate, continental configurations, global habitat area, orogenic activity, oceanic or atmospheric chemistry and even extra-terrestrial phenomenon. However, Raup and Sepkoski (1982, 1984) using extinctions in marine invertebrates demonstrated an apparently regular 26 My periodicity in the pattern of extinctions, a feature which had long been suspected in the geological record (e.g. Holmes 1927; Fischer and Arthur 1977; McCrea 1981). Figures 5.1 and 5.2 are taken from the recent work of Sepkoski and Raup (1986) from which they derive a period of 26.2 + 1.0 My. The significance of this result, if correct, is that extinction events are ultimately due to a single forcing agent rather than a plethora of individual causes (McClaren 1983).

This rough 30 My periodicity, in phase with the extinction events, has since been shown to apply to a wide range of geological indicators, as summarised by Table A.5.1 (Appendix). In particular, the discovery that episodes of mass extinction appear to correlate with the ages of dated terrestrial craters (e.g. Seyfert and Sirkin 1979; Alvarez and Muller 1984) indicates that cometary impacts could play a major role. These comet showers, in addition to causing mass extinction, could possibly trigger geomagnetic reversals (Pal and Creer 1986) and volcanic activity (Pandey and Negi 1887; Rampino 1987), although the details of this process are far from clear.

-160-



Figure 5.1: (a) The number of extinctions for marine animal families through 43 stages as measured by simple numbers of extinction per stage over the last 270My (Sepkoski and Raup 1986). (b) A model of extinction intensity from Sepkoski and Raup (1986). Background extinction, the intensity of normal intervals of time, is shown as a decreasing function of time. The heights of the 8 extinction peaks (which are left unscaled) reflect the relative magnitudes of the events at familial and stage levels.

It is important, before outlining methods for the production of periodic comet showers, to be clear about the uncertainties in the above periodicity analysis. Periodicity in mass extinctions ultimately depends on difficult decisions concerning the absolute dating of stratigraphical boundaries, the culling of the database and the definition of a mass extinction (e.g. Dingus and Sadler 1982; Stigler 1985; Hoffmann 1985, 1986; Jablonski 1986;

-161-



Figure 5.2: The cycle number against age relationship is shown for 8 significant extinction events. The solid line is a least squares fit with slope corresponding to a period of 26.2 My, while the dashed lines show the range in slopes for solutions which pass through the error bars of all points. (After Sepkoski and Raup 1986).

Tremaine 1986). Some workers (e.g. Patterson and Smith 1986) using the same data find no evidence for periodicity. Furthermore, although the record of mass extinctions may statistically show periodicity, it is possible that this might arise even if the extinctions themselves are not periodic (cf. Kitchell and Pena 1984). This point is illustrated in Figure 5.3, in which the distribution of time intervals between successive mass extinctions is compared with the exponential distribution expected for a random At first sight there is a clear inconsistency with process. the random hypothesis, the observed interval distribution resembling two delta functions broadened by noise, such as would be expected from genuine periodicity with timing errors and occasional missing cycles. However, the disparity could be illusory if an extinction event is

-162-



Figure 5.3: The frequency of the intervals between successive extinctions detailed by Sepkoski and Raup (1986). The quoted errors on the extinction dates have been allowed for in an approximate way. The arrows indicate where peaks would be expected for a 26 My periodicity (the peak at 52 My occurs when cycles are missed). The exponential curve marked 'random' relates to the expectation for a purely random process. The experimental data appear clearly to favour periodicity.

followed by a 'dead time' in which surviving species are more hardy and less prone to extinction, while new susceptible species develop more slowly (e.g. Stanley 1984).

In this way, a nearly random distribution of extinction triggers might produce an apparently periodic extinction record, though the period would correspond to the dead time and the required frequency of potential extinction causes would have to be higher.

This hypothesis would not, of course, naturally explain detected periodicity in other geological indicators. However, periodicity in these other indicators is still controversial. Uncertainties in the cratering record include its incompleteness, age determinations and the fact that not all structures used to suggest periodic comet

-163-

showers are produced by the same type of body (Weissman 1985; Grieve et al. 1986; Jacobs 1986). Doubt has also been cast on the periodicity in geomagnetic reversals (Lutz 1985; Raup 1985b; Jacobs 1986).

However, Sepkoski and Raup (1986) have reconsidered some of these questions and still believe the mass extinction record shows evidence for periodicity. The present work re-examines the case for periodicity of cometary impacts, following the suggestion of McClaren (1986) that 'surely the burden of proofs rests, now, firmly on the shoulders of those who would deny the existence of this violent phenomenon, and for them the proof must be ... evidence and argument leading to the denial of Earth-crossing asteroids and their periodic arrival on Earth'. Whilst, of course, the existence of asteroids in Earth-crossing orbits cannot be denied, this work will show that periodic 30 My variations in the cometary flux cannot be caused by encounters of the Solar System with GMCs, that is the so-called 'Galactic' model.

5.1.2 The Galactic model of terrestrial mass extinctions

It seems very unlikely that any 30 My periodicity in the terrestrial record could be due to processes in the Earth surface and atmosphere or to deep Earth processes as these have characteristic periods $10^{\circ} - 10^{5}$ y and $10^{\circ} - 10^{\circ}$ y respectively (McCrea 1981). However, the fact that the 30 My cyclicity is identical, within the errors, to the mean interval between Galactic plane crossings by the Solar System (Innanen et al. 1978; Bahcall and Bahcall 1985) has been a major argument implicating external astronomical processes as a primary cause of mass extinctions (Hatfield

-164-

A State of the second state

and Camp 1970).

In principle, a heightened flux of comets through the inner planetary system could produce significant climatic deterioration, leading to mass extinctions either by the direct effects of cometary impacts or by the passage of the Earth through streams of debris from the break up of a large comet (Clube and Napier 1986a, b, 1987). It has been suggested (Clube and Napier 1984a, b; Rampino and Stothers 1984a) that the required change in cometary flux could be caused by passages of the Solar System past molecular clouds, modulated by the Solar System oscillation, thus significantly disturbing the orbits of comets in the Oort Cloud. According to the view developed by Clube and Napier (1984a, 1986a) the present parabolic flux is primarily the result of such an event and the Oort Cloud (cf. Kresak 1977; Yabushita 1979) is regarded as an unstable system which is intermittently replenished in rare episodes of comet capture associated with the passage of the Sun through a dense molecular cloud containing comets (see Section 5.5). Alternatively, the Oort Cloud may be assumed to be primordial, with comets at large radii ultimately originating in a hypothetical dense inner core of the system, which is calculated to be stable for the lifetime of the Solar System (Hills 1981; Bailey 1983a,b,c,1986a; Shoemaker and Wolfe 1984a; Weissman 1986). Following the suggestion of Hills (1981), a major disturbance of the orbits in the inner core would produce a short-lived, intense 'shower' of comets passing through the inner planetary system. Provided that the encounters deflect enough comets and the encounter rate and degree of

그는 아내는 동안 많은 것 같아요. 지각 승규는

-165-

modulation agree with the observations, either theory in principle provides a good working hypothesis for the explanation of extinctions.

However, this (Galactic) explanation for periodic comet showers has been challenged on two grounds. First, the width of the molecular cloud layer may be too great in comparison with the amplitude of the solar motion about the Galactic plane to allow a statistically significant 30 My periodicity to be extracted from what is only a small number of dated events in the terrestrial record (Thaddeus and Chanan 1985; Thaddeus 1986). This criticism has been discussed by Stothers (1985), who holds that a Galactic signal could still be detectable in the geological record, even with the Sun's present small amplitude of oscillation. The theory might, of course, be restored by assuming that encounters between the Solar System and molecular clouds were more strongly modulated in the past, due to the amplitude of the solar motion being greater previously than now (e.g. Thaddeus 1986; Clube and Napier 1986a; Bailey et al. 1986).

The second criticism of the Galactic explanation for comet showers is that the present phase of the solar orbit does not agree with the theory's simplest predictions (Schwartz and James 1984; Davis et al. 1984). According to expectations, mass extinctions should mostly occur when the Sun is close to the Galactic plane (i.e. close to its present position, e.g. Mihalas and Binney 1981), whereas the evidence of the terrestrial record (Raup and Sepkoski 1984) indicates that we are now almost halfway between extinctions. This phase discrepancy could be attributed to

-166-

the partly stochastic nature of the extinction record, but it nevertheless does remove some of the theory's initial attraction.

In addition to examining these two problems, this Chapter will also examine the more fundamental problem, that of obtaining a large enough signal itself. Section 5.2 presents a definition of an 'extinction' shower and estimates the critical semi-major axis above which the comet velocities must be randomized in order to produce a shower which might register in the geological record. Section 5.3 then considers constraints on the masses and mass densities of perturbers, in particular GMCs capable of producing showers of sufficient strength and frequency to explain the results on mass extinctions, while Section 5.4 returns to the signal-to-noise and phase problems. Finally, Section 5.5 examines the question of the survival of the Oort Cloud over the lifetime of the Galaxy.

There are, of course, alternative astronomical suggestions for the cause of periodic mass extinctions such as Nemesis (Davis et al. 1984; Whitmire and Jackson 1984) and Planet X (Whitmire and Matese 1985). Nemesis is a postulated unseen solar companion in a highly eccentric orbit that periodically enters the Oort Cloud and perturbs comets, but such a theory has difficulties concerning irregularities in the period of revolution and dynamical instability (Hut 1984; Hills 1984; Torbett and Smoluchowski 1984; Clube and Napier 1984c; Weissman 1984). Planet X is a similar idea, with this time a planet creating comet showers whenever its perihelion passes close to the inner disc of the comet cloud, but it seems that there are major

-167-

difficulties in the planet having enough mass to scatter enough comets (Kerr 1985; Tremaine 1986). At the present time, neither of these alternatives appears to be a viable explanation for the 30 My periodicity of comet showers and so will not be considered further.

5.2 THE DEFINITION OF AN EXTINCTION SHOWER

Long-period comets entering the planetary system from the Oort Cloud are removed from their original orbits by planetary perturbations, either by ejection into interstellar space or by capture into short period orbits where they quickly decay. The flux of new comets from the Oort Cloud thus represents a steady loss of low angular momentum long period comets, which is described in velocity space by a loss cylinder in the comets' velocity distribution at large heliocentric distances. Comets with velocities inside the loss cylinder, i.e. with perihelion distances q < q_k 🕿 15AU (Bailey 1984) are effectively removed from the Oort Cloud within one orbital period. Provided that perturbations (e.g. due to stars or the Galactic tidal force) are strong enough to replenish these losses occurring on an orbital timescale with comets lying on neighbouring low angular momentum orbits, the loss cylinder is kept filled and a more or less steady influx of long period comets will result. For smaller semi-major axes, however, the orbital period and the effectiveness of external perturbations are both reduced, and at some value of the semi-major axis the loss cylinder will not be kept steadily filled (Oort 1950; Bailey 1977; Hills 1981).

-168-

Occasional close, exceptionally strong perturbations may still fill the loss cylinder, however, and it is these intermittent encounters which lead to the possibility of comet showers, the instantaneous influx of long period comets being then much greater than the steady-state value. Therefore to link mass extinctions to the nature of the perturbers, the proportion of shower to background comets above which a mass extinction may be expected to occur must be estimated. Then the critical semi-major axis, a_{crit} , down to which the loss cylinder must be filled in order for such a shower to occur may be evaluated.

5.2.1. The proportion of shower to background comets

The mechanisms by which comet showers affect the Earth are poorly understood. They range from explosions, caused by collisions of large bodies (Napier and Clube 1979; Alvarez et al. 1980), to the possibly more important long-term environmental and climatic effects brought about by interactions of the Earth with interplanetary debris on a variety of scales (e.g. Clube and Napier 1986a,b). Discussions of these questions may be found in the volumes edited by Roddy et al. (1977), Silver and Schultz (1982), Holland and Trendall (1984) and Smoluchowski et al. (1986). In fact the link between impact and biological change is far from clear (Heissig 1986; Clemens 1987), and may provide another major difficulty for comet showers as the cause of mass extinctions.

However, according to the Galactic comet shower hypothesis under consideration, the showers of comets cause mass extinctions and also an enhanced cratering rate on the Earth, this being the cause of the apparent correlation

-169-

between the ages of dated craters and times of mass extinctions (e.g. Shoemaker and Wolfe 1984b). The relative number of craters which appear to be associated with showers may thus be used to give a rough indication of the required ratio of shower to background comets necessary to cause a mass extinction.

Observations of terrestrial craters (Grieve 1982) show that the rate of production of craters, for example with diameter > 10 km, integrated over the whole earth is $(7 \pm 3).10^{-6} y^{-1}$. Observations of dated craters indicate that about 70% occur bunched in time as if associated with comet showers (Muller et al. 1984). The ratio of shower to background craters is thus $\leq 70/30$, and if crater-forming projectiles arise from the long-period comet flux, a ratio of shower to background comets (R_{Sh}) on the order of $70/30 \leq 2$ would be implied. However, if a significant proportion of terrestrial craters are caused by projectiles having an unmodulated source, such as the asteroid belt, the required modulation of the comet flux would be much greater.

Although estimates concerning the cratering record have large uncertainties, it is generally agreed that most craters can be explained as a result of impacts with bodies derived from the observed population of Earth-crossing asteroids (Grieve and Dence 1979; Shoemaker 1984). Barely 5% of the total cratering rate can be attributed to the present long period comet flux (Weissman 1982; Olsson-Steel 1987). If Earth crossing asteroids predominately arrive as a result of collisions or gravitational interactions in the asteroid belt (Wetherill and Shoemaker 1982; Wetherill 1985), a R_{8b} of about 50 would be required in order to

-170-

explain the cratering data. Alternatively, observed Earth-crossing asteroids may originate from the break up or decay of short period comets (e.g. Bobrovnikoff 1931; Öpik 1963; Shoemaker 1977; Clube and Napier

- Some the second of the secon

ية علي 1 % من ية المراجع الماني. المانية في المراجع المانية المراجع المانية

1982b,1984a,b,1986a,b), thus the terrestrial cratering record reflects changes in the population of short period comets, some of which certainly derive from the isotropic near-parabolic flux. If all short period comets originated from this flux, the cratering record might in principle be explained by $R_{sh} \approx 2$. But if some (possibly the majority e.g. Fernández and Ip 1983a,b; Bailey 1986b) originate from a hypothetical inner core to the Oort Cloud, a presumably unmodulated source not directly connected to the long period flux, the required variation in the latter would obviously be greater.

Therefore, to explain the apparent periodicity in the cratering record the comet shower hypothesis ought to predict $R_{Sh} > 2$. Depending on the ultimate source of Earth-crossing asteroids, it may be that the required value of R_{Sh} should have to exceed 2 by more than an order of magnitude, placing a correspondingly stronger constraint on the theory.

5.2.2 The critical semi-major axis

The critical semi-major axis, a_{crit} , is defined as the value for which a single filling of the loss cylinder down to a_{crit} ensures that the proportion of shower to background comets during a time interval T \cong 30 My will exceed the required value $R_{gh} > 2$.

Following Bailey (1983a,b), a spherically symmetrical model for the comet cloud is defined so that the number

-171-
N(E) dE of comets with orbital energies per unit mass E in the range E, E + dE is given by

$$N(E) = K(E/\hat{E})^{\forall}, \quad E_{min} \leq E \leq E_{mox}$$

$$= 0 \quad otherwise \qquad (5.1)$$

where $E = -GM_{\odot}/2a$, $\dot{E} = -GM_{\odot}/2R_{\odot}$, $E_{m_{\odot}K} = -GM_{\odot}/R_{\odot}$ and $E_{min} = -GM_{\odot}/2a_{\odot}$, K being a constant defined in terms of the observed near parabolic flux and other parameters of the model, a is the semi-major axis of the orbit, $R_{\odot} \simeq 2.10^{5}$ AU is the outer radius of the comet cloud and a_{\odot} is the semi-major axis corresponding to the inner boundary of the cloud. For the Oort (1950) model of the comet cloud & = 5/2but for the arguably more plausible models (Hills 1981; Bailey 1983b,1986a) we generally expect -2 < & < 0. Thus the cloud is centrally condensed having most of its mass in relatively tightly bound orbits with a $\leq 10^{4}$ AU

It has recently been shown that the dominant source for the quasi-steady background comet flux into the inner planetary system is the tidal force of the Galactic disc (Torbett 1986a,b; Morris and Muller 1986; Heisler and Tremaine 1986; Byl 1986; Delsemme 1987), although random stellar perturbations are still required to replenish the low angular momentum orbits on which the tide can operate. Following Heisler and Tremaine (1986), the semi-major axis at above which the tide keeps the loss cylinder steadily filled may be written (Bailey 1986b)

$$\mathbf{a}_{c} = \left(\frac{3\sqrt{2}}{40\pi} \frac{\mathbf{M}_{0}}{\rho_{0}} \mathbf{q}_{1c}^{1/2}\right)^{2/7}$$

(5.2)

where ρ_0 is the total mid-plane density of Galactic material. In this approximation the solar orbit about the Galactic plane obeys simple harmonic motion with the

-172-

interval between successive Galactic plane crossings

$$\mathbf{T} = \frac{\mathcal{R}}{(4\mathcal{R}G\rho_0)}\dot{\mathcal{V}}_2 \tag{5.3}$$

A STREET CONTRACT THE PERSON

Assuming T = 30 My gives $\rho_0 = 0.194 M_0 \text{pc}^{-3}$ (cf. Bahcall and Bahcall 1985), so that

 $a_{\beta} = 3.208.10^{\circ} q_{\beta 5}^{\gamma} AU$ (5.4)

where $q_{1S} = q_{1G} / 15 AU$.

Bailey (1986b) has shown that the injection rate per unit semi-major axis by the Galactic tide, for comets with $a > a_b$ and perihelia q distributed uniformly in the range $0 \leqslant q \leqslant q_{bc}$ is

$$F_{k_{e,t}}(a) = \frac{K}{4\pi} (GM_{\odot})^{1/2} R_{\odot}^{2/3} a^{3/2} J_{le} (a > a_{t}) (5.5)$$

where

$$J_{lc} = (2GM_{\odot} q_{lc})^{1/2}$$
 (5.6)

is the specific angular momentum of a nearly parabolic orbit on the loss cylinder.

Setting $x = a/a_{t}$, $x_{R} = R_{0}/2a_{t}$ and integrating (5.5) over $a \ge a_{t}$, the background injection rate of new comets into the planetary system ($a \ge a_{t}$ and $q \le q_{tc}$) becomes

$$\dot{N}_{gs} = \frac{KR_{0}}{2\pi}^{3/2} (GM_{0})^{3/2} q_{k} a_{c}^{3/2} \int_{1}^{1/2} \frac{\pi^{2}}{x^{3/2}} dx$$
 (5.7)

and the number of background comets injected during a time interval T is simply

$$N_{qs} = N_{qs} T.$$
 (5.8)

Now the number of shower comets injected as a result of filling the loss cylinder down to a semi-major axis $a_{crib} < a_{c}$ can also be calculated. From (5.1) the total number of comets in the cloud per unit semi-major axis is

$$N(a) = K \begin{pmatrix} R_0 \\ a \end{pmatrix} \qquad \frac{GM_0}{2a^2} \qquad (a_0 \leq a \leq R_0/2) \qquad (5.9)$$

and if a strong perturbation causes the loss cylinder to become filled down to a_{C^*} , the fraction of such comets with perihelia less than q_{lc} is $\cong 2q_{lC}$ /a. Hence the number of shower comets may be written

$$N_{Sh} = KR_{\Theta}^{-t}GM_{\Theta}q_{lc} a_{t}^{t-2} \int_{x_{erit}}^{t} dx \qquad (5.10)$$

where $x_{erit} = a_{erit} / a_t$. Combining (5.7), (5.8) and (5.10) the ratio of shower to background comets is obtained.

$$R_{\text{Sh}} = \frac{N_{\text{Sh}}}{N_{\text{QS}}, T} = \frac{P(a_{\text{K}})}{T} \int_{T}^{T} \frac{\sqrt[3]{x_{\text{Crit}}} \times dx}{\int_{1}^{x_{\text{R}}} \frac{\sqrt[3]{x_{\text{Crit}}} - 3}{x_{\text{Crit}} \times dx}, \quad (5.11)$$

where $P(a_{\xi}) = 2R(GM_{\odot})^{1/2} a_{\xi}^{3/2}$ is the period of an orbit with the semi-major axis a_{ξ} . Figure 5.4 shows R_{Sh} as a function of x_{crit} and the power law index \forall of the adopted model of the Oort Cloud. Setting $R_{Sh} = 2$ then defines the value of a_{crit} (for a particular value of a_{ξ}) below which the loss cylinder must temporarily be filled in order to cause an extinction shower. It can be seen that the original non-centrally condensed model of Oort (1950), having $\forall = 2.5$, is unable to produce any significant comet showers at all; and secondly if R_{Sh} is required to be greater than 2 on average, the loss cylinder must be filled down to

 $a_{crit} < 2.10^{b} AU$ (5.12) for $a_{t} = 3.2.10^{b} AU$.

Of course, the above discussion depends on the assumption (equation 5.1) that the Oort Cloud model can be described by a simple power law with constant index \Im . It

-174-

المجار ويتجرج والمحادي والمراجل ووالقرائل



Figure 5.4: The ratio of shower to background comets $R_{sh} = N_{sh} / N_{s,7}$ as a function of x_{crit} $= a_{crit} / a_t$. The curves are labelled by the corresponding spectral index, \mathcal{X} , of the power law distributions for cometary energies (equation 5.1). A value of $R_{sh} > 2$ is required to produce an 'extinction' shower. (A value of $a_t = 3.208.10^{4}$ AU has been used).

may be that Oort Cloud models are not characterised by a single value of the slope of the energy distribution (e.g. Delsemme 1977; Bailey 1983b,1986a) or even that \forall may be time dependent (Kresák 1977; Yabushita 1979; Bailey 1984; Clube and Napier 1984a,1986; Clube 1987a). We have discussed these possibilities in detail elsewhere (Bailey et al. 1987, hereafter referred to as Paper VI) and concluded that there is little effect on the above value of accit. It is this value from the condition $R_{Sh} > 2$ that puts extremely

-175-

tight constraints on the nature of any perturbers capable of causing extinction showers with a mean interval as short as 30 My.

5.3 THE SIGNAL PROBLEM

Sepkoski and Raup (1986) appear to have obtained an almost complete sequence of mass extinctions every $\cong 30$ My during the past $\cong 270$ My, specifically at least 8 events out of 10 (Figures 5.1 and 5.2). If these are due to comet showers, the bodies which disturb the Oort Cloud must have a space density and mass such that they trigger a death shower with almost unit probability during each 30 My interval. Assuming a mean mass density $\rho \equiv$ nM for the perturbers and a relative velocity V to the Solar System, both averaged along the solar orbit about the Galaxy, during the time between Galactic plane crossings we expect an average of one encounter with impact parameter

 $b < b_{1} = (\pi n VT)^{-1/2}$ (5.13)

The probability distribution of minimum impact parameters, b_{min} , which occur during each cycle is

$$P(b_{min}) db_{min} = \frac{2b_{min}}{b_{i}^{2}} \exp\left(-\frac{b_{min}^{2}}{b_{i}^{2}}\right) db_{min} \qquad (5.14)$$

so that the fraction f of cycles in which b_{min} exceeds a value bg is given by $f = \exp(-b_g^2/b_g^2)$. The fact that less than 1/5 of the cycles are missing from a complete sequence of extinctions therefore indicates that encounters with impact parameters up to

$$\mathbf{b}_{1/5} = \sqrt{\ln 5^{1}} \mathbf{b}_{1} = \left(\frac{\ln 5}{\overline{\kappa}} \frac{M}{\rho \mathrm{VT}}\right)^{1/2}$$
(5.15)

must be able to trigger an extinction shower. The question

-176-

*

is, now, is this possible for astrophysically plausible perturbations of the Oort Cloud?

For a given encounter with impact parameter b, the semi-major axis down to which the loss cylinder is filled depends on whether the impulse approximation is valid or not, and, if it is, on whether the perturber passes at a large distance or close enough to the Sun to pass directly through the Oort Cloud. Long period comets with semi-major axes a, lie at mean heliocentric distances on the order of 3a/2. Following Fernández and Ip (1987), if the impulse approximation is valid and b < 3a/2, then the loss cylinder will be filled down to a shower semi-major axis, a_{Sh} , given by

$$a_{sh} = b \left[\frac{2}{3} \frac{g_{le} V^2}{GM_{\odot}} \left(\frac{M_{\odot}}{M} \right)^2 \right]^{1/2} \qquad (b < \frac{3a}{2}) \qquad (5.16)$$

For more distant impulsive perturbations, Hills (1981) obtained

$$a_{sh} = b \left[\frac{8}{27} \frac{q_{lc}}{GM_{\odot}} V^{2} \left(\frac{M_{\odot}}{M} \right)^{2} \right]^{1/4} \left(\frac{3a < b < b_{max}}{2} \right)$$
(5.17)

where b_{max} is the impact parameter beyond which the impulse approximation breaks down.

At large values of the impact parameter, the duration 2b/V of the encounter becomes long compared with the orbital period $2\pi(GM_{\odot})^{1/2}a^{3/2}$ of the perturbed comet with semi-major axis a. This yields a rough estimate of the maximum impact parameter b_{max} above which (5.17) breaks down, i.e. $b_{max} \approx V(a^3/GM_{\odot})^{1/2}$. In this situation, a_{sh} may be estimated from equation (5.2) with the factor $4\pi\rho_{\odot}$ replaced by M/b^{\odot} (Byl 1983) which gives

-177-

$$a_{Sh} = \left(\frac{3\sqrt{2^{2}}}{10} q_{lc}^{1/2} b^{3} \frac{M_{\odot}}{M}\right)^{2/7}$$
 (b > b_{max}) (5.18)

The continuity of equations (5.17) and (5.18) at $b = b_{most}$ gives

$$b_{max}$$
 (a) = $\sqrt{\frac{400}{243}} V \left(\frac{a^3}{GM_{\odot}}\right)^{1/2}$ (5.19)

Equations (5.16) - (5.18) determine the value of a_{Sh} down to which the loss cylinder is filled by a single encounter with impact parameter b. Setting b = b_{NS} gives values of a_{Sh} for which bodies with parameters M, ρ and V are able to fill the loss cylinder down to a_{Sh} during a specified time interval T in 80% of cycles. These equations with equation (5.15) thus become

$$a_{\text{Sh}}(AU) = 5.659.10^{4} q_{15}^{1/2} v_{20}^{1/2} T_{30}^{-1/2} \rho_{0s}^{1/2} \left(\frac{M_{\odot}}{M}\right)^{1/2}$$
 (M < M₀) (5.20)

$$a_{\text{Sh}} (AU) = 3.171.10^{\text{b}} q_{15}^{1/2} T_{30}^{-1/2} \rho_{05}^{-1/2} (M_1 \leq M \leq M_2) (5.21)$$

$$a_{\text{Sh}} (AU) = 2.670.10^{\text{b}} q_{15}^{1/7} V_{20}^{-3/7} T_{39}^{-3/7} \rho_{05}^{-3/7} \left(\frac{M_1}{10^{\text{b}} M_0}\right)^{1/7} (M > M_2) (5.22)$$

where
$$M_1$$
 and M_2 are given by
 $\left(\frac{M_1}{M_{\odot}}\right) = \left(\frac{3q_{1c}V^2}{2GM_{\odot}}\right)^{1/2} = 3.184 q_{1s}^{1/2} V_{20}$
(5.23)

anđ

$$\left(\frac{M_2}{M_{\odot}}\right) = 3.331.10^{\frac{1}{12}} q_{15}^{\frac{3}{12}} V_{20}^{\frac{3}{12}} T_{20}^{-\frac{1}{2}} \rho_{05}^{-\frac{1}{2}}, \qquad (5.24)$$

with $V_{20} = V/20 \text{ kms}^{-1}$, $T_{30} = T/30$ My and $\rho_{05} = \rho/0.05 \text{ M}_{\odot} \text{pc}^{-3}$. As emphasised by Fernandez and Ip (1987), equation (5.20) should be used whenever the mean mass of the perturbers is less than M₁, that is for perturbations by typical stars and by hypothetical smaller bodies such as brown dwarfs.

These expressions show that a_{h} is least when M lies in the range $M_1 \leq M \leq M_2$ and that for this broad range of

-178-

masses the value of $a_{M_{h}}$ is independent of both M and V. Equation (5.12) means that $a_{Sh} \leq 2.10^{\circ}$ AU in order for a shower to register in the terrestrial record, and therefore perturbers, in order to produce mass extinctions at 30 My intervals, must have a mean mass density $p > 0.1 M_{\odot} pc^{-3}$ averaged along the solar orbit (equation 5.21). This depends on the assumption that $R_{Sh} \cong 2$ for detectable showers; if the required value of R_{Sh} is larger, then a_{eric} will be lower and the minimum mass density of perturbers even higher. Since the limit on the perturber density is already close to the maximum allowable local density of all Galactic material (i.e. $\rho_0 \approx 0.2 M_{\odot} pc^{-3}$, from stellar observations and in order that the mean time between Galactic plane crossings should be 🗳 30 My), it is clear that the Galactic model cannot produce shower-to-background ratios much greater than 2 for 30 My comet showers.

The above conclusions depend on the use of (5.11) to calculate the ratio of shower to background comets. Fernandez and Ip (1987) call this contribution to a shower the 'halo' component, since it comprises comets coming into the planetary system from all directions, distributed more or less isotropically about the Sun. However, comets whose orbits lie close to the path of the perturber may also be strongly affected, thereby possibly constituting a second 'clustered' contribution to a comet shower. Assuming that comets with semi-major axes a lie at heliocentric distances close to r = 3a/2, encounters with an impact parameter b will thus only affect comets in a clustered sense if a > 2b/3. Thus, as noted by Fernandez and Ip (1987), a non-zero clustered contribution will be associated with

-179-

relatively low mass perturbers, i.e. $M < M_{j}$, or (equation 5.23) less than or equal to the mass of a typical star. Since molecular clouds have $M \gg M_{\mu}$, a full derivation of this topic will not be given here, but can be found in Paper VI. It is shown there that the cluster shower to background ratio is given for various models of the Oort Cloud by

$$R_{GLMS} = 0.09 q_{1S}^{-1/2} V_{20}^{-1} T_{29}^{2} \rho_{0S} \left(\frac{M}{M_{\odot}}\right) \qquad (\Im = 0)$$

$$= 0.17 q_{1S}^{-S/1/2} V_{20}^{-1/2} T_{20}^{+3/1/2} \rho_{0S} \left(\frac{M}{M_{\odot}}\right)^{1/2} \qquad (\Im = -1) \qquad (5.25)$$

$$= 0.32 q_{1S}^{-2/1/2} T_{30}^{-S/1/2} \rho_{0S}^{-S} \qquad (\Im = -2)$$

$$= 0.57 q_{1S}^{-1/14} V_{20}^{1/2} T_{30}^{-3/1/2} \rho_{0S}^{-S} \left(\frac{M_{\odot}}{M}\right)^{1/2} \qquad (\Im = -3)$$

(४≕−3)

These expressions show that if the Oort Cloud is only
moderately centrally condensed (i.e.
$$2 > -2$$
), the maximum
value of R_{clus} , reached for high densities and $M \cong M_1 \cong 3M_{\odot}$,
is of order unity. Therefore for such models, showers of
clustered comets at 30 My intervals will be too weak to
register in the terrestrial record. If, however, the Oort
Cloud is extremely centrally condensed, $2 = -3$, R_{clus} rises
as the mass of the perturbers decreases, which could lead to
very large values of R_{clus} for small masses. For example,
one could postulate a model in which $2 = -3$ (corresponding
to a cloud with density $n_e(r) \ll r^{-7}$) and assume that all the
missing mass in the Galactic disc, resided in low mass brown
dwarfs, i.e. $\rho \simeq 0.1 M_{\odot} pc^{-3}$ and $M \simeq 0.01 M_{\odot}$. This would
imply $R_{clus} \simeq 30$. Thus showers of clustered comets might
affect the terrestrial record if (i) the missing mass in the

-180-

Galactic disc is composed entirely of low mass bodies with $M \simeq 10^{-2} M_{\odot}$ (cf. Stothers 1984) and (ii) the Oort Cloud itself is extremely massive > $10^{\circ} M_{\odot}$ and centrally condensed. This latter condition presents great difficulty and may prove prohibitive (Paper VI).

In summary, for the Galactic hypothesis to produce significant variations in the cometary flux (i.e. $R_{sh} \ge 2$) on a regular 30 My timescale, the perturbers must have a high mass density > 0.1M₀pc³, and large masses M > $3M_0$. The local density of molecular clouds ($\simeq 1-3.10^{-2} M_{\odot} pc^{-3}$; e.g. SSS 1984; Bhat et al. 1985; Dame et al. 1987; Section 4.6), even considering the uncertainty in α_{20} , is much too small to impose a recognizable 30 My modulation of the cometary flux, periodic or otherwise (cf. Thaddeus 1986; Scoville and Sanders 1986). In fact, 😋 🗠 28 would be needed to account for 30 My comet showers! Such a large value of α'_{20} is inconsistent with reasonable estimates from &-rays and other methods (Chapter 1). In addition such large GMC masses implied by such a large $\mathfrak{A}_{\lambda 0}$ would have catastrophic consequences for the survival of the whole of the Oort Cloud (Section 5.5). Therefore, no useful constraint can be derived on the mass density of GMCs from 30 My periodicities in the terrestrial record. The suggestion that known molecular clouds produce the perturbations leading to 30 My comet showers (Clube and Napier 1984a; Rampino and Stothers 1984a, 1986; Stothers 1985) cannot be valid, irregardless of arguments concerning the value of α_{20} , scale heights or phase problems. Comet showers however, with a much longer characteristic interval may still be produced by known bodies, for example

-181-

200 - 500 My for molecular clouds or stars (cf. Morris and Muller 1986).

The mass density of perturbers necessary on the Galactic model exceeds any which can be associated with known bodies. The total visible mass observed in the solar neighbourhood is $\approx 0.1 M_{\odot} \text{pc}^{-3}$ (Hill et al. 1979; Stothers 1984). The possibility, however, that the perturbations are by massive dark bodies is difficult to rule out completely, as the density limit is just within the maximum allowed for the dark matter in the solar neighbourhood (Bahcall 1984, 1986a). However, Section 4.4 has reviewed evidence for the nature of the dark matter and concluded that the dark matter (if it exists at all) is probably in the form of low mass stars (Bahcall 1986b). Observations of the existence of wide binaries in the solar neighbourhood may even place an upper limit of $\approx 2M_{\odot}$ on the masses of the dark bodies (Bahcall et al. 1985), ruling out the idea of > $3M_{\odot}$ perturbers altogether (but see Wasserman and Weinberg 1987).

It has been suggested (Clube 1987a,b) that the required dark matter may reside in the GMCs themselves, as 'primary condensations' such as brown dwarfs and giant comets. Clube (1987c) further suggests that this could explain why the mass of H₂ from virial analysis appears to be greater than derived from \forall -rays (Table A.1.1). However, Bhat et al. (1986) have argued that there is essentially little discrepancy between virial masses and \forall -ray masses when the more appropriate optically thinner lines are used. If all of the dark matter resided in GMCs (i.e. $\rho \sim 0.1 M_{\odot} pc^{-3}$), using the virial mass arguments even of SSS (1984), it is difficult to see how this could occur without significantly

-182-

increasing their internal velocity dispersion above that observed. It is also difficult to see how such a large population of brown dwarfs could be confined to GMCs close to the plane.

Low mass perturbers may constitute the dark matter, but are ineffective unless their masses are extremely small $(M < 10^{-2} M_{\odot} \text{ for } \rho \simeq 0.1 M_{\odot} \text{pc}^{-3})$ and the Oort Cloud model is both exceedingly massive and centrally condensed. Hills (1985, 1986) has considered possible limits to the space density of low mass perturbers in the Galactic disc, but for bodies with masses $\approx 10^{-2} M_{\odot}$ the available constraints are extremely weak. However, such a massive and centrally condensed Oort Cloud seems unlikely.

We thus conclude that it is most unlikely that the required perturbations of the Oort Cloud can be provided by missing mass in the Galactic disc. Whether large mass or low mass perturbers, ρ is required to be greater than 0.1 M_Opc⁻³. This value is marginally consistent with estimates of the dark matter mass density made by Bahcall (1984, 1987), but is far greater than more recent estimates made by other workers (e.g. Gilmore and Wyse 1987; Bienaymé et al. 1987; Rana 1987a). Even if the dark matter can provide $\rho > 0.1 \text{ M}_{\odot}\text{pc}^{-3}$ with suitable perturbers, the next section describes two further difficulties for the Galactic explanation of periodic comet showers: the signal-to-noise problem and the phase problem.

5.4 THE SIGNAL-TO-NOISE AND PHASE PROBLEMS

According to the Galactic hypothesis, mass extinctions

-183-

are most likely to occur when the Solar System is close to the Galactic plane. This is where the space density of perturbers is greatest, and where the likelihood of a close encounter is highest. Following Thaddeus and Chanan (1985), the assumption is made in this Section that the cometary influx is simply proportional to the rate of encounters with perturbers, which are taken to be material with gaussian scale height $h_p = \frac{1}{2}h_c$, where h_c is the corresponding scale height of known molecular clouds. (The assumption of a gaussian distribution fits the large scale CO surveys and is generally accepted by radioastronomers; the following analysis is not sensitive to this assumption and remains essentially unchanged if an exponential or other plausible form is taken). Observations of molecular clouds in the solar neighbourhood and in the region presumably covered by the Solar System during the last 270 My indicate that the half-width half maximum of the distribution is $z_{1/2} = 85 + 20$ pc (Sanders 1981; Dame and Thaddeus 1985; Bronfman et al. 1987). In particular, Dame et al (1987) find $z_{1/2} = 87pc$ from local measurements which are independent of assumptions concerning the distance to the Galactic Centre. Thus, recent measurements do not favour the value of $z_{N2} = 60$ pc advocated by Stothers (1985). In general, whatever the nature of the perturbers, their scale height hp must be at least as great as that of the young stars and molecular clouds in the Galactic disc, so 5 > 1,

The approximation of Thaddeus and Chanan (1985) favours the Galactic modulation model, because by ignoring the more or less steady background flux of comets due to distant stellar perturbations and the Galactic tide, it artificially

-184-

enhances any periodic signal in the cometary influx. It also implicitly assumes that encounters are short range, the effect of long range encounters is to reduce the modulation.

Nevertheless despite these biases to greater signal-to-noise ratios, it has proved difficult for the Galactic hypothesis to achieve a signal-to-noise ratio in the encounter rate with molecular clouds as high as that observed in the terrestrial record of mass extinctions ($\cong 2$). This difficulty arises because the present amplitude, z_0 , of the solar motion perpendicular to the Galactic plane is comparable or less than the scale height of the proposed perturbers.

A related difficulty, the so called phase problem, arises because although the record of mass extinctions is essentially complete over the past ≈ 270 My and the Sun's present position is close to the Galactic plane, we now seem (unexpectedly on the model) to be roughly halfway between extinctions (Raup and Sepkoski 1984).

A possible solution to both problems is to assume (e.g. Thaddeus 1986; Bailey et al. 1986) that the required encounters between the perturbers and the Solar System were more strongly modulated in the past than now. This may be justified because the amplitudes of oscillation of stars as old as the Sun are mostly significantly greater ($\simeq 250$ pc, Wielen 1977; Fuchs and Wielen 1987) than the present solar value ($z_0 \simeq 80$ pc). Clube and Napier (1986a) have therefore argued that over the past $\simeq 270$ My the Sun's amplitude of oscillation might have been more typical of other G stars of its age and that a recent close encounter with another body, for example Gould's Belt (Napier 1986) has reduced its

-185-

velocity to the present value, simultaneously changing the orbital phase.

In order to study this possibility, the solar motion is assumed to be approximately harmonic perpendicular to the plane, that is

 $z(t) = z_{\odot} \sin [w(t + t_{\odot})]$ (5.26) where t = 0 denotes the present time, w is the angular frequency and t_G is the time of the last crossing of the Galactic plane. If the present z-position and z-velocity of the Sun are taken to be \simeq 8pc north of the plane and \simeq 8 kms⁻¹ away from the plane (Mihalas and Binney 1981; Stothers 1985) then t_G \simeq 1 My. It should be noted that other studies especially those in relation to young stars (de Vaucouleurs and Malik 1980) suggest a present z-position of \simeq 20-30 pc, but this has little effect on the conclusions.

Now, if a close encounter with another body occurred at a time t = $-t_{Q}$, leading to a strong perturbation of the solar orbit, this itself would also produce an intense comet shower and an episode of mass extinction. According to the data of Raup and Sepkoski (1984), $t_{Q} \simeq 11$ My, but an exact determination of the date of the most recent extinction event is difficult and other estimates cover a wide range (Napier 1986). In what follows, although this is not a crucial assumption, $3 \leq t_{Q} \leq 17$ My is taken.

The orbit of the Sun prior to the encounter is therefore

 $z(t) = z_{i} \sin [w(t + t_{c}) + S]$ (5.27) where $z_{i} > z_{o}$ is the prior amplitude of oscillation and S is a phase factor introduced to describe the distance of the

-186-

Sun from the plane at the time t_{\odot} of the encounter. Therefore from (5.26) and (5.27)

$$z_{e} = z_{o} \sin \left[w(t_{e} - t_{e})\right] = z_{i} \sin \delta \qquad (5.28)$$

or
$$\sin \delta = \frac{z_0}{z_1} \sin \Theta$$
 (5.29)

The pre and post-encounter velocities of the Sun can also be derived,

$$z_2 (-t_2) = wz_4 \cos \delta$$
 (5.31)

and
$$z_{\ell}(-t_{\ell}^{\vee}) = wz_{0} \cos \Theta$$
 (5.32)

The required impulse $\operatorname{Av}_{\mathbb{Z}}$ in the z direction is

therefore,

$$\Delta v_{\chi} = w z_0 \cos \Theta - w z_1 \cos \Theta \qquad (5.33)$$

or
$$\underline{\Delta v_{z}}_{WZ_{0}} = \cos \Theta \pm (r_{\dot{a}}^{2} - \sin^{2}\Theta)^{1/2}$$
 (5.34)

where $r_{\dot{A}} = z_{\dot{A}}^{*}/z_{\Theta}$. The solution for Δv_{Z} as a function of $z_{\dot{i}}^{*}/z_{\Theta}$ thus has two branches, depending on whether the encounter merely slows the Sun down or actually reverses its direction of motion. Now for T = 30 My, the range in Θ corresponds to the range in t_{Q} , that is -1.6 $\leq \Theta \leq -0.2$. However, whatever the exact phase, it is readily verified that the solutions for $\Delta v_{Z}^{*}/wz_{\Theta}$ are bounded for all Θ by solutions corresponding to $\Theta = 0$, i.e. $\Delta v_{Z} = wz_{\Theta} (1 \pm r_{\dot{A}})$. Since $r_{\dot{A}}^{*} > 1$ (because $z_{\dot{A}}^{*} > z_{\Theta}$), the encounter must produce an impulse in the z direction whose magnitude $|\Delta v_{Z}| > wz_{\Theta}(r_{\dot{A}}^{*} - 1)$.

Thaddeus and Chanan (1985) showed that a minimum r_4° value of \cong 3 was necessary in order to achieve a signal-to-noise ratio of 2 in the encounter rate with molecular clouds. In order to be extremely conservative, it will be assumed that $r_4^{\circ} \cong 3/2$ will produce a sufficient

-187-

signal-to-noise ratio in the terrestrial record. Thus, allowing any other perturbers to have a scale height § times that of molecular clouds, for $z_0 \approx 80$ pc, the initial amplitude of the solar motion prior to the hypothetical massive encounter must be $z_i \geq 120$ § pc. For § = 1, since the present mid plane velocity, $wz_0 \approx 8$ kms⁻¹, the minimum z-velocity change necessary to produce the required change in amplitude is $V_{mun} \approx 4$ kms⁻¹. Therefore, for an encounter with a point mass M, impact parameter b and relative velocity V, the total impulsive velocity change Δv must satisfy

 $\Delta v = \frac{2GM}{bV} > |\Delta v_2| \geq \Im V_{min} = wz_0 (r_A - 1) \Im = 4 \Im km s^{-1} (5.35)$

This inequality places stringent limits on the likely mass of the body encountered. A constraint is obtained by requiring that the intensity of the accompanying comet shower should not be too great, since the evidence of the terrestrial record does not suggest that the most recent event was particularly extreme (Figure 5.1). This probably requires $a_{sh} > 10^{b}$ AU (cf. Figure 5.4), and so combining equations (5.35) and (5.17),

$$\frac{M}{M_{\odot}} > 1.681.10^{5} q_{15}^{-1/2} g^{2} V_{20} \left(\frac{V_{min}}{4 \text{kms}^{-1}}\right)^{2} \left(\frac{a_{sh}}{10^{4} \text{AU}}\right)^{2} (5.36)$$

Therefore, if the Solar System has indeed been recently decelerated in the z-direction as required for the Galactic hypothesis to satisfy the signal-to-noise and phase constraints of the terrestrial record, the body which caused the impulse must have had $M >> 1M_{\odot}$. The mass range implied, suggests that a sufficiently massive GMC may have caused such a perturbation (e.g. Clube and Napier 1986a), although if $\S > 3$ (corresponding to a dark matter scale height

-188-

comparable with that of the stellar disc) the required mass of the cloud is excessively large. Using the CO data of Dame et al. (1987), observations of molecular clouds in the solar neighbourhood (< 350 pc) do not show the existence of any sufficiently massive centrally condensed body in the right direction and in any case the required impact parameter would probably correspond to a penetrating encounter of the cloud with attendant reduced net effect. The data of Taylor et al. (1987) on Gould's Belt show it to be too large a structure and not massive enough to provide the required impulse. If the halo is composed of massive black holes (Lacey and Ostriker 1985) or dark clusters (Carr and Lacey 1987), then their mass of $\sim 10^{9} M_{\odot}$ is of the required order, but the relative velocity as they pass through the disc is # 250 kms¹ (Ipser and Semenzato 1985), which rules them out as suitable candidates. Moreover, as § increases the required change in velocity of the Sun becomes quite prohibitive.

Unless the scale height of the hypothetical perturbers is less than 200 pc (i.e. $\S < 3$), the signal-to-noise and phase problems are most unlikely to be overcome by postulating a recent close encounter of the Solar System with a GMC. Currently accepted values for the scale heights of dark matter tend to be larger than this (e.g. Bienayme et al. 1987).

Lastly, two further difficulties for the Galactic explanation of mass extinctions by periodic comet showers need to be mentioned. First, interactions with perturbers of any kind will inevitably change both the amplitude and phase of the solar orbit about the Galactic plane, so that

-189-

the extent to which strict periodicity is present in the geological record imposes a tight constraint on the theory. In general, an astronomical hypothesis of the kind considered here will not predict a constant signal-to-noise ratio or phase over a very extended period of time. Secondly, the geometry of the astronomical model naturally predicts that a few encounters will have much smaller impact parameters than the rest, and this will be reflected in the distribution of values of a_{q_b} (equations 5.20 - 5.22) which scales roughly in proportion to b. Since the Oort Cloud must be centrally condensed on the model (Figure 5.4), this implies a rapidly falling frequency distribution of individual shower intensities which is not matched at all by the terrestrial record of mass extinctions (Figure 5.1). Neither of these difficulties can be accurately quantified with present knowledge (the first because the nature of the perturbers is still unknown, the second because the link between the comet flux and the mass extinction rate has not been firmly identified), although both appear to militate against a direct astronomical link with periodic mass extinctions.

In conclusion, if the strong 30 My periodicity apparent in the geological record is confirmed, direct astronomically induced processes can be paradoxically almost certainly be ruled out, as no known observed or dark matter candidates have sufficient mass density or small enough scale height. This suggests that any periodicity, especially in the cratering record is probably spurious (c.f. Shoemaker and Wolfe 1986; Tremaine 1986). The claim that an apparent 30 My periodicity in volcanic activity is induced by

-190-

cometary showers (Pandey and Negi 1987), irregardless of the uncertain link between comets and induced volcanism (Rampino 1987), must also be incorrect. Apart from the cratering record, periodicities in the other terrestrial indicators must then be ascribed to processes intrinsic to the Earth.

If the evidence for periodicity in the geological record is shown to be false, it may still be possible to attribute global mass extinctions to a weakly modulated rate of arrival of very massive comets in near-Earth orbit, as previously suggested for periodic extinctions (Clube and Napier 1986a). The mean interval between mass extinctions would then depend, among other things, on the mass distribution of the largest comets (large enough to cause a significant terrestrial trauma), and the biological reaction to such events.

The claimed 30 My periodicity can therefore not be used to constrain the masses of GMCs. Finally, the next section briefly reviews one more aspect of the interaction of GMCs with the Oort Cloud which has in the past been used to constrain GMC masses.

5.5 THE SURVIVAL OF THE OORT CLOUD

It has been suggested that very close encounters with GMCs over the lifetime of the Solar System would lead to a complete stripping of comets from the Oort Cloud (Clube and Napier 1982b; Napier and Staniucha 1982). If this is correct, then a presently observed Oort Cloud implies replenishment of comets from some source. It is well known that comet capture from interstellar space is an inefficient

-191-

process, even taking into account three body encounters with planets (Valtonen and Innanen 1982; Torbett 1986c). Clube and Napier (1982a, 1984a) therefore suggested a three body process using GMCs. There are some arguments for believing that there exists a high space density of comets in molecular clouds (Clube and Napier 1982a, 1985; Bailey 1987), and in an encounter of the Solar System with a GMC, it was postulated that the Oort Cloud as well as being disrupted was then replenished with new comets. On this picture, although the Solar System may have had a primordial comet cloud, the presently observed comets are of interstellar origin. There are, however, many problems with this view including whether GMCs can confine comets for a long enough time (van den Bergh 1982) and whether this replenishment can occur often enough (Torbett 1986c).

If the Oort Cloud was not totally stripped over the lifetime of the Solar System, then of course there would be no need for comet replenishment. Bhat et al. (1986) pointed out that if the median mass of clouds was $1.3.10^{5}$ M₀ rather than the conventional 5.10^{5} M₀ (SSS 1984) a significant fraction of the Oort Cloud should in fact remain intact up to the present day, removing the need for Oort Cloud replenishment (cf. Hut and Tremaine 1985). This may be interpreted, in the light of problems with the replenishment mechanism as evidence once again for a lower α_{20} than advocated by SSS (1984).

However, there is another alternative, and that is to postulate a massive inner core to the Oort Cloud which is stable to even the very close encounters (Hills 1981; van den Bergh 1982; Bailey 1983a,b). The disrupted outer Oort

-192-

Cloud is then replenished with comets from the inner core. It may be that constraints on such a model can be provided by the total mass of the Oort Cloud, effect on planetary orbits and the short period comet flux (Bailey 1983a, 1986b; Paper VI) but at present such constraints only argue against extreme centrally condensed models of the Oort Cloud (Section 5.3). A model with $\Im \simeq -2$ would mean that a significant fraction of the Oort Cloud would survive even with the higher values for the masses of GMCs.

Therefore, if the model of the Oort Cloud can be better specified then the survival of the Cloud may be used to constrain GMC masses. However, considerable uncertainty in how centrally condensed the Oort Cloud is, means that at present the higher masses of SSS (1984) cannot be totally ruled out by this argument.

CHAPTER SIX

THE LIFETIME OF GIANT MOLECULAR CLOUDS AND

H₂ IN THE VIRGO CLUSTER SPIRALS

6.1 INTRODUCTION

The early surveys of CO in the Galaxy, adopting a large value of \mathscr{A}_{2O} , had indicated a large amount of H_2 in the inner Galaxy, > 5 times the mass of HI at the peak of the molecular ring (e.g. Solomon, Sanders and Scoville 1979). From this result Scoville and Hersh (1979) argued that H atoms would usually be in the form of H_2 rather than HI which suggested that the lifetime of GMCs > 10° - 10° y. In fact, early estimates of the growth of 10° - 10° M_O clouds also indicated lifetimes $\approx 10^{\circ}$ y which seemed to confirm this picture.

However, work on the chemical evolution of the Galaxy (Section 4.6) and the \forall -ray data (Bhat et al. 1985; Strong et al. 1987) has suggested that the mass of H₂ in the inner Galaxy is roughly equal to the mass of HI, the conversion factor \mathfrak{A}_{20} being \mathfrak{a} 3 in the solar neighbourhood and varying as a function of Galactocentric radius. This would indicate that the lifetime of GMCs $\leq 10^{6}$ y (Blitz and Shu 1980). It is the intention of this Chapter to ask whether this is consistent with current estimates of the lifetime of GMCs. The argument can of course be inverted and evidence for 'short' GMC lifetimes may indicate lower H₂ estimates in the inner Galaxy.

The question is quite complex and so Section 6.2 re-examines the argument of Scoville and Hersh (1979) and

-194-

its various assumptions. Evidence is presented that it is the radial dependence of the lifetime of the HI phase which controls the mass ratio of H_2/HI . Previous arguments for a long lifetime of GMCs are shown to be no longer valid and evidence favouring a short lifetime is presented.

It is against this background that recent CO surveys of galaxies in the Virgo cluster have been used to claim GMC lifetimes $\geq 10^{\circ} - 10^{\circ}$ y (Kenney and Young 1986; Stark et al. 1986). As the possibility of making an estimate of the lifetime of the molecular gas (or even putting a lower limit to it) is clearly attractive and important for the mass of gas in the inner Galaxy, these claims are critically examined in Section 6.3.

This leads on to a consideration of α_{20} in the Virgo spiral galaxies. The conventional approach in estimating the mass of H_2 in external galaxies has been to assume that the value of α_{20} determined locally, applies universally (e.g. Young and Scoville 1982a; Dickman et al. 1986). Section 6.4 suggests that the cluster environment in Virgo changes α_{20} with respect to isolated galaxies and uses evidence from SFRs and IR results to confirm this claim. Finally, the consequences of these results for the evolution of the Virgo galaxies are briefly described.

6.2 THE LIFETIME OF GIANT MOLECULAR CLOUDS

6.2.1 The mass of H2 in the inner Galaxy

Scoville and Hersh (1979) proposed that if the ISM constantly cycles gas through the H_2 , HI and HII phases, and the relative abundances are maintained in steady state,

-195-

continuity demands that the mass flow rate out of the H_2 phase $(M_{M_2}/\mathbb{T}_{M_2})$ must equal the flow rate out of the HI and HII phases into H_2 (M_{MI}/\mathbb{T}_{ME}) , that is

$$\frac{M_{H_2}}{T_{H_2}} = \frac{M_{HE}}{T_{HE}}.$$
 (6.1)

The time scale for a typical low density atomic or ionised hydrogen region to exist is $\cong 10^{\$}$ y, so that if $M_{M_2} > M_{MT}$ then $T_{M_2} > T_{MT}$. The quantity T_{M_2} was directly identified with the GMC lifetime.

However, since GMCs have different masses, the definition of their lifetime as a group, T_{M_2} , depends on the time scale of evolution of a cloud as a function of its mass, t(m) and on the mass distribution of clouds

 $f(m) dm = m^{-\psi'} dm \qquad (6.2)$

where $\sqrt[4]{2}$ 1.5 (Stark et al. 1987). If GMCs are produced from HI behind shocks (e.g. Bash et al. 1977) then \mathbb{T}_{M_2} is unambiguous, being the average mass weighted period between cloud formation and disruption. However, if GMCs are formed by cloud-cloud collisions,with larger molecular clouds growing at the expense of smaller ones, \mathbb{T}_{M_2} includes both the time span of the giant cloud (\mathbb{T}_{GMC}) and the time taken for smaller clouds to grow to that mass. Therefore, in the following \mathbb{T}_{M_2} will be termed the lifetime of the molecular phase with $\mathbb{T}_{M_2} \gg \mathbb{T}_{GMC}$.

At the peak of the molecular ring, the estimate of H_2 presented in this work (Figure 4.9) gives $M_{H_2}/M_{MX} \approx 1.0$ which implies by equation (6.1) $T_{H_2} \simeq 10^{\circ}$ y. In contrast, the distribution of SSS (1984) gives $M_{H_2}/M_{MX} \approx 5$ and therefore $T_{H_2} \simeq 5.10^{\circ}$ y. It is important to be clear about the uncertainties in this analysis. The first is the assumption that the relative abundances of HI and H_2 are

-196-

maintained in a steady state. This is not strictly true as gas is continually depleted due to star formation, some of the gas being returned to the ISM but some being locked up in long lived stars or dead stellar remnants. The aspect of star formation may be examined crudely in the following way. The rate of change of H_{\odot} is given by

$$\frac{dM_{W_2}}{dt} = -\frac{M_{W_2}}{T_{W_2}} + \frac{M_{WT}}{T_{WT}}.$$
(6.3)

Assuming that all H_2 cycles back to HI through stars, which lock up a fraction (1-et), then

$$\frac{dM_{MR}}{dt} = - \frac{M_{MR}}{U_{MR}} + e \leq M_{H_2}$$
(6.4)

Scoville and Hersh (1979) assumed $\mathfrak{A} = 1$ and $\mathrm{dM}_{W_2}/\mathrm{dt} = \mathrm{dM}_{W_2}/\mathrm{dt} = 0$ in order to obtain equation (6.1). In Section 2.4, the fraction of molecular gas to the total hydrogen was shown to be empirically correlated with the metallicity Z, such that

$$\frac{M_{\text{N2}}}{(M_{\text{N2}} + M_{\text{HZ}})} = AZ^{-b}$$
(6.5)

where b 🖆 1.3 - 1.4. This expression yields

$$\frac{1}{M_{H_2}} \frac{dM_{H_2}}{dt} = \frac{1}{M_{H\Sigma}} \frac{dM_{H\Sigma}}{dt} + 2.3b \left(\frac{d\log Z}{dt}\right) \left(\frac{M_{H_2}}{M_{M\Sigma}} + 1\right). \quad (6.6)$$

Substituting equations (6.3) and (6.4) into (6.6) gives

$$\frac{1}{\mathbf{L}_{\mathbf{H}_{2}}} \left(\frac{\mathbf{M}_{\mathbf{H}_{2}} + \mathbf{o} \mathbf{\xi} \mathbf{M}_{\mathbf{H}_{2}}}{\mathbf{M}_{\mathbf{H}_{2}} + \mathbf{M}_{\mathbf{H}_{2}}} \right)^{-2.3b} \frac{\mathrm{dlog}Z}{\mathrm{dt}}$$
(6.7)

which reduces in the limit $\mathcal{A} \longrightarrow 1$ and $d\log 2/dt \longrightarrow 0$ to equation (6.1). From Figure (4.10), $d\log 2/dt$ over the past 5Gy at $R_{\mathfrak{G}} = 6$ kpc may be estimated to be $\cong 10^{-10}$ dex y^{-1} which over $10^{\mathfrak{B}}$ y makes a negligible contribution if $M_{\mathfrak{ME}}/M_{\mathfrak{M}_2} \approx 1$. Section 4.2 estimated the returned fraction as 0.4 - 0.6, but \cong will be effectively much higher as not all the H₂ will form stars (say 50%). We therefore conclude that the effects of the relative abundances not being constant will

-197-

only alter lifetimes derived from equation (6.1) by at most 20%.

On the basis of this, we may now go on and examine equation (6.1) in the context of variation with Galactocentric radius. In Figure 6.1 the variation of $M_{M_2}/M_{M_{\overline{M}}}$ has been derived as a function of R_6 , using the H_2 distributions of SSS (1984) and the present work (Figure 4.9). Assuming that each disc annulus is closed (i.e. there are no significant gas flows), this variation also represents $T_{M_2}/T_{M_{\overline{M}}}$, which varies by at least an order of magnitude over the disc of the Galaxy.

Present theories of cloud formation, whether by large scale gravitational instability or cloud-cloud collisions are not dependent on whether the gas is H_{2} or HI, and in fact the observation that the size spectrum of molecular clouds both in the inner and outer Galaxy is very similar to the HI size spectrum as measured in the solar neighbourhood (Tereby et al. 1986) indicates that the formation mechanism is the same. The problem is therefore, why does $M_{M_2}/M_{M_{\rm L}}$ vary by over an order of magnitude. Elmegreen and Elmegreen (1987b) have found a similar trend in individual 'superclouds' of mass $\simeq 10^7 M_{\odot}$. The total cloud mass appears to be constant as a function of R_{G} (Grabelsky et al. 1987) but molecular cores become less massive and atomic envelopes become more massive with increasing R_G. Figure 6.1 suggests that the increase of M_{M_2}/M_{M_3} with decreasing R_{G_1} could be due to either increasing Tw, in the inner Galaxy or decreasing Twr.

If molecular clouds were held up from collapse longer in the inner Galaxy then T_{M_2} could be higher. However,

-198-



 $\frac{\text{Figure 6.1}}{\text{is as a function of Galactocentric radius. HI}} : The ratio of the mass of H₂ to the mass of HI as a function of Galactocentric radius. HI values have been averaged from Table A.2.1 (Appendix) while H₂ has been taken from SSS (1984) and the present work (Figure 4.9). Following the argument of Scoville and Hersh (1979) this variation also represents the ratio of the lifetime of the molecular and atomic phases.$

firstly if \mathbb{V}_{N_2} were responsible for the variation, a value of $\mathbb{V}_{N_2} \approx 10^{\circ}$ y at the 6kpc ring would imply $\mathbb{V}_{N_2} \leq 10^{\circ}$ y in the outer Galaxy. If GMCs are built up from cloud-cloud collisions (Section 6.2.2), the time-scale for forming $10^{\circ} - 10^{\circ} M_{\odot}$ clouds is > 10° y, implying that the size spectrum of molecular clouds in the outer Galaxy would be very different. However, observations of the size spectrum show no such effect (Tereby et al. 1986). This suggests that \mathbb{V}_{N_2} may be roughly constant with radius.

Secondly, evidence for a roughly constant \mathbb{T}_{W_2} may be derived from the observed SFR. The rate of flow of H_2 into HI must be connected to the SFR (Chapter 2). The SFR ($\frac{4}{5}$) will be approximately given by

$$\Psi_{\rm S} \simeq \eta \, \underline{M}_{\rm M_2} \tag{6.8}$$

where η is the efficiency of star formation, or essentially the amount of H₂ that forms stars. The results of Section 2.3.1 showed a correlation in the disc of the Galaxy of $\psi_s \propto M_{W_2}^{k}$ with k close to unity. Equation (6.8) therefore implies $\overline{w}_{M_2} \ll \eta$ as a function of R_G. Myers et al. (1986) have recently estimated the stellar content of molecular clouds in the inner Galaxy from the far-IR and Lyman et luminosities. The mass of stars (M_s) derived in this way represents only the stars currently present in the molecular cloud and for the total mass of stars it must be multiplied by a factor R'> 1 in order to account for stellar production during the whole of the cloud lifetime. There is no way to reliably estimate R' for each cloud, but using the ratio of M_s to the cloud mass (M_c) may give some indication of the trend of $\eta \notin m_{S}/M_{\odot}$) with R_☉.

-200-

Following Myers et al. (1986), deriving $M_{\mathbb{C}}$ from the CO data of Dame et al. (1986) using $\bigotimes_{20} = 4$, the quantity $M_{\mathbb{G}}/M_{\mathbb{C}}$ is shown as a function of Galactocentric radius in Figure 6.2. There is considerable spread, but there is no evidence



Figure 6.2 : The ratio of the stellar mass in molecular clouds, estimated from far-infrared and Lyman luminosities, to the mass of the cloud as a function of Galactocentric radius. Data are taken from Myers et al. (1986) following their adoption of $\mathfrak{A}_{20} = 4$ for the estimate of M_c. The horizontal bars represent mean values.

for M_S/M_C (and hence η) decreasing with increasing R_C , which would be needed to explain Ψ_S if a U_{M_Z} variation is responsible for the M_{M_Z}/M_{M_Z} variation. In fact the trend in Figure 6.2 is in the opposite sense. It is interesting in this context to explore the effect of a metallicity correction to \mathfrak{A}_{20} . Using the metallicity measurements of Figure 2.5, the difference in mean metallicity between $R_C = 10$ kpc and $R_G = 6$ kpc is $\cong 0.4$ dex. If \mathfrak{A}_{20} were inversely proportional to metallicity (Bhat et al. 1985), this would decrease the estimated cloud mass and therefore increase $\log(M_S/M_G)$ by \cong 0.4dex at 6kpc compared to 10kpc, thus flattening out any gradient. It therefore appears a variation in T_{M_2} cannot account for the M_{M_2}/M_{MX} variation, and if a metallicity correction to \mathfrak{C}_{20} is used, T_{M_2} and the star formation efficiency (assuming that this is represented by M_S/M_G) are constant with R_G .

Returning to Figure 6.1, the variation of M_{M_2}/M_{NS} must then be due to a variation in \mathbb{T}_{hS} as a function of R_{G} , that is, the lifetime of the molecular phase is fairly constant, but the rate of conversion of $HI \longrightarrow H_2$ increases with decreasing R_G . This is consistent with the suggestion of Section 2.4 that in a region of more grains per cm⁻³ more HI evolves into H_2 . Assuming that the catalytic process for $H + H \longrightarrow H_2$ is completely efficient for all H atoms arriving at a grain surface, then the rate of formation of H_2 is

$$\frac{\mathrm{dn}(\mathrm{H}_{2})}{\mathrm{dt}} = \frac{1}{2} \mathrm{n}(\mathrm{H}) \mathrm{\pi} \mathrm{a}^{2} \mathrm{n}_{\mathrm{g}} \mathrm{V}_{\mathrm{M}}$$
(6.9)

for grains of radius a, number density n_{ij} and where V_{M} is the probable velocity of H atoms (Dyson and Williams 1980). Of course for a mass M_{NE} to be all converted to H_2 , $dn(H_2)/dt \ll (M_{NE}/T_{NE})$ and therefore

$$T_{\rm HIC} \stackrel{1}{=} (6.10)$$

If U_{M_2} , a and V_M are roughly constant with R_G , then the decrease in U_{MZ} from $R_G = 10$ kpc to 6 kpc required to give M_{M_2}/M_{MZ} for the radial distribution of M_{M_2} advocated in this work (Figure 6.1) is ≈ 2.5 . This is roughly the increase in metallicity over the same region, which is consistent with the grain density following the metallicity. (Section 2.5

-202-

and Appendix A.2.5 have argued that the dust-to-gas ratio is proportional to the metallicity, but the total gas is roughly constant over this region). The variation for SSS (1984) can also be explained by an increase in metallicity combined with an increase in total gas.

The situation concerning individual clouds will no doubt be more complex. Elmegreen and Elmegreen (1987b) suggest that superclouds of mass $\approx 10^7 M_{\odot}$ could form by gravitational instabilities in the ambient interstellar gas (Elmegreen 1987d) while molecular clouds inside form differently by agglomeration of smaller clouds. They ascribe the variation in molecular fraction per cloud as due to the variation in average pressure in the disc of the Galaxy, arguing that greater pressure implies greater density and therefore molecular line shielding being more important. However, it may be simply that the dust-to-gas ratio controls the molecular fraction, not only by enhancing evolution of $HI \longrightarrow H_2$ but also by increasing opacity. Franco and Cox (1986) have shown that any of the proposed mechanisms to agglomerate interstellar clouds will be effective in the molecularization of the ISM as long as they can generate column densities above

$$N_{\overline{1}} \simeq 5.10^{20} \left(\frac{Z_{\odot}}{Z}\right) \text{ cm}^{-2}$$
(6.11)

where the metallicity has been taken proportional to the dust-to-gas ratio. Thus for higher metallicities, it is easier to form molecular clouds, as dust grains provide the opacity necessary to prevent the external UV radiation flux from penetrating.

We tentatively suggest therefore that the observed variation in $M_{H_{2}}/M_{WZ}$ with R_{G} , both in disc annuli and

-203-

individual superclouds can be understood in terms of the variation in \mathbb{T}_{NT} due to the variation in metallicity and hence dust-to-gas ratio. It should be noted that this mechanism does not differentiate between the two different H_2 distributions until \mathbb{T}_{M_2} is fixed by other means. However, caution is needed in any application of equation (6.1) as \mathbb{T}_{NT} is a function of Galactocentric radius. Any derivation of M_{M_2} from this equation probably needs as much care with evaluating \mathbb{T}_{MT} as with \mathbb{T}_{M_2} .

Nevertheless, we take $\mathbb{C}_{WI} \cong 10^{\circ}$ y at 6kpc and now consider the evidence concerning an estimate of \mathbb{C}_{W_2} . 6.2.2 The formation and growth of GMCs

One of the main arguments for a lifetime of the molecular phase > 10° - 10° y came from initial estimates of the time needed to 'grow' GMCs by cloud-cloud collisions (Scoville and Hersh 1979; Kwan 1979). The time required to build up clouds of 10° - 10° M₀ simply by random cloud-cloud collisions of smaller clouds is 2.10° - 10° y implying of course, \overline{v}_{N_2} at least greater than these values.

However, these models neglected the tendency of spiral arm gravitational fields to enhance substantially the number density of clouds and thereby increase cloud-cloud collisions. Recent numerical simulations have shown that much shorter formation times are possible when a spiral potential is imposed (Tomisaka 1986; Combes and Gerin 1987). Kwan and Valdes (1983) found that the rate of growth of massive clouds near a potential minimum such as a spiral arm was enhanced by a factor 3-6 over the simple random case, large GMCs forming within 4.10^7 y. Mutual gravitational attraction between neighbouring clouds can also increase the

-204-

rate of coalescence. Kwan and Valdes (1987) modelled the local gravitational interaction between clouds closer than 400pc and found the rate of coalescence of larger clouds increased by a factor of 3. Large scale gravitational interactions have yet to be modelled, but once again it is thought that the rate of coalescence will be increased (Roberts and Steward 1987). Elmegreen (1987a) has also shown that growth times can be reduced to $\cong 6.10^{\forall}$ y, simply by taking account of the effect of magnetically enhanced collision cross-sections.

An allied argument concerns the cloud-cloud velocity dispersion of molecular clouds. Stark (1979, 1983, 1984) found that the velocity dispersion of clouds was roughly constant over the cloud mass range 10^2 - $10^{5\cdot 5}$ M_m. Equipartition of the random kinetic energy of the clouds, due to the clouds surviving several cloud-cloud collision times would predict the velocity dispersion proportional to $M_e^{\sim 1/2}$. Since equipartition is not observed, mechanisms other than cloud agglomeration must dominate the kinetic energy of the cloud ensemble. This may be taken as evidence that clouds do not form by agglomeration or that there exists some process which injects kinetic energy into larger clouds. Jog and Ostriker (1987) have suggested that the gravitational scattering of massive clouds off each other in the differentially rotating Galactic disc constitutes an effective gravitational viscosity which could be such a process. Clouds with mass $\geq 10^5 M_{\odot}$ do show a velocity dispersion consistent with equipartition (Stark 1983; Scoville et al. 1987). However, the relevance for cloud lifetimes is that although early calculations had suggested

-205-

that equipartition through cloud-cloud collisions could be achieved only in $\geq 10^{\circ}$ - 10° y, recent calculations indicate that equipartition can be achieved in $\geq 5.10^{\circ}$ y (Scoville et al. 1987; Jog and Ostriker 1987). In fact these recent calculations do not include the effects of gravitational focusing or the effect of spiral arms, both of which would shorten this timescale even further.

If the velocity dispersion of molecular clouds, (or clouds $<10^{\$} M_{\odot}$) can be taken as evidence for formation mechanisms other than cloud-cloud collisions, these formation mechanisms have even shorter timescales. The Parker instability (Mouschovias et al. 1974; Blitz and Shu 1980) where cloud formation proceeds due to magnetic instabilities in the gas layer has a growth time of $\approx 2.10^{7}$ y (Elmegreen 1982), while formation of clouds due to large scale gravitational instability also has a growth time of $\approx 2.10^{7}$ y (Cowie 1981; Elmegreen 1987d).

We therefore conclude that earlier values of $\mathbb{V}_{\mathbb{N}_2} \cong 10^{\mathbb{G}} - 10^{\mathbb{G}}$ y from a consideration of the growth of GMCs are no longer valid; there is still no general agreement on the mechanism of growth, but all current mechanisms favour growth times < $10^{\mathbb{G}}$ y.

6.2.3 Spiral arms

The question of whether GMCs are confined and formed in spiral arms has been used to argue for both long and short lifetimes, the argument being that if GMCs are confined to spiral arms, then their lifetime must be $< 4.10^7$ y otherwise they will diffuse into the inter-arm region (e.g. Turner 1984).

Consequently, there have been extensive searches for

-206-

spiral structure with conflicting results (Liszt 1985). Solomon and Sanders (1980) from the early CO surveys claimed no spiral pattern but these observations provided little information because of random and systematic motions (Blitz and Shu 1980). On the other hand, Cohen et al. (1980) claimed GMCs were only found in spiral arms, consistent with a short lifetime.

The picture at present is much more complicated, large-scale surveys of the Galactic plane showing two distinct cloud populations, (i) clouds with warm cores (> 10K) constituting ~ 25% of the total population and ~ 50% of the total emission, clearly associated with HII regions and Galactic spiral arms, and (ii) clouds with cold molecular cores (< 10K) constituting ~75% of the total number and ≈ 50% of the total emission, distributed uniformly throughout the disc (Sanders 1981; Solomon et al. 1985; Scoville et al. 1986, 1987). Recent observations of external spiral galaxies confirm this picture (Rydbeck et al. 1985; Lo et al. 1987b). It appears that the size spectrum of clouds is not greatly different between arm and interarm regions (Scoville et al. 1987) although there is clear evidence for clustering of clouds with warm cores into aggregates (Rivolo et al. 1986). The estimation of the mass of the two populations is complicated by the temperature effect, but virial theorem mass estimates by Scoville et al. (1987) essentially show no difference between the two populations (but see Section 1.2 for uncertainties in this analysis).

Early theories of cloud formation had suggested GMCs were formed in spiral arms from HI and launched from them in

-207-
ballistic orbits (Bash et al. 1977). Stark et al. (1987) have challenged this view that molecules form from interarm atomic gas in a spiral density wave. From a study of external spiral galaxies, there appears to be no obvious correlation between the surface brightness of CO and that the arm class, which suggests that density waves are not required to form molecules. A simple interpretation therefore is that molecular clouds form throughout the disc, and that spiral density waves merely organise the existing clouds (and other galactic material) into a spiral pattern. This is consistent with molecular clouds being found in the interarm regions, an interarm/arm contrast for HI (Elmegreen and Elmegreen 1987a) and the observation that the average SFR per unit area of a galaxy is relatively independent of the presence of an obvious wave mode (Elmegreen and Elmegreen 1986).

The clustering of clouds into aggregates could explain some results that molecular clouds are brighter in density wave spiral arms (Dame et al. 1986). It is a well known property of spiral galaxies that particle orbits have a tendency to linger in spiral arms in response to the gravitational attraction toward the local potential minimum in the arm. As a result individual cloud particles will often take as much time to move across spiral arms as in between. In addition, neighbouring particle trajectories tend to converge as they enter a spiral arm. This orbit crowding phenomenon has been shown in the numerical simulations of Roberts and Steward (1987) to allow larger clouds to assemble from smaller clouds and is the primary mechanism which underlies the organisation of GMC complexes

-208-

along spiral arms. These complexes are loosely associated and are dissociated upon exciting the arm through the postarm divergence in the system's velocity field. Roberts and Steward (1987) point out on the basis of their results, determination of individual cloud lifetimes is difficult, as the individual clouds in their computed models live forever and yet they show strong concentrations in spiral arms. This counters earlier arguments that molecular clouds must be long lived because of the lack of spiral structure as well as the counter arguments that molecular clouds are short lived because they are observed primarily in spiral arms.

However, if the scenario is that clouds form throughout the disc and are built up into the largest complexes by orbit crowding in spiral arms, what does this mean for the lifetime of the molecular phase? Kwan and Valdes (1987) in detailed numerical simulations have studied this process and modelled the fragmentation of clouds. In order to fit the observed mass spectrum of clouds they find $\mathbb{G}_{N_2} \leq 2.10^6$ y, a value also found by Turner (1984) in his similar examination of the SFR in the Galaxy. This upper limit is a little higher than the $\mathbb{G}_{N_2} \simeq 10^6$ y required for the M_{N_2} in the inner Galaxy advocated by this work, but well below the $\mathbb{G}_{N_1} \simeq 5.10^6$ y required by the M_{N_2} of SSS (1984).

A number of questions are still unclear, for example the significance of warm cloud cores. In the numerical simulations, giant cloud complexes are clearly confined to spiral arms but when compared to the warm core cloud population of Solomon et al. (1985) appear to have a different distribution along the arm, suggesting that the

-209-

warm cores may not trace out the spiral arms uniformly. Nevertheless, once again we see that early arguments for long lifetime GMCs from the spiral arm distribution (e.g. Solomon and Sanders 1980) are not well founded, the evidence favouring $\mathbb{G}_{\mathbb{W}_2} \leq 2.10^{\mathbb{G}}$ y.

6.2.4 Other considerations

There are a number of additional arguments to the ones previously outlined which suggest short lifetimes, $\leq 5.10^7$ y, certainly for GMCs and possibly \mathbb{T}_{M_2} . Many are well known, but here we briefly review them for completeness.

The gravitational binding energy of a cloud of mass $\sim 2.10^{\$} M_{\odot}$, and diameter $\approx 40 \text{pc}$ is $\approx 10^{\$\odot} \text{ ergs}$. Blitz and Shu (1980) showed that $\cong 10^{\$\circ}$ ergs was available from OB star formation inside the cloud (assuming a small efficiency of $\approx 10^{-\$} - 10^{-\$}$ in converting radiant energy into expansion) over 10^{7} y to disrupt the parent cloud. Even if OB star formation is slow in beginning after GMC formation, clouds would still be disrupted in a few x 10^{7} y (Turner 1984). For clouds with mass $< 10^{4\flat} M_{\odot}$, Mazurek (1980) calculated a disruption time-scale of $\simeq 4.10^{6}$ y. Therefore it seems very unlikely that GMCs where stars are forming can exist $\gtrsim 5.10^{7}$ y.

It is now well established that GMCs appear to be loose associations of smaller clouds or clumps (e.g. Blitz and Stark 1986; Blitz 1987). Random collisions between these clumps, at rates determined by their observed random motions, will ensure that they must coalesce into a centrally condensed cloud structure in $\approx 10^7$ y (Blitz and Shu 1980). Elmegreen (1985) has re-examined this point and added into the calculation the effect of the magnetic fields

-210-

of the clumps to obtain a timescale of $\leq 2.10^{7}$ y. The observation that such centrally condensed structures are not seen in general, with the one exception of the poph cloud, once again suggests that GMCs have lifetimes of a few 10^{7} y.

An additional problem for longer lifetime GMCs $(\geq 2.10^{\%}\text{y})$ is a suitable mechanism to support these clouds against collapse. The problem for theoreticians has always been to prolong GMC lifetimes for much longer than the free fall timescale ($\approx 10^{6}\text{y}$). It has been suggested that this could be done by magnetic fields (Mouschovias 1976a,b), rotational support (Field 1978), the winds of T-Tauri stars (Norman and Silk 1980) or supersonic turbulence within the clouds (Scalo and Pumphrey 1982), but each of these mechanisms can be shown not to have enough energy to support clouds for longer than $\cong 4.10^{6}$ - 1.10^{7} y (Blitz and Shu 1980; Turner 1984; Elmegreen 1985). Even when global turbulence originating from galactic rotation provides a fresh input of energy (Henriksen and Turner 1984), GMCs can only be supported for a few x 10^{7} y before forming stars.

Finally, theoretical arguments and laboratory data support the suggestion that CO and long chain molecules can stick easily to grain surfaces at the low temperatures prevailing in GMCs (Allen and Robinson 1977; Leger 1983). If GMCs are long-lived in comparison with the time scale for absorption onto grain surfaces (< 10^7 y), it is surprising that larger depletions of such molecules are not seen. In order to have much longer GMC lifetimes, efficient mechanisms must be found to return molecules back to the gas phase, which may be difficult (Williams 1985; van Dishoeck and Black 1987).

-211-

In summary, therefore, present observational and theoretical evidence from the formation of GMCs, spiral arm structure and other considerations favour $\mathbb{T}_{GMCS} \leq 5.10^{7}$ y with the lifetime of the molecular phase $\mathbb{T}_{M_{2}} \leq 2.10^{6}$ y. From the continuity argument of Scoville and Hersh (1979), this is consistent with the H₂ distribution in the inner Galaxy proposed in Section 4.6 of this work, much lower than the H₂ distribution of SSS (1984). It is in the light of this result, that claims have been made from CO observations of the Virgo cluster of galaxies for a much longer lifetime of GMCs and the molecular phase. In the following we critically examine these claims.

6.3 THE LIFETIME OF THE MOLECULAR PHASE FROM GAS STRIPPING IN THE VIRGO GALAXIES

Recent CO surveys of galaxies in the Virgo cluster (Young 1985b; Kenney and Young 1985, 1986; Stark et al. 1986) have led to the claim that the lifetime of the molecular phase, \overline{U}_{M_2} , is $\geq 10^{\circ} - 10^{\circ}$ y. Specifically, Stark et al. (1986, hereafter referred to as S) and Kenney and Young (1986, hereafter referred to as KY) studied total galaxy contents of HI and CO as a function of angular distance from the centre of Virgo, assumed to be coincident with M87. The interesting results found were that the ratio of CO/HI increased towards the cluster centre, as did the ratio of the CO 'diameter' to that of HI. S found no evidence for differences in results for early-type galaxies (Sab and earlier) and late-type (later than Sab). These results were explained in terms of HI being removed

-212-

preferentially, compared with H_2 , by ram pressure due to the intercluster medium (ICM). The argument then proceeds that if HI is removed from the inner disc, the molecular gas must survive for at least the cluster crossing time $(3.10^{\circ} - 10^{\circ})$ y, since conversion from H_2 —>HI would have led to its disappearance, in view of HI being swept away rather readily.

This argument for long molecular lifetimes is crucially dependent on the nature of gas stripping and on the assumption that \mathfrak{A}_{29} is not affected by the cluster environment. In this Section, we examine gas stripping and in Section 6.4 the value of \mathfrak{A}_{20} in the Virgo spirals. 6.3.1 The HI deficiency

It was suggested some time ago (Davies and Lewis 1973) that galaxies in the Virgo cluster are deficient in total HI mass with respect to field galaxies of the same morphological type. Although there have been some doubts (e.g. Tully and Shaya 1984) a majority of authors now arrive at the same conclusion that a significant number of Virgo spirals are depleted of HI (e.g. Chamaraux et al. 1980; Haynes et al. 1984; Haynes 1985; Guiderdoni and Rocca-Volmerange 1985; Huchtmeier 1985; Warmels 1986). Haynes and Giovanelli (1986) found that for galaxies less than 2.5° from the centre of the cluster, 75% of the observed galaxies are HI deficient. Between 2.5° and 5° the deficiency is still observed, but the fraction falls, disappearing at $\simeq 5^{\circ}$. This deficiency has been attributed to galaxy formation in the cluster environment, or the stripping of the gas by tidal encounters or interaction with the ICM. HI deficiency has been shown to be a phenomenon in

-213-

some other clusters such as A262, A2147 and Coma but is by no means universal (Bothun et al. 1982; Giovanelli and Haynes 1985), the gas removal mechanism being efficient in some environments but not in others. The correlation of HI deficiency and cluster X-ray luminosity, rather than local galaxy density, suggests the predominance of galaxy-ICM interaction as the cause (Giovanelli and Haynes 1983; Haynes and Giovanelli 1986).

In addition, it has also been shown that HI discs of galaxies within 5^{0} of the centre of the Virgo cluster are smaller compared to isolated galaxies of the same optical size (e.g. van Gorkom and Kotanyi 1985; Warmels 1985; Haynes and Giovanelli 1986). Giovanardi et al. (1983) and Giovanelli and Haynes (1983) further claimed that the gas was depleted in such a way that the surface density remained roughly constant. Using the data of Giovanelli and Haynes (1983), Figure 6.3 illustrates this, plotting the 'surface density' of HI (M_Mg/D_Mg) for galaxies Sab and later as a function of the angular distance (③) from M87. We have also plotted the isolated galaxy sample of Hewitt et al. (1983). The surface density appears to be roughly constant, although the statistics are poor at the smallest values of Θ , where for $\Theta < 2^{\circ}$ there might be a small reduction. A similar analysis has been carried out for the independent sample of Helou et al. (1984) and the trend confirmed. This sample is not strictly directly comparable in absolute surface densities to Giovanelli and Haynes (1983) because of a different definition of D_{MX} . This result strongly indicates that if HI is removed, it is taken from the outer parts of galactic discs, thereby decreasing

-214-



Figure 6.3 : The surface density of HI $(M_{MZ} / (D_{KZ})^3)$ in spiral galaxies (Sab and later) as a function of angular distance form M87 (©). The data of Giovanelli and Haynes (1983) and Hewitt et al. (1983) define D_{MR} as twice the distance from the galactic centre within which the best fit model contains 70% of its total HI mass. Helou et al. (1984) define D_{MX} as twice the distance from the galactic centre where the HI flux falls to 1/e of the peak value. These measures are strictly not directly comparable, however for galaxies in common, <D_{WX} (Giovanelli and Haynes) $/D_{MR}$ (Helou et al.) > = 1.09 with a standard deviation of 0.14. Common galaxies are linked by vertical bars. Horizontal lines represent median values. For comparison, the isolated galaxy sample of Hewitt et al. (1983) is shown with an assumed distance to Virgo of 15Mpc.

both $M_{M\bar{l}}$ and $D_{M\bar{l}}$.

Of possible gas removal mechanisms, galaxy-galaxy interactions (Spitzer and Baade 1951; Icke 1985) can be shown to be not frequent enough in Virgo (Giovanelli and Haynes 1983) while galactic winds (Matthews and Baker 1971) preferentially deplete the central galactic regions so increasing D_{WX} rather than the observed decrease. Although the details are complex, two mechanisms, ram-pressure stripping (Gunn and Gott 1972) and thermal evaporation (Cowie and Songaila 1977) both predict stripping of the outer regions and therefore a decrease in D_{WX} . As both S and KY have suggested ram-pressure stripping in order to explain their results, the next section examines the likelihood of this mechanism in the Virgo spirals. 6.3.2 The likelihood of ram-pressure stripping

1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 - 1999 -

Since the original suggestion of Gunn and Gott (1972), ram-pressure stripping has been studied theoretically by a number of workers (Gisler 1976; Lea and de Young 1976; Kent 1980; Farouki and Shapiro 1980; Shaviv and Salpeter 1982). In its simplest form it predicts that galactic gas will be removed if its density is below n_{ISM} where

$$n_{1SM} \simeq \left(\frac{V_{g}}{V_{eSC}}\right)^{2} \cdot n_{1CM}$$
 (6.12)

 V_{g} being the velocity of the galaxy with respect to the ICM (the component perpendicular to the galactic plane), V_{CSC} , the escape velocity from the galaxy at the point in question and n_{ICM} the density of the intercluster medium.

A number of observations have suggested its occurrence in galaxies in the Virgo cluster. As described above, the existence of a radial deficiency in HI despite damping projection effects coupled with smaller HI discs can be explained in principle by ram-pressure stripping. In the core of the cluster there appears to be a marginal correlation of HI deficiency with velocity (Giovanelli and Haynes 1983) although a larger sample (Haynes and Giovanelli

-216-

ي المريز الديا ما ما أو يرأ

1986) shows no significant correlation. Deficient spirals seem to be on radial orbits (Dressler 1986; Giraud 1986) while the non-deficient are on more circular orbits and therefore avoid the high density ICM. Warmels (1985) found asymmetric HI distributions in the Virgo core which are expected for ram-pressure stripping. However, Haynes and Giovanelli (1986) found that dwarf galaxies which have a lower restoring force were not more HI deficient than spirals contrary to expectation.

Stark et al. (1986) evaluated expression (6.2) to obtain $n_{1SM} = 0.6 \text{ cm}^{-3}$ and concluded that a major part of the ISM was stripped. However, this is due to what appears to be an overestimate of both n_{ICM} and V_{\odot} . The value used for the ICM, $n_{ICM} = 3.10^{-2} \text{ cm}^{-3}$ (Stark 1987b) is even higher than the density reasonable models would predict right at the centre of Virgo. Taking the model of Fabricant and Gorenstein (1983) for the X-ray surface brightness distribution about M87, n_{ICM} may be evaluated as

$$n_{ICM} (\emptyset) = (n_{ICM})_{\emptyset} \left[1 + \left(\frac{\emptyset}{a} \right)^2 \right]^{-\emptyset \cdot \emptyset \cdot \emptyset}$$
(6.13)

where the core radius a = 1.6'. At $\bigotimes \simeq 2^{\circ}$, $n_{IGM} \simeq 10^{-10}$ cm⁻³ (Gorenstein et al. 1977; Forman et al. 1979) which gives $(n_{IGM})_{\odot} \simeq 2.7.10^{-2}$ cm⁻³. A more realistic average value for $\bigotimes < 2^{\circ}$ therefore would be at least an order of magnitude lower than that quoted by S. Secondly, S use $V_{\odot} = 1800$ kms⁻¹ which is again too large for Virgo. The usual velocity dispersion is quoted to be 720-800 kms⁻¹ (Ftaclas et al. 1984; Huchra 1985) which gives $V_{\odot} \simeq 1300$ kms⁻¹ (i.e. $\sqrt{3}$ x 760).

Using $V_{QSC} \simeq 400 \text{kms}^{-1}$ (for a typical galactic position equivalent to that of the Sun in our own Galaxy),

-217-

 $V_{\text{g}} \geq 1300 \text{ kms}^{\circ}$ and $n_{\text{IGM}} \approx 3.10^{\circ} \text{ cm}^{\circ}$ in equation (6.12) gives

 $n_{\rm NSM} \simeq 0.032 \ {\rm cm}^{-9} \simeq 7.7 \ 10^{-6} \ {\rm M_{\odot}pc}^{-9}$ (6.14) Although Bosma (1981a) has argued that $V_{\rm CSC}$ may be an upper estimate as the total mass density of the Galaxy may be higher than the total mass density found at comparable radii in many other galaxies, this effect will be less than a factor of 2.

Therefore the fact that it is only densities $\leq 10^{-2}$ cm⁻³ that are susceptible to ram-pressure loss indicates that it is only the extreme outer parts of galaxies in the sense of large galactocentric radius that are at risk. In order to illustrate this, we have taken the CO and HI radial profiles (Young 1985b; Warmels and van Woerden 1985) for the spiral galaxy NGC4254, and calculated the total gas as a function of galactocentric radius. If it were within 1° of the centre of the cluster, equations (6.12) and (6.13) imply that only gas with radius > 15 kpc will be swept by ram-pressure stripping. (As the CO falls off exponentially, even a high value of $\mathfrak{A}_{20} = 8$ for the conversion of $CO \longrightarrow H_2$ does not affect this result). The inner regions of galaxies, where the value of V_{QSC} will be higher by \approx 2 and therefore n₁₅₆₁ still lower by a factor 3-4, should be safe. It is this fact which has important consequences for the understanding of the CO results.

6.3.3 Comparison of the CO and HI results

The detailed time evolution of HI and H_2 and their inter-relation is quite a complex process (Chapter 2 and Section 6.2) and there is thus an attraction in studying the Virgo galaxies, if due to gas stripping, the HI-->H₂ process

-218-

has been turned off. As S note, if this is correct and there is no environmental effect on the CO luminosity, then the process $H_2 \longrightarrow HI$ can be studied in near-isolation.

S and KY argue that, in contrast to the HI deficiency, the CO contents are roughly normal. Further, Young (1985b) presents CO profiles as a function of galactocentric radius which are similar to the profiles of nearby spirals in that the bulk of the CO is found within 10 kpc of the galactic centre following a roughly exponential distribution.

Figure 6.4 reproduces from KY the ratio of the CO disc



Figure 6.4 : The ratio of angular diameter for CO (D_{CO}) to that for HI (D_{NE}) for spiral galaxies in Virgo versus angular distance to the centre of the cluster, reproduced from Kenney and Young (1986). The horizontal lines represent median values. Squares denote types Sbc-Sc, circles denote Sab-Sb. D_{NE} and D_{CO} are the diameters at which HI falls to a surface density of 5.10^{10} atoms cm⁻², and H₂ (using $\alpha_{20} = 8$) falls to 4.10^{20} molecules cm⁻².

to the HI disc as a function of @. The increase in this quantity may be explained as D_{CO} being constant while D_{MI} falls closer to the cluster centre. This is consistent with ram-pressure stripping removing HI from the outer disc, with

-219-

the CO, which is concentrated in the inner parts, being left alone. It is to be noted that if this explanation is correct, nothing can be inferred concerning the $H_2 \longrightarrow HI$ process and H_2 lifetimes as both HI and H_2 are unaffected in the inner parts of a galaxy. Only if HI is removed from the whole of the disc, as KY point out, do the arguments for a long lifetime of H_2 hold.

In a more detailed examination of this problem, the ratio of mean surface density of H_2 (CO) to that of HI has been derived by the present author as a function of @ and this is given in Figure 6.5. If HI were being swept out



<u>Figure 6.5</u>: Ratio of surface densities of ' H_2 ' and HI against angular distance from the centre of the Virgo cluster. The data are taken from Kenney and Young (1986) and M_{M_2} is calculated using their constant value of $\alpha_{20} = 8$. The horizontal lines represent median values.

rather uniformly in the inner parts of the galaxies while the GMCs were left unaffected due to their higher density (Kent 1980; Kritsuk 1983) this ratio would be expected to rise with decreasing $^{\odot}$. In fact, the ratio is roughly constant, and may even fall somewhat below $^{\odot}$ < 2 $^{^{\odot}}$. A possible reason for this is advanced in the next section.

Therefore, the arguments favouring lifetimes of the molecular phase larger than $(3-10) \cdot 10^{3}$ y are not well founded. In deriving these long lifetimes it is also assumed that there is no environmental effect on α_{20} . The next section shows that this assumption may also be questionable.

6.4 THE CO-++2 CONVERSION FACTOR IN THE VIRGO GALAXIES

6.4.1 The effects of the inter-cluster wind on GMC

Implicit in the arguments advanced elsewhere, and by this Chapter so far, is the assumption that the intercluster wind has no effect on the GMC. Whilst it is true that ram pressure has probably a negligible effect on the tightly bound clouds (Kent 1980; Kritsuk 1983) its effect on the dust grains, and the thermal properties of the clouds may be quite considerable. As the ¹²CO lines are optically thick, only effects in the outer layers of the GMC would be enough to change the CO/H₂ ratio. Two effects are likely, partial evaporation of grains returning more CO into the gas phase, and an increase in gas temperature which would affect α_{20} , as $\alpha'_{20} \ll T_{\rm K}^{-1.3}$ (Kutner and Leung 1985).

It has not proved possible to evaluate the magnitude of these effects, detailed modelling of GMCs (e.g. van Dishoeck and Black 1986) being outside the scope of this work. However, it is apparent that both effects result in an

-221-

apparent increase in the CO signal for the same amount of H_2 . If true, Fig. 6.5 would indicate an even greater loss of H_2 at @ < 4[©]. A consequence would be that the formation of GMC was inhibited by the ICM as an increased loss of H_2 by way of short-life clouds which cycle to HI which is then lost, cannot be invoked, without also losing the HI in the denominator of the ordinate of Fig. 6.5.

Whatever the details it appears that one should expect a lower a_{20} for many galaxies in the Virgo cluster than for isolated galaxies. Both optical and IR information are now used to examine this effect.

6.4.2 The star formation rate

Direct measurements of the SFR from H& fluxes show that for a given morphological type, the SFR in late type spirals within 6^{\oplus} of the centre of Virgo is less than in equivalent field galaxies by a factor of approximately 2 (Kennicutt 1983a, 1985).

This seems to be supported by the lack of high surface brightness disc galaxies (Phillips and Disney 1986) and smaller stellar discs (Peterson et al. 1979) again within 6[°] of the Virgo core. The observation that nuclear Het emission (Stauffer 1983) and nuclear IR emission (Devereux et al. 1987) do not appear to be significantly depressed in spirals in the same region seems to indicate once again that the cluster environment has preferentially affected the outer regions of the galaxies. The deficiency of HI is correlated with UV colours (Giovanelli and Haynes 1983), galaxies being fainter than those with normal HI contents.

The colours of spirals within 6^{\oplus} of the core are redder than those in the field but are normal relative to field

-222-

galaxies of the same HI content (Kennicutt 1983a; Stauffer 1983). Kennicutt (1983a) has used this to argue against recent gas stripping as this would make galaxies anomalously blue relative to their HI content. However, in understanding the SFR, it is the H₂ rather than HI that is important, the SFR ($\frac{1}{12}$) being well correlated with the mass of H₂ (Chapter 2). Using the recent CO data, a comparison of these two quantities is much more useful in giving information about the Virgo environment.

CO data have been compiled from KY and Knapp et al. (1987) and M_{W_2} has been derived using a datum value of $\mathfrak{A}_{20} = 8$ (following KY) in the first instance. Following Kennicutt (1983b), the Het data of Kennicutt and Kent (1983) have been used to derive W_3 . This process involves estimating extinction and the use of an IMF and the result is uncertain to at least 50%. Figure 6.6 shows the ratio W_3/M_{W_2} as a function of angular distance from M87 for galaxies Sab and later.

Turner (1984) has shown that under some circumstances a GMC may break up into smaller molecular clouds which may once again be re-cycled through galactic spiral density waves on timescales $\simeq 2.10^{\circ}$ y to form more GMCs. Having delayed star formation in this way he obtains a theoretical ratio $\sqrt[4]{8}/M_{W_2} \simeq 5.10^{\circ}$ y⁻¹. It can be seen that all of the Virgo spirals are lower than this, on average by a factor of 3.5. This is consistent of course, with the claim that the SFR is lower (Kennicutt 1983a) while the H₂ content is approximately normal (Young 1985; K; S).

It has been argued that for some spirals the SFRs estimated from Het could be underestimated, the SFRs only

-223-



Figure 6.6 : The ratio of the SFR to the mass of H_2 (calculated using $\alpha_{20} = 8$) as a function of angular distance from M87 for galaxies Sab and later. The SFR has been determined in two independent ways as outlined in the text. Also shown for comparison are the IRAS bright galaxies and the isolated galaxies of Young et al. (1986a,b) and the theoretical ratio for W_8/M_{H_2} of Turner (1984).

being correct for blue galaxies with large equivalent widths (Lonsdale-Persson and Helou 1987). Unfortunately, very few of the isolated galaxies in the H& data of Kennicutt and Kent (1983) have CO measurements, so it is at present impossible to see whether this may be the reason for the above result.

However, an independent check of this can be made using

IRAS data for the Virgo galaxies. Knapp et al. (1987) give the 60µm and 100µm fluxes taking into account extended emission. Following equations (2.12) and (2.13) of Section 2.5.1, an estimate can be made of the SFR of 0, B and A stars from the IR and blue luminosity (derived from RC2). The distance to Virgo has been taken to be 15Mpc. SFRs derived in this way are also shown in Figure 6.6, the same galaxies being joined by a vertical bar to their H¢4 determination. The trend suggested by the H¢4 determination is confirmed, with some scatter as expected.

It may be argued that the uncertainties in deriving in these ways make direct comparison with the value of Turner (1984) meaningless. Equation (2.12) obviously does not account for either the formation of low mass stars or the re-cycling of gas in the ISM, although to some extent these effects may balance. In addition, although IR emission is often associated with star formation (e.g. de Jong et al. 1984; Telesco et al. 1986) the situation may be quite complex. The IR emission may be only partially associated with star formation, a second contribution associated with the interstellar radiation field heating of diffuse material, and this fraction may vary considerably amongst galaxies (e.g. Pajot et al. 1985; Cox et al. 1986; Sekiguchi 1987; Lonsdale-Persson and Helou 1987; Walterbos and Schwering 1987; Wainscoat et al. 1987). However, it is encouraging that in Figure 6.6 the Hox and IR determinations show good agreement.

Furthermore, using exactly the same method as described above for the IR determination, we have also derived $\frac{M_{M_{2}}}{M_{M_{2}}}$ for 27 IRAS bright galaxies of Young et al. (1986a) and 13

-225-

galaxies taken from the Karachentseva Catalogue of Isolated Galaxies (1973) presented by Young et al. (1986b). The average values of each of these samples are also plotted in Figure 6.6.

If the SFR can reasonably be derived from equations (2.12) and (2.13), and if Turner's (1984) model of star formation is viable, this immediately suggests that the use of $\mathfrak{A}_{20} = 8$ for even the isolated galaxies is an overestimate by a factor of approximately two. If $\mathfrak{A}_{20} \approx 4$ were used then the IRAS bright galaxies would have a higher efficiency of star formation, a result to be expected as many are interacting systems (Kennicutt et al. 1987), and the isolated galaxy sample would be consistent with the model of Turner (1984). Due to the uncertainties involved, this may not be true, but what does seem clear is that the galaxies within 5° of the centre of the cluster have on average a much lower $\mathfrak{A}_{\mathrm{S}}/\mathrm{M}_{\mathrm{M}_{2}}$ compared to isolated galaxies of the same type.

This last result indicates either of two things. If the H₂ contents are 'normal', that is, there is no environmental effect on the conversion factor α_{20} , then the efficiency of star formation is reduced in the Virgo galaxies. The other possibility, as suggested in Section 6.4.1 above, is that α'_{20} decreases because of the ICM and so a low H₂ mass appears to be 'normal', i.e. the use of the same α'_{20} for the Virgo and isolated spirals overestimates M_{H₂} and so decreases ψ_{S}/M_{H_2} .

Concerning the first possibility, most of the theoretical studies suggest an enhancement in $\frac{1}{23}$ rather than a lower efficiency (e.g. Dressler and Gunn 1983). Bothun

-226-

and Dressler (1987) argue that the ram-pressure phenomenon, as well as stripping HI, could enhance star formation. Using Jura's (1976) relation involving the minimum mass required for cloud collapse as a function of external pressure, it appears that this mass is $20-30M_{\odot}$ for rich clusters like Coma, which is 10-100 times less than the minimum collapse mass induced by the passage of Galactic density waves or supernovae shocks (Shu et al. 1972; Turner 1984). As the external pressure increases with decreasing 9, one should see a greater enhancement closer to the core. Therefore, in clusters with high ICM densities, ram pressure induced star formation is a real possibility. Of course, Virgo is not such a rich cluster as Coma, n_{iCM} being lower by a factor of ~ 3 (Shanks 1987). Repeating the same calculation for Virgo, we obtain minimum masses of 35-50Mo which shows that even in Virgo, induced star formation is still a possibility.

In reality, of course, the situation is likely to be much more complicated and difficult to model. If ram pressure is very effective in removing the HI medium, replacing it with a hot, but thin ICM, then cold GMC may rapidly evaporate. At present the theory of thermal evaporation (e.g. Balbus and McKee 1982) does not adequately describe the effects on cold dense clouds with a hot ICM, although Section 6.3.3. argued that ram pressure was not able to remove HI in the regions of GMC. Kent (1980) speculates that the molecular clouds could even experience a drag force which causes them to collect in the centre of the Galaxy. He also shows that if a hydrodynamical code is used to model the interaction, a bow shock forms at the interface

-227-

of the galaxy with the ICM which facilitates the flow of material around rather than through the disc. This may even inhibit the excitation of cloud collapse, but it is unlikely that this would reduce the efficiency of star formation as compared to isolated galaxies.

Apart from the above uncertainties, the general view seems to be that the ICM should increase the efficiency of star formation rather than reduce it. Compression of gas in the inner portion of the galaxy may induce star formation while the outer disc is stripped of HI (Haynes and Giovanelli 1986). Therefore, the reason that $\frac{44}{15}$ is low in the Virgo galaxies is because the H₂ content is low. To explain this low value, then within $\$ = 5^{\degree}$,

 α_{20} (Virgo) $\simeq 0.5 \alpha_{20}$ (isolated galaxies) (6.15)

In fact, if the ICM does increase the efficiency of star formation with decreasing @, $@_{29}$ (Virgo) would be even lower. More data on CO contents and SFRs as a function of @, as well as a greater theoretical understanding of the efficiency of star formation, could even lead to an evaluation of $@_{20}$ as a function of @. We would expect $@_{20}$ to be lower for smaller @, but the data at present do not allow such an analysis. In particular, more CO observations are needed for galaxies in the range $@ = 6 -12^{\circ}$. The next section, further examines the IR evidence for a low $@_{20}$. 6.4.3 IR results

Figure 6.7 shows the ratio of the total FIR luminosity $(40 - 300 \mu m)$ as calculated above (equation 2.13) to M_{M_2} (with $\mathcal{C}_{20} = 8$), as a function of angular distance from M87. The mean of the isolated galaxy sample (Young et al. 1986b) is calculated in the same way. Rengarajan and Verma (1986)

-228-



Figure 6.7 : The ratio of total IR luminosity to mass of H_2 as a function of angular distance from the centre of Virgo. The isolated galaxy sample is that of Young et al. (1986b). M_{M_2} in both cases has been calculated using $\ll_{20} = 8$.

earlier had found $L_{XR}/M_{H_2} \approx 13$ (correcting their value to $cl_{20} = 8$) for a sample of interacting and isolated galaxies. It can be seen that galaxies with $\otimes < 5^{\circ}$ are on average well below the value for the isolated galaxies.

Young et al. (1986a) found that for IRAS bright galaxies L_{IR} depended roughly on the first power of the H₂ mass and a temperature defined from the ratio of the 60µm and 100µm fluxes. Tacconi and Young (1987) from CO surveys of dwarf irregular galaxies extend this relationship over three more orders of magnitude. Assuming a λ^{-1} emissivity law, the 'dust temperature' as defined by Young et al. (1986a,b) has been derived for the Virgo galaxies using the 60/100µm flux ratios of Knapp et al. (1987) and the IRAS Extragalactic Catalogue (Lonsdale et al. 1985). Although the physical interpretation of T for a galaxy is far from clear (e.g. Thronson et al. 1987b), there appears to be no

-229-

real difference between the Virgo sample and the isolated galaxy sample (Figure 6.8a). A similar result for a much larger



<u>Figure 6.8</u> : (a) Temperature derived from the ratio of the 60 μ m and 100 μ m fluxes assuming a λ^{-1} emissivity law, as a function of angular distance from the centre of Virgo, compared to the isolated galaxy sample of Young et al. (1986b). (b) Ratio of 'warm' dust (calculated from the 100 μ m flux) to total gas as a function of angular distance from the centre of Virgo, compared to the isolated galaxy sample of Young et al. (1986b). M_{H2} is calculated from the CO data using $\mathfrak{A}_{20} = 8$.

sample of Virgo spirals within 6[®] of the core has also been found by Leggett et al. (1987a). In addition, Bicay and Giovanelli (1987) also found similar dust temperatures to the ones calculated above in galaxies in seven other rich clusters.

Therefore, in the absence of any temperature difference, if $L_{\overline{XR}}$ is correlated with M_{M_2} , Figure 6.7 indicates once again that particularly within $@ = 5^{\circ}$, M_{M_2}

has been overestimated. There have been a number of recent claims that the IR properties of the Virgo galaxies are 'normal' or indistinguishable from field galaxies (e.g. de Jong 1985; Devereux et al. 1987; Leggett et al. 1987a), which would appear to contradict the above picture. Leggett et al. (1987b) have even used this and the apparent 'normal' H_{2} contents (KY;S) to argue that the far IR flux in disc galaxies is dominated by emission from dust associated with molecular gas. However, it is important to be clear about what is meant by the IR properties of Virgo galaxies being normal. For example, Devereux et al. (1987) compare the average $L_{\hat{X}\hat{R}}$ of galaxies with $\otimes < 9^{\circ}$ with field galaxies and find no real difference. As any effect on \$20 is likely to occur for $@ < 5^{\circ}$, any gross average over 9° will smooth out any differences. Leggett et al. (1987a,b) have considered galaxies with $\otimes < 6^{\circ}$. They find that (S_{100} / S_{CO}) and (L_{IR} / L_{\odot}) are similar to field galaxies, and that relationships between these two quantities and L_{XR} are similar to those found in the field galaxy sample of Rieke and Lebofsky (1986). However, the spread in absolute values of L_{RR} is so large that any comparison with the absolute value of $L_{\mathbb{KR}}$ for field galaxies is difficult. Young et al. (1986b) had in fact noticed in their study of isolated and interacting galaxies that the ranges in L_{IR} for the two galaxy samples had sufficient overlap that selection on L_{XR} alone would not distinguish between the samples. The distinction between the two samples was shown clearly by $L_{\rm IR}/M_{\rm M_{2}}$, the interacting galaxies having a mean value 6 times that of the isolated sample. The important indicator for comparisons is therefore not a gross average of L_{IR} , but the quantity

-231-

 L_{ER}/M_{M_2} . For galaxies within $@ \le 5^{\circ}$, this quantity is found to be lower than that for isolated galaxies. We do not expect any theoretical reason for a lower L_{RR} as compared to M_{M_2} for $@ < 5^{\circ}$ (unlike the situation for interacting galaxies where a greater efficiency of star formation is expected), which leads once again to the conclusion that $@_{29}$ is lower for $@ < 5^{\circ}$ than for isolated galaxies.

In addition to the total luminosity, the IR flux densities can be used to estimate the mass of warm dust in each galaxy. Using equations (2.15) and (2.16) of Section 2.5.3, the 100 μ m fluxes and T from Figure 6.8a, Figure 6.8b shows the dust-to-gas ratio (M₀/M₀) as a function of @. HI and H₂ masses have been taken from KY and Knapp et al. (1987) using $@_{20} = 8$. There appears to be an enhancement in M₀/M₀ for the Virgo galaxies over the isolated galaxies but this is to be expected due to the well known HI deficiency (Section 6.3.1).

As was noted in Section 2.5.3, the 100 μ m flux only samples the warm dust and so underestimates the total amount of dust present. For four galaxies in the Virgo sample, NGC4254, NGC4321, NGC4501 and NGC4527, Chini et al. (1986) have obtained 1300 μ m fluxes. This wavelength samples dust of all temperatures and so a total dust-to-gas ratio can be computed, but only for the central 90" of the galaxies sampled by the 1300 μ m beam. Young (1985b) shows that the CO distribution in the Virgo galaxies is roughly exponential, so following Section 2.5.3 assuming that the HI surface density is approximately constant with galactocentric radius while the CO follows an exponential distribution with scale length 5kpc (Young et al. 1985) allows the total $M_{\rm D}/M_{\rm C}$ in

-232-

the central 90" to be calculated. Using the 1300_{μ} m flux and temperature T in equation (2.16) gives a range for the four galaxies as

$$\frac{M_{B}}{M_{G}} \approx (2-6) \cdot 10^{-3}$$
 (6.16)

compared to the canonical value for the solar neighbourhood of $10.10^{\circ 3}$. This result at face value is consistent with the idea that H₂ has been overestimated especially noting the fact of the HI deficiency.

However, the major uncertainty in the above analysis is the use of a temperature derived simply from the 60/100µm flux. Chini et al. (1986) argue that their observations cannot be explained by dust emitting at a uniform temperature and adopt a warm and cold component model with corresponding temperatures T_W and T_c . The use of T_c as given by Chini et al. (1986) increases M_D by a factor 2-3 over M₀ calculated using T. This marginally brings M₀/M₀ to the solar neighbourhood value, but it is difficult to increase it any more, except of course by reducing $M_{\mbox{\scriptsize g}}$. In fact, if the dust-to-gas ratio follows the metallicity of a region (Appendix A.2.5) as the metallicity increases with decreasing galactocentric radius, M_{\odot}/M_{\odot} should be > 10.10⁻³ in the central regions of a galaxy. Therefore, although there are large uncertainties, the IR data seem to indicate a lower α_{20} in galaxies within 5° of the Virgo core than in the solar neighbourhood and isolated galaxies.

Recently, Knapp et al. (1987), by comparing the CO and HI fluxes and 100 μ m fluxes for galaxies with $@<12^{\circ}$, and assuming a constant dust-to-gas ratio concluded that $@_{20} = 12.6\pm7.0$ in the Virgo spirals. However, as pointed out above, the 100 μ m flux is not a good indicator of the

-233-

total dust content and variations in the dust-to-gas ratio could greatly affect this result.

It could be argued that the low dust-to-gas ratio in Virgo was due not to an overestimate of α_{20} but that the Virgo galaxies may have an intrinsically small dust-to-gas ratio. Although Section 6.4.1 suggested that evaporation of grains in the outer envelopes of GMC could be caused by the ICM; this effect is not expected to affect the whole of the galactic dust content (Dyson 1987). As Knapp et al. (1987) point out, it will be observations at longer wavelengths that will constrain α_{20} from the IR data.

6.4.4 The evolution of galaxies in the Virgo cluster

It has been argued in this Section, that although the CO observations are normal relative to galaxies of the same morphological type (Young 1985b), α_{20} and therefore $M_{\rm M_2}$ are lower in galaxies within 5° of the Virgo core. This means that the Virgo spirals in that region are deficient in H_2 as well as HI.

Ram-pressure stripping could be responsible for the HI deficiency, but cannot be responsible for any H₂ deficiency as the H₂ is concentrated in the inner regions and is much denser (Kent, 1980; Kritsuk 1983). The possible variation in \mathfrak{A}_{20} with \mathfrak{B} however means that Figures 6.4 and 6.5 cannot be at present interpreted as evidence for a stripping mechanism of H₂. A consideration of thermal evaporation or turbulent viscous stripping (Cowie and Songaila 1977; Nulsen 1982) in relation to molecular clouds is beyond the scope of this work, although both processes could be involved.

Finally, there is always the possibility of a primordial origin for the gas deficiency. Phillips (1987)

-234-

has shown that HI deficient galaxies have a lower surface brightness which he interprets as galaxies having been deficient in HI for a significant fraction of their lifetimes and may favour a primordial deficiency. If the cluster environment affects the formation of galactic discs in some way so that HI is deficient (e.g. Dressler 1980), this in turn would lead to smaller H_2 masses and smaller integrated SFRs (Section 4.7). HI could also be subsequently stripped from the outer regions by ram-pressure stripping, the deficiency in H_2 in the inner regions being of primordial origin. This 'primordial' deficiency picture, however, may not be consistent in Virgo with the kinematic evidence that the cluster is relatively young, late type spirals only arriving in the core some \approx 6Gy ago (Tully and Shaya 1984; Guiderdoni 1987; Binggeli et al. 1987).

Whatever the mechanism for the gas deficiency, the result that $@_{20}$ is affected by the ICM environment is important in relation to the rest of this thesis. The standard assumption that the $@_{20}$ determined locally applies universally (e.g. Young and Scoville 1982a), appears to invalid for galaxies within 5° of the Virgo core. Of course, the low SFR, L_{XR}/M_{W_2} and dust-to-gas ratio might be explained by other independent reasons, but a consistent and perhaps the simplest interpretation is that M_{W_2} and therefore $@_{20}$ has been overestimated.

-235-

CHAPTER SEVEN

CONCLUSIONS

7.1 SUMMARY OF THE PRESENT WORK

A quantitative description of the evolution of spiral galaxies is a complex problem requiring a wide ranging synthesis of observation and theory. This thesis has attempted to understand the key role played by molecular hydrogen in that evolution, in areas such as star formation, chemical evolution and the relationship between the Galaxy and the Solar System. The goal of this work was not only a description of the global physics of H_2 in galaxies, but also to use that physics to constrain the mass and radial distribution of H_2 , a problem of much contemporary debate.

The relationship of molecular hydrogen to the star formation rate (SFR) in galaxies has been examined. Although it is now well established observationally that star formation takes place primarily in the molecular gas, previous parameterisations of the SFR had not explicitly taken this into account. Following Talbot (1980), from a comparison of the radial distribution of the SFR, H₂ and HI in the Galaxy, it is concluded that the SFR ($\frac{1}{2}$) seems to be better correlated with the surface density of molecular hydrogen (Σ_{M_2}) than the total gas or atomic hydrogen, such that

$$\Psi_{\rm S} \ll \Sigma_{\rm M_2}^{\rm K} \tag{7.1}$$

with k determined empirically to be 1.2 ± 0.2 for the Bhat et al. (1985) distribution of H_2 , or 0.7 ± 0.2 for the SSS (1984) distribution of H_2 . Between surface or volume densities of gas, the data at present do not make a sharp distinction. We therefore suggest that it is the amount of molecular hydrogen, as opposed to the total amount of gas, in any particular region which determines the SFR. Spiral arms merely re-organise matter within discs and do not determine the SFR, in agreement with recent observational results from external spiral galaxies (Elmegreen and Elmegreen 1986).

The molecular fraction of the gas (p_2) is found to be correlated with the metallicity Z of a region with $p_{\gamma} \ll Z^{\flat}$, $b \cong 1.3$. In its simplest form this relationship can be understood in terms that a region of higher metallicity will have more grains and consequently more evolution of $HI \longrightarrow H_2$. In addition, metals themselves could be important in the formation of molecular clouds, acting as coolants and increasing opacity. We suggest therefore that variations in the mass ratio of H_2/HI as a function of Galactocentric radius and from galaxy to galaxy are controlled by the metal abundance (assumed to be proportional to the dust-to-gas ratio), rather than the volume densities (Tacconi and Young 1986, 1987) or spiral arms (Wyse 1986). Tacconi and Young (1987) have since suggested this qualitatively from their study of dwarf irregular galaxies. This picture is consistent with recent observations of external spirals that suggest that density waves do not trigger molecular cloud formation (Stark et al. 1987).

The problem for models of galactic evolution has always been to describe the time evolution of Σ_{M_2} and Ψ_S . Using the above correlations derived from the radial distributions in the Galaxy, a more explicit form of the SFR is obtained in terms of the total gas fraction μ and metallicity,

-237-

Ys ec (Erxyzb)k

where \mathscr{B}_{T} is the total surface density of matter and X is the chemical fraction of hydrogen in the ISM. The time dependence of this modified 'Schmidt law' of the SFR explains quite naturally the production of a metallicity gradient and the constancy of the \mathscr{B}_{S} over the lifetime of the disc, without the requirement of infall of gas from the halo.

A preliminary study of external spirals seems to support the general nature of the above 'law'. The total SFR of a galaxy is shown to be correlated with its total H_{2} content, although uncertainty concerning the $CO \rightarrow H_{2}$. conversion factor, α_{20} , in these galaxies means that this result must be interpreted at present with care. Compiling radial distributions of HI, H_{2} and Z for nine other spirals, we confirm the same correlation $p_2 \in Z^{\flat}$ as observed in the In a further attempt to study this correlation, the Galaxy. mass of dust in a galaxy has been calculated from the 60mm and 100µm IRAS fluxes and this 'warm' dust-to-total gas ratio is shown to be correlated with the total ${\rm p}_2$ of a galaxy. The 100µm flux however does not sample the total mass of dust in a galaxy and so 1300µm observations have been used in order to derive the total dust-to-gas ratio of a galaxy. This analysis can only be at present carried out for the central 90" of galaxies and there are large uncertainties concerning the temperature of the dust and the radial distribution of the gas. No correlation of this dust-to-gas ratio with p2 is found. However, there is evidence that the dust-to-gas ratio and/or α_{20} varies from galaxy to galaxy.

(7.2)

-238-

Using the above form for the SFR, a self-consistent model of the chemical evolution of the Galaxy is proposed and compared with observations. This model is presented in terms of the instantaneous recycling approximation and includes an explicit form for dark matter. Due to the observational and theoretical uncertainty concerning the existence of infall of gas to the disc, we assume gas flows to have negligible effect on the chemical evolution of the disc, and evolve each disc annulus as a closed system. The yield of metals is taken as $y_{\chi} = KZ^{n}$ to study both the constant yield case (n = 0) and variable yield. The formalism is developed for any value of n, but results are computed for n = 0 and n = 1.

The G-dwarf problem in the solar neighbourhood is solved by the requirement that the initial metallicity of the disc $Z_0 \simeq Z_1/3.7$ (Z, being the present metallicity) which finds recent support from consideration of the thick disc of the Galaxy (Gilmore and Wyse 1986). The age-metallicity relation of stars in the solar neighbourhood (Twarog 1980) is well reproduced by the same model. The case of constant yield provides a better fit to the AMR, while variable yield provides a better fit to the G-dwarf metallicity distribution, but considering present observational uncertainties neither case is ruled out. The returned fraction R is found to be in the range 0.3 - 0.65 depending on the value of n and the gas fraction in the solar neighbourhood.

For both constant and variable yield, the ratio of the present SFR (ψ_1) to the SFR when the disc formed (ψ_2) is found to be approximately unity in the solar neighbourhood,

-239-

consistent with observational determinations of the past history of the SFR. For n = 1, the gas at present in the solar neighbourhood will be depleted to zero in a short timescale of \approx 3Gy, while for the constant yield case the SFR decreases rapidly after \approx 10Gy and μ > 0 for Galactic lifetimes > 20Gy.

Given an initial density profile of the total mass distribution in the disc, it is shown that with time the Galaxy may naturally develop a radial gradient in the gas fraction and the metallicity even though it begins with a constant initial gas fraction and constant initial metallicity at all radii. The present observed metallicity gradient can be generated simply from the radial distribution of Σ_{γ} , provided that the exponent k > 1. The metallicity gradient is not generated by infall or radial gas flows or necessarily variable yield (although this helps). We conclude therefore that a consistent model of the chemical evolution of the disc of the Galaxy is possible without invoking infall or radial inflow of gas.

To further examine the consistency of this model, we have studied the returned fraction R, the dark matter fraction D and the yield of metals y_{Z} from the IMF and details of stellar nucleosynthesis. With the IMF of Miller and Scalo (1979), the usual remnant fraction and the fraction of ejected metals from Maeder (1984), we find $R \cong 0.42$, $D \cong 0.05$ and $y_{Z} = 0.024$. It seems very difficult to simulate a case where $y_{Z} \ll Z$ from current knowledge of the IMF and stellar nucleosynthesis, the case of y_{Z} = constant is much more likely.

The consequences of this constant yield model of

-240-

· · · · · ·

Galactic chemical evolution for the nature of the dark matter in the solar neighbourhood have been discussed. If from the details of stellar nucleosynthesis the yield is consistent with the form of the IMF of Miller and Scalo (1979) extended down to 0.004 $\rm M_{\odot}.$ However, if $\rm y_{\rm Z} \, \cong \, 0.025$, D/(1-R) is 🖄 0.4 for which the Miller-Scalo IMF is inadequate and a separate population of brown dwarfs are needed. The IMF of Larson (1986) can provide D/(1-R) ≥ 0.2 in the form of dead stellar remnants rather than brown dwarfs but requires a small upper mass cut-off to metal production, < 16M₀, which may be unrealistic. The IMF of Rana (1987a) derived using the SFR of equation (7.2) requires a large population of brown dwarfs and an upper cut-off < $100M_{\odot}$. The IMF of Scalo (1986) with a constant SFR cannot provide dark matter in baryonic form to solve the dark matter problem in the disc.

Assuming the applicability of the chemical evolution model to other spirals, abundance gradients in external spirals are examined. For a constant yield, no infall and k = 1.4 the trend between Z and \mathbb{Z}_{γ} of Edmunds and Pagel (1984a) is reproduced, the shape of the relationship depending on the age of the disc. We suggest that for spiral galaxies with constant Z_0 as a function of radius, the metallicity gradient may be a direct indicator of the age of the disc. The evolution of radial gradients in stars, gas and metals are illustrated. It is shown that $\frac{4}{3}$ is exponential with radius for the first 7-8 Gy of a galaxy's lifetime and the total galactic $\frac{4}{3}$ remains roughly constant. A central hole in the gas distribution is shown

-241-

to arise quite naturally without invoking gas flow into the bulge. More detailed observations and modelling for other spirals are required but these initial results are encouraging.

The relationship of the molecular gas in the Galaxy and other Galactic properties to the evolution of the Solar System has been studied, in particular the hypothesis that the observed 30My periodicity in the record of mass extinctions has a Galactic cause, namely perturbations of the Oort Cloud by known or unknown bodies in the Galaxy. Our principal result is that comet showers are not produced either with sufficient frequency or intensity by individual. known bodies, whether stars or molecular clouds. If showers do occur with a mean interval of 30My, it may be possible to modify the Galactic hypothesis by invoking perturbations of the Oort Cloud by hypothetical dark matter, but only if the dark matter has a mean density > 0.1 M_{\odot} pc⁻³ and is confined closely to the plane, both conditions being very unlikely. Therefore, if there is indeed an apparent 30 My periodicity in the mass extinction and geological records, we argue that astronomically induced processes are unlikely to be the primary cause.

Finally, we have examined the question of the lifetime of GMCs showing that previous arguments favouring long GMC lifetimes are no longer valid. We suggest that the lifetime of the molecular phase is constant with Galactocentric radius while the lifetime of the atomic phase is determined by the metallicity. This explains the previous correlation of p_{2} and Z and possibly the molecular fraction of 'superclouds' (Elmegreen & Elmegreen 1987b). A consistent

-242-

and comprehensive picture of the formation, evolution and dispersal of GMCs is still required, but it is concluded from a number of arguments that the lifetime of the molecular phase has an upper limit of 2.10^{9} y.

Arguments favouring lifetimes of the molecular phase > $(3-10).10^{\circ}$ y as deduced from the Virgo spirals are shown to be not well founded. These arguments depend on HI being stripped throughout the whole of the galactic disc, whereas in Virgo HI stripping occurs (if at all) predominantly in the outer regions of the disc.

From a study of CO, HI, optical and IR observations of late type spirals in Virgo, the effect of the intercluster medium on the $CO \longrightarrow H_2$ conversion factor has been examined and it is shown that Ol_{20} is lowered in these galaxies with respect to its value in the solar neighbourhood and in isolated galaxies.

7.2 IMPLICATIONS FOR THE $H_2 \rightarrow CO$ CONVERSION FACTOR AND THE MASS OF H_2 IN THE GALAXY.

Chapter 1 reviewed the current controversy surrounding the CO—>H₂ conversion factor (α_{20}) and the mass and radial distribution of H₂ in the Galaxy. The debate centres on the value of α_{20} locally and whether it varies as a function of Galactocentric radius and from galaxy to galaxy. Methods to determine α_{20} empirically have large uncertainties and have led to conflicting results. SSS (1984) from extinction and virial mass estimates determined $\alpha_{20} = 7.2$ locally and throughout the Galaxy. Although more recent work has reduced this value to $\simeq 6.0$ (Solomon and Rivolo 1987), such

-243-
a high and constant value of \mathscr{C}_{29} means that H_2 is the dominant gas component in the inner Galaxy. Bhat et al. (1986) from the cosmic \Im -ray evidence and other arguments claim a value of $\mathscr{C}_{20} \cong 3.0$ in the solar neighbourhood, decreasing towards the inner Galaxy. This variation in \mathscr{C}_{29} , ascribed to either metallicity or temperature effects results in the mass of H_2 in the inner Galaxy to be only one-fifth of that claimed by SSS (1984).

The intention of this thesis was to use the global physics of H_2 in the Galaxy in order to constrain its radial mass distribution and hence \ll_{20} .

In the models of the SFR and Galactic chemical evolution (Chapters 2 and 3) the H₂ surface mass distributions of SSS and Bhat et al. were compared. It was found that both gave good correlations with the SFR (equation 7.1), but leading to different values of the exponent k. The Bhat et al. values of \mathbb{E}_{M_2} led to $k = 1.2 \pm 0.2$ while for SSS $k = 0.7 \pm 0.2$. This difference in k, and different values of the gas fraction μ in the solar neighbourhood led to marginally better fits to the AMR and higher values of the returned fraction for the Bhat et al. distribution. However, uncertainty in the AMR and returned fraction means that neither H₂ distribution can be ruled out by these considerations.

The real distinction between the two H_2 distributions came in computing the Galactic metallicity gradient in a closed model of chemical evolution. We have shown that the predicted metallicity gradient depends crucially on the value of k. Only if k > 1 will the metallicity increase towards the Galactic centre as observed. Therefore, in such

-244-

a model only the Bhat et al. H_2 distribution can produce the observed gradient. Of course, a more complex chemical evolution model with infall and radial gas flows may be able to generate a metallicity gradient even with the SSS distribution of gas. It is not possible at present to rule out the existence of infall, although we have argued (Section 3.2) that current observations do not favour its occurrence in the later stages of Galactic evolution. However, in this work we have argued that infall is not required in order to give a consistent model of the chemical evolution of the Galaxy. The Bhat et al. H_2 distribution allows the adoption of a simple closed model with ψ_{S} et $\Sigma_{W_2}^{K}$.

In fact, compiling data on the radial distributions of the SFR in the Galaxy, and fixing k = 1.4 by the observed metallicity gradient, a radial distribution of H₂ has been derived independently of the CO and U-ray surveys. Using Turner's (1984) model of star formation we tentatively derived a surface density of H₂ in the solar neighbourhood to be $1.3\pm0.2 \text{ M}_{\odot}\text{pc}^{-2}$. When compared to the CO data of SSS (1984) this value gives $(\alpha_{20})_{\odot} = 2.5\pm0.5$, consistent with the value advocated by Bhat et al. (1985) from the U-ray analysis. It is also close to $\alpha_{20} = 3.6$ derived recently by Maloney and Black (1987) using chemical modelling techniques.

The radial distribution of H_2 suggested by the present work is shown in Figure 7.1 in comparison with the other distributions. In the outer Galaxy, we claim a small underestimate of H_2 , even by Bhat et al. (1985), possibly due to the incompleteness of the CO survey. In the inner Galaxy we find good agreement with Bhat et al. (1985). As

-245-



Figure 7.1 : The radial distribution of Σ_{M_2} in the Galaxy. The H₂ distributions of SSS (1984) and Bhat et al. (1985) are contrasted with the distribution derived in this work from the SFRs and observed metallicity gradient.

pointed out in Section 1.2, this is also in good agreement with the \forall -ray analysis of the Columbia/COS B group (Bloemen et al. 1986; Strong et al. 1987), who use an $\alpha_{20} = 5.6$ but refer to a different CO radial profile.

This means that the masses of HI and H₂ are roughly equivalent in the inner Galaxy. From a study of the lifetime of the molecular phase, we conclude that $T_{\rm H_2} < 2.10^8$ y which also leads to roughly equivalent amounts of HI and H₂. Arguments favouring a much larger $T_{\rm H_2}$ and

-246-

therefore larger $M_{W_{2}}$ in the inner Galaxy, in particular those from the Virgo spirals, are shown to be not well founded. Perturbations of the Oort Cloud of comets by GMCs are shown not to provide any useful constraints on their mass density.

Thus, evidence from chemical evolution and GMC lifetimes seems to confirm evidence from \Im -rays and other arguments (Bhat et al. 1986) that the mass of H_2 in the inner Galaxy is roughly one fifth that claimed by SSS (1984).

Comparing the radial distribution of H_2 in Figure 7.1 with the CO radial profile of SSS (1984) indicates that α_{20} varies as a function of Galactocentric radius. If this is due to metallicity such that $\alpha_{20} \ll 2^{-n^2}$, then we estimate $n' \ll 0.8-0.9$ as compared to Bhat et al.'s n' = 1. From a theoretical point of view it is still not clear whether metallicity and/or temperature will produce a radial gradient in α_{20} (e.g. Kutner and Leung 1985; Williams 1985; van Dishoeck and Black 1987), but nevertheless an assumption of a constant α_{20} (e.g. Young and Scoville 1982a; SSS 1984) seems to be incorrect.

Maloney and Black (1987) have recently used sophisticated modelling techniques to study the effect of variations in the gas temperature, mean density and CO abundance on \mathfrak{A}_{20} . They model the CO emission from galaxies taking into account cloud-cloud shielding, specifying the H₂ distribution a priori, and examine the effect of variations on I_{CO}. They conclude that the use of a constant \mathfrak{A}_{20} between galaxies will give very misleading results because of these variation effects. We have used IRAS and 1300 μ m

-247-

observations of spiral galaxies to suggest that the dust-to-gas ratio and/or \ll_{20} may vary by an order of magnitude between galaxies. We suggest that the effect of the ICM on GMCs in the Virgo spirals is to lower \ll_{20} with respect to the value in the solar neighbourhood and in isolated galaxies. If the IR and Ha properties truly describe the current SFR, and if Turner's (1984) model of star formation is correct we predict $\ll_{20} \ll 2$ in the Virgo spirals. Therefore any estimates of H₂ in external galaxies from CO using a constant \ll_{20} must be treated with caution.

7.3 FUTURE WORK

The present work has opened up a number of questions and avenues for further study. For example, although we have attempted to present a consistent model of Galactic chemical evolution, much work still needs to be done. As Pagel (1987) has pointed out, the subject is in a confused state, attempts to build a unique quantitative picture hampered by ambiguities in the data which lead to great freedom in theory and assumptions. Better and more comprehensive abundance data are required for both this and other galaxies. Theoretical understanding of stellar evolution, in particular the effect of binary evolution (e.g. Matteucci and Greggio 1986; Iben 1987), and the IMF (e.g. Scalo 1986; Rana 1987a) is crucial to models of galactic chemical evolution.

Perhaps the crucial parameter for chemical evolution and certainly the SFR is the mass of H_2 . Empirical determinations of \mathfrak{A}_{20} (Solomon et al. 1987a; Strong et al.

-248-

1987; Richardson and Wolfendale 1987) and theoretical work on cloud chemistry (van Dishoeck and Black 1986; Maloney 1987) and CO emission from galaxies (Maloney and Black 1987) are at present in progress. However, if \mathcal{C}_{20} depends on metallicity, then radial abundance data of galaxies are essential if the correct H₂ masses are to be inferred from the CO data. If \mathcal{C}_{20} depends on gas temperature effects, then it is necessary to obtain information on the excitation of CO, from the ratio of higher transitions (Maloney and Black 1987). The use of radial 1300µm observations in addition to an understanding of how the dust-to-gas ratio varies may provide an independent way of estimating H₂ in external galaxies.

We finally mention two other areas where H_2 seems to play a key role and where further study is required. Open clusters in the disc of the Galaxy are disrupted by mutual gravitational interactions between stars in the cluster, the stationary tidal field of the Galaxy and also by GMCs. van den Bergh and McClure (1980) and Lynga (1982) suggested qualitatively that the prevalence of massive clouds of H_2 in the inner Galaxy could be responsible for the fact that open clusters in the outer parts of the Galaxy have longer lifetimes than those in the inner parts.

Wielen (1985) has considered quantitatively the problem of encounters of open clusters with GMCs, deriving an analytic way of dealing with these encounters. The effect of GMCs in the solar neighbourhood is shown to give fairly good agreement with the age distribution of clusters, the reduction in lifetime depending on the volume density $\rho_{\rm H_2}$. We have used his formalism, examined the dependence on $\rho_{\rm H_2}$

-249-

and also find a dependence on cloud mass and radius. In the solar neighbourhood the difference between the H_2 distribution of SSS (1984) and this work would lead to a factor of ≈ 2 in total lifetime. If we naively insert the values for ρ_{W_2} at $R_{G} \cong 6$ kpc where the difference is more appreciable, we would predict a factor of 5 difference in lifetime.

Therefore, as a consequence of the shape of the H_{2} distribution one should expect the age distribution of open clusters to vary as a function of R_a. This has been suggested qualitatively but never shown quantitatively. The work on the solar neighbourhood will need to be extended by a study of the variation of the Galactic tidal field and mass and size of GMCs in order to predict open cluster lifetime as a function of R_{G} . The use of an open cluster catalogue (e.g. Lynga 1985) will give the age distribution of open clusters as a function of R_{G} , giving due care and attention to other variables affecting age such as radius, mass, z-dependence and possibly metallicity. The comparison of predictions with observations should lead to constraints on the radial distribution of H_2 , especially the volume densities at 6 kpc. Apart from acting as a test of dynamical theories of open clusters, this will allow H_2 to be estimated independently of the controversy surrounding the CO and ^U-ray data. The observation that clusters in LMC have much longer lifetimes than in the Galaxy (Elson and Fall 1985) may also provide a constraint on the mass of H_2 there.

An associated problem is the interaction of GMCs with wide binaries in the solar neighbourhood. Weinberg et al.

-250-

(1987) have recently studied the time evolution of wide binaries by stochastic encounters with field stars and GMCs in the solar neighbourhood with a hybrid Monte Carlo scheme. They find that if the molecular cloud density is different by a factor of 2, the half-lives for disruption change by a factor of 1.4. However, not only is the disruption process sensitive to cloud density, the cloud's internal structure will have a significant effect. Observations of wide binary separations may therefore constrain the mass density and internal structure of GMCs in the solar neighbourhood and also could provide constraints on the dark matter in the disc (Bahcall et al. 1985).

In conclusion, the role of H_2 in galaxies is crucial for many processes. We have attempted to outline various areas but many more await detailed study.

APPENDIX

Table A.1.1 : Recent values of the CO to H_2 conversion factor, \leq_{20} , published in the literature.

 $\ll_{20}(\text{atoms cm}^2 \text{K}^1 \text{km}^1 \text{s}) = 2N(H_2)/\int T(CO) dv.$

Determination and Authors	50	Notes
V <u>isual Extinctio</u> n (local region)		
Dickman (1975)	4.4	A _V < 4
Gordon & Burton (1976)	4.6	
Dickman (1978)	3.6	
Blitz (1978)	15.0	
Blitz & Thaddeus (1980)	15.0	
Blitz & Shu (1980)	2.0	
Frerking, Langer & Wilson (1982)	3.6	ρOph
Frerking, Langer & Wilson (1982)	9.6	Taurus
Young & Scoville (1982a)	8.0	
Liszt (1982)	10.0	ξ Oph, A _v > 5
Thaddeus (1983)	3-4	·
SSS (1984)	7.2	A _v > 5
Lebrun & Huang (1984)	2.2	Oph-Sagittarius
Bhat, Mayer & Wolfendale (1986)	2-4	
Richardson & Wolfendale (1987)	3.0	

Virial theorem

SSS (1984)						9.2	
Sander	5,	Scoville	8	Solomon	(1985)	5-9	

Scoville et al. (1986)	5-9
Bhat, Mayer & Wolfendale (1986)	3.0
Solomon et al. (1987a)	6.0
Scoville et al. (1987)	6.1

<u>v-rays</u>

Li et al. (1983)	2.3 local clouds
Hermsen & Bloemen (1983)	5.2 Orion complex
Bloemen et al. (1984)	5.2 Orion complex
Houston & Wolfendale (1985)	3.7 Orion complex
Bhat et al. (1985)	2.7 local clouds
Lebrun et al. (1983)	2-6 inner Galaxy
Bhat et al. (1984,1985,1986)	2.7 ⁺
Bloemen et al. (1986)	5.6 inner Galaxy

Other methods

Strong et al. (1987)

Solomon, Sanders & Scoville (1979) 12.0 radiative transfer Rickard et al. (1985) Kutner & Leung (1985) Bhat, Mayer & Wolfendale (1986) Bhat, Mayer & Wolfendale (1986) Parkinson et al. (1987)

4.6 standard cloud method 4.0 theoretical models 3.8 [†] X-ray absorption 2.7 galaxy counts 3.9 IR analysis

4.6 inner Galaxy

 † Value shown is the local value, but is found to be a function of Galactocentric radius.

Table A.1.2 : Total Galactic H2 mass estimates for

 $2 < R_G < 10$ kpc.

MH2 (10⁹MC)

Gordon & Burton (1976) 2.1 Scoville & Solomon (1975) 1-3 Solomon, Sanders & Scoville (1979) 3.9 Mass-Stony Brook SSS (1984) 2.6 Solomon et al. (1987a) 2.0 Lebrun et al. (1983) 0.4-1.2 Dame (1984) 0.7 Columbia/COSB Thaddeus & Dame (1984) 0.7 Bronfman et al. (1987) 1.2 Bhat et al. (1984,1985) 0.6 Bhat, Mayer & Wolfendale (1986) 0.6 Durham Parkinson et al. (1987) 0.9

(cf. $M_{HI} = 0.9.10 {}^9M_{\odot}$; Henderson, Jackson & Kerr 1982; Bloemen et al. 1986)

Table A.2.1 : Basic data for gas distributions and star

			(a)		(b)	(c)	
RG	Σ _H	I	h _{hi}	2 Bo	h _H 2	ይ፹	¢
(kpc)	(M _o p	ic ^{−2})	(pc)	(M _e pc	²) (pc)	(^M opc ²)	(^ເ ເ _o pc ² _{Gy} ⁻¹)
	(1)	(2)	(3)	(4) (5) (5)	(5)	(6)
3	0.6	1.1	150	0.5 4	.2 44	371	
4	1.6	1.5	150	0.9 6	.6 46	296	0.2
5	2.9	2.3	150	2.0 11	.8 56	235	12.0
6	3.7	2.8	150	3.4 16	.5 71	187	30.0
7	4.4	2.4	150	2.8 12	.2 73	149	14.0
8	5.1	2.8	150	2.5 8	.5 61	119	18.0
9	3.8	2.5	150	1.6 5	.1 65	94	7.9
10	3.5	2.7	150	1.4 3	.8 70	75	5.0
11.5	4,9	2.7	240	0.3 0	.7 70	53	6.0
12.5	5.5	3.9	300	0.3 0	.6 70	42	2.8
13.5	5.8	2.8	360	0.3 0	5 70	34	
14.5	5.8	1.2	420	0.1 0	.2 70	27	0.05

formation rate in the Galaxy.

- (1) Li et al. (1982)
- (2) Burton & Gordon (1978)
- (3) Kulkarni et al. (1982)
- (4) Bhat et al (1985)
- (5) SSS (1984)
- (6) SBM (1978)
- (a) Scale height of HI distribution
- (b) Scale height of H_2 distribution

(c) Total surface density with $\Sigma_T(R_G)_{\odot} = 75 M_{\odot}pc^{-2}$ (cf. Bahcall et al. 1983), which is also similar at $R_G>$ 4 kpc to the double exponential model of Caldwell & Ostriker (1981).

Table A.2.2	00	Determ:	inations of	of	the met	allicity	gradient,
			a dha cuit Lite		0		

 dlog(Z)	$/dR_{G}$ (c	lex kpc ⁻),	, for the	Galaxy.

e la station

Objects	Range of R _G	(Fe/H)	(0/H)	(2)	References
HII regions (electron temp.)	8-14		-0.08		Mezger et al. (1979)
	8-17		-0.05		Wink et al. (1983)
	4-13		-0.06		Garay & Rodriguez (1983)
(optical spectra)	8-14		-0.04		Hawley (1978)
	8-13		-0.13		Peimbert et al (1978)
	8-14	·	-0.09		Peimbert & Torres-Peimbert (1979)
	8-14		-0.11	· · · · ·	Talent & Dufour (1979)
	8-14		-0.10		Peimbert (1979)
	6-15		-0.07		Shaver et al. (1983)
Planetary Nebulae	7-14		-0.06		Torres-Peimbert & Peimbert (1977)
	8-12		-0.04		Peimbert & Serrano (1980)
	7-13		-0.07		Faundez-Abans & Maciel (1986)
Open Clusters	8-14	-0.05			Janes (1979)
	8-12			-0.10	Panagia & Tosi (1980, 1981)
	8-14	-0.13			Lynga (1985)
	8-13	-0.11			Cameron (1985)
Cepheids	5-15			-0.07	Harris (1981)
(IIOm periods)	5-14			-0.07	Harris & Pilachowski (1984)
Supergiants	9-12	-0.13			Luck (1982)
A and F stars	8-15	-0.08			Christian & Smith (1983)
Old disc stars	9-12	-0.04			Mayor (1976)
Young disc stars	9-12	-0.10			Mayor (1976)
All stars	9-12	-0.05			Mayor (1976)
		-0.05			Marsakov & Suchkov (1982)

tusing the calibration of O abundance of Shaver et al. (1983).

Table A.2.3 : Sources of data for Figure 2.8

(Section 2.5.2).

	HI	CO	(O/H)
The Galaxy	1	2	1
M51	3	4,5	6
IC342	7,8	9,10	6
NGC 69 4 6	8	10	6
M100	11	12	13
M101	8,14	15	16
M31	17	18	16
M33	19	16	16
M83	20,21	12,22	16
NGC2403	8	23	6

References.

1.	This work	13. McCall et al. (1985)
2.	SSS (1984)	14. Bosma et al. (1981)
3.	Weliachew & Gottesman (1973)	15. Solomon et al. (1983a)
4.	Scoville & Young (1983)	16. Diaz & Tosi (1984)
5.	Rydbeck et al. (1985)	17. Gottesman & Davies (1970)
6.	Tosi & Diaz (1985)	18. Coombes et al. (1977)
7.	Newton (1980b)	19. Newton (1980a)
8.	Rogstad & Shostak (1972)	20. Allen et al. (1986)
9.	Morris & Lo (1978)	21. Rogstad et al. (1974)
10.	Young & Scoville (1982a)	22. Coombes et al. (1978)
11.	van der Kruit (1972)	23. Young (1985a)
12.	Young & Scoville (1982b)	

Figure A.2.4:

The calibration curve of oxygen abundance against the observed ratio ([OII] + [OIII])/H[®] for HII regions. Shown are the curves given by Edmunds and Pagel (1984), McCall, Rybski and Shields (1985) and Dopita and Evans (1986) from which the Figure is taken.



Appendix A.2.5 : Is the dust-to-gas ratio a function of the metallicity?

It is a widely accepted assumption that the relative amount of dust is correlated with the metal abundance of a region (e.g. Koornneef 1985; Section 2.4). Both observation and theory tend to reinforce this assumption, but the situation is still far from clear.

Campbell et al. (1986) found that HII galaxies with lower abundances had less optical reddening, a trend

-258-

expected if the dust content decreased with decreasing metallicity (Z). On the basis of this picture, these low abundance galaxies should not be strong IR sources. Gondhalekar et al. (1986) from a study of the IRAS data from three low metallicity HII galaxies found this to be the case. Estimating the mass of gas (M₀) simply from the HI flux and the mass of dust (M₀) from the 100µm emission, they found $M_0/M_0 \ge 10^5$ as compared to $\approx 10^2$ in the solar neighbourhood. No CO observations were available for these galaxies so H₂ has been neglected, and of course the 100µm emission does not sample all of the dust, but nevertheless the dust-to-gas ratio is very low, in line with what would be expected for low metallicity.

Viallefond et al. (1982) found a correlation between the dust-to-gas ratio and metallicity in HII regions in M101, the higher the metallicity the higher the dust-to-gas ratio. The gas was estimated using both HI and CO (using a constant α_{20}), the dust mass being estimated from optical and radio extinction.

The main observational evidence however for $M_{\odot}/M_{\odot} \ll Z$ comes from a study of the SMC and LMC. The amount of gas in dust grains cm⁻³ can be written

$$\langle \rho_d \rangle = \frac{4\pi}{2} a^3 \rho_d n_d = 2 f_d m n_{\gamma}$$
 (A.1)

where a is the radius of the dust grains, $\rho_{\rm cl}$ is the actual mass density within the grain, $f_{\rm cl}$ is the fraction of heavy elements in dust grains, m is the gas mass per proton and $n_{\rm V}$ is the proton number density (Franco and Cox 1986). Therefore the dust-to-gas ratio is given by

$$\langle \underline{p}_{\underline{d}} \rangle = Z f_{\underline{d}},$$
 (A.2)

and if the 'efficiency' of dust formation (f_{d}) is roughly constant then the dust-to-gas ratio will be c Z. Relative to the solar neighbourhood, the dust-to-gas mass ratio (Zfd) is smaller by a factor of 4 in the LMC (Koornneef 1982) and by a factor of 8 in the SMC (Bouchet et al. 1985) whereas the heavy element abundances are smaller by factors of 4 and 10 in the LMC and SMC respectively (Dufour 1984). Thus, in spite of the large differences in heavy element abundances between the SMC, LMC and solar neighbourhood, the value of $f_{cl} \cong 0.5$ remains about constant for all three systems (Clayton and Martin 1984; Bouchet et al. 1985; Franco and Cox 1986). The above analysis of the LMC and SMC uses extinction in the visible and near IR towards particular stars, to estimate the dust mass, and the profile of the Lyman & interstellar absorption line along the same line of sight to estimate the column density of N_{MZ} . This analysis is done for lines of sight where there is little CO emission. If in the SMC and LMC there is considerably more H_2 because of a higher \ll_{20} (Bhat et al 1984) the dust-to-gas mass ratio would be even lower.

Not only does metallicity vary from galaxy to galaxy, it also varies as a function of galactocentric radius $(R_{G_{i}})$. If the dust-to-gas ratio is determined by the metallicity, we may expect M_{\odot}/M_{\odot} to increase towards the centre of a galaxy. For our own galaxy, it is difficult to interpret the variation of the IRAS 100 μ m flux as an indication of dust mass because of the possible variation of the dust temperature with galactocentric radius (Burton et al. 1986; Burton and Deul 1987). However, Sodroski et al. (1987) assuming that all of the dust along a given line of sight is

-260-

at a single temperature have used the 60µm, 100µm, HI and CO fluxes in an attempt to study the Galactocentric variation of M_{\odot}/M_{\odot} . The mass of H_2 is calculated from the Columbia CO data using a constant $\mathfrak{Cl}_{2\odot} = 4$. Deriving the dust temperature from the ratio of the 100µm and 60µm fluxes, it appears to be essentially constant (\mathfrak{A} 24K) with $R_{\mathfrak{G}}$. The M_{\odot}/M_{\odot} ratio however decreases from the inner to outer Galaxy by a factor of \approx 3 which is similar to the change in Z over the same region. This may indicate that $M_{\odot}/M_{\odot} \mathfrak{C}$ Z, but care is needed if IRAS has primarily sampled the warm component of Galactic dust and the fraction of dust in this component varies with $R_{\mathfrak{G}}$. Detailed observations at longer wavelengths (\mathfrak{A} 1300µm) are needed to decide between these possibilities.

In a similar study for M31, Walterbos and Schwering (1987) have shown that the dust temperature is also relatively constant with $R_{\mbox{G}}$. It is then shown that the ratio of the IR optical depth at 100µm to the column density of HI increases with decreasing radius for $R_{\mbox{G}}$ < 15 kpc. Taken at face value, this result suggests a significant variation in the dust-to-(atomic) gas as a function of R_{G} . One may ask whether this gradient is due to neglecting the contribution of H_2 . Star formation, and hence probably H_2 (Chapter 2) is concentrated in a ring around $R_{G} = 10 \text{kpc}$, indicating that there is not much molecular material internal to this ring, and so the neglect of H_2 is unlikely to be the dominant cause of the gradient. By examining the role of small grains, Walterbos and Schwering conclude that the variation in M_0/M_{\odot} is in the range of the observed metal abundance variation (Blair et al. 1982) and also consistent

-261-

with the upper limit to the variation of M_{\odot}/M_{\odot} derived in an independent analysis of optical data (Walterbos and Kennicutt 1987).

Observationally therefore the dust-to-gas ratio is correlated with the metallicity. We might also expect this from theoretical considerations. Although the details of the formation and evolution of dust in galaxies are not yet fully understood, we know that the formation rate of icy mantles is largely controlled by the abundance of the CNO group of elements and by the gas density (Creze and Isobe 1974). The most probable source of graphite grains is the atmosphere of low and intermediate mass red giants, while silicate grains are probably produced from type II supernova (Gondhalekar et al. 1986). Therefore, it is reasonable to believe that the amount of dust in the ISM will increase with the age of a galaxy, as does the metallicity. In a particular region stellar processing not only acts to increase the metal abundance, but it also depletes the mass of total gas. Even if the mass of dust stays constant, the dust-to-gas ratio will increase with metallicity.

Thus, to a first approximation we assume that

 $\frac{M_0}{M_0} \ll Z$ (A.3)

Observations of abundances and IR measurements at longer wavelengths are needed to confirm this relationship.

Table A.4.1 : Best fitting model parameters for different sets of S_1 and M_{Θ_1} .

> $n=0, \mu_1=0.09, \sigma_0=0.05, \log(Z_1/Z_0)=0.57, \psi_1=3\kappa_3pc^2 Gy^{-1}$ $d\log(Z_1)/dR_G = -0.06 dex kpc^{-1}$

δ ₁	0.4		0.5			0.6			
٥	0.5	0.6	0.8	0.5	0.6	0.8	0.5	0.6	0.8
G	-	72.1	83.0	•	72.1	83.0	65.2	72.1	83.0
χ^2_{GD}	-	36.1	28.7	-	31.2	24.6	29.2	25.0	19.5
δ ₀	-	0.35	0.15	-	0.35	0.15	0.45	0.35	0.15
D/(1-R)	-	0.098	0.35	-	0.29	0.49	0.37	0.49	0.63
УZ		0.015	0.018	-	0.019	0.024	0.024	0.027	0.033
χ^2_{AMR}	-	2.8	2.9	-	2.8	2.9	2.7	2.8	2.9
$(\psi_{1}/\psi_{0})_{0}$	。 -	0.76	0.49	-	0.76	0.49	0.99	0.76	0.49
R	-	0.41	0.40	-	0.41	0.40	0.44	0.41	0.40
D	-	0.054	0.202	-	0.163	0.282	0.193	0.272	0.363
k	-	1.40	1.48	-	1.40	1.48	1.37	1.40	1.48

Table A.5.1 : Short term periodicities claimed in the terrestrial record.

Period	(My)	Source
a 30		Dorman (1968)
≅32		Leggett et al. (1981)
∝30		Fischer & Arthur (1977)
∞33		Rampino & Stothers (1984b)
a33		Pandey & Negi (1987)
≃30		Holmes (1927)
≊32		Rampino & Stothers (1984b)
≊32		Fischer & Arthur (1977)
26-30		Raup & Sepkoski (1984)
30 <u>+</u> 1		Rampino & Stothers (1984a)
26 <u>+</u> 1		Sepkoski & Raup (1986)
32-34		Negi & Tiwari (1983)
≃ 30		Raup (1985a)
∝30		Stothers (1986)
≃30		Pal & Creer (1986)
≊27		Seyfert & Sirkin (1979)
31 <u>+</u> 1		Rampino & Stothers (1984a)
32 <u>+</u> 1		Rampino & Stothers (1984b)
≈28		Alvarez & Muller (1984)
	$ \begin{array}{c} $	$\begin{array}{c} 2 \\ $

References

- Adams, D.J., 1980. In: Cosmic X-ray Astronomy, ed. Meadows, A.J., p. 58, Adam Hilger, Bristol.
- Allen, M.A. & Robinson, G.W., 1977. Astrophys. J., 212, 396.
- Allen, R.J., Atherton, P.D. & Tilanus, R.P.J., 1986. Nature, 319, 296.
- Alvarez, L.W., Alvarez, W., Asaro, F. & Michel, H.V., 1980. Science, 208, 1095.
- Alvarez, W., 1987. Eros, 67, 649.
- Alvarez, W. & Muller, R.A., 1984. Nature, 308, 718.
- Alvarez, W., Asaro, F., Michel, H.V. & Alvarez, L.W., 1982. Science, 216, 886.
- Arimoto, N. & Yoshii, Y., 1986. In: Spectral Evolution of Galaxies, eds. Chiosi, C. & Renzini, A., p. 309, Reidel, Dordrecht, Holland.
- Arimoto, N. & Yoshii, Y., 1987. Astr. Astrophys., 173, 23.
- Arnett, W.D., 1978. Astrophys. J., 219, 1008.
- Audouze, J. & Tinsley, B.M., 1976. Ann. Rev. Astr. Astrophys., 14, 43.
- Azzopardi, M. & Vigneau, J., 1977. Astr. Astrophys., 56, 151.
- Baade, W. & Mayall, N.U., 1951. In: Problems of Cosmical Aerodynamics, p. 165, Control Air Document Office, Dayton, Ohio.
- Bahcall, J.N., 1984. Astrophys. J., 276, 169.
- Bahcall, J.N., 1986a. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 3, University of Arizona Press, Tucson.
- Bahcall, J.N., 1986b. In: Astrophysics of Brown Dwarfs, p. 233, eds. Kafatos, M.C., Harrington, R.S. & Maran, S.P., Cambridge University Press.
- Bahcall, J.N., 1986c. In: The Material Content of the Universe, Roy. Soc. Discussion Meeting, eds. Barrow, J.D., Peebles, P.J.E. & Sciama, D.W., Royal Society, London.
- Bahcall, J.N., 1986d. Ann. Rev. Astr. Astrophys., 24, 577.
- Bahcall, J.N., 1987. In: Dark Matter in the Universe, IAU Colloq. No. 117, ed. Kormendy, J., in press.
- Bahcall, J.N. & Bahcall, S., 1985. Nature, 316, 706.
- Bahcall, J.N., Hut, P. & Tremaine, S.D., 1985. Astrophys. J., 290, 15.
- Bahcall, J.N., Schmidt, M. & Soneira, R.M., 1983. Astrophys. J., 265, 730.
- Bailey, M.E., 1977. Astrophys. Space Sci., 50, 3.
- Bailey, M.E., 1983a. Nature, 302, 399.
- Bailey, M.E., 1983b. Mon. Not. R. astr. Soc., 204, 603.
- Bailey, M.E., 1983c. In: Asteroids, Comets, Meteors, eds. Lagerkvist, C.-I. & Rickman, H., p. 383, Uppsala Observatory, Uppsala.
- Bailey, M.E., 1984. Mon. Not. R. astr. Soc., 211, 347.
- Bailey, M.E., 1986a. Mon. Not. R. astr. Soc., 218, 1.
- Bailey, M.E., 1986b. Nature, 324, 350.
- Bailey, M.E., 1987. Icarus, in press.
- Bailey, M.E., Clube, S.V.M. & Napier, W.M., 1986. Vistas Astr., 29, 53.
- Bailey, M.E., Wilkinson, D.A. & Wolfendale, A.W., 1987. Mon. Not. R. astr. Soc., 227, 863. (Paper VI).
- Balbus, S. & McKee, C., 1982. Astrophys. J., 252, 529.
- Bally, J. & Langer, W.D., 1982. Astrophys. J., 255, 143; erratum 261, 747.
- Bash, F.N., Green, E. & Peters, W.L., 1977. Astrophys. J., 217, 464.
- Baud, B., Habing, H.J. & Oort, J.H., 1979. In: The Large Scale Characteristics of the Galaxy, IAU Symp. No. 84, ed. Burton, W.B., p. 29, Reidel, Dordrecht, Holland.
- Becker, S.A. & Iben, I.Jr., 1980. Astrophys. J., 237, 111.
- Beckwith, S., Persson, S.E., Neugebauer, G. & Becklin, E.E., 1978. Astrophys. J., 223, 464.
- Beers, T.C., Preston, G.W. & Shectman, S.A., 1986. Preprint.
- Berkhuijsen, E., 1982. Astr. Astrophys., 112, 369.
- Bertelli, G., Bressan, A., Chiosi, C. & Angerer, K., 1986. Astr. Astrophys., 168, 393.
- Bessel, M.S. & Norris, J., 1984. Astrophys. J., 285, 622.
- Bhat, C.L., Houston, B.P., Issa, M.R., Mayer, C.J. & Wolfendale, A.W., 1984. In: Gas in the Interstellar Medium, ed. Gondhalekar, P., p. 39, Rutherford Appleton Laboratory.
- Bhat, C.L., Houston, B.P., Issa, M.R., Mayer, C.J. & Wolfendale, A.W., 1985. Nature, 314, 511.
- Bhat, C.L., Mayer, C.J. & Wolfendale, A.W., 1986. Phil. Trans. R. Soc., 319, 249.
- Bicay, M.D. & Giovanelli, R., 1987. Preprint.
- Bienaymé, O., Robin, A.C. & Crézé, M., 1987. Astr. Astrophys., 180, 94.

Biermann, P. & Tinsley, B.M., 1974. Astr. Astrophys., 30, 1.

Binggeli, B., Tammann, G.A. & Sandage, A., 1987. Preprint.

Black, J.H. & Fazio, G.G., 1973. Astrophys. J., 185, L7.

Blair, W.P., Kirshner, R.P. & Chevalier, R.A., 1982. Astrophys. J., 254, 50.

Blitz, L., 1978. PhD Thesis, Columbia University.

- Blitz, L., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, eds. van Woerden, H., Allen, R.J. & Burton, W.B., p. 190, Reidel, Dordrecht, Holland.
- Blitz, L., 1987. In: Physical Processes in Interstellar Clouds, eds. Morfill, G.E. & Scholer, M., Reidel, Dordrecht, Holland, in press.
- Blitz, L. & Shu, F.H., 1980. Astrophys. J., 238, 148.
- Blitz, L. & Stark, A.A., 1986. Astrophys. J., 300, L89.
- Blitz, L. & Thaddeus, P., 1980. Astrophys. J., 241, 676.
- Blitz, L., Bloemen, J.B.G.M., Hermsen, W. & Bania, T.M., 1985. Astr. Astrophys., 143, 267.
- Blitz, L., Fich, M. & Kulkarni, S.R., 1983. Science, 220, 1233.
- Bloemen, J.B.G.M., 1985. PhD Thesis, University of Leiden, Holland.
- Bloemen, J.B.G.M., 1987. Astrophys. J., 317, L15.
- Bloemen, J.B.G.M. & Silk, J., 1987. Preprint.
- Bloemen, J.B.G.M., Caraveo, P.A., Hermsen, W., Lebrun, F., Maddalena, R.J., Strong, A.W. & Thaddeus, P., 1984. Astr. Astrophys., 139, 37.
- Bloemen, J.B.G.M., Strong, A.W., Blitz, L., Cohen, R.S., Dame, T.M., Grabelsky, D.A., Hermsen, W., Lebrun, F., Mayer-Hasselwander, H.A. & Thaddeus, P.M., 1986. Astr. Astrophys., 154, 25.
- Bobrovnikoff, N.T., 1931. Publs. astr. Soc. Pacif., 43, 324.
- Bohlin, R.C., Savage, B.D. & Drake, J.F., 1978. Astrophys. J., 224, 132.
- Bohor, B.F., Foord, E.E., Modreski, P.J. & Triplehorn, T., 1984. Science, 224, 867.
- Boisse, P., Gispert, R., Coron, W., Wynbergen, J.J., Serra, G., Ryter, G. & Puget, J.L., 1981. Astr. Astrophys., 94, 265.
- Bond, J.R. & Szalay, A., 1983. Astrophys. J., 274, 443.
- Bosma, A., 1981a. Astr. J., 86, 825.
- Bosma, A., 1981b. Astr. J., 86, 1791.
- Bosma, A., Goss, W.M. & Allen, R.J., 1981. Astr. Astrophys., 93, 106.
- Bosma, A., van der Hulst, J.M. & Sullivan, W.T. III, 1977. Astr. Astrophys., 57, 373.
- Bothun, G.D., 1985. Astr. J., 90, 1982.
- Bothun, G.D. & Dressler, A., 1987. Preprint.
- Bothun, G.D., Romanishin, W., Strom, S.E. & Strom, K.M., 1984. Astr. J., 89, 1300.
- Bothun, G.D., Schommer, R.A. & Sullivan, W.T., 1982. Astr. J., 87, 731.
- Bottinelli, L., Gougenheim, L. & Paturel, G., 1982. Astr. Astrophys., 113, 61.
- Bouchet, P., Lequeux, J., Maurice, E., Prévot, L. & Prévot-Burnichon, M.L., 1985. Astr. Astrophys., 149, 330.
- Bregman, J.N., 1980. Astrophys. J., 236, 537.
- Bronfman, L., Cohen, R.S., Alvarez, H., May, J. & Thaddeus, P., 1987. Astrophys. J., in press.
- Brosche, P., 1970. Astr. Astrophys., 6, 240.
- Brosche, P. & Leutes, F.T., 1985. Astr. Astrophys., 153, 157.
- Bruck, M.T., 1980. Astr. Astrophys., 87, 92.
- Burbidge, E.M., Burbidge, G.R., Fowler, W.A. & Hoyle, F., 1957. Rev. Mod. Phys., 29, 547.
- Burkert, A. & Hensler, G., 1987. Mon. Not. R. astr. Soc., 225, 218.
- Burton, W.B. & Deul, E.R., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 141, Reidel, Dordrecht, Holland.
- Burton, W.B. & Gordon, M.A., 1978. Astr. Astrophys., 63, 7.
- Burton, W.B., Deul, E.R., Walker, H.J. & Jongeneelen, A.A.W., 1986. In: Light on Dark Matter, Proceedings of the First IRAS Conference, ed. Israel, F.P., Reidel, Dordrecht, Holland.
- Burton, W.B., Gordon, M.A., Bania, T.M. & Lockman, F.J., 1975. Astrophys. J., 202, 30.
- Bushouse, H.A., 1986. PhD Thesis, University of Illinois, Urbana.

Byl, J., 1983. Moon & Planets, 29, 121.

- Byl, J., 1986. Earth, Moon & Planets, 36, 263.
- Caimmi, R., 1978. Astrophys. Space Sci., 59, 413.
- Caimmi, R. & Dallaporta, N., 1982. Astrophys. Space Sci., 84, 271.
- Caimmi, R. & Secco, L., 1986. Astrophys. Space Sci., 119, 315.
- Caldwell, J.A.R. & Ostriker, J.P., 1981. Astrophys. J., 251, 61.
- Cameron, L.M., 1985. Astr. Astrophys., 147, 47.

Campbell, A., Terlevich, R.J. & Melnick, J., 1986. Mon. Not. R. astr. Soc., 223, 811.

Carlberg, R.G., 1987. Astrophys. J., in press.

Carlberg, R.G., Dawson, P.C., Hsu, T. & Vandenberg, D.A., 1985. Astrophys. J., 294, 674.

- Carr, B.J. & Lacey, C.G., 1987. Astrophys. J., 316, 23.
- Cassé, M., Kunth, D. & Scalo, J.M., 1979. Astr. Astrophys., 76, 346.
- Castellani, V., Chieffi, A., Pulone, L. & Tornambè, A., 1985. Astrophys. J., 294, L31.
- Cayrel, R., 1986. Astr. Astrophys., 168, 81.
- Cesarsky, C.J. & Völk, H.J., 1978. Astr. Astrophys., 70, 367.
- Chamaraux, P., Balkowski, C. & Gerard, E., 1980. Astr. Astrophys., 83, 38.
- Chieffi, A. & Tornambè, A., 1984. Astrophys. J., 287, 745.
- Chini, R., Kreysa, E., Krugel, E. & Mezger, P.G., 1986. Astr. Astrophys., 166, L8.
- Chiosi, C., 1980. Astr. Astrophys., 83, 206.
- Chiosi, C. & Caimmi, R., 1979. Astr. Astrophys., 80, 234.
- Chiosi, C. & Jones, B.J.T., 1983. In: The Origin and Evolution of Galaxies, eds. Jones, B.J.T. & Jones, J.E., p. 197, Reidel, Dordrecht, Holland.
- Chiosi, C. & Matteucci, F., 1982. Astr. Astrophys., 105, 140.
- Chiosi, C. & Matteucci, F., 1983. In: Proc. of ESO Workshop on Primordial Helium, eds. Shaver, P.A., Kunth, D. & Kjär, K., p. 77, ESO Publications, Garching, F.R.G.
- Chiosi, C. & Matteucci, F., 1984. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 359, Reidel, Dordrecht, Holland.
- Christian, C.A. & Smith, H.A., 1983. Publs. astr. Soc. Pacif., 95, 169.
- Clayton, D.D., 1984. Astrophys. J., 285, 411.
- Clayton, D.D., 1985a. Astrophys. J., 288, 569.
- Clayton, D.D., 1985b. In: Nucleosynthesis, eds. Arnett, W.D. & Truran, J.W., p. 65, University of Chicago Press, Chicago.
- Clayton, D.D., 1986. Publs. astr. Soc. Pacif., 98, 968.
- Clayton, D.D., 1987. Astrophys. J., 315, 451.
- Clayton, D.D. & Pantelaki, I., 1986. Astrophys. J., 307, 441.
- Clayton, G.C. & Martin, P.G., 1984. In: Structure and Evolution of the Magellanic Clouds, IAU Symp. No. 108, eds. van den Bergh, S. & de Boer, K.S., p. 403, Reidel, Dordrecht, Holland.
- Clegg, R.E.S., Lambert, D.L. & Tomkin, J., 1981. Astrophys. J., 250, 262.
- Clemens, W.A., 1986. In: Dynamics of Extinction, ed. Elliot, D.K., p. 63, Wiley-Interscience, New York.
- Clemens, W.A., 1987. Private communication.
- Clube, S.V.M., 1978. Vistas Astr., 22, 77.
- Clube, S.V.M., 1987a. Private communication.
- Clube, S.V.M., 1987b. In: Similarity and Diversity of Comets, ESA.SP-278, in press.
- Clube, S.V.M., 1987c. Private communication.
- Clube, S.V.M. & Napier, W.M., 1982a. Q. J. R. astr. Soc., 23, 45.
- Clube, S.V.M. & Napier, W.M., 1982b. Earth Planet. Sci. Lett., 57, 251.
- Clube, S.V.M. & Napier, W.M., 1982c. The Cosmic Serpent, Faber, London.
- Clube, S.V.M. & Napier, W.M., 1984a. Mon. Not. R. astr. Soc., 208, 575.
- Clube, S.V.M. & Napier, W.M., 1984b. Mon. Not. R. astr. Soc., 211, 953.
- Clube, S.V.M. & Napier, W.M., 1984c. Nature, 311, 635.
- Clube, S.V.M. & Napier, W.M., 1985. Icarus, 62, 384.
- Clube, S.V.M. & Napier, W.M., 1986a. In: The Galaxy & the Solar System, eds Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 260, University of Arizona Press, Tucson.
- Clube, S.V.M. & Napier, W.M., 1986b. Interdisciplinary Sci. Rev., 11, 236.
- Clube, S.V.M. & Napier, W.M., 1987. Mon. Not. R. astr. Soc., 225, 55P.
- Cohen, R.S., Cong, H., Dame, T.M. & Thaddeus, P., 1980. Astrophys. J., 239, L53.
- Cohen, R.S., Thaddeus, P. & Bronfman, L., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, eds. van Woerden, H., Burton, W.B. & Allen, K.J., p. 199, Reidel, Dordrecht, Holland.
- Combes, F. & Gerin, M., 1987. Astr. Astrophys., in press.
- Comins, N.F., 1984. Astrophys. J., 284, 90.
- Coombes, F., Encrenaz, P.J., Lucas, R. & Weliachew, L., 1977. Astr. Astrophys., 61, L7.
- Coombes, F., Encrenaz, P.J., Lucas, R. & Weliachew, L., 1978. Astr. Astrophys., 67, L13.
- Cowie, L.L., 1981. Astrophys. J., 245, 66.
- Cowie, L.L. & Songaila, A., 1977. Nature, 266, 501.
- Cowie, L.L., Henriksen, M. & Mushotzky, R., 1987. Astrophys. J., 317, 593.
- Cowie, L.L., McKee, C.F. & Ostriker, J.P., 1981. Astrophys. J., 247, 908.

Cox, D.P. & Smith, B.W., 1976. Astrophys. J., 203, 361.

Cox, P., Krugel, E. & Mezger, P.G., 1986. Astr. Astrophys., 155, 380.

Crawford, A.R., 1985. Q. J. R. astr. Soc., 26, 53.

Crézé, M. & Isobe, S., 1974. A. Tokyo Astr. Obs., 14, 270.

Crutcher, R.M., Rogstad, D.E. & Chu, K., 1978. Astrophys. J., 225, 784.

D'Antona, F. & Mazzitelli, I., 1985. Astrophys. J., 296, 502.

D'Antona, F. & Mazzitelli, I., 1986. Astr. Astrophys., 162, 80.

Dame, T.M., 1984. PhD Thesis, Columbia University.

Dame, T.M. & Thaddeus, P., 1985. Astrophys. J., 297, 751.

Dame, T.M., Elmegreen, B.G., Cohen, R.S. & Thaddeus, P., 1986. Astrophys. J., 305, 892.

Dame, T.M., Ungerechts, H., Cohen, R.S., de Geus, E., Grenier, I.A., May, J., Murphy, D.C., Nyman, L-A. & Thaddeus, P., 1987. Astrophys. J., submitted.

Davies, R.D., 1974. In: Galactic Radioastronomy, eds. Kerr, F.J. & Simonsen, S.C., p. 599, Reidel, Dordrecht, Holland.

Davies, R.D. & Lewis, B.M., 1973. Mon. Not. R. astr. Soc., 165, 231.

Davis, M., Hut, P. & Muller, R.A., 1984. Nature, 308, 715.

de Jong, T., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 111, ESO, Garching.

de Jong, T., Clegg, P.E., Soifer, B.T., Rowan-Robinson, M., Habing, H., Houck, J.R., Aumann, H.H. & Raimond, E., 1984. Astrophys. J., 279, L67.

de Vaucouleurs, G. & Malik, G., 1980. Mon. Not. R. astr. Soc., 142, 387.

de Vaucouleurs, G., de Vaucouleurs, A. & Corwin, H.G., 1976. Second Reference Catalogue of Bright Galaxies, University of Texas Press, Austin. (RC2).

DeGoia-Eastwood, K., Grasdalen, G.L., Strom, S.E. & Strom, K.M., 1984. Astrophys. J., 278, 564.

Delsemme, A.H., 1977. In: Comets, Asteroids, Meteorites: Interrelations, Evolution and Origins, IAU Colloq. No. 39, ed. Delsemme, A.H., p. 453, University of Toledo Press, Toledo.

Delsemme, A.H., 1987. Astr. Astrophys., in press.

Despois, D. & Baudrey, A., 1983. In: Surveys of the Southern Galaxy, eds. Burton, W.B. & Israel, F.P., p. 173, Reidel, Dordrecht, Holland.

Devereux, N.A., Becklin, E.E. & Scoville, N.Z., 1987. Astrophys. J., 312, 529.

Diaz, A.I. & Tosi, M., 1984. Mon. Not. R. astr. Soc., 208, 365.

Diaz, A.I. & Tosi, M., 1986. Astr. Astrophys., 158, 60.

Dickman, R.L., 1975. Astrophys. J., 202, 50.

Dickman, R.L., 1978. Astrophys. J. Suppl., 37, 407.

Dickman, R.L., Snell, R.L. & Schloerb, F.P., 1986. Astrophys. J., 309, 326.

Dingus, L. & Sadler, P.M., 1982. Syst. Zool., 31, 400.

Donas, J. & Deharveng, J.M., 1984. Astr. Astrophys., 140, 325.

Donas, J., Deharveng, J.M., Laget, M., Milliard, B. & Huguenin, D., 1987. Astr. Astrophys., 180, 12.

Donovan, S.K., 1987. Nature, 328, 109.

Dopita, M.A., 1985. Astrophys. J., 295, L5.

Dopita, M.A. & Evans, I.N., 1986. Astrophys. J., 307, 431.

Dorman, F.H., 1968. J. Geol., 76, 297.

Dressler, A., 1980. Astrophys. J., 236, 351.

Dressler, A., 1986. Astrophys. J., 301, 35.

Dressler, A. & Gunn, J., 1983. Astrophys. J., 270, 7.

Dufour, R.J., 1984. In: Structure and Evolution of the Magellanic Clouds, IAU Symp. No. 108, eds. van den Bergh, S. & de Boer, K.S., p. 353, Reidel, Dordrecht, Holland.

Duley, W.W. & Williams, D.A., 1984. Interstellar Chemistry, Academic Press, London.

Dyson, J.E., 1987. Private communication.

Edmunds, M.G., 1984. In: Gas in the Interstellar Medium, ed. Gondhalekar, P., p. 84, Rutherford Appleton Laboratory.

Edmunds, M.G. & Pagel, B.E.J., 1984a. Mon. Not. R. astr. Soc., 211, 507.

Edmunds, M.G. & Pagel, B.E.J., 1984b. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 341, Reidel, Dordrecht, Holland.

Eggen, O.J., Lynden-Bell, D. & Sandage, A.R., 1962. Astrophys. J., 136, 748. (ELS).

Einsato, J., 1972. Astrophys. Letters, 11, 195.

Elmegreen, B.G., 1982. Astrophys. J., 253, 634.

Elmegreen, B.G., 1985. Astrophys. J., 299, 196.

Elmegreen, B.G., 1987a. In: Interstellar Processes, eds. Hollenbach, D. & Thronson, H., in press. Elmegreen, B.G., 1987b. In: Physical Processes in Interstellar Clouds, eds. Morfill, G.E. & Scholer,

M., Reidel, Dordrecht, Holland, in press.

Elmegreen, B.G., 1987c. In: Star Forming Regions, IAU Symp. No. 115, eds. Peimbert, M. & Jugaku, J., p. 457, Reidel, Dordrecht, Holland, in press.

Elmegreen, B.G., 1987d. Astrophys. J., 312, 626.

Elmegreen, B.G. & Elmegreen, D.M., 1986. Astrophys. J., 311, 554.

Elmegreen, B.G. & Elmegreen, D.M., 1987a. Astrophys. J., 320, in press.

Elmegreen, B.G. & Elmegreen, D.M., 1987b. Astrophys. J., in press.

Elmegreen, B.G., Elmegreen, D.M. & Morris, M., 1980. Astrophys. J., 240, 455.

Elson, R.A.W. & Fall, S.M., 1985. Astrophys. J., 299, 211.

Emerson, D.T., 1974. Mon. Not. R. astr. Soc., 169, 607.

Evans, I.N., 1986. Astrophys. J., 309, 544.

Faber, S.M. & Gallagher, J.S., 1979. Ann. Rev. Astr. Astrophys., 17, 135.

Fabricant, D. & Gorenstein, P., 1983. Astrophys. J., 267, 535.

Fall, S.M., 1987. Private communication.

Farouki, R. & Shapiro, S.L., 1980. Astrophys. J., 241, 928.

Faundez-Abans, M. & Maciel, W.J., 1986. Astr. Astrophys., 158, 228.

Fernández, J.A. & Ip, W.-H., 1983a. In: Asteroids, Comets, Meteors, eds. Lagerkvist, C.-I. & Rickman, p. 387, Uppsala Observatory, Uppsala.

Fernández, J.A. & Ip, W.-H., 1983b. Icarus, 54, 377.

Fernández, J.A. & Ip, W.-H., 1987. Icarus, in press.

Fich, M. & Blitz, L., 1982. In: Kinematics, Dynamics and Structure of the Milky Way, ed. Shuter, W.L., p. 151, Reidel, Dordrecht, Holland.

Field, G.B., 1978. In: Protostars and Planets, ed. Gehrels, T., p. 153, University of Arizona Press, Tucson.

Field, G.B. & Saslaw, W., 1965. Astrophys. J., 142, 568.

Fischer, A.G. & Arthur, M.A., 1977. In: Deep Water Carbonate Environments, eds. Cook, H.E.
& Enos, P., p. 19-50, Society of Palaeontologists and Mineralogists Special Publication 25.

Fisher, J.R. & Tully, R.B., 1981. Astrophys. J. Suppl., 47, 139.

Forman, W., Schwarz, J., Jones, C., Liller, W. & Fabian, A.C., 1979. Astrophys. J., 234, L27.

Fowler, W.A., 1972. In: Cosmology, Fusion and Other Matters, George Gamov Memorial Volume, ed. Reines, F., p. 67, University of Colorado Press.

Franco, J. & Cox, D.P., 1986. Publs. astr. Soc. Pacif., 98, 1076.

Freedman, W.L., 1986. In: Spectral Evolution of Galaxies, eds. Renzini, A. & Chiosi, C., p. 183, Reidel, Dordrecht, Holland.

Freedman, W.L. & Madore, B.F., 1984. Astrophys. J., 280, 592.

Frerking, M.A., Langer, W.D. & Wilson, R.W., 1982. Astrophys. J., 262, 590.

Ftaclas, C., Fanelli, M.N. & Struble, M.F., 1984. Astrophys. J., 282, 19.

Fuchs, B. & Wielen, R., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 375, Reidel, Dordrecht, Holland.

Fujimoto, M.Y., Iben, I. Jr., Chieffi, A. & Tornambè, A., 1984. Astrophys. J., 287, 749.

Gallagher, J.S., Hunter, D.A. & Tutukov, A.V., 1984. Astrophys. J., 284, 544.

Ganapathy, R., 1982. Science, 216, 885.

Garay, G. & Rodriguez, L.F., 1983. Astrophys. J., 266, 263.

Garmany, C.D., Conti, P.S. & Chiosi, C., 1982. Astrophys. J., 263, 777.

Garnett, D.R. & Shields, G.A., 1987. Astrophys. J., 317, 82.

Gerola, H. & Seiden, P.E., 1978. Astrophys. J., 223, 129.

Gilmore, G., 1984. Mon. Not. R. astr. Soc., 207, 223.

Gilmore, G. & Reid, I.N., 1983. Mon. Not. R. astr. Soc., 202, 1025.

Gilmore, G. & Wyse, R.F.G., 1985. Astr. J., 90, 2015.

Gilmore, G. & Wyse, R.F.G., 1986. Nature, 322, 806.

Gilmore, G. & Wyse, R.F.G., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 247, Reidel, Dordrecht, Holland.

Gilmore, G., Reid, I.N. & Hewitt, P.C., 1985. Mon. Not. R. astr. Soc., 213, 257.

Giovanardi, C., Helou, G., Krumm, N. & Salpeter, E.E., 1983. Astrophys. J., 267, 35.

Giovanelli, R. & Haynes, M.P., 1983. Astr. J., 88, 881.

Giovanelli, R. & Haynes, M.P., 1985. Astrophys. J., 292, 404.

Giraud, E., 1986. Astr. Astrophys., 85, 623.

Gisler, G.R., 1976. Astr. Astrophys., 51, 137.

Goldreich, P. & Kwan, J., 1974. Astrophys. J., 189, 441.

- Goldreich, P. & Lynden-Bell, D., 1965. Mon. Not. R. astr. Soc., 130, 97.
- Gondhalekar, P.M., Morgan, D.H., Dopita, M. & Ellis, R.S., 1986. Mon. Not. R. astr. Soc., 219, 505.
- Gordon, M.A. & Burton, W.B., 1976. Astrophys. J., 208, 346.
- Gorenstein, P., Fabricant, D., Topka, K., Tucker, W. & Harnden, F.R. Jr., 1977. Astrophys. J., 216, L95.
- Gott, J.R., 1981. Astrophys. J., 243, 140.
- Gottesman, S.T. & Davies, R.D., 1970. Mon. Not. R. astr. Soc., 149, 263.
- Grabelsky, D.A., Cohen, R.S., May, J., Bronfman, L. & Thaddeus, P., 1987. Astrophys. J., submitted.
- Gratton, R.G., 1985. Astr. Astrophys., 148, 105.
- Greggio, L. & Tosi, M., 1986. Astr. Astrophys., 156, L1.
- Grieve, R.A.F., 1982. Geol. Soc. Amer. Spec. Pap., 190,25.
- Grieve, R.A.F. & Dence, M.R., 1979. Icarus, 38, 230.
- Grieve, R.A.F., Sharpton, V.L., Goodacre, A.K. & Garvin, J.B., 1986. Earth Planet. Sci. Lett., 76, 1.
- Grindlay, J.E. & Hertz, P., 1984. In: Proceedings of the 7th Workshop on Cataclysmic Variables, eds. Lamb, D. & Patterson, J., Reidel, Dordrecht, Holland.
- Guibert, J., Lequeux, J. & Viallefond, F., 1978. Astr. Astrophys., 68, 1.
- Guiderdoni, B. & Rocca-Volmerange, B., 1985. Astr. Astrophys., 151, 108.
- Guiderdoni, B., 1987. Astr. Astrophys., 172, 27.
- Gunn, J.E. & Gott, J.R., 1972. Astrophys. J., 276, 1.
- Güsten, R., 1986. In: Spectral Evolution of Galaxies, eds. Chiosi, C. & Renzini, A., p. 449, Reidel, Dordrecht, Holland.
- Güsten, R. & Mezger, P.G., 1982. Vistas Astr., 26, 159.
- Güsten, R. & Ungerechts, H., 1985. Astr. Astrophys., 145, 241.
- Guth, A.H., 1981. Phys. Rev., D23, 347.
- Hamajima, K. & Tosa, M., 1975. Publs. Astr. Soc. Japan, 27, 561.
- Harris, H.C., 1981. Astr. J., 86, 707.
- Harris, H.C. & Pilachowski, C.A., 1984. Astrophys. J., 282, 655.
- Hartwick, F.D.A., 1971. Astrophys. J., 163, 431.
- Hartwick, F.D.A., 1976. Astrophys. J., 209, 418.
- Hartwick, F.D.A., 1983. Mem. Soc. astr. It., 54, 51.
- Hatfield, C.B. & Camp, M.J., 1970. Geol. Soc. America Bull., 81, 911.
- Hawkins, M.R.S., 1986. Mon. Not. R. astr. Soc., 228, 845.
- Hawley, S.A., 1978. Astrophys. J., 224, 417.
- Haynes, M.P., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 45, ESO, Garching.
- Haynes, M.P. & Giovanelli, R., 1984. Astr. J., 89, 758.
- Haynes, M.P. & Giovanelli, R., 1986. Astrophys. J., 306, 466.
- Haynes, M.P., Giovanelli, R. & Chincarini, G.L., 1984. Ann. Rev. Astr. Astrophys., 22, 445.
- Hegyi, D.J. & Olive, K.A., 1986. Astrophys. J., 303, 56.
- Hegyi, D.J., Kolb, E.W. & Olive, K.A., 1986. Astrophys. J., 300, 492.
- Heisler, J. & Tremaine, S.D., 1986. Icarus, 65, 13.
- Heissig, K., 1986. Modern Geol., 10, 171.
- Helou, G., Hoffman, G.L. & Salpeter, E.E., 1984. Astrophys. J. Suppl., 55, 433.
- Henderson, A.P., Jackson, P.D. & Kerr, R.J., 1982. Astrophys. J., 263, 116.
- Henriksen, R.N. & Turner, B.E., 1984. Astrophys. J., 287, 200.
- Hermsen, W. & Bloemen, J.B.G.M., 1983. In: Surveys of the Southern Galaxy, eds. Burton, W.B. & Israel, F.P., p. 65, Reidel, Dordrecht, Holland.
- Hewitt, J.N., Haynes, M.P. & Giovanelli, R., 1983. Astr. J., 88, 272.
- Hildebrand, R.H., 1983. Q. J. R. astr. Soc., 24, 267.
- Hill, G., Hilditch, R.W. & Barnes, J.V., 1979. Mon. Not. R. astr. Soc., 186, 813.
- Hillebrandt, W., Nomoto, K. & Wolff, G., 1985. Astr. Astrophys., 133, 175.
- Hills, J.G., 1977. Astr. J., 82, 626.
- Hills, J.G., 1981. Astr. J., 86, 1730.
- Hills, J.G., 1984. Nature, 311, 636.
- Hills, J.G., 1985. Astr. J., 90, 1876.
- Hills, J.G., 1986. Astr. J., 92, 595.

Hoffmann, A., 1985. Nature, 315, 659.

- Boffmann, A., 1986. Nature, 321, 535.
- Holland, H.D. & Trendall, A.F. (eds), 1984. Patterns of Change in Earth Evolution, Report of the Dahlem Workshop on Patterns of Change in Earth Evolution, Springer-Verlag, Berlin.
- Hollenbach, D. & Salpeter, E.E., 1971. Astrophys. J., 163, 155.
- Holmes, A., 1927. The Age of the Earth, Benn, London.
- Houston, B.P. & Wolfendale, A.W., 1985. J. Phys. G, 11, 407.
- Hoyle, F. & Wickramasinghe, N.C., 1978. Astrophys. Space Sci., 53, 523.
- Hoyle, F., 1981. Ice, Hutchinson, London.
- Hsu, K.J., 1980. Nature, 285, 201.
- Huang, Y-L. & Thaddeus, P., 1986. Astrophys. J., 309, 804.
- Huchra, J., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 181, ESO, Garching.
- Huchtmeier, W.K., 1982. Astr. Astrophys., 110, 121.
- Huchtmeier, W.K., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p.23, ESO, Garching.
- Humphreys, R.M. & McElroy, D.B., 1984. Astrophys. J., 284, 565.
- Hunt, R., 1971. Mon. Not. R. astr. Soc., 154, 141.
- Hunt, R., 1975. Mon. Not. R. astr. Soc., 173, 465.
- Hunter, D.A. & Gallagher, J.S., 1986. Publs. astr. Soc. Pacif., 98, 5.
- Hunter, D.A., Gallagher, J.S. & Rautenkranz, D.J., 1982. Astrophys. J. Suppl., 49, 53.
- Hut, P. & Tremaine, S.D., 1985. Astr. J., 90, 1548.
- Hut, P., 1984. Nature, 311, 638.
- Iben, I. Jr., 1980. In: Effects of mass loss on stellar evolution, IAU Coll. No. 59, eds. Chiosi, C. & Stalio, R., Reidel, Dordrecht, Holland.
- Iben, I. Jr., 1983. Mem. Soc. astr. It., 54, 321.
- Iben, I. Jr., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 365, Reidel, Dordrecht, Holland.
- Iben, I. Jr. & Renzini, A., 1983. Ann. Rev. Astr. Astrophys., 21, 271.
- Icke, V., 1985. Astr. Astrophys., 114, 115.
- Innanen, K.A., Patrick, A.T. & Duley, W.W., 1978. Astrophys. Space Sci., 57, 511.
- Ipser, J.R. & Semenzato, R., 1985. Astr. Astrophys., 149, 408.
- Israel, F.P. & Burton, W.B., 1986. Astr. Astrophys., 168, 369.
- Israel, F.P., 1984. In: Structure and Evolution of the Magellanic Clouds, IAU Symp. No. 108, eds. van den Bergh, S. & de Boer, K.S., p. 319, Reidel, Dordrecht, Holland.
- Israel, F.P., de Graauw, T., van de Stadt, H. & de Vries, C.P., 1986. Astrophys. J., 303, 186.
- Issa, M.R. & Wolfendale, A.W., 1981. Nature, 292, 430.
- Ito, K., Matsumoto, T. & Uyama, K., 1977. Nature, 265, 515.
- Jablonski, D., 1986. In: Dynamics of Extinction, ed. Elliot, D.K., p. 183, Wiley-Interscience, New York.
- Jacobs, J.A., 1986. Nature, 323, 296.
- Janes, K.A., 1979. Astrophys. J. Suppl., 39, 135.
- Janes, K.A. & Demarque, P., 1983. Astrophys. J., 264, 206.
- Jensen, E.B., Strom, K.M. & Strom, S.E., 1976. Astrophys. J., 209, 748.
- Jog, C.J. & Ostriker, J.P., 1987. Astrophys. J., submitted.
- Jura, M., 1975. Astrophys. J., 197, 581.
- Jura, M., 1976. Astr. J., 81, 178.
- Jura, M., 1985. Preprint.
- Kaelbe, A., de Boer, K.S. & Grewing, M., 1985. Astr. Astrophys., 143, 408.
- Kahn, F.D., 1974. Astr. Astrophys., 37, 149.
- Karachentseva, V.E., 1973. Comm. Spec. Ap. Obs., Vol. 8.
- Keller, G., D'Hardi, S. & Vallier, T.L., 1983. Science, 221, 150.
- Kenney, J. & Young, J.S., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 165, ESO, Garching.
- Kenney, J. & Young, J.S., 1986. Astrophys. J., 301, L13. (KY).
- Kennicutt, R.C., 1983a. Astr. J., 88, 483.
- Kennicutt, R.C., 1983b. Astrophys. J., 272, 54.
- Kennicutt, R.C., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 227, ESO, Garching.
- Kennicutt, R.C. & Kent, S.M., 1983. Astr. J., 88, 1094.

Kennicutt, R.C., Keel, W.C., van der Hulst, J.M., Hummel, E. & Roettiger, K.A., 1987. Astrophys. J., submitted.

Kent, S.M., 1980. PhD Thesis, California Institute of Technology.

Kerr, R.A., 1985. Science, 227, 1451.

- King, D.J., 1986. Private communication.
- Kinman, T.D. & Davidson, K., 1981. Astrophys. J., 243, 127.
- Kitchell, J.A. & Pena, D., 1984. Science, 226, 689.
- Knapp, G.R. & Jura, M., 1976. Astrophys. J., 209, 699.
- Knapp, G.R., Helou, G. & Stark, A.A., 1987. Astr. J., 94, 54.
- Knapp, G.R., Kuiper, T.B.H., Knapp, S.L. & Brown, R.L., 1977. Astrophys. J., 214, 78.
- Koester, D. & Reimers, D., 1985. Astr. Astrophys., 153, 260.
- Koornneef, J., 1982. Astr. Astrophys., 107, 247.
- Koornneef, J., 1984. In: Structure and Evolution of the Magellanic Clouds, IAU Symp. No. 108, eds. van den Bergh, S. & de Boer, K.S., p. 333, Reidel, Dordrecht, Holland.
- Kresák, L., 1977. In: Comets, Asteroids, Meteorites: Interrelations, Evolution and Origins, IAU

Colloq. No. 39, ed. Delsemme, A.H., p. 93, University of Toledo Press, Toledo.

- Kritsuk, A.G., 1983. Astrofizika, 19, 263.
- Kulkarni, S.R., 1987. In: Interstellar Processes, eds. Hollenbach, D. & Thronson, H., in press.
- Kulkarni, S.R., Blitz, L. & Heiles, C., 1982. Astrophys. J., 259, L63.
- Kutner, M.L. & Leung, C.M., 1985. Astrophys. J., 291, 188.
- Kutner, M.L. & Mead, K.N., 1981. Astrophys. J., 249, L15.
- Kwan, J., 1979. Astrophys. J., 229, 567.
- Kwan, J. & Valdes, F., 1983. Astrophys. J., 271, 604.
- Kwan, J. & Valdes, F., 1987. Astrophys. J., 315, 92.
- Lacey, C.G., 1984. Mon. Not. R. astr. Soc., 208, 687.
- Lacey, C.G. & Fall, S.M., 1983. Mon. Not. R. astr. Soc., 204, 791.
- Lacey, C.G. & Fall, S.M., 1985. Astrophys. J., 290, 154.
- Lacey, C.G. & Ostriker, J.P., 1985. Astrophys. J., 299, 633.
- Laird, J., 1985. Astrophys. J., 289, 556.
- Langer, W.D., 1977. Astrophys. J., 212, L39.
- Larson, R.B., 1972. Nature, 236, 21.
- Larson, R.B., 1976. Mon. Not. R. astr. Soc., 176, 31.
- Larson, R.B., 1977. In: Evolution of Galaxies and Stellar Populations, eds. Tinsley, B.M. & Larson, R.B., p. 97, Yale University Observatory, New Haven.
- Larson, R.B., 1981. Mon. Not. R. astr. Soc., 194, 809.
- Larson, R.B., 1986. Mon. Not. R. astr. Soc., 218, 409.
- Larson, R.B. & Tinsley, B.M., 1978. Astrophys. J., 219, 46.
- Larson, R.B., Tinsley, B.M. & Caldwell, C.N., 1980. Astrophys. J., 237, 692.
- Lea, S.M. & de Young, D.S., 1976. Astrophys. J., 210, 647.
- Lebrun, F. & Huang, Y.-L., 1984. Astrophys. J., 281, 634.
- Lebrun, F., Bennett, K., Bignami, G.F., Bloemen, J.B.G.M., Buccheri, R., Caraveo, P.A., Gottwald, M., Hermsen, W., Kanbach, G., Mayer-Hasselwander, H.A., Montmerle, T., Paul, J.A., Sacco, B., Strong, A.W. & Willis, R.D., 1983. Astrophys. J., 274, 231.
- Leger, A., 1983. Astr. Astrophys., 123, 271.
- Leggett, J.K., McKerrow, W.S., Cocks, L.R.M. & Rickard, R.B., 1981. J. Geol. Soc., 138, 167.
- Leggett, S.K., Brand, P.W.J.L. & Mountain, C.A., 1987b. Mon. Not. R. astr. Soc., in press.
- Leggett, S.K., Clowes, R.G., Kalafi, M., MacGillivray, H.T., Puxley, P.J., Savage, A. & Wolstencroft, R.D., 1987a. Mon. Not. R. astr. Soc., 227, 563.
- Lequeux, J., 1984. In: Structure and Evolution of the Magellanic Clouds, IAU Symp. No. 108, eds. van den Bergh, S. & de Boer, K.S., p. 67, Reidel, Dordrecht, Holland.
- Lequeux, J., Martin, N., Prévot, L., Prévot-Burnichon, M-L., Rebeirot, E. & Rousseau, J., 1980. Astr. Astrophys., 85, 305.
- Lequeux, J., Peimbert, M., Rayo, J.F., Serrano, A. & Torres-Peimbert, S., 1979. Astr. Astrophys., 80, 155.
- Lester, D.F., Dinnerstein, H.L., Werner, M.W., Harvey, P.M., Evans, N.L. & Brown, R.L., 1985. Astrophys. J., 296, 565.
- Li, T.P., Riley, P.A. & Wolfendale, A.W., 1982. J. Phys. G, 8, 141.
- Li, T.P., Riley, P.A. & Wolfendale, A.W., 1983. Mon. Not. R. astr. Soc., 203, 87.
- Liebert, J. & Dahn, C.C., 1986. In: Astrophysics of Brown Dwarfs, eds. Kafatos, M.C., Harrington, R.S. & Maran, S.P., Cambridge University Press.

Liebert, J., Dahn, C.C., Gresham, M. & Strittmatter, P.A., 1979. Astrophys. J., 233, 226.

- Liszt, H.S., 1973. PhD Thesis, Princeton University.
- Liszt, H.S., 1982. Astrophys. J., 262, 198.
- Liszt, H.S., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, eds. Van Woerden, H., Allen, R.J. & Burton, W.B., p. 283, Reidel, Dordrecht, Holland.
- Liszt, H.S., Burton, W.B. & Xiang, D., 1984. Astr. Astrophys., 140, 303.
- Liszt, H.S., Xiang, D. & Burton, W.B., 1981. Astrophys. J., 249, 532.
- Lo, K.Y., Ball, R., Masson, C.R., Phillips, T.G., Scott, S. & Woody, D.P., 1987b. Astrophys. J., submitted.
- Lo, K.Y., Cheung, K.W., Masson, T.G., Phillips, T.G., Scott, S.L. & Woody, D.P., 1987a. Astrophys. J., 312, 574.
- Lockman, F.J., 1976. Astrophys. J., 209, 429.
- Lonsdale, C.J., Helou, G., Good, J.C. & Rice, W., 1985. Catalogued Galaxies and Quasars Observed in the IRAS Survey, Jet Propulsion Laboratory.
- Lonsdale-Persson, C.J. & Helou, G., 1987. Astrophys. J., 314, 513.
- Low, C. & Lynden-Bell, D., 1976. Mon. Not. R. astr. Soc., 176, 367.
- Luck, J.M. & Turekian, K.K., 1983. Science, 222, 613.
- Luck, R.E., 1982. Astrophys. J., 256, 177.
- Lutz, T.M., 1985. Nature, 317, 404.
- Lynden-Bell, D., 1975. Vistas Astr., 19, 299.
- Lyne, A.G., Manchester, R.N. & Taylor, J.H., 1985. Mon. Not. R. astr. Soc., 213, 613.
- Lynga, G., 1982. Astr. Astrophys., 109, 213.
- Lyngå, G., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, eds. Van Woerden, H., Allen, R.J. & Burton, W.B., p. 133, Reidel, Dordrecht, Holland.
- Maddox, J., 1985. Nature, 316, 761.
- Madore, B.F., 1977. Mon. Not. R. astr. Soc., 178, 1.
- Madore, B.F., van den Bergh, S. & Rogstad, D., 1974. Astrophys. J., 191, 317.
- Maeder, A., 1980. In: Effects of mass loss on stellar evolution, IAU Coll. No. 59, eds. Chiosi, C. & Stalio, R., Reidel, Dordrecht, Holland.
- Maeder, A., 1981. Astr. Astrophys., 101, 385.
- Maeder, A., 1984. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 115, Reidel, Dordrecht, Holland.
- Mallik, D.C.V., 1980. Astrophys. Space Sci., 69, 133.
- Mallik, D.C.V., 1981. J. Astrophys. Astr., 2, 171.
- Mallik, D.C.V. & Mallik, S.V., 1985. J. Astrophys. Astr., 6, 113.
- Maloney, P.R., 1987. In preparation.
- Maloney, P.R. & Black, J.H., 1987. Astrophys. J., submitted.
- Marsakov, V.A. & Suchkov, A.A., 1982. Astrofizika, 18, 367.
- Mathis, J.S., 1959. Astrophys. J., 129, 259.
- Mathis, J.S., 1986. Publs. astr. Soc. Pacif., 98, 995.
- Matteucci, F., 1986. Astrophys. J., 305, L81.
- Matteucci, F. & Chiosi, C., 1983. Astr. Astrophys., 123, 121.
- Matteucci, F. & Greggio, L., 1986. Astr. Astrophys., 154, 279.
- Matteucci, F. & Tornambè, A., 1985. Astr. Astrophys., 142, 13.
- Matteucci, F. & Tosi, M., 1985. Mon. Not. R. astr. Soc., 217, 391.
- Matthews, W.G. & Baker, J.C., 1971. Astrophys. J., 170, 241.
- Mayer, C.J., 1986. In: Cosmical Gas Dynamics, ed. Kahn, F.D., p. 21, VNU Science Press, Utrecht, The Netherlands.
- Mayer, C.J., Richardson, K.M., Rogers, M.J., Szabelski, J. & Wolfendale, A.W., 1987. Astr. Astrophys., 180, 73.
- Mayer-Hasselwander, H.A., 1983. Space Sci. Rev., 36, 223.
- Mayor, M., 1976. Astr. Astrophys., 48, 301.
- Mayor, M. & Martinet, L., 1977. Astr. Astrophys., 55, 221.
- Mayor, M. & Vigroux, L., 1981. Astr. Astrophys., 98, 1.
- Mazurek, T.J., 1980. Astr. Astrophys., 90, 65.
- McCall, M.L., 1982. PhD Thesis, University of Texas, Austin, Texas.
- McCall, M.L., 1986. Publs. astr. Soc. Pacif., 98, 992.
- McCall, M.L. & Schmidt, F.H., 1986. In: Highlights of Astronomy, ed. Swings, J.P., p. 585, Reidel, Dordrecht, Holland.
- McCall, M.L., Rybski, P.M. & Shields, G.A., 1985. Astrophys. J. Suppl., 57, 1.

McClaren, D.J., 1983. Geol. Soc. America Bull., 94, 313.

McClaren, D.J., 1986. In: Dynamics of Extinction, ed. Elliot, D.K., p. 37, Wiley-Interscience, New York.

McClure, R.D., Vandenberg, D.A., Smith, G.H., Fahlman, G.G., Richer, H.B., Hesser, J.E., Harris, W.E., Stetson, P.B. & Bell, R.A., 1986. Astrophys. J., 307, L49.

- McCrea, W.H., 1981. Proc. R. Soc. Lond., A375, 1.
- McDowell, J., 1985. Mon. Not. R. astr. Soc., 217, 77.

Melnick, J. & Terlevich, R.J., 1986. Observatory, 106, 69.

Meusinger, H., 1985. Astr. Nachr., 306, 123.

Meyer, B.S. & Schramm, D.N., 1986. Astrophys. J., 311, 406.

- Mezger, P.G., 1978. Astr. Astrophys., 70, 565.
- Mezger, P.G., Pankonin, V., Schmid-Burgk, J., Thum, C. & Wink, J., 1979. Astr. Astrophys., 80, L3.

Mihalas, D. & Binney, J., 1981. Galactic Astronomy, Freeman & Co., San Francisco.

- Miller, G.E. & Scalo, J.M., 1979. Astrophys. J. Suppl., 41, 513.
- Mirabel, I.F. & Morras, R., 1984. Astrophys. J., 279, 86.
- Monnet, G. & Simien, F., 1977. Astr. Astrophys., 56, 173.
- Morris, D.E. & Muller, R.A., 1986. Icarus, 65, 1.
- Morris, M. & Lo, K.Y., 1978. Astrophys. J., 223, 803.
- Mould, J.R., 1978. Astrophys. J., 226, 923.
- Mould, J.R., 1984. Publs. astr. Soc. Pacif., 96, 773.
- Mouschovias, T. Ch., 1976a. Astrophys. J., 206, 753.
- Mouschovias, T. Ch., 1976b. Astrophys. J., 207, 141.
- Mouschovias, T.Ch., Shu, F.H. & Woodward, P., 1974. Astr. Astrophys., 33, 73.
- Muller, R.A., Hut, P., Davis, M. & Alvarez, W., 1984. Nature, 312, 381.
- Myers, P.C., 1985. In: Protostars & Planets II, eds. Black, D.C. & Matthews, M.S., University of Arizona Press, Tucson.
- Myers, P.C., Dame, T.M., Thaddeus, P., Cohen, R.S., Siverberg, R.F., Dwek, E. & Hauser, M.G., 1986. Astrophys. J., 301, 398.
- Napier, W.M., 1986. Private communication.
- Napier, W.M. & Clube, S.V.M., 1979. Nature, 282, 455.
- Napier, W.M. & Staniucha, M., 1982. Mon. Not. R. astr. Soc., 198, 723.
- Negi, J.G. & Tiwari, R.K., 1983. Geophys. Res. Lett., 10, 713.
- Newton, K., 1980a. Mon. Not. R. astr. Soc., 190, 689.
- Newton, K., 1980b. Mon. Not. R. astr. Soc., 191, 169.
- Nissen, P.E., Edvardsson, B. & Gustafsson, B., 1985. In: Proceedings of the ESO Workshop on the Production and Evolution of C,N,O Elements, eds. Danziger, I.J., Matteucci, F. & Kjär, K., ESO, Garching.
- Nomoto, K., 1984a. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 205, Reidel, Dordrecht, Holland.
- Nomoto, K., 1984b. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 239, Reidel, Dordrecht, Holland.
- Nomoto, K., 1984c. Astrophys. J., 277, 791.
- Norman, C.A. & Silk, J., 1980. In: Interstellar Molecules, ed. Andrew, R.H., p. 137, Reidel, Dordrecht, Holland.
- Norman, D., 1986. Private communication.
- Norris, J., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 297, Reidel, Dordrecht, Holland.
- Norris, J., Bessel, M.S. & Pickles, A.J., 1985. Astrophys. J. Suppl., 58, 463.
- Nulsen, P.E.J., 1982. Mon. Not. R. astr. Soc., 198, 1007.
- O'Donnel, E. & Watson, W.D., 1974. Astrophys. J., 191, 89.

Officer, C.B., Hallam, A., Drake, C.L. & Devine, J.D., 1987. Nature, 326, 143.

- Olive, K.A., 1986. Astrophys. J., 309, 210.
- Olive, K.A., Thielemann, F.-K. & Truran, J.W., 1987. Astrophys. J., 313, 813.
- Olsson-Steel, D., 1987. Mon. Not. R. astr. Soc., 227, 501.
- Oort, J.H., 1950. Bull. astr. Inst. Neth., 11, 91.
- Oort, J.H., 1970. Astr. Astrophys., 7, 381.
- Opik, E.J., 1958. Irish Astr. J., 5, 34.
- Öpik, E.J., 1963. Adv. Astr. Astrophys., 2, 219.

Osborne, J.L., Parkinson, M.L., Richardson, K.M. & Wolfendale, A.W., 1987. In: Physical Processes in Interstellar Clouds, eds. Morfill, G.E. & Scholer, M., Reidel, Dordrecht, Holland, in press.

Ostriker, J.P. & Thuan, T.N., 1975. Astrophys. J., 202, 353.

- Ostriker, J.P., Peebles, P.J.E. & Yahil, A., 1974. Astrophys. J., 193, L1.
- Padian, K. & Clemens, W.A., 1985. In: Phanerozoic Diversity Patterns, ed. Valentine, J.W., p. 41, Princeton University Press, Princeton.
- Pagel, B.E.J., 1981. In: Structure & Evolution of Normal Galaxies, eds. Lynden-Bell, D. & Fall, S.M., p. 211, Cambridge University Press.
- Pagel, B.E.J., 1986a. In: Cosmogonical Processes: Papers dedicated to A.G.W. Cameron, eds. Arnett, W.D. et al., VNU Science Press, Utrecht, Holland.
- Pagel, B.E.J., 1986b. Private communication.
- Pagel, B.E.J., 1986c. Publs. astr. Soc. Pacif., 98, 1009.
- Pagel, B.E.J., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 341, Reidel, Dordrecht, Holland.
- Pagel, B.E.J. & Edmunds, M.G., 1981. Ann. Rev. Astr. Astrophys., 19, 77.
- Pagel, B.E.J. & Patchett, B.E., 1975. Mon. Not. R. astr. Soc., 172, 13.
- Pagel, B.E.J., Edmunds, M.G., Fosbury, R.A.E. & Webster, B.L., 1978. Mon. Not. R. astr. Soc., 184, 569.
- Pagel, B.E.J., Terlevich, R.J. & Melnick, J., 1986. Publs. astr. Soc. Pacif., 98, 1005.
- Pajot, F., Boisse, P., Gispert, R., Lamare, J.M., Puget, J.L. & Serra, G., 1985. Astr. Astrophys., 157, 393.
- Pal, P.C. & Creer, K.M., 1986. Nature, 320, 148.
- Palla, F., Salpeter, E.E. & Stahler, S.W., 1983. Astrophys. J., 271, 632.
- Panagia, N. & Tosi, M., 1980. Astr. Astrophys., 81, 375.
- Panagia, N. & Tosi, M., 1981. Astr. Astrophys., 96, 306.
- Pandey, O.P. & Negi, J.G., 1987. Geophys. J. R. astr. Soc., 89, 857.
- Parkinson, M.L., Richardson, K.M. & Wolfendale, A.W., 1987. Q. J. R. astr. Soc., in press.
- Patterson, C. & Smith, A.B., 1986. Preprint.
- Peimbert, M., 1978. In: Planetary Nebulae: Observations and Theory, IAU Symp. No. 76, ed. Terzian, Y., p. 215, Reidel, Dordrecht, Holland.
- Peimbert, M., 1979. In: The Large Scale Characteristics of the Galaxy, IAU Symp. No. 84, p. 307, ed. Burton, W.B., Reidel, Dordrecht, Holland.
- Peimbert, M., 1984. In: Gas in the Interstellar Medium, ed. Gondhalekar, P.M., p. 50, Rutherford Appleton Laboratory.
- Peimbert, M., 1985. In: Star Forming Dwarf Galaxies, eds. Kunth, D., Thuan, T.X. & Tran Thanh Van, J., p. 403, Frontieres, Paris.
- Peimbert, M., 1986. Publs. astr. Soc. Pacif., 98, 1057.
- Peimbert, M. & Serrano, A., 1980. Rev. Mexicana Astr. Astrofis., 5, 9.
- Peimbert, M. & Serrano, A., 1982. Mon. Not. R. astr. Soc., 198, 563.
- Peimbert, M. & Torres-Peimbert, S., 1974. Astrophys. J., 193, 327.
- Peimbert, M. & Torres-Peimbert, S., 1979. Paper presented at the XVIIth IAU General Assembly, Montreal, Canada.
- Peimbert, M., Pena, M. & Torres-Peimbert, S., 1986. Astr. Astrophys., 158, 266.
- Peimbert, M., Torres-Peimbert, S. & Rayo, J.F., 1978. Astrophys. J., 220, 516.
- Pérault, M., Boulanger, F., Falgarone, E. & Puget, J.L., 1987. In: Star Formation in Galaxies, ed. Persson, C.J., Government Printing Office, Washington, in press.
- Peterson, B.M., Strom, S.E. & Strom, K.M., 1979. Astr. J., 84, 735.
- Petrou, M., 1981. Mon. Not. R. astr. Soc., 196, 933.
- Phillips, S., 1987. Preprint.
- Phillips, S. & Disney, M., 1986. Mon. Not. R. astr. Soc., 221, 1039.
- Pilachowski, C., 1984. Astrophys. J., 281, 614.
- Playford, P.E., McClaren, D.J., Orth, C.J., Gilmore, J.S., Goodfellow, W.D., 1984. Science, 226, 1161.
- Prantzos, N., Doom, C., Arnould, M. & de Loore, C., 1986. Astrophys. J., 304, 695.
- Pudritz, R.E., 1986. Publs. astr. Soc. Pacif., 98, 709.
- Quirk, W.J. & Tinsley, B.M., 1973. Astrophys. J., 179, 69.
- Rampino, M.R., 1987. Nature, 327, 468.
- Rampino, M.R. & Stothers, R.B., 1984a. Nature, 308, 709.
- Rampino, M.R. & Stothers, R.B., 1984b. Science, 226, 1427.

Rampino, M.R. & Stothers, R.B., 1986. In: The Galaxy & the Solar System, eds Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 241, University of Arizona Press, Tucson.

- Rana, N.C., 1987a. Astr. Astrophys., in press.
- Rana, N.C., 1987b. Astrophys. J., submitted.
- Rana, N.C. & Wilkinson, D.A., 1986a. Mon. Not. R. astr. Soc., 218, 497. (Paper I).
- Rana, N.C. & Wilkinson, D.A., 1986b. Mon. Not. R. astr. Soc., 216, 721. (Paper II).
- Rana, N.C. & Wilkinson, D.A., 1986c. Il Nuovo Cimento, 9C, 714. (Paper III).
- Rana, N.C. & Wilkinson, D.A., 1987a. Mon. Not. R. astr. Soc., 226, 395. (Paper IV).
- Rana, N.C. & Wilkinson, D.A., 1987b. In: Astrochemistry, IAU Symp. No. 120, eds. Vardya,
 M.S. & Tarafdar, S.P., p. 323, Reidel, Dordrecht, Holland. (Paper V).
- Rana, N.C. & Wilkinson, D.A., 1987c. Mon. Not. R. astr. Soc., submitted. (Paper VII).
- Ratcliff, S.J., 1987. Astrophys. J., 318, 196.
- Raup, D.M., 1985a. Nature, 314, 341.
- Raup, D.M., 1985b. Nature, 317, 384.
- Raup, D.M. & Sepkoski, J.J., 1982. Science, 215, 1051.
- Raup, D.M. & Sepkoski, J.J., 1984. Proc. Nat. Acad. Sci. U.S.A., 81, 801.
- Reakes, M., 1980. Mon. Not. R. astr. Soc., 192, 297.
- Reddish, V.C., 1978. Stellar Formation, p. 208, Pergamon Press, Oxford.
- Rees, M.J., 1976. Mon. Not. R. astr. Soc., 176, 483.
- Rees, M.J., 1985. Proc. R. Soc. Lond., 400, 183.
- Reid, I.N., 1987. Mon. Not. R. astr. Soc., 225, 873.
- Rengarajan, T.N., 1984. Astrophys. J., 287, 671.
- Rengarajan, T.N., 1986. Private communication.
- Rengarajan, T.N. & Verma, R.P., 1986. Astr. Astrophys., 165, 300.
- Renzini, A., 1977. In: Advanced Stages in Stellar Evolution, IAU Colloq. No. 59, eds. Bouvier, P. & Maeder, A., p. 149, Geneva Observatory, Sauverny, Switzerland.
- Renzini, A., 1984. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 99, Reidel, Dordrecht, Holland.
- Renzini, A., Bernazzani, M., Buonanno, R. & Corsi, C. E., 1985. Astrophys. J., 294, L7.
- Rich, R.M., 1986. PhD Thesis, California Institute of Technology.
- Richardson, K.M. & Wolfendale, A.W., 1987. Astr. Astrophys., submitted.
- Rickard, L.J. & Blitz, L., 1985. Astrophys. J., 292, L57.
- Rickard, L.J. & Harvey, P.M., 1984. Astr. J., 89, 1520.
- Rickard, L.J., Palmer, P., Morris, M., Zuckerman, B. & Turner, B.E., 1975. Astrophys. J., 199, L75.
- Rickard, L.J., Turner, B.E. & Palmer, P., 1985. Astr. J., 90, 1175.
- Rieke, G.H. & Lebofsky, M.J., 1986. Astrophys. J., 304, 326.
- Riley, P.A., Wolfendale, A.W., Chun-Xiang, X., Manchester, R.W., Robinson, B.J., Whiteoak, J.B., 1984. Mon. Not. R. astr. Soc., 206, 423.
- Rivolo, A.R., Solomon, P.M. & Sanders, D.B., 1986. Astrophys. J., 301, L19.
- Roberts, W.W. & Burton, W.B., 1976. Paper presented at the XVIth General Assembly of the IAU, Grenoble, France.
- Roberts, W.W. & Steward, G.R., 1987. Astrophys. J., 314, 10.
- Roberts, W.W., 1969. Astrophys. J., 158, 123.
- Robin, A. & Crézé, M., 1986. Astr. Astrophys., 157, 71.
- Robinson, B.J., Manchester, R.N., Whiteoak, J.B., Sanders, D.B., Scoville, N.Z., Clemens, D.P. & Solomon, P.M., 1984. Astrophys. J., 283, L31.
- Rocca-Volmerange, B., 1984. In: Structure and Evolution of the Magellanic Clouds, IAU Symp. No. 108, eds. van den Bergh, S. & de Boer, K.S., p. 91, Reidel, Dordrecht, Holland.
- Rocca-Volmerange, B., Lequeux, J. & Maucherat-Joubert, M., 1981. Astr. Astrophys., 104, 177.
- Roddy, D.J., Pepin, R.O. & Merrill, R.B. (eds), 1977. Impact and Explosion Cratering, Pergamon Press, New York.
- Rogstad, D.H. & Shostak, G.S., 1972. Astrophys. J., 176, 315.
- Rogstad, D.H., Lockhart, I.A. & Wright, M.C.H., 1974. Astrophys. J., 193, 309.
- Rouan, D. & Leger, A., 1984. Astr. Astrophys., 132, L1.
- Rubin, V.C., Ford, W.K., Thonnard, N. & Burstein, D., 1982. Astrophys. J., 261, 439.
- Rydbeck, G., 1985. In: Proceedings of the ESO-IRAM-Onsala Workshop on Submm Astronomy, Onsala Space Observatory Preprint, No. 85:28, Onsala.
- Rydbeck, G., Hjalmarson, A. & Rydbeck, O.E.H., 1985. Astr. Astrophys., 144, 282.
- Saito, T., Yamanoi, T. & Kaiho, K., 1986. Nature, 323, 253.

Salpeter, E.E., 1959. Astrophys. J., 129, 608.

- Salpeter, E.E., 1977. Annals N. Y. Acad. Sci., 302, 681.
- Sandage, A., 1986. Astr. Astrophys., 161, 89.
- Sandage, A., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 321, Reidel, Dordrecht, Holland.
- Sanders, D.B., 1981. PhD Thesis, State University of New York at Stony Brook.
- Sanders, D.B. & Mirabel, I.F., 1985. Astrophys. J., 298, L31.
- Sanders, D.B., Clemens, D.P., Scoville, N.Z. & Solomon, P.M., 1986a. Astrophys. J. Suppl., 60, 1.
- Sanders, D.B., Scoville, N.Z. & Solomon, P.M., 1985. Astrophys. J., 288, 373.
- Sanders, D.B., Scoville, N.Z., Young, J.S., Schloerb, F.P., Rice, W.L. & Danielson, G.E., 1986b. Astrophys. J., 305, L45.
- Sanders, D.B., Soifer, B.T., Neugebauer, G., Scoville, N.Z., Madore, B.F., Danielson, G.E., Elias, J.H., Matthews, K., Persson, C.J. & Persson, S.E., 1987. Preprint.
- Sanders, D.B., Solomon, P.M., & Scoville, N.Z., 1984. Astrophys. J., 276, 182. (SSS).
- Sanders, R.H., 1977. In: Chemical and Dynamical Evolution of our Galaxy, IAU Coll. No. 45, eds. Basinska-Grzesik, E. & Mayor, M., p. 103, Geneva Observatory, Geneva.
- Sanduleak, N., 1969. Astr. J., 74, 47.
- Savage, B.D. & Mathis, J.S., 1979. Ann. Rev. Astr. Astrophys., 17, 73.
- Savage, B.D., Bohlin, R.C., Drake, J.F. & Budich, W., 1977. Astrophys. J., 216, 291.
- Scalo, J.M., 1986. Fund. Cosmic Phys., 11, 1.
- Scalo, J.M. & Pumphrey, W.A., 1982. Astrophys. J., 258, L29.
- Schild, H. & Maeder, A., 1985. Astrophys. J., 143, L7.
- Schmidt, K.H., 1967. In: Interstellar Grains, eds. Greenberg, J.M. & Roark, T.P., NASA Publications.
- Schmidt, M., 1959. Astrophys. J., 129, 243.
- Schmidt, M., 1963. Astrophys. J., 137, 758.
- Schmidt, M., 1975. Astrophys. J., 202, 22.
- Schwartz, R.D. & James, P.B., 1984. Nature, 308, 712.
- Sciama, D.W., 1972. Nature, 240, 456.

Scoville, N.Z., 1983. In: Galactic and Extragalactic IR Spectroscopy, Proceedings of the XVIth ESLAB Symp., ed. Phillips, J.B., p. 443, Reidel, Dordrecht, Holland.

- Scoville, N.Z. & Hersh, K., 1979. Astrophys. J., 265, 148.
- Scoville, N.Z. & Sanders, D.B., 1986. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 69, University of Arizona Press, Tucson.
- Scoville, N.Z. & Sanders, D.B., 1987. In: Interstellar Processes, eds. Thronson, H. & Hollenbach, D., in press.
- Scoville, N.Z. & Solomon, P.M., 1974. Astrophys. J., 187, L67.
- Scoville, N.Z. & Solomon, P.M., 1975. Astrophys. J., 199, L105.
- Scoville, N.Z. & Young, J.S., 1983. Astrophys. J., 265, 148.
- Scoville, N.Z., Sanders, D.B. & Clemens, D.P., 1986. Astrophys. J., 310, L77.
- Scoville, N.Z., Yun, M.S., Clemens, D.P., Sanders, D.B. & Waller, W.H., 1987. Astrophys. J. Suppl., 63, 821.
- Searle, L., 1971. Astrophys. J., 168, 327.
- Searle, L., 1972. In: L'age des Etoiles, p. L11, Meudon.
- Searle, L. & Sargent, W.L.W., 1972. Astrophys. J., 173, 25.
- Searle, L., Sargent, W.L.W. & Bagnuolo, W., 1973. Astrophys. J., 179, 427.
- Seiden, P.E., 1983a. Astrophys. J., 266, 555.
- Seiden, P.E., 1983b. In: Kinematics, Dynamics and Structure of the Milky Way, ed. Shuter, W.L.H., p. 259, Reidel, Dordrecht, Holland.
- Seiden, P.E. & Gerola, H., 1982. Fund. Cosmic Phys., 7, 241.
- Sekiguchi, K., 1987. Astrophys. J., 316, 145.
- Sepkoski, J.J. & Raup, D.M., 1986. In: Dynamics of Extinction, p. 3, ed. Elliot, D.K., Wiley-Interscience, New York.
- Serrano, A. & Peimbert, M., 1981. Rev. Mexicana Astr. Astrofis., 5, 109.
- Serrano, A. & Peimbert, M., 1983. Rev. Mexicana Astr. Astrofis., 8, 117.
- Seyfert, C.K. & Sirkin, L.A., 1979. In: Earth History & Plate Tectonics, 2nd ed., Harper & Row, New York.
- Shanks, T., 1987. Private communication.
- Sharov, A.S. & Lyutyi, V.M., 1984. Soviet Astr., 28, 143.

Shaver, P.A., McGee, R.X. & Pottasch, S.R., 1979. Nature, 280, 476.

Shaver, P.A., McGee, R.X., Newton, L.M., Danks, A.C. & Pottasch, S.R., 1983. Mon. Not. R. astr. Soc., 204, 53.

Shaviv, G. & Salpeter, E.E., 1982. Astr. Astrophys., 110, 300.

Shoemaker, E.M., 1977. In: Impact & Explosion Cratering, eds. Roddy, D.J., Pepin, R.O. & Merrill, R.B., p. 617, Pergamon Press, New York.

Shoemaker, E.M., 1984. In: Patterns of Change in Earth Evolution, eds. Holland, H.D. & Trendall, A.F., p. 15, Springer-Verlag, Berlin.

Shoemaker, E.M. & Wolfe, R.F., 1984a. Lunar Planet. Sci., XV, 780.

Shoemaker, E.M. & Wolfe, R.F., 1984b. Meteoritics, 19, 313.

Shoemaker, E.M. & Wolfe, R.F., 1986. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 338, University of Arizona Press, Tucson.

Shostak, G.S., 1978. Astr. Astrophys., 68, 321.

Shu, F., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, ed. van Woerden, H., Allen, R.J. & Burton, W.B., p. 561, Reidel, Dordrecht, Holland.

Shu, F., Milione, V., Gebel, W., Yuan, C., Goldsmith, D. & Roberts, W., 1972. Astrophys. J., 173, 557.

Shull, J.M. & Beckwith, S., 1982. Ann. Rev. Astr. Astrophys., 20, 163.

- Silk, J., 1977. Astrophys. J., 214, 718.
- Silk, J., 1982. Astrophys. J., 256, 514.

Silver, L.T. & Schultz, P.H., (eds) 1982. Geological Implications of Impacts of Large Asteroids & Comets on the Earth, Geol. Soc. America, Special Paper 190.

- Sloan, R.E., Rigby, J.K., Van Valen, L.M. & Gabriel, D., 1986. Science, 232, 629.
- Smit, J. & Kyte, F.T., 1984. Nature, 310, 403.

Smith, L.F., Biermann, P. & Mezger, P.G., 1978. Astr. Astrophys., 66, 65. (SBM).

Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., (eds) 1986. The Galaxy & the Solar System, University of Arizona Press, Tucson.

Sodroski, T.J., Dwek, E., Hauser, M.G. & Kerr, F.J., 1987. Preprint.

Solomon, P.M. & Rivolo, A.R., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 105, Reidel, Dordrecht, Holland.

Solomon, P.M. & Sanders, D.B., 1980. In: Giant Molecular Clouds in the Galaxy, eds. Solomon, P.M. & Edmunds, M.G., p. 41, Pergamon Press, Oxford.

Solomon, P.M. & de Zafra, R., 1975. Astrophys. J., 199, L79.

Solomon, P.M., Barrett, J., Sanders, D.B. & de Zotra, R., 1983a. Astrophys. J., 266, L103.

Solomon, P.M., Rivolo, A.R., Barrett, J.W. & Yahil, A., 1987a. Astrophys. J., submitted.

Solomon, P.M., Rivolo, A.R., Mooney, T.J., Barrett, J.W. & Sage, L. J., 1987b. In: Star Formation in Galaxies, ed. Persson, C.J., Government Printing Office, Washington, in press.

Solomon, P.M., Sanders, D.B. & Rivolo, A.R., 1985. Astrophys. J., 292, L19.

Solomon, P.M., Sanders, D.B. & Scoville, N.Z., 1979. In: The Large Scale Characteristics of the Galaxy, IAU Symp. No. 84, ed. Burton, W.B., p. 35, Reidel, Dordrecht, Holland.

Solomon, P.M., Scoville, N.Z. & Sanders, D.B., 1979. Astrophys. J., 232, L89.

Solomon, P.M., Stark, A.A. & Sanders, D.B., 1983b. Astrophys. J., 267, L29.

Spitzer, L., 1958. Astrophys. J., 127, 17.

Spitzer, L., 1978. Physical Processes in the Interstellar Medium, J. Wiley, New York.

Spitzer, L. & Baade, W., 1951. Astrophys. J., 113, 413.

Stanley, S.M., 1984. Sci. Amer., 250, No. 6 (June), 46.

Stark, A.A., 1979. PhD Thesis, Princeton University.

- Stark, A.A., 1983. In: Kinematics, Dynamics and Structure of the Milky Way, ed. Shuter, W.L.H., p. 127, Reidel, Dordrecht, Holland.
- Stark, A.A., 1984. Astrophys. J., 281, 624.
- Stark, A.A., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, ed. van Woerden, H., Allen, R.J. & Burton, W.B., p. 445, Reidel, Dordrecht, Holland.
- Stark, A.A., 1987. Private communication.
- Stark, A.A. & Blitz, L., 1978. Astrophys. J., 225, L15.

Stark, A.A., Elmegreen, B.G. & Chance, D., 1987. Astrophys. J., in press.

- Stark, A.A., Knapp, G.R., Bally, J., Wilson, R.W., Penzias, A.A. & Rowe, H.E., 1986. Astrophys. J., 310, 660. (S).
- Stauffer, J.R., 1983. Astrophys. J., 264, 14.
- Stigler, S.M., 1985. Nature, 313, 159.
- Stothers, R.B., 1984. Nature, 311, 17.

Stothers, R.B., 1985. Nature, 317, 338.

Stothers, R.B., 1986. Nature, 322, 444.

Strömgren, B., 1987. In: The Galaxy, Proceedings of the NATO Advanced Study Institute, eds. Gilmore, G. & Carswell, R.F., p. 229, Reidel, Dordrecht, Holland.

Strong, A.W., Bloemen, J.B.G.M., Dame, T.M., Grenier, I.A., Hermsen, W., Lebrun, F., Nyman, L-A., Pollock, A.M.T. & Thaddeus, P., 1987. In: Proceedings of the XXth International Cosmic Ray Conference, Moscow.

- Struck-Marcell, C. & Scalo, J.M., 1987. Astrophys. J. Suppl., 64, 39.
- Tacconi, L.J. & Young, J.S, 1985. Astrophys. J., 290, 602.
- Tacconi, L.J. & Young, J.S, 1986. Astrophys. J., 308, 600.
- Tacconi, L.J. & Young, J.S, 1987. Astrophys. J., in press.
- Talbot, R.J., 1980. Astrophys. J., 235, 821.
- Talbot, R.J. & Arnett, W.D., 1971. Astrophys. J., 170, 409.
- Talbot, R.J. & Arnett, W.D., 1973. Astrophys. J., 186, 69.
- Talbot, R.J. & Arnett, W.D., 1975. Astrophys. J., 197, 55.

Talent, D.L. & Dufour, R.J., 1979. Astrophys. J., 233, 888.

- Tayler, R.J., 1986. In: The Material Content of the Universe, Roy. Soc. Discussion Meeting, eds. Barrow, J.D., Peebles, P.J.E. & Sciama, D.W., Royal Society, London.
- Taylor, D.K., Dickman, R.L. & Scoville, N.Z., 1987. Astrophys. J., 315, 104.
- Taylor, J.H., 1979. In: The Large Scale Characteristics of the Galaxy, IAU Symp. No. 84, ed. Burton, W.B., p. 119, Reidel, Dordrecht, Holland.
- Telesco, C.M., Drecher, R. & Gatley, I., 1986. Astrophys. J., 302, 632.
- Tereby, S., Fich, M., Blitz, L. & Henkel, C., 1986. Astrophys. J., 308, 357.
- Terlevich, E., 1987. Mon. Not. R. astr. Soc., 224, 193.
- Terlevich, R.J., 1982. PhD Thesis, University of Cambridge.
- Terlevich, R.J., 1985. In: Star Forming Dwarf Galaxies, eds. Kunth, D., Thuan, T.X. & Tran Thanh Van, J., p. 395, Editions Frontieres, Paris.
- Terlevich, R.J. & Melnick, J., 1984. Preprint.
- Thaddeus, P., 1983. Private communication to S.M. Fall.
- Thaddeus, P., 1986. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 61, University of Arizona Press, Tucson.
- Thaddeus, P. & Chanan, G.A., 1985. Nature, 314, 73.
- Thaddeus, P. & Dame, T.M., 1984. In: Proceedings of Workshop on Star Formation, Occ. Reports R. Obs. Edin., No. 13, ed. Wolstencroft, R., ROE.
- Thielemann, F-K. & Truran, J.W., 1985. In: Nucleosynthesis and its Implications for Nuclear and Particle Physics, eds. Audouze, J. & Mathieu, N., p. 373, Reidel, Dordrecht, Holland.
- Thronson, H.A., Hunter, D.A., Telesco, C.M., Harper, D.A. & Drecher, R., 1987a. Astrophys. J., 317, 180.

Thronson, H.A., Walker, C.K., Walker, C.E. & Maloney, P.R., 1987b. Astrophys. J., 318, in press.

- Tinsley, B.M., 1975. Astrophys. J., 197, 159.
- Tinsley, B.M., 1977. In: Chemical and Dynamical Evolution of our Galaxy, IAU Coll. No. 45, eds. Basinska-Grzesik, E. & Mayor, M., p.309, Geneva Observatory, Geneva.
- Tinsley, B.M., 1978. In: Planetary Nebulae: Observations and Theory, IAU Symp. No. 76, ed. Terzian, Y., p. 341, Reidel, Dordrecht, Holland.
- Tinsley, B.M., 1980. Fund. Cosmic Phys.,, 5, 287.
- Tinsley, B.M., 1981. Astrophys. J., 250, 758.
- Tinsley, B.M. & Danly, L., 1980. Astrophys. J., 242, 435.
- Tinsley, B.M. & Larson, R.B., 1978. Astrophys. J., 221, 554.
- Tomisaka, K., 1986. Publs. astr. Soc. Japan, 38, 95.
- Toomre, A., 1981. In: The Structure and Evolution of Normal Galaxies, eds. Fall, S.M. & Lynden-Bell, p. 111, Cambridge University Press.
- Torbett, M.V., 1986a. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 147, University of Arizona Press, Tucson.
- Torbett, M.V., 1986b. Mon. Not. R. astr. Soc., 223, 885.
- Torbett, M.V., 1986c. Astr. J., 92, 171.
- Torbett, M.V. & Smoluchowski, R., 1984. Nature, 311, 641.
- Tornambè, A., 1984. Mon. Not. R. astr. Soc., 206, 867.
- Tornambè, A. & Chieffi, A., 1986. Mon. Not. R. astr. Soc., 220, 529.
- Tornambè, A. & Matteucci, F., 1985. Astr. Astrophys., 151, 447.
- Torres-Peimbert, S. & Peimbert, M., 1977. Rev. Mexicana Astr. Astrofis., 2, 181.
Tosi, M., 1982. Astrophys. J., 254, 699.

Tosi, M. & Diaz, A.I., 1985. Mon. Not. R. astr. Soc., 217, 571.

Tremaine, S.D., 1986. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 409, University of Arizona Press, Tucson.

Truran, J.W., 1984. In: Formation & Evolution of Galaxies and Large Scale Structures in the Universe, eds. Audouze, J. & Tran Thanh, J., p. 391, Reidel, Dordrecht, Holland.

Truran, J.W. & Cameron, A.G.W., 1971. Astrophys. Space Sci., 14, 179.

Tully, R.B. & Shaya, E.J., 1984. Astrophys. J., 281, 31.

- Turner, B.E., 1984. Vistas Astr., 27, 303.
- Twarog, B.A., 1980. Astrophys. J., 242, 242.
- Twarog, B.A. & Wheeler, J.C., 1982. Astrophys. J., 261, 636.
- Twarog, B.A. & Wheeler, J.C., 1987. Astrophys. J., 316, 153.
- Upgren, A.R. & Armandroff, T.E., 1981. Astr. J., 86, 1898.
- Urey, B.C., 1973. Nature, 242, 32.
- Vader, J.P. & de Jong, T., 1981. Astr. Astrophys., 100, 124.
- Valtonen, M.J. & Innanen, K.A., 1982. Astrophys. J., 255, 307.
- van den Bergh, S., 1958. Astr. J., 63, 492.
- van den Bergh, S., 1962. Astr. J., 67, 486.
- van den Bergh, S., 1981. Astr. J., 86, 1464.
- van den Bergh, S., 1982. J. R. astr. Soc. Canada, 76, 303.
- van den Bergh, S. & McClure, R.D., 1980. Astr. Astrophys., 88, 360.
- van der Kruit, P.C., 1972. Astrophys. J., 186, 807.
- van Dishoeck, E.F. & Black, J.H., 1986. Astrophys. J. Suppl., 62, 109.
- van Dishoeck, E.F. & Black, J.H., 1987. In: Physical Processes in Interstellar Clouds, eds. Morfill, G.E. & Scholer, M., Reidel, Dordrecht, Holland, in press.
- van Genderen, A.M., 1969. Bull. Astr. Inst. Netherlands Suppl., 3, 221.
- van Gorkom, J. & Kotanyi, C., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 61, ESO, Garching.
- van Woerden, H., Schwarz, U.J. & Hulsbosch, A.N.M., 1985. In: The Milky Way Galaxy, IAU Symp. No. 106, eds. van Woerden, H., Burton, W.B. & Allen, K.J., p. 387, Reidel, Dordrecht, Holland.
- Verschur, G.L., 1975. Ann. Rev. Astr. Astrophys., 13, 257.
- Verter, F., 1983. PhD Thesis, Princeton University.
- Verter, F., 1985. Astrophys. J. Suppl., 57, 261.
- Viallefond, F., Goss, W.M. & Allen, R.J., 1982. Astr. Astrophys., 115, 373.
- Vigroux, L., Chieze, J.P. & Larareff, B., 1981. Astr. Astrophys., 98, 119.
- Villumsen, J.V., 1985. Astrophys. J., 290, 75.
- von Hoerner, S., 1960. Fortschritte der Physik, 8, 191.
- Wagoner, R.V., 1973. Astrophys. J., 179, 343.
- Wainscoat, R.J., de Jong, T. & Wesselius, P.R., 1987. Astr. Astrophys., 181, 225.
- Waller, W.H., Clemens, D.P., Sanders, D.B. & Scoville, N.Z., 1987. Astrophys. J., 314, 397.
- Walterbos, R.A.M. & Kennicutt, R.C., 1987. In preparation.
- Walterbos, R.A.M. & Schwering, P.B.W., 1987. Astr. Astrophys., 180, 27.
- Warmels, R.H., 1985. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 51, ESO, Garching.
- Warmels, R.H., 1986. PhD Thesis, University of Groningen, Holland.
- Warmels, R.H. & van Woerden, H., 1984. In: Groups and Clusters of Galaxies, eds. Mardirossian, F., Giuricin, G. & Mezzetti, p. 251, Reidel, Dordrecht, Holland.
- Wasserman, I. & Weinberg, M.D., 1987. Astrophys. J., 312, 390.
- Weidemann, V., 1984. Astr. Astrophys., 134, L1.
- Weidemann, V. & Koester, D., 1983. Astr. Astrophys., 121, 77.
- Weinberg, M.D., Shapiro, S.L. & Wasserman, I., 1987. Astrophys. J., 312, 367.
- Weissman, P.R., 1982. Geol. Soc. Amer. Spec. Pap., 190, 15.
- Weissman, P.R., 1984. Nature, 312, 380.
- Weissman, P.R., 1985. Nature, 314, 517.
- Weissman, P.R., 1986. In: The Galaxy & the Solar System, eds. Smoluchowski, R., Bahcall, J.N. & Matthews, M.S., p. 204, University of Arizona Press, Tucson.
- Weliachew, L. & Gottesman, S.T., 1973. Astr. Astrophys., 24, 59.
- Wetherill, G.W., 1985. Meteoritics, 20, 1.
- Wetherill, G.W. & Shoemaker, E.M., 1982. Geol. Soc. Amer. Spec. Paper, 190, 1.

White, S.D.M. & Audouze, J., 1983. Mon. Not. R. astr. Soc., 203, 603.

- Whitmire, D.P. & Jackson, A.A., 1984. Nature, 308, 713.
- Whitmire, D.P. & Matese, J.J., 1985. Nature, 313, 36.
- Wielen, R., 1977. Astr. Astrophys., 60, 263.
- Wielen, R., 1985. In: Dynamics of Star Clusters, IAU Symp. No. 113, eds. Goodman, J. & Hut, P., p. 449, Reidel, Dordrecht, Holland.
- Wiklund, T., 1987. In: Star Forming Regions, IAU Symp. No. 115, eds. Peimbert, M. & Jugaku, J., Reidel, Dordrecht, Holland, in press.
- Williams, D.A., 1985. Q. J. R. astr. Soc., 26, 463.
- Williams, D.A. & Hartquist, T.W., 1984. Mon. Not. R. astr. Soc., 210, 141.
- Wilson, J.R., Mayle, R., Woosley, S.E. & Weaver, T., 1986. Ann. N.Y. Acad. Sci., 470, 267.
- Wilson, R.W., Jefferts, K.B. & Penzias, A.A., 1970. Astrophys. J., 161, L43.
- Wink, J.E., Wilson, T.L. & Bieging, J.H., 1983. Astr. Astrophys., 127, 211.
- Wolbach, W.S., Lewis, R.S. & Anders, E., 1985. Science, 230, 167.
- Woosley, S.E. & Weaver, T.A., 1986a. In: Nucleosynthesis and its Implications in Nuclear and Particle Physics, eds. Audouze, J. & Mathieu, N., p. 145, Reidel, Dordrecht, Holland.
- Woosley, S.E. & Weaver, T.A., 1986b. In: Radiation Hydrodynamics in Stars and Compact Objects, eds. Mihalas, D. & Winkler, K.H.A., p. 91, Springer-Verlag, Berlin.
- Woosley, S.E., Axelrod, T.S. & Weaver, T.A., 1984. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 263, Reidel, Dordrecht, Holland.
- Wyse, R.F.G., 1986. Astrophys. J., 311, L41.
- Wyse, R.F.G. & Gilmore, G., 1986. Astr. J., 91, 855.
- Wyse, R.F.G. & Silk, J., 1985. Astrophys. J., 296, L1.
- Wyse, R.F.G. & Silk, J., 1987. Astrophys. J., 313, L11.
- Xu, D., Ma, S., Chain, Z., Mao, X., Sun, Y., Zhang, Q. & Yang, Z., 1985. Nature, 314, 154.
- Yabushita, S., 1979. Mon. Not. R. astr. Soc., 187, 445.
- Yahil, A., 1984. In: Stellar Nucleosynthesis, eds. Chiosi, C. & Renzini, A., p. 251, Reidel, Dordrecht, Holland.
- Yang, J., Schramm, D.N., Steigman, G. & Rood, R.T., 1979. Astrophys. J., 227, 697.
- Yang, J., Turner, M.S., Steigman, G., Schramm, D.N. & Olive, K.A., 1984. Astrophys. J., 281, 493.
- Yokoi, K., Takahashi, K. & Arnould, M., 1983. Astr. Astrophys., 117, 65.
- Yoshii, Y., 1984. Astr. J., 89, 1190.
- Yoshii, Y. & Saio, H., 1986. Astrophys. J., 301, 587.
- Yoshii, Y., Ishida, K. & Stobie, R.S., 1987. Astr. J., 93, 323.
- Young, J.S., 1985a. In: The Milky Way Galaxy, IAU Symp. No. 106, eds. van Woerden, H., Allen, R.J. & Burton, W.B., p. 183, Reidel, Dordrecht, Holland.
- Young, J.S., 1985b. In: ESO Workshop on the Virgo Cluster of Galaxies, eds. Richter, O-G. & Binggeli, B., p. 151, ESO, Garching.
- Young, J.S., 1986. In: Star Forming Regions, IAU Symp. No. 115, eds. Peimbert, M. & Jugaku, J., Reidel, Dordrecht, Holland, in press.
- Young, J.S. & Sanders, D.B., 1986. Astrophys. J., 302, 680.
- Young, J.S. & Scoville, N.Z., 1982a. Astrophys. J., 258, 467.
- Young, J.S. & Scoville, N.Z., 1982b. Astrophys. J., 260, L11.
- Young, J.S., Gallagher, J.S. & Hunter, D.A., 1984. Astrophys. J., 276, 476.
- Young, J.S., Kenney, J.D., Lord, S.D. & Schloerb, F.P., 1985a. Astrophys. J., 287, L65.
- Young, J.S., Kenney, J.D., Tacconi, L.J., Clausen, M.J., Huang, Y.-L., Tacconi-Garman, L., Xie, S. & Schloerb, F.P., 1986b. Astrophys. J., 311, L17.
- Young, J.S., Schloerb, F.P., Kenney, J.D. & Lord, S.D., 1986a. Astrophys. J., 304, 443.
- Young, J.S., Scoville, N.Z. & Brady, E., 1985b. Astrophys. J., 288, 487.
- Young, J.S., Tacconi, L. & Scoville, N.Z., 1983. Astrophys. J., 269, 136.

Zinn, R., 1985. Astrophys. J., 293, 424.

ACKNOWLEDGEMENTS

It is a pleasure to thank my supervisor, Professor A.W. Wolfendale, FRS, for the opportunity to commence this work, and for his constant guidance and encouragement.

Various parts of this work were carried out in collaboration with Dr. N.C. Rana and Dr. M.E. Bailey, who are thanked for their co-operation. Dr. Rana with great patience and friendship introduced me to the excitement of astrophysics research.

I wish to thank members of the theoretical astrophysics group in the Durham Physics Department, C.L. Bhat, A. Broadbent, B.P. Houston, M.R. Issa, C.J. Mayer, J.L. Osborne, A. Parvaresh, M.L. Parkinson, K.M. Richardson, M.J. Rogers and E.C.M. Young, for helpful discussions and their friendship.

I have valued discussions with and suggestions from Profs. W.D. Arnett, S.M. Chitre, R.S. Ellis, S.M. Fall, F.D.A. Hartwick, I. Iben Jr., B.J.T. Jones, D. Lynden-Bell, J.P. Ostriker, B.E.J. Pagel, T.N. Rengarajan, G.D. Rochester, P.M. Solomon, R. Wielen, H. van Woerden, Drs. F. Ashton, A. Burkert, S.V.M. Clube, J.E. Dyson, C.S. Frenk, G. Gilmore, W.M. Napier, S. Phillips, L. Secco, A.A. Stark, T. Shanks and A.T. Sinclair. Prof. W.A. Clemens, Drs. D. Norman and C. Patterson are thanked for correspondence on the biological aspects of this work.

The provision of facilities by the Department of Physics, University of Durham is gratefully acknowledged. In particular, the help of Mrs. M. Chipchase, P. Russell and

-282-

K. Gettings in typing and diagrams. Part of this work was completed at the Tata Institute of Fundamental Research, Bombay, whose hospitality is gratefully acknowledged. All computing was carried out on STARLINK.

Financial support was provided by a University of Durham Studentship, supplemented by the Department of Physics, Tata Institute of Fundamental Research, and a NATO grant for the conference 'The Galaxy' (Cambridge 1986).

Finally, my thanks to my parents, whose love, support and prayers have been invaluable to the completion and enjoyment of this work.

